

# Examining Physical Activity and Sedentary Behaviour Changes in Late Adolescence

AUTHOR(S)

Jennifer Roseanne Hatt

PUBLICATION DATE

01-12-2019

HANDLE

10779/DRO/DU:22670872.v1

Downloaded from Deakin University's Figshare repository

Deakin University CRICOS Provider Code: 00113B

### Examining Physical Activity and Sedentary Behaviour Changes in Late Adolescence

by

Jennifer Roseanne Hatt BEx&SS(Hons)

Submitted in fulfilment of the requirements for the degree of

Doctor of Philosophy

Deakin University

December 2019



## DEAKIN UNIVERSITY CANDIDATE DECLARATION

I certify the following about the thesis entitled

Examining Physical Activity and Sedentary Behaviour Changes in Late Adolescence

submitted for the degree of Doctor of Philosophy

- a. I am the creator of all or part of the whole work(s) (including content and layout) and that where reference is made to the work of others, due acknowledgment is given.
- b. The work(s) are not in any way a violation or infringement of any copyright, trademark, patent, or other rights whatsoever of any person.
- c. That if the work(s) have been commissioned, sponsored or supported by any organisation, I have fulfilled all of the obligations required by such contract or agreement.
- d. That any material in the thesis which has been accepted for a degree or diploma by any university or institution is identified in the text.
- e. All research integrity requirements have been complied with.

'I certify that I am the student named below and that the information provided in the form is correct'

Full Name: Jennifer Roseanne Hatt

Signed: Signature Redacted by Library

Date: 11/05/2020



### DEAKIN UNIVERSITY ACCESS TO THESIS - A

I am the author of the thesis entitled

Examining Physical Activity and Sedentary Behaviour Changes in Late Adolescence

submitted for the degree of Doctor of Philosophy

This thesis may be made available for consultation, loan and limited copying in accordance with the Copyright Act 1968.

'I certify that I am the student named below and that the information provided in the form is correct'

Full Name: Jennifer Roseanne Hatt

Signed: Signature Redacted by Library

Date: 11/05/2020

### Acknowledgements

To my supervisors Professor Anna Timperio, Professor David Crawford and Doctor Verity Cleland: thank you for the research training that you poured in and your time and guidance.

To the ProjectADAPT chief investigators Professor Anna Timperio, Professor David Crawford, Associate Professor Jim Dollman, Doctor Verity Cleland and Professor Sarah McNaughton: thank you for your vision and input. To all who worked on ProjectADAPT, especially Jacqui Della Gatta, as well as Michelle Jackson, Doctor Lauren Arundell, Emily Thomas, Lisa Lucas, Stacey Eyres, Jasmine Everist, Doctor Ashley Ng, Doctor Maree Thorp and Doctor Gavin Abbott: thank you for your time and efforts. To the ProjectADAPT and LEAP informers: thank you for your responses.

Thank you to all of the funding bodies that supported this research: the Australian Research Council for a Discovery Project grant (DP130101078) that supported ProjectADAPT; the Australian Government Research Training Program for a Stipend Scholarship; the School of Exercise and Nutrition Sciences (SENS) at Deakin University for some Higher Degree by Research (HDR) student funding; and the Faculty of Health at Deakin University for a HDR student Conference Grant.

To my colleagues and peers within Deakin University, Institute of Physical Activity and Nutrition (IPAN), SENS, Behavioural Epidemiology Group (BEG) and mini-BEG: your high standards and output are inspirational. To Doctor Gavin Abbot and Doctor Karen Lamb: thank you for your data analysis support. To my peers, including Doctor Melissa Mialon, Linda Komarek, Keren Flanagan, Doctor Dai Barnett, Doctor Flick Pendergast, Doctor Maree Thorp, Doctor Melissa Burton, Doctor Katherine Downing, Doctor Jane Willcox and Doctor Rebecca Leech: thank you for your advice.

To my A-team Kea, Maple, Chestnut, Dove, the boss Buddina and Medika: thank you for carving a space that I look forward to and where I love being; thank you for our time together, laughs and smiles. To Fi, Amy, Jenna, Jess, Catherine, Amber, Caz, Dee, Jess, Nicole, Erin, Ash and Lauren: thank you for our conversations.

i

To my mother, Claire: thank you for your support; this thesis is dedicated to you; you continue to inspire me, and you deserve never-ending love, admiration and recognition. To Dad and my family: thank you for your encouragement. To my son: thank you for gifting me with joy. To my husband: thank you for your support, for being my rock and a good choice.

"If you don't go after what you want, you'll never have it. If you don't ask, the answer is always no. If you don't step forward, you're always in the same place." - Nora Roberts

#### Rationale

Physical activity declines from childhood to adulthood, on average. In late adolescence, prevalence of physical activity is low and sedentary behaviour is high. This is a concern because physical activity in adolescence is positively associated with health outcomes during adolescence and adulthood, and sedentary behaviour in adolescence is associated with negative immediate and later health outcomes. Further, during the normative transition out of secondary school, young adults are experiencing physical and mental development and situational changes in their place of residence, employment and study status, leading to altered social and financial independence. These adaptations to context and circumstances involve changes in habitual routines that may play a role in disrupting trajectories of physical activity and sedentary behaviour. The final secondary school years are, therefore, a critical time to intervene to influence future adulthood health. However, changes in physical activity and sedentary behaviour during the transition out of secondary school are poorly understood, and little is known about the underlying influences on these behaviours during this unstable period.

#### Objective

This thesis aimed to:

- 1. Examine physical activity and sedentary behaviour changes during the transition out of secondary school; and
- 2. Examine individual, social and environmental influences on physical activity and sedentary behaviour during the transition out of secondary school.

#### Methods

A pragmatic paradigm that employed both qualitative and quantitative methods was used to study the physical activity and sedentary behaviour of older adolescents. A narrative review of the literature was conducted to determine the literature gaps about how physical activity and sedentary behaviour change during the transition out of secondary school and underlying influences on these behaviours. This informed a qualitative study that used semi-structured one-on-one interviews via telephone or face-to-face (n=29; 19-years-old; 55% women). Participants were recruited via social media and convenience sampling of recent school leavers. Interviewees were asked about their physical activity and sedentary behaviour before and after transitioning out of secondary school, and what they felt was influential. Verbatim transcripts were analysed using emergent coding. Thematic analyses identified perceived changes in and influences on physical activity and sedentary behaviour during the transition out of secondary school.

That inductive study helped to inform a prospective study that collected quantitative data from a cohort of Year 11 students (second last year of secondary school). Participants were recruited via social media and secondary schools. Three surveys were conducted, either interviewer-administered via telephone or self-administered online, spaced 12-months apart (baseline: n=1 022, 17-years-old, 74% girls; two-year follow-up: n=852, 19-years-old, 74% women). Regression models and generalised estimating equations were used to analyse associations of daily means of physical activity and sedentary behaviour with multiple individual, social and environmental attributes and changes in behaviour were examined according to situational transitions. Cross-sectional correlates, longitudinal determinants, and moderators of associations between situational transitions and changes in physical activity and sedentary behaviour were identified.

#### **Overall findings**

The narrative literature review found that studies focused on the transition between late adolescence and adulthood have examined sedentary behaviour less frequently than physical activity, examined behaviours retrospectively, had long follow-up periods, and included a wide age range that included children. Therefore, the specific transition out of secondary school is not well understood and may result in unique changes in physical activity and sedentary behaviour and underlying influences.

The qualitative exploration had a sample bias towards recent school leavers commencing tertiary education and not changing their place of residence post-school. Most interviewees perceived that they changed their physical activity and sedentary behaviour post-school, with few conveying that they maintained their physical activity (n=5) or sedentary behaviour (n=4). The directions of change were mixed (n=13 increased and n=11 decreased physical activity; n=13 increased and n=12decreased sedentary behaviour). Recent school leavers described that the types of behaviour that changed the most were active transport, occupational physical activity, sport, technology use, studying at a desk, sedentary transport and occupational sedentary behaviour. The main influences on physical activity during their transition out of secondary school were believed to be time-use and social support, and, for sedentary behaviour, tertiary education and social support.

The test-retest reliability of the ProjectADAPT survey was tested in a separate sample of Australian 17-year-olds (*n*=83) and was acceptable. Compliance with the National Physical Activity (53%) and Sedentary Behaviour (34%) Guidelines was poor, although higher than other studies. Mean discretionary physical activity was 77 mins/day and total sedentary behaviour was 9 h/day in Year 11. Cross-sectional correlates of physical activity in late adolescence were physical activity goal setting, self-efficacy and enjoyment, social network count and sedentary behaviour discouragement from friends or colleagues (all positive). In multivariate analyses, the only correlate of sedentary behaviour identified in late adolescence was physical activity goal setting (negative).

After participants transitioned out of secondary school, compliance with the National Physical Activity and Sedentary Behaviour Guidelines decreased. There were decreases in mean discretionary physical activity (-9 mins/day; p<0.001) and total sedentary behaviour (-44 mins/day; p<0.001). There were differences in changes in sedentary behaviour according to situational transitions, with recent school leavers who worked  $\geq$ 20 h/wk, exclusively worked, did not study, lived independently, or studied while living independently decreasing sedentary behaviour more than respective comparison groups. Situational transitions did not affect changes in physical activity.

Longitudinally, physical activity enjoyment and goal setting at baseline predicted higher physical activity following the transition, while baseline television avoidance self-efficacy predicted lower sedentary behaviour. Numerous moderators of the associations between situational transitions and physical activity (12) or sedentary behaviour (2) during the transition out of secondary school were found. Common moderators were baseline levels of sedentary behaviour discouragement from family, friends or colleagues, co-participation in physical activity, electronic games, television and digital versatile discs (DVDs) with family, friends or colleagues. An example is those living with parents post-school had higher physical activity if they reported a high amount of physical activity equipment at home in Year 11.

v

#### Conclusion

Physical activity and sedentary behaviour patterns in late adolescence changed over time. Physical activity declined after Year 11, whereas sedentary behaviour declined after leaving the structure of secondary school. Working  $\geq 20$  h/wk and/or exclusively working resulted in the greatest declines in total sedentary behaviour during the transition. Correlates and determinants of physical activity and sedentary behaviour were mostly individual attributes, including physical activity goal setting, enjoyment and self-efficacy. In contrast, predominantly social attributes moderated associations between situational transitions and physical activity and sedentary behaviour during the transition out of secondary school, including sedentary behaviour discouragement and co-participation in physical activity with family. Therefore, to insulate older adolescents against the impact of the transition out of secondary school, initiatives are needed that aim to increase physical activity goal setting, enjoyment and selfefficacy, sedentary behaviour discouragement, and co-participation in physical activity with family. Future research is needed to identify effective, feasible and scalable strategies to inform interventions in multiple settings, including secondary schools, tertiary institutions, families and communities. This may result in the adoption and maintenance of higher physical activity and lower sedentary behaviour during time which are associated with positive health outcomes in adolescence and later adulthood.

# **Table of contents**

Acknow	vledgementsi	
Abstrac	tiii	
Table of	f contentsvii	
List of t	ablesx	
List of f	figuresxii	
List of a	appendicesxiv	
Glossar	y of key abbreviationsxv	
Glossar	y of key termsxvi	
Chapter	1	
Thesis of	overview1	
Chapter	· 2	
Literatu	re review	
2.1	Introduction	
2.2	Overview of adolescence	
2.3	Definitions, measurement and health outcomes10	
2.4	Guidelines	
2.5	Changes in physical activity and sedentary behaviour in late adolescence.20	
2.6	Explaining physical activity and sedentary behaviour in late adolescence .24	
2.7	Conclusion and thesis aims	
Chapter	• 3	
A qualit	tative exploration of influences on physical activity and sedentary behaviour the transition out of secondary school 41	
3.1	Introduction	
3.2	Chapter aims	
3.3	Methods	
3.4	Results	
3.5	Discussion	
Chapter 4		
Project/	ADAPT memodology	
4.1	nuouucuon	
4.2	Chapter aims	

4.3	Methods	84
4.4	Results	03
4.5	Discussion	10
Chapter	51	19
Physical	l activity in late adolescence1	19
5.1	Introduction1	20
5.2	Chapter aims1	20
5.3	Methods1	21
5.4	Results	23
5.5	Discussion	27
Chapter	61	35
Sedenta	ry behaviour in late adolescence1	35
6.1	Introduction1	36
6.2	Chapter aims1	37
6.3	Methods1	37
6.4	Results	39
6.5	Discussion	43
Chapter	71	51
Changes	s in physical activity and sedentary behaviour during the transition out of ry school	51
7.1	Introduction 1	52
7.1	Chapter aims	53
7.2	Methods 1	53
7.5	Results 1	57
7.5	Discussion	57 73
7.5		15
Chapter	81	79
Determi transitio	nants and moderators of physical activity and sedentary behaviour during the nout of secondary school	he 79
8.1	Introduction 1	80
8.2	Chapter aims1	81
8.3	Methods1	81

8.4	Results		
8.5	Discussion		
Chapter	9		
Conclus	ions	203	
9.1	Introduction		
9.2	Overview of findings		
9.3	Strengths and limitations		
9.4	Implications		
9.5	Conclusion	214	
References			
Appendi	Appendices		

# List of tables

### Chapter 2

Table 2.1	Physical activity information
Table 2.2	Sedentary behaviour information
Table 2.3	Assessment tools for physical activity and sedentary behaviour
Table 2.4	Summary of proportion of populations who meet guidelines in various high-income countries
Table 2.5	Summary of key correlates of physical activity and sedentary behaviour in late adolescence
Table 2.6	Summary of determinants of physical activity and sedentary behaviour in late adolescence
Chapter 3	
Table 3.1	Sample profile ( <i>n</i> =29)
Table 3.2	Examples and the emergent hierarchical order of the perceived influences on changes in physical activity and sedentary behaviour post-school
Table 3.3	Major, minor and uncommon themes of the perceived influences on changes in physical activity and sedentary behaviour post-school
Table 3.4	Summary of individual, social and environmental influences on changes in physical activity and sedentary behaviour post- school, by direction of influence
Chapter 4	
Table 4.1	Treatment and scoring of individual, social and environmental survey items, and internal reliability
Table 4.2	Sample profile ( <i>n</i> =1 022)
Table 4.3	Test-retest reliability results of the survey items for physical activity and sedentary behaviour variables
Table 4.4	Test-retest reliability results of the survey items for the independent variables
Table 4.5	Baseline distribution (mean±SD) for independent variables
Chapter 5	
Table 5.1	Duration of discretionary physical activity, LTPA and active transport (mean mins/day±SD) in late adolescence
Table 5.2	School-related physical activity in late adolescence
Table 5.3	Occupational physical activity in late adolescence
Table 5.4	Partially- and fully-adjusted associations between independent variables and discretionary physical activity (mins/day) in late adolescence

Chapter 6	
Table 6.1	Proportion of participants reporting 'mainly sitting' during school breaks and at work in late adolescence
Table 6.2	Duration of total sedentary behaviour and recreational screen time (mean h/day±SD) in late adolescence
Table 6.3	Partially- and fully-adjusted associations between independent variables and total sedentary behaviour (mins/day) in late adolescence
Chapter 7	
Table 7.1	Baseline characteristics of study completers and non- completers
Table 7.2	Changes in physical activity during the transition out of secondary school
Table 7.3	Changes in sedentary behaviour during the transition out of secondary school
Table 7.4	Study characteristics of participants at the two-year follow-up
Table 7.5	Employment characteristics of participants at the two-year follow-up
Table 7.6	Other characteristics of the participants at the two-year follow- up
Chapter 8	
Table 8.1	Additional treatment of baseline independent variables in preparation for moderation analyses and the proportion of the cohort who had high baseline levels
Table 8.2	Comparison of the distribution of independent variables at baseline between study completers and non-completers
Table 8.3	Baseline determinants of discretionary physical activity and total sedentary behaviour after leaving secondary school

# List of figures

### Chapter 2

-	
Figure 2.1	The ecological model's categories of influences on behaviour
Figure 2.2	Life transition model
Figure 2.3	The step method with a dichotomised moderator altering the strength of the association between the independent variable (situational transition) and the dependent variable (physical activity/sedentary behaviour)
Figure 2.4	The step method with a dichotomised moderator altering the direction of the association between the independent variable (situational transition) and the dependent variable (physical activity/sedentary behaviour)
Chapter 3	
Figure 3.1	Study flow diagram
Figure 3.2	The 20 words most frequently said by the overall sample
Figure 3.3	The 20 words most frequently said by participants who identified as women
Figure 3.4	The 20 words most frequently said by participants who identified as men
Chapter 4	
Figure 4.1	Study design and timeline
Figure 4.2	Participant recruitment flow diagram
Chapter 7	
Figure 7.1	Study design and timeline
Figure 7.2	Retention proportions (RP) among participants
Figure 7.3	Change in marginal means (95% confidence intervals) for total sedentary behaviour according to tertiary student status at two-year follow-up ( $n=796$ )
Figure 7.4	Change in marginal means (95% confidence intervals) for total sedentary behaviour according to substantial weekly work hours at two-year follow-up ( $n$ =804)
Figure 7.5	Change in marginal means (95% confidence intervals) for total sedentary behaviour according to living with parents at the two-year follow-up ( $n$ =805)
Figure 7.6	Change in marginal means (95% confidence intervals) for total sedentary behaviour according to study and work combinations at the two-year follow-up ( $n=795$ )

Figure 7.7	Change in marginal means (95% confidence intervals) for total sedentary behaviour according to whether participants were studying while living with or without parents at the two-year follow-up ( $n=785$ )
Figure 7.8	Change in marginal means (95% confidence intervals) for total sedentary behaviour according to whether participants were working while living with or without parents at the two-year follow-up ( $n$ =795)
Chapter 8	
Figure 8.1	Moderation of associations between tertiary student status and discretionary physical activity (adjusted marginal means) at the two-year follow-up
Figure 8.2	Moderation of associations between substantial weekly work hours and total sedentary behaviour at the two-year follow-up, adjusted marginal means
Figure 8.3	Moderation of associations between substantial weekly work hours and discretionary physical activity at the two-year follow-up, adjusted marginal means
Figure 8.4	Moderation of associations between living with parents and discretionary physical activity or sedentary behaviour at the two-year follow-up, adjusted marginal means

#### Chapter 2

Appendix 2.1	Summary of prospective longitudinal studies of tracking of
	physical activity during the transition from adolescence to
	adulthood

#### Chapter 3

Appendix 3.1	Consolidated criteria for reporting qualitative research
	(COREQ) guidelines

Appendix 3.2 Qualitative study interview topic guide

Appendix 3.3 Qualitative study coding tree

#### Chapter 4

Appendix 4.1	ProjectADAPT two-year follow-up telephone survey
Appendix 4.2	Sample profile ( <i>n</i> =1 022) according to remoteness and area- level SEP
Appendix 4.3	Sample profile ( $n=1$ 022) according to survey delivery mode and recruitment method
Appendix 4.4	Individual survey items' test-retest reliability scores
Appendix 4.5	Baseline distribution (mean±SD) for independent variables according to remoteness and area-level SEP
Chapter 7	
Appendix 7.1	Time by gender, remoteness or area-level SEP interactions for changes in physical activity and sedentary behaviour between baseline and the two-year follow-up
Appendix 7.2	Interactions between situational transitions and changes in discretionary physical activity and total sedentary behaviour
Chapter 8	
Appendix 8.1	Interaction effects between baseline levels of the independent variables and situational transitions on discretionary physical activity (mins/day) after leaving secondary school
Appendix 8.2	Interaction effects between baseline levels of the independent variables and situational transitions on total sedentary

Appendix 8.3 Associations between situational transitions and difference in adjusted marginal means of discretionary physical activity and total sedentary behaviour at the two-year follow-up, stratified by significant baseline moderators

behaviour (h/day) after leaving secondary school

# **Glossary of key abbreviations**

Abbreviation	Term
±	Standard deviation, unless otherwise noted
AIC	Akaike's information criteria
ANOVA	Analysis of variance
В	Correlation coefficient
CI	Confidence interval
DVD	Digital versatile disc
E-device	Electronic device
E-game	Electronic game
Н	Hours
ICC	Intraclass correlation coefficient
IPAQ	International Physical Activity Questionnaire
IPAQ-A	International Physical Activity Questionnaire for Adolescents
LPA	Light-intensity physical activity
LTPA	Leisure-time physical activity
Max.	Maximum
Mins	Minutes
MPA	Moderate-intensity physical activity
MVPA	Moderate- to vigorous-intensity physical activity
N/A	Not applicable
No.	Number
PA	Physical activity
PC	Personal computer
PE	Physical education
Ref.	Reference group
SB	Sedentary behaviour
SD	Standard deviation
SE	Standard error
SEIFA	Socio-Economic Indexes for Areas
SEP	Socio-economic position
TV	Television
UK	United Kingdom
USA	United States of America
VCE	Victorian Certificate of Education
VPA	Vigorous-intensity physical activity
Vs	Versus
Wk	Week

Note: This glossary is intended to be referred to concurrently when reviewing chapters, especially alongside the tables therein.

# Glossary of key terms

Key terms	Definition
Correlate	An independent variable that is associated with a dependent variable <sup>1</sup> . Derived from cross-sectional data <sup>2</sup> . Not necessarily on the causal pathway between the independent and dependent variables, as the direction of association cannot be established with cross-sectional data <sup>2</sup> . Can suggest what may partially explain behaviour that warrants further examination in longitudinal studies.
Cross-sectional study	Cross-sectional studies are based on data collected from participants at one point in time; in effect, taking a snapshot of the participants' behaviour.
Determinant	An independent variable that is a causal factor that results in differences in a dependent variable <sup>2</sup> . Derived from longitudinal studies and can be used to predict prospective changes <sup>2</sup> . They provide stronger evidence of factors that facilitate or inhabit dependent variables over time than correlates, since determinants precede the behaviour <sup>2</sup> .
Late adolescence	For the purposes for this thesis, late adolescence is defined as 16-21-years-old. The World Health Organisation states the age range for adolescence is 10-19 years <sup>3</sup> . By 19-years- old, almost every Australian (98%) has left secondary school <sup>4</sup> , signalling young adulthood and potential commencement of multiple situational transitions.
Longitudinal study	Longitudinal studies follow a cohort of participants over a period of time, taking periodic or pre-post measurements.
Moderator	A third variable that is an effect modifier. Different levels of the moderator result in different relationships between the independent variable's effect on the dependent variable <sup>5</sup> . Derived from longitudinal studies and intervenes with levels of a dependent variable over time.
Physical activity	Physical activity can be broadly described as movement by muscles that causes an increase in energy <sup>6</sup> .
- Discretionary physical activity	Discretionary physical activity is one of the primary outcome variables in Chapters 4-8. It is the sum of moderate- to vigorous-intensity physical activity in leisure- time and transport domains. It does not include light- intensity, school-related or occupational physical activity time because these were not assessed by the ProjectADAPT survey. It represents non-utilitarian physical activity because active transport is arguably not discretionary for some, for example, young people, disadvantaged groups, those without driver/motor bicycle license or those without access to a car/motor bicycle.

\_

Key terms	Definition
Sedentary behaviour	Sedentary behaviour is an activity that involves sitting, reclining, or limited movement, and low energy expenditure during waking hours <sup>7</sup> . Sedentary behaviour excludes sitting while active such as cycling.
- Total sedentary behaviour	Total sedentary behaviour is the other primary outcome variable in Chapters 4-8. It is the sum of sedentary behaviour in all domains (leisure-time, transport, education, work and domestic). Although the behaviours sought by the International Physical Activity Questionnaire (IPAQ) refer to 'sitting', the survey item asked about other postures as well (time lying down) which is more reflective of the term 'sedentary behaviour' <sup>7</sup> .
Situational transition	These include life events and post-school pathways, such as changes in place of residence, employment, education, health and relationships. It is hypothesised that routines and habits around physical activity and sedentary behaviour may be disrupted as a result of situational transitions <sup>8</sup> . Although childbirth and marriage can affect physical activity <sup>9, 10</sup> , these will not be examined because the median age of occurrence is older than the focus age range for this thesis (28.9 and 29.2-years-old for females, respectively <sup>11</sup> ).
Year 11; Year 12	The penultimate year and final year of secondary schooling in the Australian school system, respectively.

Chapter 1

Thesis overview

t is well-known that physical activity has a dose-response relationship with health outcomes, with inadequate physical activity associated with coronary heart disease, type two diabetes mellitus, colon and breast cancers, and premature mortality<sup>12</sup>. Additionally, growing evidence indicates sedentary behaviour (sitting, reclining or limited movement, and low energy expenditure during waking hours<sup>7</sup>) is associated with cardiometabolic outcomes and all-cause mortality risk<sup>13, 14</sup>. Most adolescents do not meet Physical Activity and Sedentary Behaviour Guidelines<sup>15</sup>. This is compounded by physical activity tending to decrease and sedentary behaviour tending to increase with age<sup>16</sup> and, specifically, an average decrease in physical activity of 7% per year during adolescence<sup>17</sup>. Therefore, ensuring young people are sufficiently active and limit sedentary behaviour is critical.

How and why physical activity and sedentary behaviour change toward the end of adolescence during the transition out of secondary school is poorly understood. Late adolescence is a life stage where there is high variability and instability due to the wide range of life changes that occur, such as leaving secondary school, commencing tertiary education or commencing full-time employment. The life transition model<sup>18</sup> posits that transitions are a time when harmful changes in health behaviours patterns may occur that lay the foundation for the development of chronic health conditions and diseases. The model also hypothesises that those going through a transition are highly receptive to education, presenting an opportunity to embed resilience to harmful changes in behaviour and lifestyle prior to leaving secondary school. Therefore, intervening prior to the transition out of secondary school may set up health-promoting habits regarding physical activity and sedentary behaviour for the lifespan. To do so, and in keeping with the Behavioural Epidemiology Framework<sup>19</sup>, key influences need to be identified that may protect against inadequate physical activity and excessive sedentary behaviour. The ecological model<sup>20, 21</sup> suggests assessing a range of potential individual, social and environmental influences. Despite there being studies that have reported influences on physical activity and sedentary behaviour in adolescence, including longitudinal studies, very little research has sought to examine influences on physical activity, and especially sedentary behaviour, during the transition out of secondary school in late adolescence (16-22-year-olds).

This thesis aims to examine how physical activity and sedentary behaviour change in late adolescence and to identify the individual, social and environmental influences underlying those two health behaviours. This thesis focusses on the time when students are in the final years of secondary school and in the immediate years after, ensuring the impact of the normative transition of leaving school on physical activity and sedentary behaviour is observed. The aims of this thesis are fulfilled using a sequential progression of data collection and analysis methods. Correspondingly, each chapter develops the evidence from previous chapters, expanding what is known about how and why physical activity and sedentary behaviour change during the transition out of secondary school. A mixed-method approach is used, including both qualitative (Chapter 3) and quantitative (cross-sectional and longitudinal; Chapters 4-8) analyses.

Chapter 2 presents a narrative review of existing literature on physical activity and sedentary behaviour in late adolescence. Descriptive epidemiology of the two health behaviours are discussed, as are the various influences. The gaps in the literature are highlighted throughout this chapter. Chapter 3 presents qualitative data from the LEAP (LEaving school and your Activity Patterns) study. The LEAP study involved interviews that captured the narratives, viewpoints and perceptions of recent secondary school leavers (18-22-year-olds). The topic guide focussed on changes in physical activity and sedentary behaviour, the impact of situational transitions, and individual, social and environmental influences on physical activity and sedentary behaviour during the transition out of secondary school.

Findings from the LEAP study were used to inform survey development for a twoyear prospective study of school leavers named ProjectADAPT. Chapter 4 presents the baseline participant characteristics and test-retest reliability from ProjectADAPT. Chapters 5-6 outline cross-sectional baseline data from ProjectADAPT that provides a snapshot of behaviour patterns and identifies individual, social and environmental correlates of physical activity and sedentary behaviour in late adolescence.

In Chapter 7, longitudinal changes in physical activity and sedentary behaviour during the transition out of secondary school within the ProjectADAPT study are described. This chapter represents the first of two chapters involving quantitative longitudinal analyses, as the type of evidence examined in this thesis shifts from qualitative (Chapter 3) and quantitative cross-sectional data (Chapters 5-6).

Identifying how physical activity and sedentary behaviour change during the transition out of secondary school is important, and yet there have been few studies examining this period. This is despite exposure to multiple simultaneous changes in lifestyle and circumstance which could impact health behaviours. This chapter also describes the proportion who experienced various situational transitions after leaving secondary school, such as changes in education, employment and housing, and whether these are associated with changes in physical activity and sedentary behaviour.

Chapter 8 identifies longitudinal determinants of physical activity and sedentary behaviour during the transition out of secondary school. As such, the chapter builds on previous findings presented in this thesis by examining whether influences on physical activity and sedentary behaviour identified in the qualitative (Chapter 3) and cross-sectional quantitative (Chapters 5 and 6) studies are also important longitudinally. This chapter expands on the findings of Chapter 7 by identifying which baseline individual, social and environmental attributes moderate associations between situational transitions (for example, living independently from family) and physical activity and sedentary behaviour two years later, after transitioning out of secondary school. This chapter is novel because it examines a wide range of potential moderators, including environmental characteristics, and also includes sedentary behaviour; whereas, previous research of moderators during the transition out of secondary school has been limited to psychosocial variables and physical activity in one domain only  $(leisure-time)^{22}$ . This research is important because it will inform interventions as to individual, social and environmental attributes on which to base strategies for school leavers, with further targeting possible for those who experience specific situational transitions.

The final chapter provides a concise synthesis of the overall findings and outlines a critical analysis of the research methods undertaken for this thesis by summarising key strengths and limitations. Recommendations are provided for preventative interventions that aim to promote the adoption and maintenance of physical activity and avoidance of excessive sedentary behaviour during the transition out of secondary school. Implications of the thesis findings for future research and practice are proposed.

Chapter 2

Literature review

#### 2.1 Introduction

he aim of this chapter is to present a narrative literature review on physical activity and sedentary behaviour in late adolescence and the transition out of secondary school and the influences on these behaviours. This chapter begins with an overview of adolescence, then presents how physical activity and sedentary behaviour are defined, measured and affect health outcomes. The current National Physical Activity and Sedentary Behaviour Guidelines for adolescence are outlined, along with compliance estimates and a review of how physical activity and sedentary behaviour change over late adolescence. Theoretical models for understanding influences on physical activity and sedentary behaviour are described and correlates of these behaviours are reviewed. Associations between situational transitions and changes in physical activity and sedentary behaviour during the transition out of secondary school are introduced. Lastly, longitudinal determinants and moderators of the associations between situational transitions and physical activity and sedentary behaviour during the transition out of adolescence are summarised. This chapter concludes with the overall thesis aims that stem from the gaps in the literature.

#### 2.2 Overview of adolescence

Adolescence has been defined as the life stage between 10-19-years-old that bridges childhood and adulthood<sup>23</sup>. The transition out of secondary school during late adolescence (15-19-year-old) has been identified as a critical time and important life stage for growth and development<sup>23</sup>. It is a time of change and instability due in part to a range of key life events that occur over this time<sup>23</sup>. Common situational transitions include leaving school, moving out of the family home for the first time, travelling, or commencing full-time employment or tertiary study. The transition is associated with a range of life changes, such as social, independence, setting, economic, developmental and lifestyle changes<sup>23</sup>.

#### 2.2.1 Social changes

Adolescence is a time when much maturation, character formation and emotional development occurs<sup>23, 24</sup>. During adolescence, there is generally an increase in social behaviour<sup>23, 25</sup>. A study reported that 15-19-year-olds adolescents spend an average of 1h 23mins socialising in the home and with friends on a weekday, and 4h 27mins on a Saturday<sup>26</sup>. This is opposed to 19-48 mins/day for parents<sup>27</sup>. Adolescent peer

groups are unstable since <50% of friendships last >1yr and between 33-50% of dissolve by adulthood<sup>28</sup>.

#### 2.2.2 Changes in independence

Adolescents are usually dependent on parents for food, shelter, transport and money. Compared to adults, adolescents spend more time studying, more time in recreation and less time driving<sup>29</sup>. Generally speaking, as adolescents transition into adulthood authority figures are questioned, autonomy increases, and dependence on parental figures is reduced<sup>24</sup>. Adults, such as parents and school teachers, typically impose boundaries on adolescents on what is acceptable to do and not acceptable. From approximately 18-years-old, gaining a license to drive unaccompanied is possible, which decreases dependence on parental figures for transport and increases independence. Also, approximately from 18-years-old, individuals are now legally able to access entertainment venues, such as bars, night clubs, and gambling facilities<sup>30</sup>, and Australian citizens are required by law to vote in parliamentary elections (enrolment may occur from 16-years-old)<sup>31</sup> which may also increase independence.

#### 2.2.3 Changes in setting

The secondary school setting is structured in nature; it is regimented, there are bells to prompt students to change locations, with timetabled classes, curricular and extracurricular opportunities for sport and creative pursuits, and scheduled breaks for recess and lunch. However, post-secondary school, the structure of the weekday may change from early starts, scheduled breaks, and mid-afternoon end-times. An introduction to other settings may be a reason for the shift in structure and change in routine and behaviour during the transition out of secondary school.

In the workplace setting, sedentary jobs are an increasing population trend, with more persons employed in desk-based jobs than in previous decades<sup>32, 33</sup>. Thorp et al.<sup>34</sup> reported sedentary time accounted for 77% of time spent at work in office-based occupations, with almost half of this time spent in prolonged bouts of sitting  $\geq$ 20 mins. Additionally, some desk-based occupations may have set business hours (for example, 9am-5pm) with one break for lunch, which is conducive to prolonged sitting bouts of up to 4h duration. However, the primary employment options for

young adults include part-time occupations that typically involve more standing than sitting, such as in the retail trade industry<sup>35</sup>.

For those combining study and work, there are opportunities for physical activity during travel to and from the workplace, place of study and home. In higher education settings (universities and vocational training), the main activity is sedentary behaviour during classes, but there are opportunities for activity during breaks and when traveling to and from home. In the home setting, some young adults might experience changes post-secondary school. New expectations may include more domestic chores, financial contributions via rent or board, grocery shopping for the household, meal preparation and responsible pet ownership.

#### 2.2.4 Economic changes

In Australia, to be able to work without significant restriction on the type and duration of employment individuals must be  $\geq 15$ -years-old<sup>36</sup>. As of February, 2014, 12% of Australian 15-19-year-olds worked full-time and 32% worked part-time<sup>37</sup>. Gaining employment may increase responsibility and accountability. Increased access to income leads to economic independence from parents<sup>24</sup>. Adolescents with economic independence have increased opportunities for additional progress and change, such as being able to afford a car, to participate in a larger range of social activities, to travel, to rent and, therefore, move out of the family home.

#### 2.2.5 Developmental changes

Biological changes occur during adolescence as the individual's body matures. Physiologically, adult-like features and functions become more prominent during late adolescence, as a consequence of puberty and high hormone levels<sup>38</sup>. Cognitive neuroscience research has shown that dramatic brain development occurs during adolescence, especially of the prefrontal cortex<sup>25</sup>. This part of the brain is involved in high-level cognitive functions, such as making decisions, planning ahead, social interactions, becoming self-aware, feeling self-consciousness, showing restraint, and being able to control impulses and inhibit behaviours<sup>39</sup>. Prior to the development of these high-level cognitive functions, adolescent brains have a heightened responsiveness to incentives<sup>40</sup>. This partially explains typical adolescent behaviours, such as spontaneity, unconsidered decision-making, expecting actions will be inconsequential, and nonlinear changes in behaviour<sup>40</sup>.

#### 2.2.6 Changes in lifestyle health behaviours

Lifestyle health behaviours continue to be learnt and practised during adolescence, including behaviours associated with social health, mental health, eating patterns, weight status, physical activity and sedentary behaviour. Experimentation and initiation of consumption of alcohol and cigarettes is common during adolescence<sup>41</sup> and from 18-years-old, Australians are able to legally purchase these substances. Further, risk-taking and novelty-seeking behaviours can be prevalent during adolescence<sup>25</sup> with the more serious examples being binge-drinking alcohol, illegal activities, illegal drug use, unsafe sex, and reckless driving<sup>42</sup>. Risk-taking behaviours may increase during adolescence and then rapidly decrease during the transition into early adulthood as impulse control develops<sup>42</sup>. Risk-taking behaviours tend to differ by sex in a similar way to puberty, with females starting and peaking earlier than males, males peaking at higher levels than females, and males reducing risk-taking behaviours more gradually than females<sup>42</sup>.

How much people move and sit and what people eat are key modifiable lifestyle health behaviours. Eating patterns during adolescence mostly do not meet nutrition guidelines, with most adolescents not eating sufficient fruit and vegetables, and skipping breakfast on  $\geq$ 5 days/wk<sup>43, 44</sup>. However, as adolescents transition into adulthood, eating behaviours become more aligned with nutrition guidelines<sup>45</sup>. Physical activity and sedentary behaviour are important and influential on aspects of social, physical and mental health, as described in the follow section. It is generally accepted that physical activity declines over adolescence<sup>15, 17, 23</sup>, although less is known about how sedentary behaviour changes. This narrative literature review focuses on physical activity and sedentary behaviour because little is known about how and why these important health behaviours change during the transition out of secondary school.

#### 2.3 Definitions, measurement and health outcomes

Although physical activity and sedentary behaviour are on the same movement and energy spectrum and, therefore, are interdependent, there are distinct implications that stem from each<sup>46</sup> that pose positive outcomes or negative burdens on public health. The health benefits of physical activity are well established and have a positive dose-response relationship, beginning with increments in activity  $>0^{47}$ . Compared to research on physical activity and health, research on the deleterious health outcomes of sedentary behaviour are more recent and still emerging. For clarity, how physical activity and sedentary behaviour are defined and measured are summarised below.

#### 2.3.1 Defining and measuring physical activity and sedentary behaviour

Physical activity is movement by the body by the skeletal muscles that causes an increase in energy expenditure<sup>6</sup> and is a behaviour that is directly modifiable. There are various terms to describe physical activity. Firstly, health-enhancing physical activities are performed in addition to 'baseline' activities (light-intensity activities of daily living, such as walking, standing and lifting light objects)<sup>48</sup>. Secondly, inadequate physical activity is an insufficient amount of health-enhancing physical activity<sup>49</sup>, whereby recommended amounts for general health benefits are not achieved<sup>50</sup>. Physical activity may be categorised into domains, intensity, frequency, duration and type (Table 2.1).

Category	Definition	
Domain	Leisure-time, occupational (including school or educational), transportation, and domestic <sup>51</sup> .	
Туре	Walking, running, swimming, playing sport, cycling, taking public transport, gardening, vacuuming, and sweeping <sup>52</sup> .	
Intensity	Light, moderate or vigorous intensities are, respectively, 1.6-2.9, 3- 5.9 and 6-9 times the metabolic equivalent of rest <sup>53</sup> . Examples are stretching, walking and running, respectively <sup>52</sup> .	
Frequency	How often it is performed.	
Duration	Can be separated into bouts, such as a short, 10 mins bout or longer bouts. Total duration is the sum of bouts $>10 \text{ mins}^{54}$ .	

**Table 2.1**Physical activity information

In the past, people who achieved low amounts of physical activity were considered 'sedentary'<sup>50</sup>. Now, however, sedentary behaviour is known as distinct from physical activity<sup>50</sup>, although arguably interdependent. Sedentary behaviour involves sitting, reclining, or limited movement, and low energy expenditure during waking hours<sup>50</sup> and is also a behaviour that is directly modifiable. Recreational screen time (as opposed to educational screen time) includes watching television, watching a movie at a cinema, using the internet on a computer, using a mobile phone, using handheld devices (tablets) or using consoles, such as Xbox (Microsoft, Redmond, Washington, United States of America (USA)) or PlayStation<sup>®</sup> (Sony Computer Entertainment Europe, London, United Kingdom (UK)). Educational screen time includes using devices at school to enhance learning<sup>55</sup> and to prepare for technology-use in the workplace<sup>56</sup>. Sedentary behaviour may also be categorised into domains, intensity, frequency, duration and type (Table 2.2).

Category	Definition		
Domain	Leisure-time, occupational (including school or educational),		
	transportation, and domestic <sup>51</sup> .		
Туре	Reading, sitting at a desk, driving a car, sitting on public transport,		
	playing piano, sewing, and screen time.		
Intensity	These activities have a metabolic equivalent range of $\leq 1.5^{50}$ .		
Frequency	Can be measured by bouts, whereby a bout can be started by sitting		
	and finished or interrupted by standing or stepping <sup>57</sup> .		
Duration	Usually dichotomised into prolonged sitting, whereby sitting is		
	uninterrupted for $\geq 20$ mins <sup>58</sup> or interrupted sitting, whereby standing		
	or stepping for $\geq 1$ min interrupts the bout of sitting <sup>57</sup> .		

**Table 2.2**Sedentary behaviour information

Common tools for assessing physical activity and sedentary behaviour are presented in Table 2.3. When choosing a suitable assessment tool for a population study, criteria to consider include cost, age-appropriateness and practicality for the sample size<sup>59</sup>.

Tools	Strengths	Limitations	
Self-report: survey,	Cost efficient <sup>60</sup> , requires	Potential bias from	
interview, log book, diary,	short term memory, and	over-reporting and	
recall questionnaire	captures domain and	under-reporting <sup>62</sup> .	
	context <sup>61</sup> .		
Device-based: pedometer	Automatically collects	Cannot capture	
(stepping), accelerometer	activity information that is	domain, context and	
(sedentary, light, moderate	time and date stamped. co-participation		
and vigorous activity),	Doesn't involve recall	information <sup>63</sup> . Devices	
inclinometer (sitting,	bias <sup>61</sup> . Better validity than	may be expensive <sup>61</sup> .	
stepping and standing)	self-report.		

Table 2.3	Assessment tools	for physical	l activity and	sedentary ]	behaviour
1 abic 2.5	Tissessment tools	ioi physica	i activity and	sedentary	ocnavioui

Self-reported assessment of physical activity and sedentary behaviour is common in population studies and involves participants reporting to researchers. Self-reporting captures data that devices cannot, such as the domain, the specific type of physical activity and sedentary behaviour, and non-ambulatory physical activity information. A major strength of self-reporting is cost efficiency<sup>60</sup>. It has been reported that self-reporting tools have fair to modest reliability and validity<sup>61, 64, 65</sup>, and good reliability and validity of participant's recall of the frequency and ranking of physical activity, but typically not the duration<sup>62, 66</sup>. Confounding issues around bias, over-reporting, under-reporting, giving socially desirable answers and ability to recall are limitations of self-reported assessment tools for physical activity and sedentary behaviour<sup>62, 66</sup>. Favourably, statistical adjustments may be used to remove these confounders and allow the usual behaviour levels to be estimated<sup>60</sup>.

Examples of self-reported assessment tools include log books, diaries and recall questionnaires. Log books and diaries ask participants to record details about behaviour, such as the frequency, intensity, duration, type and context of physical activity and sedentary behaviour each day. A major strength of a log book or diary is the cost efficiency<sup>60, 67</sup>; however, coding may be costly for large population studies<sup>61</sup> and some participants may find them burdensome to complete. Log books or diaries require short-term memory which improves the accuracy of self-reporting<sup>67</sup>. Recall questionnaires may be self- or interviewer-administered. Delivery options include face-to-face, telephone, mobile phone, online, computer, or on pen and paper.

Several self-reported assessment tools for large population studies are internationally compatible with versions for various age groups, such as the International Physical Activity Questionnaire (IPAQ) (http://www.ipaq.ki.se/). The IPAQ is for use with 15-69-year-olds. It is a recall questionnaire (for the past seven days) that has acceptable validity and reliability<sup>54, 68</sup>. It collects rich behavioural information on physical activity, including the frequency (days/wk), intensity (moderate and vigorous), duration (mins or h/occasion), type (including specific questions on walking and cycling) and domains of physical activity (domestic, transport, occupational and leisure-time). The IPAQ also includes some questions about sedentary behaviour. It has long, short, telephone- or self-administered versions. There has been a version developed specifically for adolescents (IPAQ-A) which additionally asks about physical activity at school<sup>69</sup>. A limitation of the IPAQ is that it does not find all health outcome associations previously reported using devices and underestimates some association strengths<sup>70</sup>.

#### 2.3.2 Physical activity and health

More evidence of the association between physical activity and health have been published in samples of adults and children, than adolescence. In Australian adolescents, youth obesity is at its highest rate<sup>71</sup> and type two diabetes mellitus is emerging at younger ages<sup>72</sup>. Physical activity is a modifiable risk factor for both of these conditions<sup>73</sup>. In a large study of healthy 4-18-year-olds (*n*=20 781), device-assessed moderate- to vigorous-intensity physical activity (MVPA) adjusted for sedentary time was negatively associated with waist circumference, systolic blood pressure, fasting insulin and fasting triglycerides, and positively associated with high-density lipoprotein cholesterol (HDL-C)<sup>74</sup>. Physical activity during adolescence in the settings of school<sup>75</sup> and active transport<sup>76, 77</sup> is associated with improved physical fitness and central fatness and notably, a large longitudinal study from the USA (*n*=1 718) estimated attribute risk and found obesity would decrease by 22% if 15-17-year-olds engaged in active transport  $\geq 4$  days/wk<sup>78</sup>.

Physical activity during childhood and adolescence in the form of sport participation is associated with improved psychological and social health, particularly from team sports compared to individual sports and in an amount that is higher than what leisure-time physical activity (LTPA) elucidates<sup>79</sup>. For example, a large cross-sectional study of 10-19-year-olds from Iceland (n=32 456) reported MVPA is
associated with less depressive symptoms, especially from organised sport among girls<sup>80</sup>. Lastly, it is estimated that overweight or obesity would decrease by 11% and obesity would decrease by 26% if older adolescents played on  $\geq$ 2 sports teams/year<sup>78</sup>.

Importantly, health during adolescence tracks into adulthood. Formative work by Berenson et al.<sup>81</sup> studied children and young adults (2-39-years-old) and reported that the presence of risk factors, such as body-mass index (BMI), elevated blood pressure, dyslipidaemia and triglycerides increased with age. Further, these were associated with evidence of early-onset coronary and aortic atherosclerosis. This implies that multiple risk factors in childhood and adolescence may increase risk of cardiovascular disease in adulthood<sup>81</sup>. A large study by Venn et al.<sup>82</sup> found overweight and obesity in 7-15-year-old Australians (*n*=4 571) predicted obesity in early adulthood. Other cardiovascular disease risk factors, such as elevated blood pressure and atherosclerosis development also track from youth to adulthood<sup>83, 84</sup>. Studies also show that adolescent physical activity is associated with adult health outcomes of HDL-C (positive), total serum cholesterol to HDL-C ratio (negative), the sum of four skinfolds (negative), and obesity (negative)<sup>85, 86</sup>.

Burden of disease is the loss in health from living with or dying early due to a disease or injury<sup>87</sup>. Worldwide, it is estimated that inadequate physical activity is associated with 9% of premature mortality worldwide and 6% of the burden of disease from coronary heart disease, 7% from type two diabetes mellitus, 10% from colon cancer, and 10% from breast cancer<sup>12</sup>. In Australia, the proportions are higher (11% from coronary heart disease, 19% from type two diabetes mellitus, 16% from bowel cancer, and 11% for breast cancer), with a total burden of disease contribution of 2.6% in 2011 that increases with age<sup>87</sup>. Inadequate physical activity is the second main, modifiable health behaviour risk factor (behind tobacco) contributing to the burden of disease, mainly via its association with coronary heart disease development<sup>87, 88</sup>. The total cost of inadequate physical activity in Australia in 2013 was estimated to be \$805 million (approximately 80% direct costs to health-care system and 20% productivity), excluding the costs from dementia and uterine cancers<sup>89</sup>.

### 2.3.3 Sedentary behaviour and health

Compared to physical activity, less evidence has accumulated to date that relates sedentary behaviour to health; however, the number of studies is growing. Currently, posture and energy expenditure are thought to be the casual mechanisms for the negative cardiometabolic outcomes of sedentary behaviour<sup>90</sup>, with genetic expression also suggested<sup>91-96</sup>. Additionally, engaging in sedentary behaviour during leisure time may displace more active and healthier pursuits, such as light-intensity physical activity (LPA)<sup>97</sup>. The way that sedentary behaviour is accumulated has been associated with health outcomes, with a study finding that participants had better health outcomes for waist circumference, BMI, triglycerides and 2h plasma glucose if sedentary behaviour was broken up compared to engaging in prolonged bouts, adjusting for total sedentary time and MVPA<sup>57</sup>. Of the studies relating sedentary behaviour to health, most focused on adults, and very few focus on total sedentary behaviour and instead include indicators, such as screen time, sedentary transport, reading, sitting at school or sedentary occupations<sup>98</sup>.

Some studies of sedentary behaviour and adverse health consequences, especially initially, have not accounted for physical activity to distinguish that health implications found are not caused by a lack of physical activity. Traditionally, studies used adjustments to control for physical activity but there is a recent shift to use compositional analyses, as our understanding develops of how physical activity and sedentary behaviour are on the same movement and energy spectrum and are interdependent. A recent, large dose-response and intensity-weighted time-use meta-analysis reported that the adult all-cause mortality risk threshold starts at seven uncompensated sedentary behaviour metabolic equivalent hours (USMh) for sitting (n=258 688) or 3USMh for television viewing (n=156 593)<sup>14</sup>. This study used data from six countries (China, Qatar, Australia, Denmark, USA and Spain) to consider the interdependence of sedentary behaviour and physical activity and argued that decreasing sitting time and increasing physical activity are equally important to offset all-cause mortality risk.

There is evidence that sedentary behaviour in adolescence is negatively associated with health outcomes with adjustment for physical activity in six studies<sup>99-104</sup>. Notably, two studies found sedentary behaviour was negatively associated with metabolic health<sup>99, 100</sup>. Firstly, a cross-sectional study of 13-17-year-olds (49% girls) found self-reported console game use positively associated with diastolic blood

pressure, mean arterial pressure, triglyceride and a score for cardiometabolic risk<sup>99</sup>. Secondly, in a study of children and adolescents (mean age 12-years-old; n=4581), accelerometer-measured sedentary time was positively associated with blood pressure and fasting blood glucose, insulin, triglycerides and HDL-C<sup>100</sup>. These associations have serious implications since cardiovascular disease risk factors present during adolescence track into adulthood<sup>84, 105, 106</sup>.

Additional evidence from studies that adjusted for physical activity includes two cross-sectional surveys of 11-15-year-olds in Europe and the USA that found screen time during adolescence was negatively associated with mental health, specifically self-image, life satisfaction, perceived health status and family relationships<sup>101, 102</sup>. A cross-sectional survey with a large sample of adolescents (n=1 234) and small samples of teachers (n=10) and parents (n=10) in Lima, Peru, found television viewing was negatively associated with self-rated health and academic performance<sup>103</sup>. A study of 16-year-olds from Iceland found that those who engaged in <5.3 h/day of self-reported screen time have a lower relative risk of life dissatisfaction, irrespective of frequency of participation in self-reported vigorousintensity physical activity (VPA) more (risk ratio, RR: 0.30; 95% CI: 0.15, 0.61; p < 0.005) or less (RR: 0.26; 95% CI: 0.08, 0.80; p < 0.05) than four times/wk<sup>104</sup>. Specific to adolescent girls, a systematic review found 16 additional studies that adjusted for physical activity and reported screen-based sedentary behaviour is positively associated with BMI, waist circumference, body fat, energy intake, soft drink consumption, depression, and musculoskeletal pain<sup>107</sup>.

In contrast, some positive benefits during adolescence from computer use, particularly on social and mental health have been found. A large study by Durkin and Barber<sup>108</sup> (n=1 304) reported that adolescents who played computer games, compared to adolescents who did not, had better interpersonal skills concerning engagement at school and relationships with family and friends, better intrapersonal skills (mental health and self-identity), less drug and alcohol use, and more involvement in sports and clubs outside of school. Further, screen-based sedentary behaviour is positively associated with family relationships in Western Europe and the quality of peer relationships in the USA and Europe<sup>102</sup>. However, evidence that sedentary behaviour has positive health outcomes is relatively rare. Future research is needed into the mechanisms as the findings may be related to specific aspects of computer use, such as restrictions on time or co-participation with parents.

### 2.3.4 Summary

Physical activity and sedentary behaviour are clearly defined and distinct health behaviours. There are many benefits to health from engaging in physical activity and negative consequences of performing inadequate physical activity and excessive sedentary behaviour. The negative consequences include significant economic burdens on health care costs and workplace productivity. This highlights the need to set recommendations and monitor how much physical activity and sedentary behaviour people engage in.

## 2.4 Guidelines

### 2.4.1 Physical Activity and Sedentary Behaviour Guidelines

The World Health Organisation, many governments and other peak organisations disseminate age-specific and evidence-based guidelines for physical activity and sedentary behaviour. The Australian Commonwealth Government's Physical Activity and Sedentary Behaviour Guidelines for adolescents<sup>109</sup> are generally consistent with guidelines internationally from the World Health Organisation<sup>110</sup> and governments in the UK<sup>111, 112</sup>, USA<sup>48</sup>, and Canada<sup>113, 114</sup>. It is currently recommended that 13-17-year-olds attain 60 mins or up to several hours of MVPA every day, and muscle and bone strengthening physical activity on  $\geq 3$  days/wk<sup>109</sup>.

Sedentary behaviour guidelines for adolescents advise to break up prolonged periods of sedentary behaviour at home, in school, at work, in the community and when travelling as often as possible and in as many ways as possible<sup>109</sup>. Limiting the use of electronic media for entertainment to  $\leq 2$  h/day is advised<sup>109</sup>. In the UK, sedentary behaviour guidelines for adolescents advise minimising the amount of time spent sitting for extended periods<sup>111</sup>. Canadian sedentary behaviour guidelines for adolescents advise to limit recreational screen-time to  $\leq 2$  h/day and to minimise prolonged sitting, time indoors during the day, and motorised transport<sup>113</sup>. The USA does not currently have sedentary behaviour guidelines for adolescents endorsed by the government; however the American Academy of Pediatrics has media and screen time guidelines<sup>115</sup>. These are similar to guidelines from Australia and the UK in that it is recommended to limit recreational screen time to <2 h/day, no screen time for those <2-years-old, no screen time in bedrooms, and parents are encouraged to monitor, co-participate, model, plan and set rules and boundaries for screen-time behaviours.

## 2.4.2 Compliance

In 2013, the Australian Bureau of Statistics released data on compliance with the Australian Physical Activity and Sedentary Behaviour Guidelines among adolescents and adults, derived from the large, cross-sectional Australian Health Survey<sup>15</sup> (2011-2012; n=31 837). The survey compared its findings to the previous version of the guidelines (1999), as opposed to the latest guidelines (2019). Self-reported assessment of physical activity using a 7-day recall survey was used, which is susceptible to over-reporting<sup>62, 66</sup>. Table 2.4 presents the findings, as well as those from other high-income countries.

**Table 2.4**Summary of proportion of populations who meet guidelines in varioushigh-income countries

	Age		Proportion of population meeting age-appropriate guideline
Country	(years)	Guideline	(%)
Australia <sup>15</sup>	15-17	≥60 mins/day of MVPA	5.8 (RSE: 18.5)
	15-17	<2 h/day of recreational screen time	18.5 (RSE: 12.5)
	18-24	150-300 mins/wk of	59 (males),
		MPA	48 (females)
Canada	12-16 <sup>116</sup>	<2 h/day of recreational	14 (male),
		screen time	18 (female)
	18-25 <sup>117</sup>	150 mins/wk of MPA	71.6
	18-19 <sup>118</sup>		66 during school
			41 post-school
UK <sup>119</sup>	'Youth': 2-18		30-40
$USA^{120}$	12-17	150 mins/wk of MPA	16.3
	18-24	and PA that strengthen	26.1
		muscles and bones on $\geq 2$	
		days/wk	

RSE: relative standard error.

Among Australian 15-17-year-olds, very few complied with the adolescent physical activity guideline and almost a fifth met the adolescent sedentary behaviour guideline<sup>15</sup>. The proportion of older adolescents who did not comply with both guidelines on all 7 days/week was 98.1%<sup>15</sup>. In 18-24-year-olds, two-thirds of males and almost a half of females achieved the Adult Physical Activity Guidelines from 1999 of 30 mins of moderate-intensity physical activity (MPA) on most days/wk<sup>15</sup>. As there was no specific duration limit for adult sedentary behaviour, there was no direct comparison available.

Consistent with Australia, <20% of Canadian adolescent girls and boys met the screen time guidelines of <2 h/day<sup>116</sup>. A considerably higher proportion of Canadian 18-25-year-olds met the physical activity guidelines (72%<sup>117</sup>) than Australian young adults. The difference is most likely due to the heterogeneity of measures, which impedes direct comparisons but still allows for evaluating trends. A narrative review by Ekelund et al. reported that a considerably higher proportion of youth in the UK (around a third) met the physical activity guidelines compared to Australian estimates, based on self-reported data<sup>119</sup>. In the 1999–2006 National Health and Nutrition Examination Survey in the USA, a higher proportion of a broad group of adolescents met the aerobic guideline only (14.7%) and the strengthening guideline only (21.3%) than in Australia<sup>120</sup>. Higher compliance than Australia was also found in young adults, as 52.4% met the aerobic activity guideline (similar to Australia) and 29.3% met the strengthening guideline<sup>120, 121</sup>.

## 2.4.3 Summary

The Australian Guidelines are consistent with those in other developed countries. They include the frequency, intensity, duration, and type of physical activity required to achieve health benefits and prevent chronic diseases, and recommendations about breaking up prolonged sedentary behaviour and reducing sedentary behaviour. Compliance is very low among Australian older adolescents and young adults and was lower than other high-income countries including the Canada, UK and USA.

# 2.5 Changes in physical activity and sedentary behaviour in late adolescence

This section reviews studies that assess the type and volume of physical activity and sedentary behaviour in late adolescence via cross-sectional studies that contain participants of various ages or via longitudinal studies that follow-up a cohort over time. These studies are required to understand if and by how much physical activity and sedentary behaviour changes during the transition out of secondary school.

## 2.5.1 Cross-sectional studies

A narrative review by Sallis<sup>122</sup>, summarised the evidence about age-relate declines in physical activity, and highlighted that the steepest declines were seen during adolescence. Another study confirmed age-related declines in physical activity, where young adults performed less physical activity than older adolescents<sup>123</sup>. This was a cross-sectional study of various age groups by Caspersen et al. and reported that 15-year-olds had higher LTPA compared to 18-year-olds, with large to very large (3-8%) annual declines. LTPA was similar for 18- and 21-year-olds suggesting that over these few years, LTPA amounts may have declined but then stabilised. Gender differences were noted, as men tended to engage in more physical activity than women. However, in addition to being cross-sectional, this study was based on a non-validated self-reported survey instrument, considered physical activity during leisure time only, and had incongruent data sources, as data collection assessment methods used for adolescence and adulthood physical activity changed between 15-21-years-old.

Other cross-sectional studies reported data on mean duration of physical activity and sedentary behaviour for older adolescents and young adults (18-24-year-olds) from other high-income countries. Cross-sectional data from the large Australian Health Survey  $2011-2012^{15}$  (*n*=31 837) showed that as age increased in groups of 15-17-year-olds vs 18-24-year-olds, self-reported MVPA was lower (62 vs 39 mins/day) and sedentary behaviour was higher (3 vs 5.5 h/day). This was consistent with previous studies over time<sup>122, 123</sup>. A European study<sup>124</sup> used devices and reported that adolescents spent more time than Australians engaging in sedentary behaviour (9 h/day) and similar time for MVPA (55 mins/day). A Canadian retrospective recall study<sup>118</sup> of 18-19-year-olds reported that the proportion of students who complied

with the recommended physical activity guidelines 2-months into the first year of university was lower than the proportion who complied in the final 2-months of secondary school, and VPA was also lower. Few qualitative studies in this area have been published. One such small study sampled first-year university students in the USA (n=19) and reported that those who used to be involved in sport during secondary school had an absence of regular physical activity after leaving school<sup>125</sup>.

Cross-sectional studies are research projects that collect data from participants at one point in time; in effect, taking a snapshot of the participants' behaviours. Cross-sectional studies may not accurately reveal individual variations in physical activity and sedentary behaviour over time but do allow for between-group comparisons<sup>123</sup>. Strengths of cross-sectional study design are the ability to collect data from a large sample size containing a broad range of ages in a brief period of time. Further, the findings are more likely to have external validity and study drop out and attrition are not a concern<sup>123</sup>. Notably, cross-sectional studies do not follow the same cohort of people over time so cannot describe adequately or examine in-depth age-related behaviour change<sup>126, 127</sup>.

## 2.5.2 Longitudinal studies

Longitudinal studies follow a cohort of participants over time, taking periodic or prepost measurements. Longitudinal designs allow for the examination of how behaviours track, which involves monitoring the group mean, or whether an individual's rank or position within a cohort is maintained or changed over time<sup>128</sup>. Tracking correlation coefficients have been categorised as small (0.10-0.29), moderate (0.30-0.49) or large ( $\geq 0.5$ )<sup>129</sup>. A recent systematic literature review of physical activity and sedentary behaviour during the transition from adolescence to early adulthood reported low to moderate tracking of behaviours, physical activity generally decreased and sedentary behaviour usually increased<sup>130</sup>. It also reported few studies assessed sedentary behaviour, most studies used self-reported measures and studies often categorised behaviours differently<sup>130</sup>.

## Physical activity

In adolescence, physical activity decreases on average 7% per year, according to a systematic review and pooled analysis of 26 international longitudinal studies of 10-19-year-olds<sup>17</sup>. Appendix 2.1 presents a summary of prospective longitudinal studies published since 1994 that reported the tracking of and/or changes in physical activity as children and adolescents transition into adulthood. In summary, 15 prospective longitudinal studies from seven countries assessed physical activity across the transition; specifically, Australia, Canada, Finland, Netherlands, Northern Ireland, Norway and Sweden. Most studies did not additionally assess sedentary behaviour. The sample sizes ranged from 105 to 7 794, age at baseline ranged from 10-18-years-old, and the duration of follow-up intervals ranged from 7 to 25 years. There was a pattern for weak or fair tracking of physical activity<sup>106, 131-133</sup>, with stronger tracking in males than females<sup>106, 131</sup>. Most studies demonstrated a general trend for decreasing physical activity over time<sup>85, 134, 135</sup>.

Some longitudinal studies noted intra-individual seasonal variances<sup>106, 131, 132</sup> but not all studies stated whether adjustment was made for this confounder. The studies tended to begin in mid-late adolescence and have a long interval to follow-up (up to 25 years). This large follow-up interval included other age-related transitions that might mask the effects of the specific transition out of secondary school and would not capture short-term and interim changes in behaviour. With the exception of one<sup>136</sup>, most studies were limited by the use of self-report assessment tools and few<sup>137</sup> reported the reliability and validity of the assessment tool used. Direct comparison of data was impeded by studies analysing data differently and therefore reporting correlation coefficients, odds ratios and/or mean change; however, general trends were able to be described and physical activity mostly declined over time.

Reporting correlation coefficients is a limitation, as not all of the studies reported that they considered common confounding variables of weather, smoking, area-level socio-economic position (SEP), individual-level SEP (maternal and paternal education), remoteness, or English as the primary language spoken at home. Furthermore, correlation coefficients report the strength of association and comparative within-group rank, which could distort the variation in behaviour over time<sup>127</sup>.

## Sedentary behaviour

Compared to physical activity, fewer prospective longitudinal studies report the tracking of, and changes in, sedentary behaviour as adolescents leave secondary school. A study that followed Estonian 9-year-olds for 9-10 years and Swedish 15year-olds for six years and used devices (accelerometers) found no change in sedentary time from adolescence to early adulthood<sup>138</sup>. A New Zealand birth cohort study<sup>139</sup> followed 1 013 individuals until 26-years-old, assessing time spent watching television by self- and proxy-report (parent). A daily mean±SD of television time at 15-years-old was 3.58±1.79h for males, and 3.19±1.71h for females. At 21-years-old, there was a decrease in both sexes, to 3.07±1.73h for males and 3.07±1.79h for females. Television time was correlated between 15-21-year-olds (r=0.42, p<0.001). Using the same cohort, Landhuis et al.<sup>140</sup> reported a decrease in the mean weekday television time of 5-15-year-olds from 2.3±0.9h to 1.9±1.4h for 32-year-olds (r=0.33, p < 0.001). In contrast, two longitudinal studies reported on screen-time more broadly and found increases from adolescence into adulthood. A seven-year follow-up study of 11-21-year-olds (at baseline) from the USA found the proportion who achieved ≤14h/wk of screen-time decreased by17%<sup>141</sup>. Lastly, a 10-year follow-up study of 10-year-old Belgians found the proportion who did >2h/day of screen-time increased on weekdays (12 to 51%) and weekend days (41 to 64%)<sup>142</sup>.

Limitations of longitudinal studies of sedentary behaviour include a lack of diversity of type of sedentary behaviour assessed, since watching television may be overshadowed now by concurrent sedentary behaviour and prevalent use of recreational screen time on portable electronic devices. More longitudinal studies of sedentary behaviour in adolescence are needed that have varied countries of origin, smaller intervals of assessment rather than large intervals that may have masked changes within assessment periods, and contemporary measures of youth sedentary behaviour.

## 2.5.3 Summary

Cross-sectional studies reported differences in average volume between age groups that suggest lower physical activity and higher sedentary behaviour with age, with young adults spending more time in sedentary behaviour and almost half the amount of time in physical activity per day compared to older adolescents. Longitudinal studies have reported poor or fair tracking of physical activity from adolescence into adulthood, and a decline in the volume of physical activity. There is a lack of prospective longitudinal studies that observe sedentary behaviour or the specific years during the transition out of secondary school. A better understanding of the tracking and changes in physical activity and sedentary behaviour is needed, as well as what the underlying influences are on these behaviour changes.

# 2.6 Explaining physical activity and sedentary behaviour in late adolescence

To understand physical activity and sedentary behaviour during the transition out of secondary school, theories and models can be tested and various types of cross-sectional and longitudinal associations may be examined.

## 2.6.1 Theories and models of behaviour

Theoretical and conceptual frameworks and models from various fields of study may help understand health behaviours and why they change<sup>143</sup>. Frameworks and models may be used to guide research development by identifying key areas, such as particular settings, correlates, determinants and moderators of health behaviours<sup>143</sup>. Different theories and models often have overlapping constructs<sup>144</sup>. However, in regard to health behaviour, constructs may be separated into choice-driven or choice-enabling, such as intrapersonal theories and macro-environment models<sup>144</sup>. Theories have evolved from being based on an individual, choices and preferences that were limited in scope and did not consider external influences. There are now comprehensive theories and models with multiple constructs. Two such theories and models that may be applied to understand health behaviour in adolescence are outlined below.

The ecological model (Figure 2.1) may be applied to health behaviours and has separate categories of individual, social and environmental<sup>20, 21</sup>. This model identifies organisations, communities, and public policy as potential influences on health behaviours. It is widely used, especially in health behaviour research, due to the extensive range of influences on behaviour. This model is inter-disciplinary in its formation because it encompasses theories from multiple fields of sociology, psychology and public health. It hypothesises a reciprocal relationship between influences across the levels. Although, the specific relationship between the constructs are not well addressed within the model<sup>20</sup>, it may provide a framework for research whereby influences on behaviour change from multiple levels would be examined.

**Figure 2.1** The ecological model's categories of influences on behaviour. *Adapted from Owen et al.*<sup>21</sup>, 2011



While the ecological model is all-inclusive, there are some models that incorporate broad, systemic influences for groups or life stages that include more specificity. For example, the life transition model (Figure 2.2) was developed to assist public health nurses in understanding and influencing women and children's health and health behaviours over life transitions<sup>18</sup>. This model highlights contributing influences during a transition while also suggesting that individual and environmental constructs may impede the quality of the transition and the adoption of new health-promoting behaviours. It lists examples of constructs as obesity, race, age, mood, relationships, support, physical facilities and transportation. Kaiser et al.<sup>18</sup> defines a transition as an experience containing a period of time, change, variability and

uncertainty. Transitions are a disturbance, interruption and a shift from one stable period of time to another, where the individual has to adapt to changes. Some individuals have better outcomes post-transition than others. A goal of health providers is for the uptake of a healthier choice of behaviour to be adopted prior to or over a transition and then maintained. This is one of the few models that is specific to transitions and provides a useful framework for understanding behaviour changes during the transition out of secondary school.

## Figure 2.2 Life transition model. Adapted from Kaiser et al.<sup>18</sup>, 2009



## 2.6.2 Key correlates of physical activity and sedentary behaviour

In this section, the key correlates of physical activity and sedentary behaviour in late adolescence are categorised according to the ecological model<sup>20, 21</sup> and a summary is presented in Table 2.5. The majority of studies presented in Table 2.5 are quantitative as few qualitative studies in this area have been published. Two such qualitative studies sampled older adolescents but were not focussed on the transition out of secondary school specifically<sup>145, 146</sup>.

Category	Physical activity	Sedentary behaviour
Individual	Sex (+; male) <sup>147</sup>	Age $(+)^{148}$
	Perceived competence $(+)^{147}$	Ethnicity (+; non-white) <sup>148, 149</sup>
	Physical self-perceptions $(+)^{145, 150}$	Parental education (-) <sup>149</sup>
	SEP $(+)^{151}$	SEP (-) <sup>148, 149</sup>
	PA enjoyment $(+)^{146}$	
	Lack of time and technology (-) <sup>145</sup>	
	Desire to achieve academically (-) <sup>146</sup>	
Social	Support from parents and family (+) <sup>147</sup>	Parental modelling (+) <sup>148</sup>
	Teachers, peers and	Parental restriction (+;
	competition (-) <sup>145</sup>	enforcement of rules and limits) <sup>148</sup>
		Parental discouragement (-) <sup>148</sup>
Environmental	Land-use mix $(+)^{152}$	Cul-de-sac density $(+)^{153}$
	Residential density (-) <sup>152</sup>	TV in bedroom $(+)^{154}$
	Presence of high-quality sport facilities (+) <sup>155</sup>	TVs and computers at home $(+)^{148}$
	Shop walkability $(+)^{155}$	Reported crime $(+)^{153}$
	Sport facilities at school (+) <sup>155</sup>	Avoidance of risk (+; outdoor safety concerns) <sup>148</sup>
	Distance to public recreational	Parental perception of good
	facilities from home (-) <sup>145, 155</sup>	neighbourhood sport facilities
	Access to shops and the size of shops (+; males) <sup>155</sup>	(-) <sup>153</sup>
	PA equipment at home	
	(+; white females) <sup>155</sup>	
	Cost of using facilities (-) <sup>145</sup>	

**Table 2.5**Summary of key correlates of physical activity and sedentarybehaviour in late adolescence

## Individual correlates

Individual correlates may include demographic, biological, psychosocial and behavioural characteristics. A systematic review by Bauman et al.<sup>147</sup> included three systematic reviews and identified sex (male) and perceived competence as the key correlates of higher physical activity at the individual level in 10-18-year-olds. Additionally, it was reported that BMI or anthropometry were examined but there was consistently no association with physical activity in adolescence, and that there was inconclusive evidence that ethnicity (white), self-efficacy, barriers to physical activity and smoking were correlates of physical activity in adolescence. Since those systematic reviews, there has been some original research that has reported physical self-perceptions<sup>150</sup> and SEP<sup>150, 151</sup> as individual correlates of physical activity in 12-17-year-olds.

Individual correlates of sedentary behaviour were reported by a systematic review of 2-18-year-olds<sup>148</sup> and one of 13-18-year-olds<sup>149</sup>. Key individual correlates of sedentary behaviour in adolescence were age (older), ethnicity (non-white), parental education and SEP. No association was consistently found for birth weight and sedentary behaviour. Mixed findings were found for the association of BMI and sex with sedentary behaviour.

### Social correlates

The social environment encompasses who an individual resides with and spends time with, including parents, friends, siblings, and peers<sup>156</sup>. A systematic review by Bauman et al.<sup>147</sup> included four systematic reviews on adolescence (10-18-year-olds) and identified support from parents and family for physical activity as the key social correlate of higher physical activity. Additionally, parental activity and perceived parental role models were examined but there was consistently no association with physical activity<sup>147</sup>. Additionally, a systematic review by Pate et al.<sup>148</sup> reported correlates of sedentary behaviour in children and adolescents (2-18-year-olds) were parental modelling, avoidance of risk (outdoor safety concerns), parental restriction (enforcement of rules and limits) and parental discouragement. There was consistently no association between parental concerns for excess television and sedentary behaviour<sup>148</sup>.

## **Environmental correlates**

The environment is also referred to as the physical environment, built environment (man-made structures), natural environment, domain, setting, or context. Environments that physical activity and sedentary behaviour takes place in include home, school, work, and neighbourhood<sup>21, 157</sup>. Both perceived and objective characteristics of the environment are useful to understanding behaviour, which is especially true for parental perceptions of the household environment, due to parent's ability to impose boundaries on adolescents<sup>158</sup>. Systematic reviews by Ding et al.<sup>152</sup> and Davison and Lawson<sup>155</sup> found environmental correlates of higher physical activity in adolescence (12-18-years-old) were land-use mix, residential density, sport facilities, the ability to walk to shops, sport facilities at school, short distance to public recreational facilities from home, access to shops and the size of shops (males), physical activity equipment at home (white females), and the number of recreation facilities and parks close to home (females). There was consistently no association between perceived presence of paths or street connectivity, and greater access to local shops (females) and physical activity in adolescence. Inconsistent evidence was found for access to parks, recreation facilities, street connectivity, social incivilities, walking and biking facilities, traffic speed and volume, and unspecified traffic safety.

There are limited data on environmental correlates of sedentary behaviour in adolescence. Some studies have reported that environmental correlates of higher sedentary behaviour are the number of televisions and computers per household, having a television in the bedroom, cul-de-sac density, reported crime, and poor parental perception of neighbourhood sport facilities<sup>148, 153, 154</sup>. Some of these studies had wide ranges that included children so may not reflect unique correlates of older adolescents.

## 2.6.3 Determinants of physical activity and sedentary behaviour

For the purpose of this review, determinants have been categorised according to the ecological model<sup>20, 21</sup>. No studies were found that assessed individual, social and environmental determinants of both physical activity and sedentary behaviour in the one study. Studies used samples that also included childhood and early and mid-adolescence. Table 2.6 summarises findings and highlights the higher amount of studies that focus on a wider variety of determinants of physical activity compared to sedentary behaviour.

Category	Physical activity	Sedentary behaviour
Individual	VPA planning (+) <sup>159</sup>	Age of puberty onset
	Intentions $(+)^{161}$	$(-)^{160}$
	Planning $(+)^{161}$	
	Self-efficacy $(+)^{161}$	
Social	Perception of parental modelling and	Perception of maternal
	physical activity (+; female) <sup>162</sup>	encouragement to be active (-; female) <sup>163</sup>
	Family support (+) <sup>159, 163</sup>	
	Perception of peer physical activity	
	levels $(+)^{159}$	
	Perception of paternal care for fitness	
	$(+; male)^{163}$	
Environmental	Number of gymnasium facilities	Television in
	$(+; males)^{164}$	adolescent's bedroom
	Crime (-) <sup>164</sup>	$(+)^{165}$
	Landscape diversity (-; in those who	
	did not move to a new house) <sup>164</sup>	

**Table 2.6**Summary of determinants of physical activity and sedentarybehaviour in late adolescence

## Individual determinants

One study<sup>159</sup> assessed individual determinants of changes in meeting physical activity guidelines during the transition from late adolescence to early adulthood and found planning of VPA was important. Another study reported intentions, planning and self-efficacy were individual determinants of higher physical activity<sup>161</sup>. Limitations of this study were a wide age range of the sample that included younger adolescents that might have masked variances in determinants in late adolescence and a short follow-up of <3-months<sup>161</sup>.

While no studies were found that considered the individual determinants of sedentary behaviour during the transition out of secondary school, one study had a sample that included older adolescents<sup>160</sup>. It reported age of puberty onset was an individual determinant of higher sedentary behaviour. Limitations of this study were the wide age range of the sample that included younger adolescents which might mask variances in determinants in late adolescence, and limited range of sedentary behaviour considered: only television, videos or video games on the computer on school days.

## Social determinants

Three studies have reported social determinants of physical activity, one of which also measured sedentary behaviour, and had a sample that included older adolescents and young adults<sup>159, 162, 163</sup>. Social determinants of higher physical activity were child perceptions of parental modelling (female)<sup>162</sup>, parental physical activity (female)<sup>162</sup>, peer physical activity<sup>159</sup> and paternal care for fitness (male)<sup>163</sup>. Perceptions of maternal encouragement to be active (female) was a social determinant of lower sedentary behaviour<sup>163</sup>. Limitations of these studies included a wide age range of the sample as one study also included children<sup>162</sup> that might mask variances in determinants during the transition from late adolescence into early adulthood. Additionally, most had long follow-up periods that might miss behaviour change due to the specific transition out of secondary school<sup>162, 163</sup> and only a small range of social variables were considered.

## Environmental determinants

There is a lack of studies on the environmental determinants of physical activity and sedentary behaviour across all ages groups<sup>166</sup>. No studies have considered the environmental determinants during the transition out of secondary school of changes in sedentary behaviour nor changes in both physical activity and sedentary behaviour. One study followed-up 12-15-year-olds and reported that the presence of a television in the adolescent's bedroom was a determinant of higher self-reported television viewing during leisure-time over two years<sup>165</sup>. Another study used National Longitudinal Study of Adolescent Health data from the USA and found determinants of higher physical activity were the number of gymnasium facilities (males), low crime and low landscape diversity in the neighbourhood (in those who did not move out of the family home)<sup>164</sup>. Limitations of these data were the long seven-year follow-up period, only the neighbourhood environment was considered (physical attributes of university campus residences, schools, homes or workplaces were excluded), only one aspect of physical activity was considered (leisure-time frequency) and the baseline sample included secondary students as well as some who had already transitioned out of secondary school.

## **2.6.4** Situational transitions and changes in physical activity and sedentary behaviour

The transition out of secondary school has been highlighted as important, although not well understood or well-studied<sup>167</sup>. This is a normative transition and coincides with multiple simultaneous situational transitions that occur over a short time period<sup>168</sup>. Possible consequences of situational transitions include new support bases (both social and financial); altered access to neighbourhood facilities; reformed social environments, such as meeting colleagues at a new job, meeting peers during tertiary studies, and a shift in friendship groups away from secondary school friends; altering relationships, such as starting a relationship with a significant other or separating; and, uncommonly, changing physical status due to an illness, injury, or a disability<sup>9, 10</sup>.

A systematic review by Allender et al.<sup>10</sup> reported the associations between situational transitions and changes in physical activity from 19 observational studies that were mainly cross-sectional and used self-reported recall assessment tools. Allender et al. reported that commencing paid work, having an illness in childhood, changing place of residence and unemployment was associated with lower physical activity. Further, those who started work earlier (at 18-years-old) were less likely to be as physically active than those who started work later (at 25-years-old). A large four-year longitudinal study by Brown and Trost<sup>9</sup> similarly examined associations between the situational transition of entering the workforce on physical activity among 18-23year-old Australian women (n=7281) at baseline. Women had 15% higher odds of not complying with physical activity guidelines at follow-up if employed. In contrast, those experiencing the situational transition of returning to study or changing work hours had 18-23% lower odds of not complying with physical activity guidelines at follow-up. In comparison to physical activity, there is a dearth of studies of the associations between situational transition and sedentary behaviour, television viewing, sitting, screen time, or electronic media use.

## Commencing tertiary studies

A narrative review by Nelson et al.<sup>169</sup> reported the transition from secondary school to tertiary study was a unique developmental stage. Dramatic changes in lifestyle were found, weight gain was highly prevalent, and the proportion meeting physical activity guidelines dropped from 34% to 13%. Nelson et al. found that most commencing tertiary students had declines in aerobic fitness, which were unlikely to reverse with age. Nelson et al. also reported higher levels of watching television, studying or using a computer for  $\geq$ 30 h/wk. Few qualitative studies in this area have been published. One such study used focus groups of first-year university students from the USA and found that those who described being involved in sport during secondary school perceived an absence of regular physical activity after leaving school<sup>125</sup>.

Of the few quantitative studies assessing change in physical activity and sedentary behaviour during the transition from secondary school to tertiary study, directions of change were mixed. However, decreasing television viewing was a consistent finding. A 7-month follow-up study by Edmonds et al.<sup>170</sup> reported 17-20-year-old Canadian women in the first year of studying decreased self-reported television use<sup>170</sup>. An 8-month follow-up study by Pullman et al.<sup>171</sup> reported 17-20-year-old Canadian men did not change slow aerobic physical activity, strength training and flexibility, but increased computer use and study time, and decreased television use<sup>171</sup>. A two-year follow-up study reported 18-20-year-old Americans self-reported no change in total exercise, decreased aerobic exercise and increased stretching<sup>172</sup>. Lastly, a cross-sectional retrospective study found 18-19-year-old Canadians self-reported a decrease in VPA and compliance with physical activity guidelines<sup>118</sup>.

### Simultaneously commencing tertiary studies and moving out of home

Quantitative studies in the USA and Canada<sup>170, 171, 173</sup> have researched associations between simultaneously commencing tertiary education and moving out of the family home into on-campus student residences and changes in physical activity and sedentary behaviour. Butler et al.<sup>173</sup> surveyed females in the USA at the start of the first semester of university and 5-months later and reported a decline in self-reported total physical activity. The study by Pullman et al.<sup>171</sup> surveyed males in Canada in the holidays prior to commencing university, at the end of the first semester (2-5months later) and at the end of the second semester (6-8-months later). At the final follow-up, computer use and time studying increased, fast aerobic physical activity and television time decreased and no change was seen in slow aerobic physical activity, strength training, and flexibility training. From the same university in Canada, the study of first year women by Edmonds et al.<sup>170</sup> collected survey data in the holidays prior to commencing university, 3-months later and 7-months later<sup>170</sup>. VPA and computer use increased, MPA and watching television declined, and strength-building physical activity did not change.

In their qualitative study in the USA, Cluskey and Grobe<sup>125</sup> explored students' experiences using focus groups at the end of the first year of university. Students described struggling to adapt previous physical activity into a new routine and perceived that physical activity decreased. Students described a change in social roles and responsibilities, modelled the behaviour of new roommates, and were influenced by social norms. Participants described that lack of experience, changes in routines and the absence of family support influenced poor choices regarding physical activity. A common limitation of the above four studies<sup>125, 170, 171, 173</sup> is that physical activity and sedentary behaviour were not assessed before the transition out of secondary school.

## **2.6.5** Moderators of associations between situational transitions and physical activity and sedentary behaviour

The following section will review how moderators may help to understand how situational transitions and influences work together to understand physical activity and sedentary behaviour during the transition out of secondary school.

## Conceptualising moderating variables

Three ways that moderators vary the effect of an independent variable on a dependent variable are linear (positive or negative association), quadratic (the association exponentially increase or decrease) and step<sup>5</sup>. The step method of variance is when a moderator partitions the independent variable's effect size on the dependent variable into groups<sup>5</sup>. In this case, moderation has occurred if the strength (Figure 2.3) or direction (Figure 2.4) of the association between a situational transition and physical activity or sedentary behaviour is dissimilar for a dichotomised moderator.





**Figure 2.4** The step method with a dichotomised moderator altering the direction of the association between the independent variable (situational transition) and the dependent variable (physical activity/sedentary behaviour). *Adapted from Bauman et al.*<sup>2</sup>, 2002



The step method of moderation will be tested in this thesis because it clearly identifies whether high or low levels of influences impact behaviour over time in atrisk or protected population groups. The findings in relation to effect modifiers can inform interventions as it helps to pinpoint what to target for different people. The findings elucidate what adolescents need to be equipped with prior to the transition out of secondary school to protect against lower physical activity and higher sedentary behaviour associated with specific situational transitions.

Conceptually, the identification of protective attributes aligns with a resilience model of health promotion<sup>174</sup>. Resilience is defined as a being able to constructively progress, even though substantial challenges are present<sup>175</sup>. The transition out of secondary school may pose substantial challenges since adolescents typically experience multiple simultaneous changes in circumstance. Determining what promotes resilience is important for informing interventions<sup>174</sup>. Implementing

strategies for specific target groups or tailoring techniques to a group or individual risk profile<sup>176</sup> is an evidence-based way to maximise the efficacy of a health behaviour change intervention and has been used in interventions aimed at diet, smoking, alcohol consumption and breast cancer screening<sup>177, 178</sup>.

### **Moderators**

A two-year study from Australia reported moderators for the associations between situational transitions and changes in LTPA during the transition of leaving secondary school<sup>22</sup>. It reported those who did not belong to a sporting club in Year 12 decreased LTPA over time more if they worked full-time post-school compared to those not working full-time. As belonging to a sporting club may be, arguably, part of the outcome, future research needs to examine whether there are other moderators for physical activity. There is a research gap for longitudinal studies examining individual, social or environmental moderators of the association between situational transitions and sedentary behaviour during the transition out of secondary school. Because of the lack of studies in the older adolescent population group, studies of other population groups are reviewed next to inform potential moderators of physical activity.

In other population groups, moderators of physical activity include social support from friends and family in Hispanic women<sup>179</sup>, habits moderating the association between intention and total MVPA in older adults<sup>180</sup>, weight status moderating the association between motivation and physical activity in disadvantaged youth<sup>181</sup>, and retirement status moderating the association between park proximity to home and recreational walking in mid-older aged adults<sup>182</sup>. Comparatively, moderators of changes in sedentary behaviour in any population group includes positive communication with parents in African American adolescents<sup>183</sup>, and urban location moderated the association between co-participation in physical activity with family and sedentary behaviour in children (5-12-years-old)<sup>184</sup>. Variables examined were categorised according to categories of the ecological model<sup>20, 21</sup> of individual<sup>180, 181</sup>, social<sup>179, 183, 184</sup> and environmental<sup>179, 181, 184</sup>. These findings provide an idea of potential moderators during the transition out of secondary school to investigate. As a comparison, in other fields of study, such as psychology, common moderators to situational transitions or stressful life events include self-efficacy and competence<sup>185</sup>.

### 2.6.6 Summary

This section highlighted how theories and models can be applied and tested to explain behaviour, including the ecological model<sup>20, 21</sup> and the life transition model<sup>18</sup>. There are also various types of associations that can partially explain behaviour including correlates, determinants and moderators. Individual, social and environmental correlates of physical activity and sedentary behaviour in adolescence include SEP, parental support, neighbourhood sport facilities and neighbourhood walkability. Individual, social and environmental longitudinal determinants of physical activity and sedentary behaviour include SEP, perception of parental support and the number of sport facilities in the neighbourhood. Situational transitions that appear to influence physical activity include commencing work and changing place of residence, while commencing tertiary study appears to influence physical activity and sedentary behaviour. Limited studies have examined moderators of associations between situational transitions and physical activity or sedentary behaviour during the transition from late adolescence out of secondary school.

A limitation of the body of correlates, determinant and moderator research is that most of the studies incorporated early, mid- and late adolescence samples, making it difficult to understand why physical activity and sedentary behaviour change from late adolescence during the transition out of secondary school. Additionally, previous research has not comprehensively examined a range of modifiable individual, social and environmental influences, meaning there may be undiscovered unique explanations for physical activity and sedentary behaviour during the transition out of secondary school. Further, there are few longitudinal studies whereby physical activity and sedentary behaviour were assessed prospectively from before leaving secondary school, with most asking participants to reflect and rely on memory (retrospective). Therefore, there is a need to examine the individual, social and environmental correlates, determinants and moderators of physical activity and sedentary behaviour during the transition out of secondary school. Prospective research is also needed to determine effects on physical activity and sedentary behaviour of different simultaneous situational transitions, such as changes in study, employment, and place of residence.

## 2.7 Conclusion and thesis aims

This chapter presented a narrative literature review on physical activity and sedentary behaviour in late adolescence and the underlying influences. Late adolescence is regarded as a time of great change and behavioural instability. Simultaneous life changes occur, such as an increased social life, more independence, economic growth, and physical development. Because inadequate physical activity and excessive sedentary behaviour are associated with negative health outcomes in adolescence and adulthood, more research is needed into understanding these behaviours during the transition out of secondary school. Cross-sectional studies have reported that as age increases during the transition out of secondary school, physical activity is lower and sedentary behaviour is higher. Longitudinal studies have observed similar findings but sometimes assessed change over large intervals that missed the specific transition out of secondary school. Very little is understood about sedentary behaviour trajectories. There is a need to better characterise changes in physical activity and sedentary behaviour, as the transition out of secondary school may be a key time to intervene to influence adult health and the associated costs to the public health system and burden of chronic disease.

While there are some known correlates of physical activity and sedentary behaviour in adolescence and adulthood, correlates in late adolescence may be unique. Transitioning out of secondary school is a life event with multiple, simultaneous and rapid changes that may stabilise or destabilise health behaviours. Situational transitions of commencing tertiary education or full-time employment may be associated with changes in physical activity and sedentary behaviour during the transition, but little research has been done on this so far. Further, there are limited longitudinal data on determinants of physical activity and sedentary behaviour during the transition out of secondary school, and while identifying moderators of these relationships is important to inform interventions, few studies have examined individual, social and environmental attributes that may moderate these associations. Findings will inform the development of tailored strategies in school settings or within interventions that aim to prevent declines in physical activity and increases in sedentary behaviour and improve adolescent and adult health. This thesis' purpose is to understand physical activity and sedentary behaviour during the transition out of secondary school by addressing the research gaps highlighted in this chapter. To achieve this, a mixed methods approach will be utilised. Specifically, this thesis aims to:

- 1. Examine how physical activity and sedentary behaviour change during the transition out of secondary school; and
- 2. Examine individual, social and environmental influences on physical activity and sedentary behaviour during the transition out of secondary school.

Chapter 3

A qualitative exploration of influences on physical activity and sedentary behaviour during the transition out of secondary school

## 3.1 Introduction

hysical activity and sedentary behaviour are important health behaviours, as argued in the previous chapter; however, 15-17-year-old Australians engage in low physical activity, excessive sedentary behaviour and do not meet recommendations<sup>15</sup>. Most longitudinal studies have demonstrated a trend for physical activity to decrease over time and many studies report low physical activity tracks from childhood/adolescence into adulthood<sup>130, 186</sup>. These trends are concerning and warrant exploration into why they occur, especially since physical activity and sedentary behaviour during adolescence are early-life exposures that may partially explain some health outcomes in adulthood<sup>16, 85, 86</sup>.

Leaving secondary school and entering into adulthood is considered a critical time for growth and development<sup>23</sup>. Changes in physical activity over time may be influenced by situational transitions<sup>8, 10</sup>. Examples of situational transitions are leaving school, travelling, changing place of residence by moving out of the family home for the first time, commencing casual, part-time or full-time employment, changing relationships, and commencing further education. For each of these, various personal and social contextual factors may exist in alternate ways and have varying impact on the amount physical activity and sedentary behaviour that individuals engage in.

Very little research has been conducted about the transition out of secondary school and looking to quantitative research provides limited clarity. A two-year follow-up study of final-year Flemish students reported household composition moderated associations between changes in individual variables (self-efficacy and health-related barriers) and a decrease in leisure-time sports<sup>187</sup>. Other quantitative studies<sup>160-165, 187-</sup> <sup>189</sup> have measured influences on changes in physical activity and sedentary behaviour during the transition out of secondary school; however, they lacked depth and a detailed understanding of the context of physical activity and sedentary behaviour; therefore, qualitative exploration is needed to gain further insight.

Few qualitative studies in this area have been published<sup>125, 190-192</sup>. These studies mainly explored the transition to tertiary education<sup>125, 190, 191</sup> or to work<sup>192</sup>. As such, they are limited in their ability to provide insights regarding influences on changes in physical activity and sedentary behaviour for other situational transitions, such as changing place of residence, travelling or commencing further study while working.

The studies by Cluskey and Grobe<sup>125</sup> and Nelson et al.<sup>190</sup> focussed on weight gain and physical activity, but not sedentary behaviour. The study by Deliens et al.<sup>191</sup> framed the study around behaviours related to weight and eating but also provided insights in physical activity and sedentary behaviour. The study by Koehn et al.<sup>192</sup> explored influences on changes in physical activity during the transition out of school into work but excluded sedentary behaviour and other situational transitions of further education and travelling. Also, most of those (88%) participants had left school only 2-months ago and this brief window may not reflect enough time for post-school changes in physical activity to occur. Other qualitative studies of influences on physical activity have only sampled older adolescents<sup>145, 146, 193</sup> or young adults<sup>194</sup> and have not focussed on the transition out of secondary school or explored sedentary behaviour.

In summary, the few qualitative studies that explored influences on physical activity and sedentary behaviour during the transition out of secondary school mainly focussed on physical activity and limited the situational transitions to commencing further education.

## 3.2 Chapter aims

The specific aims of this chapter are to:

- 1. Explore perceived changes in physical activity and sedentary behaviour during the transition out of secondary school among recent school leavers;
- Explore how situational transitions impact physical activity and sedentary behaviour during the transition out of secondary school among recent school leavers; and
- Explore perceptions of individual, social and environmental influences on changes in physical activity and sedentary behaviour during the transition out of secondary school among recent school leavers, and how these may differ by situational transitions.

## 3.3 Methods

Common types of qualitative methods include one-on-one interviewing and focus groups. One-on-one interviews have several advantages over focus groups, including facilitating explorations<sup>195</sup> and providing an opportunity for participants to express personal views free from constraint. In contrast, a group setting may not allow equal chance to speak and differing opinions in the group may cause participants to conform or not be forthcoming<sup>196</sup>. Semi-structured one-on-one interviews is a commonly adopted type of interview method providing a scaffolding of key topics for the interviewer to broach, rather than closed questions<sup>197</sup>. This allows participants to steer the interview so that it remains focussed on what is most relevant to them and allows the researcher to ask follow-on questions that expand on points made by the participant<sup>197</sup>.

### 3.3.1 Study design

The LEAP (LEaving school and your Activity Patterns) study was a qualitative investigation that collected data on the perceptions of recent secondary school leavers (<3-years post-school). Perceptions were gained via one-on-one semi-structured interviews that explored experiences and views about how physical activity and sedentary behaviour had changed during the transition out of secondary school, and what these changes were attributed to. Interviews were conducted via the telephone or face-to-face. Ethical approval was obtained from Deakin University's Human Ethics Advisory Group - Health.

Three theories or models underpinned this study. The methodological orientation initially was grounded theory<sup>198</sup>, which is inductive and guided what type of coding to use during data preparation (emergent). However, the ecological model<sup>20, 21</sup> guided how the influences were grouped during thematic analysis and reporting, which does not align with a grounded theory approach. Lastly, the life transition model<sup>18</sup> highlighted that there are cognitive-behavioural indicators of transitions that are assets or risks to adaptations. That model posits that situational transitions may disrupt roles and was used in this chapter to guide the analysis, reporting and interpretations of the findings.

The consolidated criteria for reporting qualitative research (COREQ) guidelines<sup>195</sup> (Appendix 3.1) was used for this study. It is a 32-item checklist that ensured inclusion of theories in the study design; noting potential bias from the research team, interview setting and the presence of non-participants; count and reasons for participant withdrawals; the number of coders; a coding tree; noting software used; quotes that illustrate the themes; and clear major and minor themes.

#### **3.3.2** Participant recruitment

As this was an exploratory qualitative study, a target sample size could not be predetermined. The sample size was guided by data saturation: the point at which no new themes or ideas emerged from the interviews<sup>193</sup>. However, it was anticipated that between 27-32 participants would be needed, based on similar previous studies<sup>146, 193</sup>. The inclusion criteria for participants were that they: 1) had left secondary school <3-years ago; 2) were 18-21-year-olds; 3) could speak and comprehend adequate English; and 4) currently resided in Australia.

Recruitment commenced in November, 2014 and continued until no new information was raised in the interviews (rolling recruitment until data saturation) which occurred in June, 2015. Participants were recruited via a combination of methods, specifically, social media, convenience sampling, flyers and snowball sampling. For each of the four recruitment methods described in detail below, potential participants contacted the student investigator directly via email to express interest.

Social media: A study Facebook page was made, and an advertisement was placed on Facebook to be delivered to Australian 18-21-year-olds with a Facebook account. The advertisements 'boosted' (shared) either the page or a post from the page that included the study email address. In Australia, social networking sites were the number one online activity for most (83%) Australian 16-29-year-olds in 2010 and most (93%) social networking site users had a Facebook account in 2011<sup>199</sup>. When the advertisement was clicked on, potential participants were forwarded to the study page. By the end of recruitment (June, 2015) the page had 71 likes and 26 posts on the page. The posts received, in total, two likes, three comments and two shares. The page provided the study email address as the means of expressing interest in participating. Visitor posts were disabled in the page's security settings to limit the chance of spam or inappropriate posts. The

advertisements were run for 21 days, cost \$65.90/day, and there was an average of 137 impressions/day delivered (2 877 total impressions). Impressions refer to the number of times an advertisement is shown and do not necessarily reflect the number of individuals who saw it.

- 2. Convenience sampling: Existing personal and professional networks of the student investigator were emailed information about this study, a flyer, the link to the study page and the study email address to request further information. Networks were asked to forward the information on to people who they thought might be eligible and interested.
- 3. Flyers: Flyers were displayed in public areas such as the Burwood campus of Deakin University and community noticeboards. The flyers directed potential participants to the study page and email address for more information about this study. The initial flyer was succeeded by flyers that only sought male participants.
- 4. Snowball sampling: the study flyer, Facebook page link and email address were provided within emails to the student investigator's social network and to participants after completing their interviews. The email asked for the information to be forwarded on to people who they thought might be eligible and interested.

When an expression of interest was received, potential participants were emailed a detailed plain language statement and consent form, which included permission to be recorded and for de-identified quotes to be used. Potential participants were asked to complete the form and return it via email. Participants were not formally screened by the researcher, as the plain language statement outlined the inclusion criteria and participants confirmed they met these criteria on the consent form. Within the consent form, participants nominated whether they wished to be interviewed via the telephone or face-to-face in a private room at Deakin University's Burwood campus, and provided their availability. Only those who returned a completed consent form that confirmed eligibility took part in this study (Figure 3.1). Consent to participate was freely provided and participants were informed that they could withdraw at any time. The consent form collected some basic demographic information about the participants including age, sex, and the year that they left secondary school.

## Figure 3.1 Study flow diagram



## 3.3.3 Protocol

The topic guide (Appendix 3.2) was pilot-tested on three young adults to gauge the duration and the flow of the interview topic guide, as per recommendations for qualitative studies<sup>197</sup>. Although there was a slight risk that participants may have felt self-conscious during a face-to-face interview, this was minimised by adopting a friendly, welcoming and calm approach during the interview. Participants were reassured that they did not need to answer questions that they would rather not and that they could end the interview at any time, for any reason.

Participants were sent an email with details of when (and where, if necessary) the interview was scheduled, according to their availabilities. Participants received a reminder email either the day before or on the morning of the interview, depending on whether the scheduled interview was early or late in the day.

The interviews took place between November, 2014 and June, 2015. The interviews were one-on-one with the student investigator. The duration of the interviews ranged from 11-30 mins and averaged  $20.9\pm6.1$  mins (SD). All interviews were recorded using a digital voice recorder. The participant's name was not mentioned during the interview. A \$20 gift card for a major department store was offered to participants at the completion of the interview to compensate for their time involved in participating.

The digital recordings and transcripts of the interviews were labelled with the participant's unique identifier code for storage, cleaning, coding and analysing. This re-identifiable form meant illustrative quotes could be reported with minimal descriptive information, including age, gender, duration since leaving secondary school, employment status (work/study), and whether they had moved out of the family home post-school. Identifiable data (consent forms and participant database) and re-identifiable data (recordings and transcripts) were stored separately.

## 3.3.4 Measures

Participants were taken through a semi-structured interview topic guide (Appendix 3.2) that was comprised of open-ended questions to generate discussion, surrounded by a scaffolding of key topic questions. The topic guide included an opening, introduction, key questions and an ending. General questions asked participants for their perceptions, views and experiences with physical activity and sedentary behaviour during secondary school and post-school, and to expand on any changes. If changes had occurred, participants were asked why and encouraged to explain influences on the changes. Lastly, participants were asked what could be actioned in various settings (university, workplace, home or socially) to prevent a decrease in physical activity and an increase in sedentary behaviour.

The final part of the interview (Appendix 3.2) involved noting third parties present and the setting the interviewee was in, as per COREQ guidelines<sup>195</sup> (Appendix 3.1). This is because bystanders may have biased the informer's narrative. It has been recommended<sup>197</sup> that participants of qualitative studies should be in a comfortable environment in order to feel unrestricted when speaking and to willingly and honestly share information without hesitation with the interviewer.

## 3.3.5 Data analysis

NVivo version 12 Plus (QSR International) was used for data preparation and analyses.

## Data preparation

The recordings were transcribed at a level of intelligent verbatim (no utterances such as ah, um and err). Six recordings were sent to a professional service called Transcriber Online (www.transcriberonline.com) for transcribing and the student investigator transcribed the remaining recordings. Rather than applying preconceived codes to the interview transcripts, the codes were formed based on the participant narratives (emergent coding). A coding tree (Appendix 3.3) was developed from the interview transcripts that represented frequent topics. The tree was used to code the paragraphs, sentences and phrases within the transcripts, as applicable. Codes were then grouped into categories.

A second researcher cross-coded 10% of the interview transcripts. To ensure appropriate code allocation and data interpretation, discussions took place between investigators to provide consensus when decisions were not clear. This approach to qualitative analysis is commonly recommended<sup>200-203</sup>, including within the COREQ guidelines<sup>195</sup>.

## Descriptive analyses

Demographic characteristics were summarised; age and duration since leaving secondary school were collected on the consent form and analysed using means and standard deviations in Stata (version 12 for Windows, 2012, StataCorp LP).

## Thematic analyses

For an overview of the recent school leavers' perspectives, the most frequently mentioned words were visualised using word clouds. Word clouds give insight into exploratory textual analysis by quickly clearly communicating the most mentioned points that might reveal patterns and steer further analysis. Word frequency queries
on the transcripts overall and by gender were used. Words were grouped with stemmed words (as opposed to other groupings: exact, synonyms, specialisations or generalisation) and no minimum word length was chosen. Exclusions were discourse markers (oh, well, now, then, you know, I mean), discourse connectives (so, because, and, but, or), interjections, exclamations, fillers and utterances. The twenty words most used were identified and graphically presented.

Thematic analyses of interview transcripts from semi-structured interviews allows meaning to emerge authentically from participants' discourses<sup>204</sup>. Nodes were added to descriptions of static or changed physical activity and sedentary behaviour and summarised using *n* and proportions (Aim 1). Coding queries (compound coding query, coding comparison query and group query) were used on the reason nodes to identify emerging themes (Aims 2 and 3). Reasons were ranked by the most frequently mentioned. Major themes were mentioned by >14 participants. The next seven most-mentioned reason nodes were categorised as the minor themes (mentioned by  $\leq$ 14 participants), and the rest of the nodes were categorised as uncommon.

Matrix coding queries were used which cross-tabulated nodes of interest to determine links between attributes and influences. Firstly, a matrix coding query was used between the direction (positive or negative) and the influences. This allowed examination of the main direction of each influence and variances in the direction. Secondly, a matrix coding query was used between the influences and multiple personal attributes. Attributes explored were gender, area-level SEP, age when interviewed, years since leaving school, whether participants changed residence postschool, whether participants went straight into tertiary education post-school, current employment status, change in physical activity and sedentary behaviour, and change in friendship groups. Area-level SEP was determined using the postcode provided on the consent form and national census data of Socio-Economic Indexes of Areas (SEIFA) of relative advantage and disadvantage to stratify into lowest, mid and highest tertiles of socio-economic area. Following the ecological model<sup>20, 21</sup>, influences were grouped when reporting the findings into individual, social and environmental. In the following section, the results of the multiple coding queries are presented, along with pertinent quotes selected that illustrate themes or queries.

50

# 3.4 Results

# 3.4.1 Description of participants

Table 3.1 describes the demographic characteristics, situational transitions and activity of the participants post-school. Women and men were approximately evenly represented in the sample. The average age of participants was 19-years-old and most had finished secondary school >1-year ago. All participants were living in an urban area and most participants were living in a least-disadvantaged neighbourhood (the highest tertile of socio-economic area). A third of participants were living in a different residence or had changes in household composition post-school. Most of the participants were concurrently working and studying.

During the interviews, participants reported that most were alone except for four; one participant was sitting in a public setting (on a train), one had his mother sitting next to him at home, and two had one other person present at home. An almost even proportion of the sample described increases or decreases in physical activity and sedentary behaviour post-school, and less than a fifth of the sample discussed that they had maintained physical activity and sedentary behaviour.

## **Table 3.1**Sample profile (n=29)

Characteristics	
Demographic characteristics	
Women, $\%(n)$	55% (16)
Age, mean years±SD	19.1±1.0
Duration since leaving secondary school, <i>mean years</i> ±SD	$1.4{\pm}0.8$
Lowest tertile of socio-economic area, $\%(n)$	17% (5)
Mid tertile of socio-economic area, $\%(n)$	31% (9)
Highest tertile of socio-economic area, $\%(n)$	52% (15)
Situational transitions, % (n)	
Changed residence	35% (10)
Commenced tertiary education	86% (25)
Studying only	31% (9)
Studying and working	62% (18)
Working only	79% (2)
Perceived physical activity post-school, % (n)	
Increased	45% (13)
No change	17% (5)
Decreased	38% (11)
Perceived sedentary behaviour post-school, % (n)	
Decreased	41% (12)
No change	14% (4)
Increased	45% (13)

#### 3.4.2 Word clouds

Word clouds for the overall sample and by gender are presented in Figures 3.2-3.4. The word cloud for the overall sample suggests that recent school leavers perceived multiple settings of physical activity and sedentary behaviour to be important, since the words 'university', 'home', 'work', and 'schools' were frequently used.

Figure 3.2 The 20 words most frequently said by the overall sample

active around going high home hours lot much now probably really SChOOIS sitting think time university walk well work

year

**Figure 3.3** The 20 words most frequently said by participants who identified as women

activity changed classes different exercise friends home hours job minutes people School sitting

time university walk watch work **Figure 3.4** The 20 words most frequently said by participants who identified as men

active breaks classes gym high home hour less minutes physical play school sitting sport standing studying time university walk work

The words most frequently used differed by gender and warrant further exploration, since word clouds may provide a snapshot of findings and reveal patterns to inform thematic analysis. Although men and women frequently described 'walking', terms specific to physical activity such as 'gymnasiums', 'playing' and 'sport' were among the top 20 words mentioned most by men but not women. This may reflect the common types of physical activity men engage in during the transition out of secondary school. 'People' and 'friends' were among the top 20 words mentioned most by women but not men. This may be indicative of social aspects of physical activity and sedentary behaviour being more important and influential to women than men. 'Breaks' and 'standing' were among the top 20 words mentioned most by men

but not women. This may suggest that men intentionally break up sedentary behaviour with standing or moving to decrease sedentary behaviour and increase physical activity more than women. Although men and women frequently described 'sitting' in general and a specific type of sedentary behaviour 'studying', 'television' was among the top 20 words mentioned most by women but not men.

## 3.4.3 General life changes

Some participants described a range of general life changes since they had left secondary school. These included having more independence, less stress, and changes in social networks. Other general life changes experienced by participants post-school were more responsibilities (notably financial responsibilities), getting a driver's license, getting a car, moving out of the family home, getting a job, and having more responsibilities at home.

I've the responsibility of cleaning the house and cooking and ... paying bills. Having to work more to be able to afford that. I guess getting my license as well, I got that a year and a half/two years ago. So, with that, petrol and car bills (Woman, 20-years-old, 2.5-years post-school, changed residence post-school, university student, working).

*I'm expected to do a bit more because I'm home more* (Man, 19-years-old, 1.5-years post-school, living with parents, university student, working part-time).

Frequently, participants mentioned that they experienced changes in social circles post-school, with most participants describing losing or drifting apart from friends post-school. Some participants described making new friends at the place of tertiary education or work.

Basically, I stopped seeing everyone from high school. You just kind of drift apart and you make new friends in your classes at university or through work instead. It's a bit different because you're not seeing those people every day (Woman, 19years-old, 1.5-years post-school, living with parents, university student, working two part-time jobs).

I still have all my old friendships, but I have a lot of new friendships now and I probably prefer them more in a way (Woman, 19-years-old, 0.5-years post-school, living with parents, university student, working two jobs).

Some participants described having more independence post-school and how the strict boundaries that were imposed on them by parents and teachers during secondary school had been lifted.

I am a little bit more independent now in the sense that [my parents] are not necessarily worried if I stay up until two or three. Whereas in VCE [Victorian Certificate of Education], yeah (Man, 18-years-old, 0.5-years post-school, living with parents, university student, working).

Some participants described feeling less pressure and stress post-school because they perceived that university was more flexible than secondary school, and they experienced more enjoyment from studies than what they felt at secondary school.

Once school finished, I feel that it was a lot less stressful than it was, because exams take a toll on you after a while. You have a bit of a clean slate (Woman, 18-years-old, 0.5-years post-school, living with parents, university student, working casually).

University is way more flexible with my workload. ... I am only in two or three times a week. ... I got a bit more lazy with my courseload. Certainly a lot less stressed ... because I enjoy what I do and there is not as much pressure (Man, 20-years-old, 1.5-years post-school, living with parents, university student).

# **3.4.4** Descriptive exploration of physical activity and sedentary behaviour

Participants, generally, were better able to articulate, had more things to say, and were quicker to answer the questions about physical activity, than questions about sedentary behaviour. During interviews, most participants asked for clarification, examples or repetition of the definition of sedentary behaviour, but did not do the same for physical activity.

I think a lot of people don't really realise. I don't think people think as much about sedentary behaviour and that being a bad thing. I think the education and how things are sort-of constructed is more very much focussed on how much physical activity to get and not much on educating people on sedentary behaviour (Man, 20-years-old, 2.5-years post-school, changed residence post-school, university student). Participants were asked to describe physical activity and sedentary behaviour while they attended secondary school and post-school, then to summarise changes (Aim 1). Five domains of physical activity and sedentary behaviour, leisure-time, domestic, transport, occupational and school, emerged from the data.

## Physical activity and sedentary behaviour during the school years

Commonly, participants described participating in less physical activity during senior years when they were studying towards receiving their leaving certificate - Victorian Certificate of Education (VCE) - compared to earlier years of secondary school. Some participants described the various reasons why physical activity reduced and sedentary behaviour increased during school years. Reasons included physical education classes were only compulsory for students in earlier years of secondary school, they felt like they did not have to prioritise school workload over physical activity pursuits in the earlier years, they needed to study more in the later school years, and some had a personal preference for sitting.

Well, after Year 10 we no longer had PE anymore. ... I would probably go for walks maybe two or three days a week, but that sort of died down a bit as I got into Year 12 (Woman, 18-years-old, 1-year post-school, living with parents, student, working).

[I was] sedentary for most of the time. Other than scheduled sport time, pretty much all of it was sedentary (Man, 18-years-old, 0.5-years post-school, living with parents, university student, working).

*It was just difficult doing an hour of physical activity [per day] because of my VCE studies* (Woman, 18-years-old, 0.5-years post-school, living with parents, university student, working casually).

Once I got to Year 11, my study load obviously increased a little bit and I had to pick between either the sport or creative arty stuff and I went for the creative side. And then, VCE, I had to drop that as well; so, Year 12 I did nothing but focus on my school work and I was school captain at the time, so I focussed on that (Woman, 20-years-old, 2.5-years post-school, changed residence post-school, working).

*Towards the end [of school, I was] not very [active]. I used to sit there* (Man, 19years-old, 1.5-years post-school, changed residence post-school, university student). Video game use during secondary school years was discussed by two participants (both men) as being 15h/wk or 10h/wk, on average. One participant spoke about how the main type of sedentary behaviour that he engaged in during the earlier years of school was playing video games, but this changed to studying during his senior years.

Not really in the later years of high school. I did it [engaged in video game use] up to Year 11 a fair bit. But after that, I focussed on study (Man, 18-years-old, 0.5-years post-school, changed residence post-school, university student).

Some participants described preferentially choosing sedentary behaviour pursuits during leisure-time during the school years such as mobile phone use, rather than engaging in physical activity. Some participants raised that their sedentary behaviour habits at school continued on at home, saying that they engaged in a high amount of sedentary behaviour both during school hours and after school hours at home. These participants also commonly discussed that they were tired, that school was tiring, and that they felt like they needed to rest.

Pretty high [mobile phone use], to be fair. I should have been a fatty. ... I think every teenager does [use of mobile phones a lot during late adolescence] (Man, 18-years-old, 0.5-years post-school, living with parents, university student).

Mostly it was pretty bad. If I didn't have much homework, I would be really happy about it and be like 'Oh yeah, I can sit on my computer for eight hours', or something. I did become pretty addicted to going online and things. And time spent watching movies on my laptop is another thing I liked to do as well. So yeah, it wasn't good (laughs) (Woman, 18-years-old, 1-year post-school, living with parents, university student, working).

Because I was just tired from school, I just wanted to come home and sit and do nothing. Even though I had been sitting but it still takes it out of you. I'd come home and just sit and relax for the rest of the night (Woman, 20-years-old, 2.5-years post-school, living with parents, university student, working).

It was amusing: you'd sit down all day at school and come home and feel like you needed a rest and sit down a bit more (Man, 20-years-old, 2.5-years post-school, changed residence post-school, university student, volunteering).

*I just mainly sat* (Man, 19-years-old, 1.5-years post-school, living with parents, university student, working).

## Physical activity and sedentary behaviour changes post-school

Findings for whether physical activity and sedentary behaviour changed post-school were mixed (Table 3.1), as similar proportions of participants perceived that they had increases and decreases. Only five respondents felt that physical activity had not changed post-school and only four described maintaining sedentary behaviour.

#### Leisure-time physical activity and sedentary behaviour

Structured physical activity, such as a sport, was mentioned frequently by participants. Many participants discussed that they engaged in sport during and post-secondary school around 1-2 times/wk, but some did not at either stage, and a few did during school only. Notably, there was no mention of an uptake of structured physical activity post-school among participants. The sports that the participants were involved in were swimming, netball, La Crosse, mixed martial arts, basketball and Australia Rules Football. Mostly, those who said that they were engaged in sport did not usually engage in walking for transport. Some participants considered that they had maintained sport post-school, while few described a reduction. Other LTPA described by participants commonly included strength training at a community or home gymnasium.

Types of leisure-time sedentary behaviour in the home environment described were playing video games, using tablets and mobile phones, and watching television or movies. The frequency of using these types of sedentary behaviour was mostly described as daily. Sedentary screen use and technology were mentioned frequently. The participants who raised mobile phone use mostly perceived that this reduced post-school. The time spent using video games post-school decreased for one participant but increased for another who, when asked to give an example duration of time spent playing video games post-school, replied with:

Lots. Lots and lots (Man, 20-years-old, 1.5-years post-school, living with parents, university student).

Watching television was not popular. It was common to discuss low or no television watching during and post-school or a reduction post-school. The reasons given included a lack of interest in what was being programmed on television, a preference for watching videos on YouTube, some did not feel the need to use screen-time to 'destress' because they had less stress post-school, watching movies on laptops instead, or watching a television series via a streaming service. The one participant who said that he watched more television post-school reasoned that this was due to the new streaming services for movies and television shows that were not available during secondary school.

I reduced television because there really isn't that much on that I like now, so I just don't find anything that I have an interest in (Man, 19-years-old, 1.5-years post-school, living with parents, student, working).

Probably less [sitting] now. Purely because of workload and it was just crazy in Year 12. ... I think during Year 12, because I was fairly stressed, I would go on my mobile to de-stress, or what I thought was de-stressing. Nowadays, I kind of think, 'Well, I've got to do that. Well, sure.' There's just not that much pressure. I'll just do the work and then I'll do something else. Yeah, I think more technology then than now (Woman, 18-years-old, 0.5-years post-school, living with parents, university student).

#### Study-related sedentary behaviour

Most participants described that they frequently studied during secondary school and that it was almost exclusively a sedentary behaviour. Those who commenced tertiary education (n=25) described similar sedentary experiences at school and university. Tertiary students spoke about sedentary behaviour twice as often as physical activity. An inference from this that a greater amount of sedentary behaviour than physical activity may take place at tertiary institutes. Respondents reflected that sedentary behaviour at university occurred mostly during lectures and while in the library. Participants who stood for 2h during laboratory classes at university were a rare exception.

It's still pretty bad at the moment. So, university, whenever I'm at university, as long as it's not practical, I'm usually sitting down. Whether that be studying in the library, in a lecture, in a tute; I'd generally be in the sedentary position (Man, 20years-old, 2.5-years post-school, changed residence post-school, university student, volunteering). Some participants, when prompted, described prolonged sitting (>30 mins of sitting time that is not broken up by standing or moving) during secondary school and immediately post-school, with instances more common post-school. Prolonged sitting was mostly referred to when describing studying and was repeatedly perceived as bad but unavoidable. One participant described an estimated increase in prolonged sitting to 2-3h per occasion while at university, compared to <1.5h per occasion while at secondary school. Frequently described settings for prolonged sitting were secondary school, university and home. The dominant setting differed depending on age; prolonged sitting was commonly reported to take place in the classroom during secondary school or at home for tertiary students. Some tertiary students described intentionally trying to break up prolonged sitting regularly by standing, stretching or making cups of tea. It was common for respondents to believe that long periods of uninterrupted sitting were part of normal university life because of the hours of study required.

At school it was always prolonged periods. Not so much during the breaks, but in classes you would be sitting down for at least an hour. At home, similar sort-of thing, you'd sit down for probably about 1-hour bouts, I reckon, maybe even 1.5-hour bouts. ... [and compared that to now, at university] During the rest of the day, so, a lot of the time, I'd probably sit down and study. That is probably the biggest portion of my sedentary time. I try and get up about once every half hour, if I can remember. Sometimes I'd be sitting down for 2-3 hours straight (Man, 20-years-old, 2.5-years post-school, changed residence post-school, university student, volunteering).

I try and do that thing where you have a break every thirty minutes in your studying, that's meant to help [you] study better. That doesn't always happen, basically, if I'm on a roll (Woman, 19-years-old, 1.5-years post-school, living with parents, university student, working two jobs).

I know that you're meant to get up and stretch every so often, but sometimes you just sort of get pretty engrossed in what you're doing. Even if I'm studying and then want a break, I might lay down on my bed which is more sedentary activity (Man, 20-years-old, 2.5-years post-school, changed residence post-school, university student).

#### Transport and domestic-related physical activity and sedentary behaviour

When transport methods were described by participants, active modes, such as walking or cycling, were discussed as often as sedentary modes, such as driving. As mentioned earlier, those who said that they played a sport said that they did not usually engage in walking for transport. Participants raised that the overall frequency of commuting was mostly reduced post-school and the most common change was substituting active transport with driving. When prompted, participants described engaging in domestic physical activity, such as doing chores (sweeping and washing up dishes), gardening, and standing to cook.

#### Work-related physical activity and sedentary behaviour

Almost all of the participants held employment during secondary school and/or postschool. Post-school, some respondents described ceasing work to study exclusively. However, most had paid or volunteer employment (three held two jobs and one held three jobs) and the number of work shifts ranged from 1-5/wk. Some respondents who continuously worked during the transition out of secondary school perceived changes in physical activity and/or sedentary behaviour at work. There was an increase in those who held sedentary occupations over time (n=1 during secondary school vs n=4 post-school), one of whom transferred from an active job to a sedentary job.

Overall, participants had more active jobs than sedentary jobs during and postschool. The number of participants with active occupations declined over time (n=16 during secondary school vs n=13 post-school). Participants reflected that active jobs involved walking more often than standing. Some participants increased the h/wk at an active job post-school and few participants changed from not working to having an active job. These changes are not necessarily positive, as a physically active job has previously been associated with a high prevalence of depression and high stress levels<sup>205</sup> perhaps due to a lack of choice to be active at work.

#### 3.4.5 Situational transitions during the transition out of secondary school

Participants experienced multiple situational transitions post-school. Some participants perceived that these situational transitions were somewhat responsible for changes in physical activity and sedentary behaviour they experienced (Aim 2). The main situational transitions discussed were: 1) commencing tertiary education, 2) changing residence (moving from a rural area to an urban locality, moving from the family home into share houses within the university student residence buildings on campus or community share houses), and 3) commencing full-time work. The impact on physical activity and sedentary behaviour from these three situational transitions are mentioned briefly in this section and will be explored in the following section. Additionally, simply leaving secondary school was also considered influential on changes in physical activity.

I feel like you stop being as active as you used to be. There used to be badminton clubs and table tennis, swimming, carnivals, practise and I used to go to everything. Then once high school finishes, it just stops (Woman, 18-years-old, 0.5-years post-school, living with parents, university student, working casually).

Several participants discussed their perceptions of why situational transitions were partially responsible for changes in physical activity and sedentary behaviour (Aim 3). For instance, of those who commenced tertiary education (n=25), some attributed changes in physical activity and sedentary behaviour to this situational transition. Also, an active or sedentary degree choice such as exercise science or arts/laboratory-based science was thought to be influential on physical activity and sedentary behaviour changes.

Just by the nature of my degree, I would definitely say that it [the amount of physical activity that I do] is more than what I would have done at school (Man, 20-years-old, 2.5-years post-school, changed residence post-school, university student, volunteering).

Changing residence by moving from a rural location into an urban suburb was thought by some participants to influence physical activity and sedentary behaviour. Specific examples they gave of reasons why included more co-participation in physical activity with neighbours, more parks, better neighbourhood aesthetics, less access to physical activity facilities and less traffic in rural compared to urban areas. Further, moving out of the family home was also said to influence physical activity and sedentary behaviour due to more independence from parents, fewer rules and boundaries from parents, and more responsibilities such as domestic chores and having to pay for internet and mobile phone bills. The directions of changes in physical activity and sedentary behaviour because of these situational transitions differed between participants. I came from the country. I did all my primary school in the country. Thinking back then, we still had the internet, still had that kind of stuff. Probably because I'm [now] in more of the dense area, I don't know my neighbours that much. I used to go out and play with my neighbours pretty much every day and we had parks right next to each other and you could go and kick a soccer ball. Because I live in a city [now], it's less physical activity (Man, 19-years-old, 0.5-years post-school, living with parents, university student, working part-time).

If I'm ever bored or anything I just go for a walk. I don't have to tell anyone where I'm going, I can just decide: 'Oh, I'm going to go somewhere, or go and do something'. I'm responsible for cleaning the house, doing the washing, domestic chores. ... A lot more independence (Woman, 19-years-old, 1-year post-school, changed residence post-school, university student, working).

A change in employment was a situational transition that was perceived to change physical activity and sedentary behaviour post-school by some participants, in positive or negative directions. Active or sedentary job types were described as a key reason and participant occupations included retail (cafés, supermarkets and department stores), sport team coach, pizza delivery driver, door-to-door charity donation collector and debating adjudicator.

# **3.4.6** Influences on changes in physical activity and sedentary behaviour during the transition out of secondary school

This section reveals the findings from thematic analyses of the perceived influences on changes in physical activity and sedentary behaviour during the transition of leaving secondary school (Aim 3). Table 3.2 lists examples that represented the themes and Table 3.3 lists the major, minor and uncommon themes. Social support was a major theme for physical activity and sedentary behaviour and the other themes differed minimally. For physical activity, time use was as a unique major theme and weight was as a unique minor theme. For sedentary behaviour, tertiary education was a unique major theme and technology was a unique minor theme. Although the impact of tertiary education, remoteness and occupation have been briefly summarised already in this chapter, these will be discussed more in this section alongside multiple illustrative quotes.

First order	Second order	Third order
(Examples)	(Codes/	(Categories)
	nodes)	
Friends	Social support	Social support
Family		
Housemates		
School peers		
Health practitioners		
Size and quality of grounds, play equipment,	School	Education
and facilities		
School sport teams		
Studying, assignments, exams, and homework		
Studying	Tertiary	
Time on campus (contact hours)	education	
Choice of course		
Grounds and facilities of campus		_
Computer use	Technology	Technology
Mobile phone use		
Television watching		
Playing video games		_
Free time	Time use	Independence
Spare time		
Routines and regimented days		
Holidays		
Driving	Transport	
Public transport use		
Commute duration and distance		
Car ownership		
Active transport (walking/cycling for transport)		
Moving out of home	Home life	
More responsibilities at home		
Doing more chores		
Paying board		_
Choice of occupation	Occupation	Occupation
Changes in occupation		
Body image	Weight	Internal
Weight status (overweight/underweight)		
Weight loss		
Weight gain		
Maturity and attitude	Internal	
Feelings of shame/embarrassment		
Mental health and depression		
Personal preference (likes, dislikes, being		
bored, attention span, interest, and enjoyment)		

**Table 3.2**Examples and the emergent hierarchical order of the perceivedinfluences on changes in physical activity and sedentary behaviour post-school

	Physical activity	Sedentary behaviour
Major themes	Time use	Tertiary education
	Social support	Social support
Minor themes	School	Time use
	Tertiary education	Transport
	Transport	School
	Home life	Occupation
	Occupation	Internal
	Weight	Technology
	Internal	Home life
Uncommon themes	Income	Income
	Availability of facilities	Remoteness
	Pet ownership	Availability of facilities
	Remoteness	Priorities
	Priorities	Independence
	Independence	Habits
	Age	Food
	Food	Age
	Habits	Pet ownership

**Table 3.3**Major, minor and uncommon themes of the perceived influences on<br/>changes in physical activity and sedentary behaviour post-school

Participants discussed the direction of the perceived influences on changes in physical activity and sedentary behaviour post-school, which are summarised in Table 3.4.

**Table 3.4**Summary of individual, social and environmental influences on<br/>changes in physical activity and sedentary behaviour post-school, by direction of<br/>influence

Influences	Positive perceived impact	Negative perceived impact
Individual		
Physical	Desire to lose/maintain weight	Different priorities
activity	Physical activity type preferences	Poor time management
	Physical activity enjoyment	Routine change
	Tertiary course (science	Tertiary course (art degree)
	degree)	
	More responsibilities	Less sleep
Sedentary	None mentioned.	More free time
behaviour		Change in transport mode from active to car
		Car ownership

#### Social

Physical	Behavioural modelling (parent	Parental rules
activity	weight loss)	
	Desire to gain weight (men)	
	Encouragement	
	Independence	
	Social support	
Sedentary	None mentioned	Behavioural modelling
behaviour		Co-participation in sitting
		Encouragement
		Fewer rules post-school
		Independence
		Social support
		Subjective social norms

## Environmental

Physical	Occupation (active)	Unstructured university
activity	Increased working hours	contact hours
	Remoteness change	
	(rural/urban)	
Sedentary	Fewer hours on campus	Increased study requirements
behaviour	Lack of internet access	Increased technology
		access/use
	Flexible university classes	Increased working hours
		Increased commute duration

## Individual influences

## Individual influences on changes in physical activity

Reasons described by this sample for changes in physical activity post-school that may be considered individual influences included priorities, time management, changes in routines, independence, preference for types of physical activity, physical activity enjoyment, and desire to lose weight (all in a positive direction). Poor time management, prioritising work/study over being active, poor sleep, and changes in routines were perceived to negatively influence physical activity for some.

Not having the time to fit it in as well as going to work, so yeah, because everything else just kind of topples, topples down (Woman, 19-years-old, 1-year post-school, changed residence post-school, university student, working).

Work and trying to fit in exercise into all that is really hard, and often I will exercise on my own time or say: 'Okay, I have free time now, I'll go for a run' ... so, time management could probably be better, I think (Woman, 19-years-old, 1.5-years post-school, living with parents, university student, working two jobs).

It is hard to find the time to go to the gym in between classes, and things like that, because it is not as structured - nine to three thirty - as school. Here, you could have an hour here or there, but it is not really enough time to go to the gym to do a good session (Man, 18-years-old, 0.5-years post-school, changed residence post-school, university student).

An increase in responsibilities and independence post-school was mentioned to positively influence physical activity. Some participants described how they took on the responsibility for being active themselves post-school, as they could no longer rely on the free opportunities provided by schools to be physically active.

At high school ... it would just sort-of happen, but now, I go out of my way to do it. ... Now it is only something that I get if I pursue it (Man, 20-years-old, 1.5-years post-school, living with parents, university student).

*It's all just consciously deciding to be active rather than having to be* (Man, 19years-old, 1.5-years post-school, changed residence post-school, university student).

*I do what I want now. My personal preferences have changed since high school* (Man, 18-years-old, 0.5-years post-school, living with parents, university student, working).

Some participants believed that age, personal preference and enjoyment of types of physical activity were positive contributors to their physical activity post-school. One participant mentioned that her age positively influenced changes in her physical activity as she was now able to access a gymnasium alone since turning 18 so she bought and used a gymnasium membership.

Going out to town at night is very active if you're dancing as well. Being 18, that's a fun way to be active too (Man, 19-years-old, 1.5-years post-school, living with parents, university student, working part-time).

I've gotten more motivated to do physical activity now. In high school, it was just like: 'I'm going 'cause I have to go', and now I'm like: 'I've started to do it, I enjoy doing it, I want to get fitter and get more healthy now'. So, I suppose my mentality around it has changed a bit (Woman, 19-years-old, 1.5-years postschool, living with parents, university student, working two jobs).

I just like being active. That's pretty much it. Even when I am sitting, I always fidget, move my legs and things like that, to try and keep moving as much as possible (Man, 18-years-old, 0.5-years post-school, changed residence post-school, university student).

Some participants mentioned that the desire to lose weight was an impetus for increasing physical activity. One said that since starting to drink alcohol, he has put on weight so is now more active to try to rectify that.

*Because I was a bit overweight ... so now I exercise to keep that off* (Man, 18-years-old, 0.5-years post-school, changed residence post-school, university student).

#### Individual influences on changes in sedentary behaviour

Of the multiple reasons participants gave for changes in sedentary behaviour postschool, individual influences were sometimes mentioned, such as having more time available, obtaining access to a car and getting a driver's license, and more independence.

*Probably more [sitting] because I have got more free time* (Man, 20-years-old, 1.5-years post-school, living with parents, university student).

Beforehand, when I first started going to the gym, I used to walk or ride my bike. Now I drive. Especially during winter because it's cold. ... Getting to my girlfriend's place, I used to have to get the bus and walk nearly half an hour to get there, ... now I just drive. ... Basically, anywhere where I used to walk, whether it be a friend's place, or gym, or girlfriend's place, or whatever, I basically drive now, I don't walk (Man, 20-years-old, 1.5-years post-school, had not changed residence, university student, working in retail).

When you get public transport, obviously you're walking quite a lot and everything. But driving, you just drive straight there, so you don't have to (Woman, 19-years-old, 0.5-years post-school, living with parents, university student, working two jobs).

Independence was a perceived influence on physical activity and sedentary behaviour. Some participants felt that having more independence meant that they faced consequences of a decrease in physical activity and an increase in sedentary behaviour if they were not responsible, were not organised, or did not get enough sleep.

If you get less amount of sleep, then you're probably less likely to get up and go for a run or get up and maybe go somewhere. As well as that, if you plan to ride your bike to, say, work, or something like that, however, you slept in or you've had a late night, or something like that, then you have to take the car and that's definitely affected physical activity and sedentary as well (Man, 19-years-old, 0.5years post-school, living with parents, university student, working part-time).

## Social influences

Several common reasons participants gave for changes in physical activity and sedentary behaviour post-school may be considered social influences, including social support, behavioural modelling, encouragement, co-participation in physical activity or sedentary behaviour, and subjective social norms. Of those, support from social networks was often described as influential on changes in physical activity and sedentary behaviour post-school. The social network was described as including friends, family, housemates, peers, and health practitioners. Some participants considered social support and social networks to positively or negatively influence physical activity or sedentary behaviour, while others did not consider them influential at all. Some participants mentioned that their social network provided positive or negative support at different times regarding co-participation; for example, one participant's social network included housemates who he sat with, but also friends who he engaged in physical activity with.

## Social influences on changes in physical activity

Participants described a number of perceived social influences on changes in physical activity post-school. These included behavioural modelling (parent weight loss), a desire to gain weight (in men) due to subjective social norms, encouragement from social networks, social support, independence from parents (described by those no longer living with parents), and parental rules (negative; described by those still living with parents).

As an example, a change in dependence on parents for transport was perceived to positively influence physical activity. Two participants described how their parents used to drive them to school, but their parents would not drive them to university. This meant that they felt forced to become more self-reliant and independent. They believed that this change had resulted in an increase in physical activity. A change in dependence on parents was an influence that appeared to be universal, regardless of situational transitions experienced post-school.

*I don't have my parents driving me around, so I just have got public transport* (Woman, 18-years-old, 0.5-years post-school, changed residence post-school, university student).

At high school, I didn't do as much activity. I was driven to school. I didn't really participate in sports. But now that I have taken up tertiary education, I am responsible for my own transport, so I walk everywhere and sort myself out (Man, 19-years-old, 1.5-years post-school, living with parents, student, working).

Sometimes the same person within the participant's social network provided both positive and negative support at different times; for example, a participant's mother initially encouraged her to go for a walk frequently but then banned it because her poor mental health was exacerbated by time that she spent alone.

The main influence originally was probably my mum. She, like, lost a fair lot of weight through exercise, and that, which [made me] start doing that (Man, 18-years-old, 0.5-years post-school, changed residence post-school, university student).

A few men described that a desire to gain weight so that they were not skinny anymore resulted in more physical activity post-school. Previous work suggests that this motivation may stem from subjective norms creating a desire in some men to look masculine and muscular<sup>206</sup>. This appeared to be a universal influence regardless of situational transitions experienced post-school.

I was quite small when I was in high school; I was rather thin and underweight because of the running I was doing. From there, I think I changed my perspective and thought I've got to put some weight on now. So, I went and started doing that. At the gym, I'll do weights. I'll eat more (Man, 20-years-old, 1.5-years postschool, had not changed residence, university student, working in retail).

*I turned 19, 20, and I was still skinny* (Man, 19-years-old, 1.5-years post-school, changed residence post-school, university student).

*I've probably put on a little bit of weight since school. But I've sort of been aiming to do that. I've been quite a thin build* (Man, 20-years-old, 2.5-years post-school, changed residence post-school, university student).

## Social influences on changes in sedentary behaviour

Social influences specific to changes in sedentary behaviour post-school included having to be more independent, behavioural modelling, encouragement, co-participation in sitting, social support, fewer parental and school rules, and seasonal changes. Social networks appeared to play a key role in sedentary behaviour. One participant described co-participation in television viewing with family and some participants, all men, said that they play video games with friends. They perceived that co-participation, behavioural modelling, subjective norms, and encouragement from friends increased their video game use. This appeared to be universal influence regardless of situational transitions experienced post-school.

I only watch a little bit of [television] in the evening when my parents have the television on (Woman, 18-years-old, 1-year post-school, living with parents, student, working).

It's, like, a social thing for me as well because a lot of my friends do the same thing (Man, 20-years-old, 1.5-years post-school, living with parents, university student).

Sometimes, a mate will call me and tell me to jump on a game and I'll go and sit down for the next few hours and play with him (Man, 19-years-old, 1.5-years post-school, changed residence post-school, university student).

Participants described experiencing rules and boundaries set by secondary schools about certain types of sedentary behaviour such as mobile phones use that resulted in less sedentary behaviour. One participant described how mobile phones would be confiscated if they were seen at school. Some participants described experiencing an increase in mobile phone use post-school because these rules and boundaries no longer applied. This also appeared to be a universal influence regardless of situational transitions experienced post-school.

If you got busted [using your mobile phone], you had to hand it in and only get it back at the end of the week; so, that was a big deterrent, I guess (Man, 19-years-old, 1.5-years post-school, living with parents, university student, working part-time).

Seasonal changes (e.g. winter) and being on holidays were believed to result in more sedentary behaviour because of a reduction in people available to socialise with. This was a minor finding and appeared to be limited to those who experienced the situational transition of commencing tertiary education.

Well, because I'm on holidays, it's more than what it should be. ... I actually think that I sat more during school holidays because everyone was busy at the same time (Woman, 19-years-old, 1-year post-school, changed residence post-school, university student, working).

If maybe some of my friends ask me out then I would go out with them. But if I don't have any dates, I would just sit at home (Woman, 22-years-old, 2.5-years post-school, living with parents, university student, working part-time).

## Environmental influences

Some of the topics in this section, such as tertiary education, employment and living independently or with parents were briefly mentioned in section 3.4.5; however, more information and alternate examples for these topics are presented here.

## Environmental influences on changes in physical activity

Participants reported that the choice of university course had influenced physical activity. For example, some participants studied exercise and sport science, science, or nursing degrees. They reported that they stood during laboratory classes, were sometimes active during class time or were active while working because of the nature of the course's content.

Another big reason has been just, I've learnt how important any daily exercise is for health. Because I am studying nursing, so when we were learning about different conditions like arthritis, osteoporosis and obesity, I just really caught onto to how simple it is to prevent or reduce the effects of things like that by just using your body and going for walks (Woman, 18-years-old, 1-year post-school, living with parents, university student, working).

Some participants spoke about how they believed that they did more physical activity at secondary school because it had more contact hours than university. They mentioned large ovals at school, more sport clubs, and compulsory sport carnivals that influenced more physical activity. Some participants said that their tertiary institute did not have these same attributes or offer these opportunities and, therefore, did not encourage as much physical activity as their school did.

*I was at school more than university, so there was a lot more walking there* (Man, 19-years-old, 1.5-years post-school, changed residence post-school, university student).

Conversely, other participants mentioned that because there was more flexibility with the university classes compared to structured secondary school timetable, they were able to do more physical activity and the presence of staircases on the grounds of universities and the sprawled layout of the university buildings encouraged more physical activity.

You get more opportunity to get up and do stuff than there was in high school. In high school it was sort of limited what you're doing, in university you choose to – I'm going to get up and walk around, I'm going to do this instead of going to the lecture or I'm going to do this after the lecture so I can move around (Woman, 19-years-old, 0.5-years post-school, living with parents, university student, working two jobs).

Some participants commented about how they believed that moving from a rural area to a city influenced physical activity. Specifically, they mentioned that in rural communities, they had less traffic, had better neighbourhood aesthetics, but had less physical activity equipment and facilities.

For example, in terms of running, even moving up to the city, there's not many running tracks. It's crowded. You've got to wait for traffic a lot of the time. Whereas in the country there was a lot less traffic, there was more running spots, there was more interesting areas so there was more motivation to be involved in physical activity in that means. I guess, in terms of gym equipment and facilities, it's probably a lot more limited but it just depends on your training and your goals (Man, 20-years-old, 2.5-years post-school, changed residence post-school, university student, volunteering).

Employment changes were believed to influence changes in physical activity with increased hours at work post-school described as a reason for less physical activity in some participants.

... generally, workloads are a lot higher so I'm less likely to get away at a decent hour [from work] to kind-of go to the gym and do that stuff (Woman, 20-years-old, 2.5-years post-school, changed residence post-school, working).

Conversely, the choice of an active occupation was raised as influential to increasing physical activity. One participant with an active occupation as a team coach and in the fitness industry partially attributed an increase in his physical activity to the nature of his employment. However, because he was tired from working, he decreased his time spent at the gymnasium during his leisure-time after business hours. This experience may be consistent with a 'compensation' hypothesis<sup>207, 208</sup>.

I think the main thing for me is that it's part of my career and so, I'm motivated in that sense because I get paid for it, does that make sense? That would be the biggest motivator. ... The main [reason] was that fitness, or work[ing] in the fitness industry, for me has meant that my physical activity is directly associated with my career. That's the main reason really. I'm inclined to be active because of that (Man, 20-years-old, 2.5-years post-school, changed residence post-school, university student, volunteering).

#### Environmental influences on changes in sedentary behaviour

Several participants studying at university believed that they had increased their sedentary behaviour compared to when they were at secondary school. It was reasoned that this was because of more assessments, they studied more, exams were harder, the commute was longer, and they now cared more about doing well academically because they had chosen the subjects themselves. One participant said that she always sat during her classes because she was studying an art degree that required her to sit.

Studying more than at high school. At high school there was only a little bit of homework to do and anything extra is on you. ... Exam period for university is more intense than exam period for high school. I definitely do way more work than I would a few years ago. I think the workload is a big factor. I think the units I've chosen are ones I want to do good in, rather than those that have been assigned to me. They probably make me sedentary a bit more because I want to do better, so I'll sit for more time (Man, 19-years-old, 1.5-years post-school, living with parents, university student, working part-time).

I'd say more-so at university. Definitely a lot more sitting now. I think that I'm taking education a bit more seriously now that I'm at university (Man, 20-years-old, 2.5-years post-school, changed residence post-school, university student).

The travel on university days and that, sitting on the train. So many hours wasted on the train (laughs). In high school, you didn't have that (Woman, 19-years-old, 1.5-years post-school, living with parents, university student, working two part-time jobs).

One participant described how she felt that universities had expectations that she should sit more and that sitting more was normal for her peers. Subjective norms about an increase in sedentary behaviour was common for those who experienced the situational transition of commencing tertiary education.

I would say expectations. I have to live with the expectations that is set from my university so that is probably why I probably have to sit more than I would in high school. But I feel that I sit more, but I don't really feel the change even. I feel like it is just normal for university (Woman, 18 years, 0.5 years post-school, no residential change post-school, university student, working casually).

Conversely, a few participants who had commenced tertiary education talked about sitting less compared to when they were at secondary school, and partly attributed this to more flexibility and less contact hours at university compared to secondary school.

I think [my sitting time] has decreased somewhat [since] school just because the contact hours [of university] have decreased (Man, 18-years-old, 0.5-years post-school, living with parents, university student, working).

A few participants mentioned that an increase in the access to and use of technology (laptops, tablets, mobile phones, social media, and the internet) influenced an increase in sedentary behaviour post-school. Conversely, a participant mentioned that having no internet access after moving out of the family home resulted in less television watching and mobile phone use.

I feel like a lot of people just, if you have a break, they just grab their phones. ... You know what, a big thing is technology. Just because there's technology there, I mean that's like, a massive reason (Woman, 18-years-old, 0.5-years post-school, living with parents, university student, working casually).

As soon as you go on [social media], you get hooked and you stay on there for a little while. Definitely, instead of either going out. I'm sitting at home either on the laptop or on the phone (Man, 19-years-old, 0.5-years post-school, living with parents, university student, working part-time).

I don't have internet at home at the moment, so I can't sit for hours watching television and shows. ... I think that I'm starting to use [my mobile phone] less, especially because I have got no internet on my phone (Woman, 18-years-old, 0.5-years post-school, changed residence post-school, university student).

Both participants who commenced full-time work post-school described increases in sedentary behaviour and partially attributed this to the sedentary nature of their work. One participant talked about the increased use of a mobile phone while sitting that was associated with her job, as a contributor to her increased sedentary behaviour post-school.

I think because I have a desk job and [my mobile phone] is just there. I work in an office (Woman, 20-years-old, 2.5-years post-school, living with parents, working).

It depends on obviously how busy work is. I am in a corporate environment and we've gone through an organisational restructure. There's a lot less staff ... I work in an office, so I do spend quite a lot of time [during] the day sitting. Usually that is for longer periods, but I do try and stand up, walk around a little bit, go outside when I can. But it is quite sedentary (Woman, 20-years-old, 2.5years post-school, changed residence post-school, working).

## 3.5 Discussion

The aims of this chapter were to explore perceptions of how and why changes in physical activity and sedentary behaviour occur during the transition out of secondary school among recent school leavers. This sample had mixed findings for how physical activity and sedentary behaviour changed, with an almost even number of responders reporting increases and decreases. This finding is inconsistent with most of the previous literature which has reported an average trend for a decline in physical activity post-school using large-scale surveys<sup>15, 85, 106, 118, 123, 132-135, 186</sup>. This highlights that although there may be an overall decline in physical activity on average, not all people have the same experience.

Thematic analyses revealed that social support was a major perceived influence on physical activity and sedentary behaviour changes post-school, though not for all participants. Minor themes for physical activity were changes in time use, transport, home life, occupation and weight, and minor themes for sedentary behaviour were tertiary education, time use, transport, home life, occupation and technology. Themes for physical activity and sedentary behaviour were mainly consistent and differed minimally. The unique theme for physical activity was weight and the unique theme for sedentary behaviour was technology. Of note, no influences were raised that supported a maintenance of physical activity and sedentary behaviour over time; participants only discussed influences that were perceived to result in changes (increases or decreases).

The direction of influences on physical activity and sedentary behaviour were sometimes mixed. The perceived changes in social support and increase in free time and independence post-school were believed to influence physical activity and sedentary behaviour. Some participants believed that these influences increased physical activity, while others said that they increased sedentary behaviour. This suggests that in different people, changes in free time, independence and social support may facilitate physical activity or sedentary behaviour. This finding is consistent with previous research that similarly reported that peers and free time were facilitators and barriers to health behaviours<sup>125</sup>. Instead of one clear reason for these interpersonal differences between participants, possible explanations include characteristics of the social support network and peer group profile, gender, age, geographic region and what situational transitions they experienced post-school.

The three main situational transitions participants mentioned experiencing were commencing tertiary education, changing residence and employment changes. Some participants partially attributed changes in physical activity and sedentary behaviour to these situational transitions. Participants discussed multiple reasons behind the perceived associations between situational transitions experienced post-school and changes in physical activity and sedentary behaviour. A novel finding was that these perceived influences were specific to the situational transitions. If students commenced full-time employment post-school, influences were what kind of job they had (whether they now had physically demanding work), working hours (consistent with a previous study<sup>188</sup>) or technology access/use. These were novel compared to a qualitative study about the transition from school to work that alternatively found a motivation for health, peer norms, and independence from parents were important<sup>192</sup>. If students changed residence post-school, the influences were increased traffic, domestic responsibilities and independence from parents (less rules and boundaries), less access to physical activity facilities or poorer neighbourhood aesthetics. These findings are novel compared to a previous quantitative study about household composition of tertiary students that found selfefficacy and health-related barriers were important<sup>187</sup>. Lastly, if students commenced tertiary education post-school, common influences were choice of active or sedentary tertiary course (novel), and increased free time, commute time or hours studying. Additional comparisons of these findings with other studies about transitions to tertiary education<sup>125, 189-191</sup> are discussed in the subsequent paragraph.

The findings of this study were mostly consistent with the few previous qualitative research studies in this area, and these studies mainly focussed on those who commenced tertiary education. Cluskey and Grobe<sup>125</sup> conducted focus groups about weight gain and, similar to the current study, they reported a perceived decline in physical activity due to less social support and structured routines, and that good family role modelling during school years influenced a perceived increase in physical activity post-school. Nelson et al.<sup>190</sup> conducted focus groups and interviews about weight gain in students and similarly reported a decline in physical activity post-school by a lack of time, poor time management, a lack of motivation, seasonal changes such as winter, and poor social support. Deliens et al.<sup>191</sup> conducted focus groups of third-year university students and, consistent with the current study, they reported that the influences on physical activity and sedentary

77

behaviour were perceived enjoyment, time management, convenience, parental control, modelling, social support, availability and accessibility of facilities, and commute time and distance. They reported that media and advertising were influential, which was inconsistent with the current study. Lastly, Deforche et al.<sup>189</sup> conceptualised that moderators of associations between situational transitions and changes in physical activity and sedentary behaviour could be extrapolated from qualitative cross-sectional data. Consistent with the current study, they reported that the associations between influences and changes in physical activity and sedentary behaviour activity and sedentary behaviour appeared to be moderated by living on campus, an increase in social life, and the academic pressure to study. Although qualitative findings may generate hypotheses, these proposed moderating relationships need to be confirmed using quantitative longitudinal data<sup>5</sup>.

In general, some participants gave the impression that they had not thought that sedentary behaviour was important to health, a health behaviour to be aware of, or needed to be reduced. More insights were given on physical activity than sedentary behaviour, potentially because of a lack of familiarity with sedentary behaviour. A lack of cognizance may be due to the relatively recent addition of sedentary behaviour recommendations within the National Physical Activity Guidelines<sup>109, 209</sup> and the paucity of corresponding public awareness campaigns.

One of the underpinning theories of this study, the life transition model<sup>18</sup>, aids understanding of the findings in reference to the process of achieving measurable optimal transition outcomes. The model posits that there are cognitive-behavioural health indicators of transitions that are assets or risks for a transition experience, such as personal efficacy, engagement, health behaviours and service use. Some attributes that were perceived by the participants to positively or negatively influence physical activity and sedentary behaviour fit the description of the model's indicators of assets (physical activity enjoyment, social support, gymnasium use) or risks (independence, time management). The model further postulates that situational transitions are personal experiences that involve disruption and changes in roles that require adaptation. All the participants in this sample perceived that they experienced one or more situational transitions such as changes in education, residence, employment and relationships, and most described disruption to their physical activity and sedentary behaviour routines because of these. Future application of this model should include testing moderating effects of the components of cognitive-behavioural health indicators of transitions (assets or risks) on the association between changes in physical activity or sedentary behaviour over time and situational transitions postschool.

#### **Strengths and limitations**

This chapter addresses key gaps in the literature, including providing understanding via exploration of changes in physical activity and sedentary behaviour during the transition out of secondary school. Advantages of a qualitative approach include the extrapolation of detailed data that highlights understudied viewpoints<sup>210</sup>. A rich discussion was provided by the one-on-one interview format. Rolling recruitment until data saturation ensured the data provided a thorough understanding of experiences, perceptions and viewpoints<sup>195</sup>. Importantly, participants were engaged and were not hesitant to divulge information. There may have been selection bias from health-conscious individuals who volunteered for a study about physical activity and sedentary behaviour, whose offered views may have been influenced by social desirability of a physically active lifestyle. However, some participants openly discussed participating in low physical activity and high sedentary behaviour.

The COREQ guidelines<sup>195</sup> (Appendix 3.1) were followed, which add rigour to qualitative research and require double-coding, collecting setting information such as presence of non-participants, clearly reporting major and minor themes, and acknowledging bias. As stated in the COREQ guidelines<sup>195</sup>, the possible reflexivity must be noted. This stemmed from interviewer characteristics and potential biases that may have affected the direction of interviews and the interpretation of the findings. These include the interviewer's gender (woman), credentials, limited experience and brief interview training. Further, the nine participants recruited via convenience sampling either had an established direct relationship with the interviewer or an indirect relationship, that the interviewer had established relationships with those participants' relatives or friends. To counter this and reflexivity, future qualitative research is recommended to include multiple interviewers<sup>204</sup>, which also includes the process of reaching consensus, which in-turn fosters reliability.

The key limitation of this study stems from a minimally diverse sample. Participants lived in urban areas and most lived in a least-disadvantaged neighbourhood. Participants were a homogenous group with university students overrepresented,

79

very few full-time employees, and no early school leavers, apprentices or trainees. The findings may differ from a random sample since those who work full-time or left school early may have unique experiences and influences on their behaviour that differ to others in this sample, such as income, demands on time, and their peer group profile. Although there were participants employed full-time, their views may not be well represented in this study. Another limitation was collecting data via retrospective recall, including the levels of and influences on physical activity and sedentary behaviour during secondary school. Relying on memory and the potential presence of human recall bias would be alleviated in future qualitative studies that use a prospective longitudinal design to follow-up school leavers during their transition into early adulthood.

A variety of recruitment methods (convenience sampling, flyers, snowball sampling and social media) were used until data saturation was achieved. The social media strategy was the least successful (n=2), was expensive (\$1 383.90AUD) and had poor engagement (few likes, comments or shares). This is despite social networking sites being very popular in this age group in Australia and almost all users of social networking having a Facebook account<sup>199</sup>. However, there may have been too many steps involved to express interest in participating in the study, as the advertisements and page provided the study email address only. Future studies of this age group are recommended to use a variety of recruitment strategies, and if using social media, to reduce the number of steps/'clicks' to express interest, such as having advertisements redirect to an online registration form or directly to a website to find out more and complete a consent form.

This qualitative research study provided explorations of viewpoints and experiences that can complement and inform quantitative research<sup>204</sup>. An ideal quantitative sample would have variation in SEP and situational transitions and include early school leavers, apprentices and trainees (minority groups<sup>211</sup>), full-time workers, tertiary students and those living independently from family. This would allow further investigation into what this study found<sup>212</sup>. In particular, a prospective longitudinal research study is needed to examine changes in physical activity and sedentary behaviour during the transition out of secondary school. This may identify how changes in physical activity and sedentary behaviour differ according to multiple situational transitions and importantly, a range of underlying modifiable influences.

80

## Conclusion

This study provided rich information about recent school leavers' viewpoints on perceived changes in physical activity and sedentary behaviour during the transition out of secondary school and the influences. As qualitative data can be used to generate hypotheses, the following could be tested using longitudinal data. Firstly, findings about perceived physical activity and sedentary behaviour suggest that sedentary behaviour will be common in late adolescence and the direction of changes after leaving school will be mixed and without a clear pattern, and few will maintain levels. The findings of this chapter may point to altered social support, commencing tertiary education and new demands on time as the main influences underlying changes in physical activity and sedentary behaviour. Another hypothesis from this chapter's findings is that some influences will be specific to the situational transitions experienced post-school such as full-time study or work and moving out of the family home. Future research needs to test these hypotheses by comprehensively examining how and why physical activity and sedentary behaviour change during the transition out of secondary school using a prospective cohort study.

Chapter 4

# ProjectADAPT methodology

## 4.1 Introduction

rojectADAPT was a prospective study that observed a cohort of older adolescents annually during the transition out of secondary school. Physical activity and sedentary behaviour during the specific transition out of secondary school is rarely studied. Previous research has employed retrospective study designs<sup>125, 170, 171, 173</sup> or included a long follow-up period that missed the specific transition out of secondary school<sup>106, 136, 137, 213, 214</sup>. In Chapter 3, recent school leavers described perceived social support as a major influence on changes in physical activity and sedentary behaviour during the transition out of secondary school. Time use emerged as potentially more influential for physical activity than sedentary behaviour, and aspects of tertiary study appeared as more influential for sedentary behaviour than physical activity. The participants perceived similar minor influences on changes in physical activity and sedentary behaviour of school, transport, home life and occupation. Chapters 4-8 will use data from ProjectADAPT to build on Chapter 3 to identify quantitative cross-sectional and longitudinal influences on physical activity and sedentary behaviour. Understanding these associations will inform interventions and translational research, identify key groups for policy makers to target, and pinpoint strategies that promote resilience to inadequate physical activity and excessive sedentary behaviour.

# 4.2 Chapter aims

The specific aims of this chapter are to:

- Describe the demographic characteristics of ProjectADAPT participants at baseline;
- 2. Identify the test-retest reliability of key survey items; and
- 3. Examine the distribution of individual, social and environmental independent variables at baseline.

# 4.3 Methods

ProjectADAPT was funded by a Discovery Project grant from the Australian Research Council (DP130101078). ProjectADAPT was a collaboration between researchers from Deakin University, University of Tasmania and University of South Australia. ProjectADAPT aimed to understand changes in physical activity, sedentary behaviour and eating behaviours (excluded from this thesis) during the transition out of secondary school. The following sections describe the study design, sampling, protocol and measures for ProjectADAPT. Ethical approval was obtained from Deakin University's Human Ethics Advisory Group - Health.

#### 4.3.1 Study design

ProjectADAPT was a two-year prospective cohort study of secondary school students in Victoria, Australia. A baseline survey was completed during Year 11 (second-last year of secondary school in the Australian school system) and two follow-up surveys were conducted annually at the same time of year as the baseline measurement (Figure 4.1). This design negates potential intra-individual seasonal variations on outcomes. Further, this overall design captured the experiences of early school leavers or those who completed Year 12, and various post-school pathways (situational transitions). Surveys were either interviewer-administered via a telephone or self-administered online. Test-retest reliability of survey measures was established (methods are described in section 4.3.5).

#### Figure 4.1 Study design and timeline



#### 4.3.2 Participant recruitment

Participants were recruited via State, Catholic and Independent secondary schools in the state of Victoria, Australia or via social media advertising. Secondary schools were identified through a listing on a website (www.education.vic.gov.au). Schools were stratified as urban (major city) and rural (inner regional, outer regional and remote), based on the Remoteness Structure of the Australian Statistical Geography Standard according to postcode<sup>215</sup>. Schools were also stratified by lowest, mid and highest tertiles of socio-economic area based on national census data of SEIFA of relative advantage and disadvantage<sup>216</sup> at the level of the school postcode. Schools within the resulting six categories were randomised using the random number
allocator in Statistical Package for the Social Sciences then approached to participate in the study.

School recruitment (Figure 4.2) was from July, 2013 to September, 2014. In total, 232 Victorian secondary schools (out of 571) were provided with information about the study and invited to participate, excluding special needs schools and schools with <50 Year 11 students. Of these, 47 principals provided consent. Recruitment packs were either: 1) delivered by post to be handed out by a teacher; or 2) distributed by project staff to interested students following a short presentation about the study. Recruitment packs contained a plain language statement and consent form, reply paid envelope and a pen branded with the study name. As shown in Figure 4.2, the response was low (<5%).

During 2013, participants only had the option of completing the survey via a telephone interview. From January, 2014, participants had the option to complete the surveys online or via telephone. Additional changes were made to make it simpler to return consent forms (returned by mail or emailing a photo or scan) and schools were asked to distribute not only hardcopy recruitment packs but also a digital version to parents via email or the school intranet. Despite improvements in response, participation targets according to power calculations (described in section 4.3.6) were still not being met. To address this, paid social media advertising was placed on Facebook (www.facebook.com) as an additional recruitment strategy over five months from September-November, 2014 and April-May, 2015. Advertisements were run over a 42-day period from in 2014 and a 43-day period in 2015 (these were not run every day).



Figure 4.2 Participant recruitment flow diagram

Social media advertisements were restricted to 16-17-year-olds from Victoria, Australia. Clicking on the advertisement redirected individuals to a study webpage hosted by Deakin University. The webpage provided brief information about the study and asked participants to register interest and confirm eligibility (Year 11 student; live in Victoria, Australia). On receipt of registrations of interest, the plain language statement and consent form were emailed to the potential participants. The name of the secondary school was recorded on the consent form. The response and the number of completed valid consent forms returned was higher than recruitment via schools (Figure 4.2).

# 4.3.3 Protocol

The survey was hosted by an online survey software tool called Qualtrics (www.qualtrics.com). The survey was programmed into Qualtrics, along with question skips, prompts and clarifications. The survey was designed to be short and was confirmed in piloting to take 30 mins by telephone to complete. Participants who elected to complete the survey online were emailed a unique link to the baseline survey and instructions. If these participants had not completed the online survey within two weeks, a reminder email and text message were sent every two weeks (maximum three reminders).

Participants who elected to complete the survey via telephone provided availability within a consent form and the interview was scheduled based on those preferences. A

letter was posted to these participants two weeks prior to the scheduled survey day and a reminder text message or email was sent the day before. The reminders asked the participants to contact the researchers to reschedule the telephone interview if necessary. On the scheduled day, participants received a reminder text message in the morning and no response at the interview time resulted in follow-up calls at 30minute intervals (maximum three attempts). During telephone interviews, participant responses were noted by research staff either directly into an online version of the survey or onto hard-copies of the survey then later entered online.

### Follow-up methods

Data were collected from August, 2014 to June, 2017 for the one- and two-year follow-ups. The follow-up surveys were scheduled for a weekday as close as possible to the baseline anniversary. A two-page newsletter of interesting facts about the cohort that were not relevant to the main aims of the study accompanied a reminder letter or email sent to participants two weeks prior to the scheduled survey day. If the reminder was returned as undeliverable, the participant was telephoned. If the participant was unreachable, the friends and family nominated on the consent form were asked to provide updated contact details of the participant. The procedures at the follow-ups otherwise mirrored those at baseline.

#### Cohort maintenance

Multiple strategies were used to maximise retention. Firstly, participants received a \$20 gift voucher for a major department store after completing the first two surveys, and \$25 after completing the final survey, as compensation for time. Secondly, as mentioned above, the consent form requested contact details of friends or relatives who could be contacted if participants were unreachable by telephone, email or mail prior to the one- and/or two-year follow-ups. Lastly, annual birthday and season greetings cards were posted to participants, accompanied by a small gift. Gifts were branded with the study name and included a pen, fridge magnet to prompt participants to update contact details (between baseline and the first follow-up) and Universal Serial Bus (USB; between first and second follow-up). All correspondence included a reminder to email project staff if contact details changed.

## 4.3.4 Measures

The survey (telephone; two-year follow-up version) is presented in Appendix 4.1. The survey included items on physical activity, sedentary behaviour, and diet. However, items relevant to diet are not described here as they are beyond the scope of the thesis and not used in analyses. Most questions were repeated at each timepoint. The one- and two-year follow-up surveys included additional items regarding whether the participant had left secondary school (question 55a), was studying at a tertiary institution (question 55c), and what was mostly occupying their time (question 57c). Additionally, multiple survey items based on qualitative findings from the LEAP study (Chapter 3) were added. These asked about contact hours at the tertiary institution (questions 55d-f), main mode of transport (public transport, active transport and driving; question 61b) and changes in the main mode of transport (question 61c). The following section describes the relevant measures from the survey used in this thesis. The test-retest reliability of the variables was tested in a sub-study using a separate sample to ProjectADAPT, which is described in section 4.3.5.

#### Physical activity variables

The survey included items assessing LTPA (walking, other MPA and VPA during leisure-time), active transport (transport-related cycling and walking), school-related physical activity and occupational physical activity. Discretionary physical activity was a primary outcome variable and was the sum of physical activity in leisure and transport domains.

Survey items were adapted from the IPAQ telephone-administered long form version (www.ipaq.ki.se) and adolescent version (IPAQ-A)<sup>217</sup>. The IPAQ survey was chosen as it separates walking, moderate- and vigorous-intensity physical activity undertaken in different domains (leisure-time; transport; home/household & occupational). Two modules from IPAQ were included in this current survey: parts four "recreation, sport and LTPA" and five "time spent sitting". Two modules from IPAQ-A were included: parts one "school-related activity" and three "transportation physical activity". Instead of using the occupational activity IPAQ module, a more concise survey item was used from the Behavioral Risk Factor Surveillance System Questionnaire by the Centers for Disease Control and Prevention (www.cdc.gov/brfss/questionnaires/pdf-ques/2001brfss.pdf)<sup>218</sup>. The housework,

house maintenance, caring for family and gardening modules of IPAQ and IPAQ-A were only included in the final follow-up survey and were, therefore, not included in any analysis in this thesis. The domestic activity modules have been found to collect over-reported physical activity<sup>219</sup> and were considered not entirely appropriate for the focus of this study and age group.

Before answering the following questions, participants were provided with the definitions from the IPAQ of physical activity and the intensities. Participants were asked how many days in a usual week they walked or cycled to go from place to place for  $\geq 10$  mins at a time (questions 20c-f), and walked and did VPA and MPA in leisure-time for  $\geq 10$  mins at a time (questions 21a-f). Participants were offered 10 response options, being 0-7 days, 'Don't Know' and 'Refused'. Secondly, participants were asked how much time in total was usually spent doing those physical activities on one of those days and were offered units of per day or per week. Reliability and validity of these questions have been tested in adults and are substantial (0.8) and fair (0.3), respectively<sup>54</sup>. However, these questions were adapted for this study to try to improve compliance and completeness, including replacing the open-ended response options with categorical response options for the questions on the frequency of types of physical activity.

IPAQ guidelines were applied for data cleaning (www.ipaq.ki.se). Responses with 'hour and/or minutes' units were converted to minutes. Frequency ('days/week') responses were multiplied by the duration ('time spent on one of those days') to compute total mins/wk, then divided by seven to provide mean mins/day. Mean mins/day <10 were excluded and >180 were truncated to 180 because it was deemed over-reporting, as per the IPAQ scoring protocol (www.ipaq.ki.se). Even though walking is usually categorised as MPA<sup>52</sup>, it was assessed and reported separately to MPA.

Three variables were generated by summing respective components. Total LTPA was generated by summing walking, other MPA and VPA. Total active transport was generated by summing cycling for transport and walking for transport. Lastly, discretionary physical activity was generated by summing total LTPA and total active transport. Discretionary physical activity was dichotomised at  $\geq 60$  mins/day to indicate meeting National Guidelines<sup>109</sup> at each timepoint. Although the participants

would have been >18-years-old at the time of the two-year follow-up, the adolescent cut-point applied to allow comparison with baseline and the one-year follow-up.

## School-related and occupational physical activity

The survey at baseline and the first follow-up assessed four components of schoolrelated physical activity. Participants were asked: "How many physical education or PE classes do you usually have per week?"<sup>217;</sup> "Do you belong to any school sport teams or participate in any other organised school sport activities?"; "How many school sport teams or squads do you belong to?"; and "During morning recess and your school lunch break, would you say you mostly sit, mostly stand, mostly walk, mostly do moderate activity (e.g. shooting hoops) or mostly do vigorous activity (e.g. running)?"<sup>218</sup>. The validity of the IPAQ-A, including the question about physical education classes (or physical activity at school), has been tested in 15-18-year-olds and was fair (0.22)<sup>217</sup>. However, in this survey, that question had wording and response option modifications to be more concise, namely, not asking for the time spent doing physical activity during physical education classes. The response options were "None", "1 class", "2 classes", "3 classes", "4 classes", then "Other". The item assessing physical activity during school breaks was adapted from a question about occupational physical activity (described below)<sup>218</sup>.

The survey item that assessed occupational physical activity (question 56i) asked participants what best described their activity levels at work. The four discreet response options were whether participants mostly sat, stood, walked or did heavy labour or physically demanding work. It was tested in a sample of adults and test-retest reliability was moderate  $(0.40-0.45)^{218}$ .

#### Sedentary behaviour variables

#### Total sedentary behaviour

Participants were provided the definition from the IPAQ to consider before answering the following questions, that said to include time spent sitting or lying down while awake. Two items (questions 26a and b) requested total time spent in a usual week sitting on a weekday and weekend day, respectively, including time spent sitting at home, school, work and leisure-time (free time). These items were based on the IPAQ (www.ipaq.ki.se) and excluded sitting in a motor vehicle. Continuous data that included minutes were converted to hours, as recommended in the IPAQ guidelines (www.ipaq.ki.se). Sitting on weekdays for >20 h were divided by five, and on weekend days >20 h were divided by two, as it was deemed over-reporting. The h/day sitting on a weekday was multiplied by five and h/day sitting on a weekend day was multiplied by two, then summed to generate total h/wk, then divided by seven to provide mean h/day.

Sedentary transport (questions 20a-b) was also sourced from IPAQ (www.ipaq.ki.se) and defined in the survey as sitting in a motor vehicle, such as a train, bus, car and tram. Participants were asked how many days in a usual week they travelled in a motor vehicle. Response options were modified compared to its source, as the frequency response options were changed from open-ended to 10 options (0-7 days, 'Don't Know' and 'Refused') to improve compliance and completeness. Participants were also asked how much time on one of those days was usually spent travelling in a motor vehicle and the response option units were hours and/or mins per day/wk. Frequency/wk was multiplied by the duration/day, to generate total h/week, then divided by seven to provide mean h/day. Data cleaning involved exclusion of results <10 mins/day and truncating >3 h/day to 3h because it was deemed over-reporting.

These IPAQ survey items had substantial (0.7) test-retest reliability and fair-tomoderate (0.3-0.5) validity in comparison to accelerometers in adults<sup>220</sup>.

# Recreational screen time

The survey also included items related to recreational screen time (h/day). Total recreational screen time was comprised of three categories of behaviours, specifically, watching television, digital versatile discs (DVDs) and videos (questions 27a-b), using a computer, laptop or tablet (questions 28a-b), and playing electronic games (questions 29a-c). The survey specified to report only time spent sitting while doing these behaviours. If participants did not play electronic games while sitting, question 29 b-c was coded as 0 mins/h. Participants were asked to report the total time from Monday-Friday, and total time Saturday-Sunday, usually spent engaging in each of the three categories of recreational screen time. Previously reported reliability in adults was moderate for total recreational screen time (Kappa 0.41; 95%CI: 0.25-0.54), high for watching television, DVDs and videos (Intraclass correlation coefficient (ICC): 0.82; 95%CI: 0.75-0.87) and substantial for using a computer (ICC: 0.62; 95%CI: 0.48-0.73)<sup>221</sup>.

For each of the three categories of screen time behaviours, reported time was converted to hours and truncated to 10 h/day. Next, each screen time behaviour was summed to create total recreational screen time, which was truncated to 18 h/day or excluded if >24 h/day. This variable was dichotomised at  $\leq$ 2 h/day to indicate meeting National Guidelines<sup>109</sup>. As there are no national dose-specific sedentary behaviour recommendations for adults, the recreational screen time data for all three time-points were compared to the adolescents guidelines<sup>109</sup>.

### School-related and occupational sedentary behaviour

The school-related sedentary behaviour measure was part of the survey item that assessed the main activity level during school breaks that was adapted from the occupational sedentary behaviour survey item<sup>218</sup>. The occupational sedentary behaviour measure was part of the survey items that assessed the main activity level at work (question 53i). Data were each dichotomised to indicate if participants mostly sat during school breaks and as part of their work. It was tested in a sample of adults and test-retest reliability was moderate (0.40-0.45)<sup>218</sup>.

#### Independent variables

Table 4.1 describes the independent variables assessed in the survey and used in this thesis, including data treatment (summed or averaged; if applicable), sources, scoring of response options and internal reliability (if applicable). Some of the variables were identified in Chapter 3, specifically, social support, internal influences (enjoyment, competence and self-efficacy), priorities and habits (goal setting), technology access (electronic devices and televisions) and facilities. These variables were used to identify cross-sectional correlates of physical activity and sedentary behaviour in late adolescence (Chapters 5-6), longitudinal determinants of physical activity and sedentary behaviour during the transition out of secondary school (Chapter 8), and moderators of associations between situational transitions and physical activity and sedentary behaviour during the transition out of secondary school (Chapter 8).

The independent variables include characteristics traditionally specific to physical activity or sedentary behaviour, for example, physical activity enjoyment and goal setting, television avoidance self-efficacy, and the number of physical activity equipment and electronic devices at home. The rationale for examining this varied range of influences on both physical activity and sedentary behaviour is the interdependency of these two health behaviours. Firstly, previous literature has reported these health behaviours are associated with each other<sup>74, 222</sup>. Secondly, while there is time for both of these health behaviours across the day, device-assessed sedentary time and LPA is strongly negatively correlated in adults<sup>223</sup> and children<sup>224</sup>, that is, commonly displaced by each other. It is therefore important to examine the complex relationship that exists between physical activity and sedentary behaviour and the possibilities of additional explaining influences.

Var	iable names, survey items and original sources	Scale/response options and scoring	Cronbach's alpha coefficient
Individual variables (scores)			
PA enjoyment	I enjoy doing physical activity. <sup>225</sup>	5 part: 1=strongly	0.74
(3 items summed)	I enjoy playing sport.	disagree, 5=strongly agree	
	I enjoy going for walks or walking places.		
PA goal setting	How many times in the past month did you	6 part: 1=never/rarely,	0.72
$(3 \text{ items summed})^{226}$	Set a goal for how much physical activity you would like to do?	2=<1/wk, 3=1/wk, 4=2-	
	Plan particular days on which you would do physical activity?	3/wk, 5=4-6/wk, 6=every	
	Meet someone to do physical activity with?	day	
PA competence	I am good at physical activity.	5 part: 1=strongly	-
(1 item)		disagree, 5=strongly agree	
PA barrier self-efficacy	How confident are you that you could do physical activity even when	5 part: 1=not at all,	0.85
$(5 \text{ items summed})^{227}$	you	5=extremely	
	Are tired?		
	Feel sad, stressed or in a bad mood?		
	Are on holiday?		
	Feel you don't have time?		
	When it's raining?		
TV avoidance self-efficacy	How confident are you that you could		0.84
$(5 \text{ items summed})^{227}$	Watch less TV even when you are bored?		
	Turn off the TV even when there is a program you enjoy?		
	Watch less TV even when it is raining?		
	Watch less TV even if others want to watch it?		
	Turn off the TV when you are doing something else?		

# **Table 4.1**Treatment and scoring of individual, social and environmental survey items, and internal reliability

Table 4.1Continued			
Social variables			
Family (scores):			
E-games co-participation (1 item) <sup>228</sup>	During the past 6 months, how often did a family member play electronic games with you?	5 part: 1=never,	-
PA co-participation (1 item) <sup>225, 228</sup>	During the past 6 months, how often did a family member do physical activity with you?	5=very often	-
TV/DVDs co-participation (1 item) <sup>228</sup>	During the past 6 months, how often did a family member watch TV or DVDs with you?		-
PA social support (3 items summed) <sup>225, 228</sup>	During the past 6 months, how often did a family member Do physical activity with you? Encourage you to be physically active? Take you to places where you can be active?		0.72
SB discouragement (1 item) <sup>229, 230</sup>	During the past 6 months, how often did a family member discourage you from sitting too much (e.g. watching too much TV)?		-
Friends/colleagues (scores):			
E-games co-participation (1 item) <sup>228</sup>	During the past 6 months, how often did friends or work colleagues play electronic games with you?		-
PA co-participation (1 item) <sup>225, 228</sup>	During the past 6 months, how often did friends or work colleagues do physical activity with you?		-
TV/DVDs co-participation (1 item) <sup>228</sup>	During the past 6 months, how often did friends or work colleagues watch TV/DVDs with you?		-
PA social support (3 items summed) <sup>225, 228</sup>	During the past 6 months, how often did friends or work colleagues Do physical activity with you? Encourage you to be physically active? Take you to places where you can be active?		0.81
SB discouragement (1 item) <sup>229, 230</sup>	During the past 6 months, how often did friends or work colleagues discourage you from sitting too much?		-
Social network count	How many adults aged 18 years or over usually live in your household? What about children or	3 part:	-
(3 items summed)	young people aged less than 18 years? How many close friends would you say you have?	n+n+n	
Gym membership	Over the past 6 months, have you been	2 part:	-
$(1 \text{ item})^{226}$	A member of a gym or fitness club?	0=no,	
		1=yes	

Table 4.1Cont	nued		
Environmental val	iables		
Home environmen	:		
E-devices, no. of (count of 6 items	<sup>231</sup> Do you have access to the following at home mobile phone, laptop, tablet/iPad, desktop computer (PC or Mac), active games console (Wii or Kinect), other games console?	2pt: 0=no, 1=yes	-
PA equipment, n (count of 5 items	b. of Do you have access to the following at home treadmill or other gym equipment, basketball/netball ring, bicycle, tennis court <sup>232</sup> , swimming pool?		-
TVs, no. of (1 ite	m) <sup>231</sup> How many TVs do you have in your house?	1 part: <i>n</i>	-
Neighbourhood en	vironment (scores):		
Noise $(1 \text{ item})^{233}$	There is a lot of noise in my neighbourhood.	5 part:	-
Walking environ (4 items summed	<ul> <li>Local sports clubs and other facilities in my neighbourhood offer many opportunities to get exercise.</li> <li>I often see other people walking in my neighbourhood.</li> <li>I often see other people exercising (for example, jogging, bicycling, playing sports) in my neighbourhood.</li> <li>In my neighbourhood, it is easy to walk places.</li> </ul>	1=strongly disagree, 5=strongly agree	0.73
Safety (2 items summed	I feel safe walking in my neighbourhood, day or night. <sup>233</sup> My neighbourhood is safe from crime.	-	0.78
Social cohesion (2 items summed	People around here can be trusted. People around here are willing to help their neighbours.	-	0.76
Land use mix div (count of 8 destin within 10 mins) <sup>22</sup>	ersity ationsIf you walked to them, about how long would it take to get from your home to the nearest milkbar, convenience or small grocery store, supermarket, fruit and vegetable shop or market, fast food outlet, café or restaurant, public transport stop (bus, tram, train), your school, and your work if you have one?	5 part: 1=1-5 mins, 1=6-10 mins, 0=11-20 mins,	0.81
Recreation facilit (count of 5 destin within 10 mins) <sup>24</sup>	If you walked to them, about how long would it take to get from your home to the nearest park, bicycle or walking track, indoor recreation or exercise facility such as a gymnasium, swimming pool, and playing fields such as football ovals or tennis courts?	0=21-30 mins, 0=>30 mins, 0=N/A, 0=don't know	0.64

#### Adaptations

Most independent variables were adapted. This was to simplify, apply a Victorian- or Australian-context or vernacular, increase applicability to the late adolescence age group, aide interpretation or improve clarity. Most of the individual variables were adapted. Physical activity enjoyment<sup>225</sup> included newly developed survey items about sport and walking. Physical activity goal setting<sup>226</sup> was originally called 'Behavioural skills in the past month', had minor wording modifications for clarity, an additional response option was added to one item and other items had higher frequencies added as response options (2-3/wk, 4-6/wk and daily). Physical activity self-efficacy<sup>227</sup> had minor wording and response option modifications that changed the statements into questions, 'exercise' was changed to 'physical activity', 'vacation' was changed to 'holiday' and 'snowing' was removed. Television avoidance self-efficacy was adapted from physical activity self-efficacy score and the choice of wording aimed to similarly represent overcoming barriers and relapse resistance.

Most social variables were adapted. Co-participation in electronic games, physical activity and television/DVD, and social support for physical activity from family, friends or colleagues<sup>225, 228</sup> had modified response options of 'very often' rather than 'every day'. Of these, the sedentary behaviour measures were adapted from the physical activity measures. The discouragement of sitting from social networks was added to a scale<sup>230</sup> to represent social sabotage<sup>229</sup>. Gym membership<sup>226</sup> was originally named 'Club membership' and included one survey item about engagement in 'sport, exercise, or outdoor recreational group or club'. This was reduced to only 'gym or fitness club' and a timeframe of the previous six months was added.

Most environmental variables were adapted to simplify and apply context-specific vernacular. The wording and response options for the number of electronic devices at home<sup>231</sup> item were changed from continuous and open-ended to binary ('yes/no'), examples were added, only six of the 13 original response options were included, and 'cell phone' was changed to 'mobile phone'. The number of physical activity equipment at home<sup>231</sup> item included only five of the 14 original response options and tennis court was added. Neighbourhood noise<sup>233</sup> was originally part of a six-item measure for aesthetic quality. Lastly, neighbourhood walking environment included only four of the 10 original components, neighbourhood safety included two of the

three original components and neighbourhood social cohesion included two of the four original components<sup>233</sup>.

#### Known reliability and validity

Few of the independent measures have known reliability and validity from testing in other samples that did not always include adolescents. Firstly, the reliability of enjoyment of physical activity via computer administer survey was poor<sup>235</sup> (ICC: 0.13, 95%CI: 0.00, 0.45)<sup>225</sup> in a small sample of adolescents (*n*=32) from the USA. Secondly, the number of electronic devices, physical activity equipment and televisions in the home were tested in a sample of 15-year-olds (50.7% girls) and were moderate-to-high<sup>235</sup> for the scales (ICC: 0.60-0.87) and fair-to-high<sup>235</sup> for the individual items (ICC: 0.38-0.88)<sup>231</sup>. Thirdly, neighbourhood environment variables of noise, walking environment, safety and social cohesion had high<sup>235</sup> internal consistency (0.73-0.77) in a large sample of 44-year-olds (*n*=5 988; 54% women) from the USA, and moderate-to-high<sup>235</sup> test-retest reliability using a subset from the same study (0.60, 95%CI: 0.47, 0.71; to 0.88, 95%CI: 0.83, 0.91)<sup>233</sup>. Lastly, the neighbourhood environment scores for land use mix diversity and recreation facilities had acceptable internal consistency and substantial-to-high<sup>235</sup> test-retest reliability (ICC: 0.72-0.87) in a sample of 12-18-year-olds from the USA<sup>234</sup>.

## Internal reliability

Internal reliability of the scales is presented in Table 4.1 and was identified using Cronbach's Alpha<sup>236</sup>. Four of these variables initially had unacceptable internal reliability ( $<0.7^{237}$ ). Of those, recreation facilities score was used in further analyses unmodified to maintain heterogeneity with its source<sup>234</sup> and because access to recreation facilities reflects physical conditions and not a not a psychosocial construct. The remaining three variables were modified. Firstly, the newly developed scores for co-participation in sedentary behaviour with 1) family or 2) friends or colleagues were split into the individual survey items (electronic games and television/DVDs). Secondly, traffic was dropped from the walking environment score, consistent with its source<sup>233</sup>, because it improved the internal reliability ( $\ge 0.7^{237}$ ).

#### Socio-demographic variables

Gender and school attended were noted on participant consent forms. The survey (Appendix 4.1) collected additional socio-demographic information. Dates of birth were used to calculate age in years along with the surveys' completion dates. Primary place of residence postcodes were collected to identify remoteness as urban or rural and area-level SEP as lowest, mid or highest tertiles. Data were collected on employment status and type (volunteer, casual or part-time), the number of jobs held, and income (five categories ranging from <\$50/wk to >\$550/wk). Lastly, information on the highest level of school of participants' parents or carers, the birth country of participants and parents or carers, and whether English was the most commonly spoken language at home were ascertained. Data cleaning of the socio-demographic measures included checking for implausible responses.

#### 4.3.5 Test-retest reliability sub-study

Survey development and robust standards of data quality are explored within the field of psychometrics<sup>238</sup>. Although all assessment tools contain some degree of measurement error, a reliable survey is imperative for ensuring that data reflects what is being measured<sup>238</sup>. One type of reliability testing is repeatability via test-retest which assesses the ability of a survey to yield compatible results<sup>239</sup>.

To address Aim 2, the test-retest reliability of survey items was identified. There were four key reasons for this. Firstly, it will aid critical assessment of the survey and result interpretation<sup>240</sup>. Secondly, previous reliability and/or validity testing of some survey items mostly sampled adults<sup>54, 218, 221, 233, 241</sup>. Thirdly, as outlined in the previous section, some survey items were newly developed and most other survey items were adapted compared to the sources. Lastly, testing the sedentary behaviour survey items is important because a limited amount of sedentary behaviour surveys have undergone psychometric testing<sup>242</sup> and of those that have, few have acceptable reliability<sup>243</sup>.

#### Study design and procedures

Ethical approval for the test-retest reliability study was obtained from Deakin University's Human Ethics Advisory Group - Health. The test-retest study design included participants completing an online survey on two occasions, approximately two weeks apart (15.8 days on average). The survey was a consolidated version of the baseline, one- and two-year follow-up surveys of the main study. It contained all unique survey items, excluding inappropriate questions about leaving secondary school.

A sample of Year 11 students separate to the main cohort was recruited via two waves of advertising on social media in September-November, 2015 and May-July, 2016. The advertisement was delivered to 16-17-year-olds in states of New South Wales and South Australia in Australia. After seeing an advertisement, interested individuals clicked the advertisement and were taken to a Deakin University webpage to fill out an online form to register interest. Individuals who registered interest were emailed a plain language statement and a consent form for themselves and a parent/guardian to complete and return. Upon receipt of the completed consent forms (n=90), participants were emailed a unique link generated within Qualtrics to the first survey and instructions to follow. Two weeks after completing the first survey, participants were emailed a unique link for the follow-up survey and asked to complete it as soon as possible and within two weeks. If the baseline or final survey was not completed within one week of emailing the unique link, a reminder email was sent weekly (maximum two reminders). As compensation for time, a gift card redeemable at several stores was posted to the participants after the final survey was completed.

#### 4.3.6 Data analysis and power calculations

For this and the remaining chapters, data cleaning, management and analyses were conducted using Stata (version 12 for Windows, 2012, StataCorp LP) and statistical significance was set to p<0.05 for all analyses. Survey data were downloaded from the Qualtrics website in .csv files and the three survey data files were merged. Chapters 5 and 6 used all available baseline data, whereas complete case analysis took place for Chapters 7 and 8. Missing data were not imputed.

Participant characteristics at baseline are described using cross-tabulations or means and the distribution of independent variables at baseline was described using means and SD. Significance testing of continuous variables used t-tests for binary subgroups (gender and remoteness) and one-way analysis of variance (ANOVA) for area-level SEP. Lastly, significance testing of categorical variables used Pearson's chi-squared statistics. For the test-retest reliability study, to detect an ICC of 0.8, based on alpha=0.05 and beta=0.2, a sample size of approximately 39 was required<sup>244</sup>. Psychometric literature steered the choice of reliability tests used<sup>245-248</sup>. Firstly, dichotomised and non-ordinal categorical variables were assessed using percentage agreement<sup>245</sup> and Kappa statistics. Secondly, ordinal categorical variables were assessed using percentage agreement<sup>245</sup>, weighted Kappa and ICC<sup>246</sup>. Lastly, continuous variables were assessed using ICC. Reliability rating was based on the highest score of these tests. Agreement was classified as poor ( $\leq 0.2$ ), fair (0.21-0.4), moderate (0.41-0.6), substantial (0.61-0.8) or high (0.81-1.0)<sup>235</sup>.

Power calculations for ProjectADAPT revealed that to detect an R-Squared of 0.04 attributed to 14 independent variables using an F-Test with a significance level of 0.05 and 80% power, a sample size of approximately 500 at the two-year follow-up was required<sup>129</sup>. This accounted for the design effect based on clustering by baseline school (using the cluster size of seven and assuming 70 clusters in total with an ICC of 0.1 for continuous variables). The sample obtained at baseline  $(n=1\ 022)$  included participants from 299 schools, and the two-year follow-up sample (n=852) were from 273 schools at baseline. The study is, therefore, adequately powered.

# 4.4 **Results**

## 4.4.1 Description of participants at baseline

Table 4.2 presents the participant characteristics at baseline (Aim 1) for the overall sample and according to gender, remoteness and area-level SEP. A higher proportion of participants identified as girls (74%), lived in an urban area (71%), lived in the highest tertile of socio-economic area (46%), were recruited via social media (63%) and took the surveys online (93%). Common (n>20) occupations were retail assistant, fast food team member, cashier, waitress and customer service officer.

There was only one significant gender difference found, and that was between the recruitment methods of schools and social media, with a higher proportion of boys than girls recruited via schools and a higher proportion of girls than boys recruited via social media. Demographic characteristics by remoteness and area-level SEP are presented in Appendix 4.2. Most demographic characteristics differed by remoteness. A higher proportion of those who lived in an urban area also lived in the highest tertile of socio-economic area, and a higher proportion of those who lived rurally also lived in the lowest tertile of socio-economic area. A higher proportion of those who lived in an urban area, compared to the rural counterparts, completed the survey online, were unemployed, had tertiary-educated parents, were recruited via social media, had  $\leq 1$  job, did not primarily speak English at home, were born in a country other than Australia and had parents also born in a country other than Australia. Some demographic characteristics differed significantly between area-level SEP, with a higher proportion of those who lived in the highest tertile recruited via social media, completed the survey online, lived in an urban area, had tertiary-educated parents, and earned <\$150/wk, compared to those living in the lowest and mid tertiles.

Demographic characteristics by mode of the survey (telephone/online) and recruitment method (school/social media) are presented in Appendix 4.3. A higher proportion of participants who took the survey via the telephone were recruited via schools, were younger, lived rurally, and had parents born in Australia, compared to those who took the survey online. A higher proportion of those recruited via social media took the survey online, were older, identified as girls, lived in an urban area, and were employed, compared to those recruited via schools.

		Gender	
	Overall	Girls	Boys
	%	%	%
<i>(n)</i>	(1 022)	(754)	(267)
Recruitment via schools, $n=1$ 022	37.4	35.4*	43.1
Survey delivery, online, $n=1$ 022	92.6	92.6	92.5
Age (mean years $\pm$ SD), $n=1008$	$16.9\pm0.4$	16.9±0.4	$16.9 \pm 0.4$
Gender, girl, $n=1$ 021	73.9	-	-
Remoteness, urban, $n=1$ 022	70.6	72.0	43.8
Tertiles of area-level SEP, $n=1$ 021			
Lowest	23.8	22.7	27.0
Mid	30.1	29.8	30.3
Highest	46.1	47.3	42.7
Parent tertiary educated			
Mother, $n=1$ 018	54.4	54.2	54.7
Father, <i>n</i> =1 017	43.8	42.3	47.9
Employment, n=1 016t			
Not employed	41.6	40.5	44.9
Casual	42.4	43.0	40.8
Part-time	20.8	21.5	18.7
Income <\$150/wk, n=805	68.4	67.9	69.7
Birth country (Australia)			
Participant, <i>n</i> =1 015	84.4	84.1	85.8
Mother, $n=1$ 011	61.6	60.9	64.0
Father, <i>n</i> =1 004	61.2	60.6	62.9
English as primary language			
spoken at home, n=1 018	85.2	85.3	85.0

# **Table 4.2**Sample profile (n=1 022)

\*p < 0.05: Pearson's  $\chi^2$  test of significance.

# 4.4.2 Test-retest reliability of survey items

Of the 90 consent forms returned for the test-retest reliability sub-study, 89 completed the first survey and 83 (73% girls) completed both surveys. Participants were 16.9±0.6-years-old. The test-retest reliability scores of the physical activity and sedentary behaviour variables are summarised in Table 4.3 and Table 4.4 presents the test-retest reliability of the independent variables. Correspondingly, Appendix 4.4 presents the test-retest reliability for each individual survey item. The test-retest reliability scores were mostly high. Five individual survey items had poor test-retest reliability. These were co-participation with friends or colleagues in watching television and DVDs, discouragement of sedentary behaviour from friends or colleagues, the number of close friends, access to a mobile phone, and access to physical activity facilities. However, the latter three were part of sub-scales with acceptable test-retest reliability.

	Test-retest reliability statistics					
	K	appa/				
	weighted Kappa		ICC		Percentage	
	Sub-	Individual	Sub-	Individual	agreement	
	scale	items	scale	items	%	
Physical activity:						
Total LTPA						
(mins/day) <sup>1</sup>	0.59	0.46-0.70	0.58	0.57-0.79	-	
Total active transport						
(mins/day) <sup>1</sup>	0.62	0.59-0.68	0.54	0.39-0.66	-	
School PA:						
Main activity level						
during breaks (%)	-	0.65	-	-	90.0	
PE classes/wk (%)	-	0.82	-	-	95.4	
Participate in						
school sport (%)	-	0.76	-		88.0	
School sport teams,						
n (%)	-	0.62	-	-	75.0	
Occupational PA:						
Main activity level						
at work (%)	-	0.54	-	-	74.1	
Sedentary behaviour:						
SB (excluding						
transport) (h/day)	-	-	0.72	0.60-0.74	-	
Sedentary transport						
$(h/day)^1$	-	0.55	-	0.34-0.83	-	
Total recreational						
screen time (h/day)	-	-	0.95	0.04-0.78	-	
School SB:						
Main activity level						
during school				0.84 (0.76,		
breaks (%)	-	0.65	-	0.90)	90.0	
Occupational SB:				,		
Main activity level						
at work (%)	-	$0.54\pm0.09$	-	-	74.1	

**Table 4.3**Test-retest reliability results of the survey items for physical activityand sedentary behaviour variables

<sup>1</sup> ICC and Kappa statistics are presented for the continuous and categorical components (Appendix 4.4).

*Note: all were significant* (p < 0.05).

	Test-retest reliability statistics				
	ŀ	Kappa/			
	weighted Kappa		ICC		Percentage
	Sub-	Individual	Sub-	Individual	agreement
	scale	items	scale	items	%
Individual variables (scores	)				
PA enjoyment	0.54	0.51-0.73	0.72	0.57-0.77	88.7
PA goal setting	0.57	0.54-0.59	0.87	0.70-0.86	88.6
PA competence	0.72	0.72	-	-	88.9
PA self-efficacy	0.57	0.41-0.59	0.86	0.71-0.84	88.1
TV avoidance self-					
efficacy	0.52	0.44-0.56	0.75	0.63-0.74	85.1
Social variables					
Family (scores):					
E games co participation	0.37	0.37	0.58	0.58	86.0
PA co-participation	0.57	0.57	0.38	0.38	80.0
TV/DVDs co-	0.01	0.01	0.71	0.71	05.7
narticipation	0.47	0.47	0.72	0.72	82.2
PA social support	0.77	$0.47_{-}0.61$	0.72	0.72	86.4
SB discouragement	0.37	0.47-0.01	0.80	0.75-0.87	85.5
Friends/colleagues	0.25	0.25	0.71	0.71	05.5
(scores):					
F-games co-participation	0.50	0.50	0.39	0 39	84.4
PA co-participation	0.30	0.50	0.39	0.39	79.7
TV/DVDs co-	0.15	0.15	0.75	0.79	12.1
participation	0 35	0.35	0.13	0.13	79.4
PA social support	0.42	0.34-0.43	0.77	0.64-0.76	82.7
SB discouragement	0.33	0.33	0.15	0.15	86.6
Social network count	-	-	0.95	0.00-0.96	-
Gym membership	-	0.76	-	-	91.5
Environmental variables					
Home environment (counts)	:	0.55.0.00			<i>c</i> 1 <i>c</i>
E-devices, no. of	0.49	0.55-0.88	-	-	61.5
PA equipment, no. of	0.56	0.36-0.79	-	-	00.3
I VS, NO. OI	-	-	0.96	0.96	95.0
Neighbournood					
environment (scores):	0.20	0.20	0.44	0.44	80.0
Noise	0.30	0.30	0.44	0.44	80.0
Walking environment	0.39	0.19-0.46	0.74	0.51-0.78	83.2
Salety Social ashesist	0.49	0.50-0.55	0.70	0.60 - 0.68	85.9
Social conesion	0.45	0.43 - 0.40	0.72	0.05-0./1	88.0
Land use mix diversity	0.65	0.54-0.80	0.92	0.76-0.92	91.3
Recreation facilities	0.62	0.58-0.68	0.86	0./1-0.86	89.7

**Table 4.4**Test-retest reliability results of the survey items for the independent<br/>variables

*Note: all were significant* (p < 0.05).

### 4.4.3 Distribution of independent variables at baseline

The distribution of the baseline independent variables is presented in Table 4.5 (Aim 3) for the overall sample and according to gender. Most means were around half of the maximum possible score or count; however, some biases were present. Girls reported significantly higher scores than boys for physical activity goal setting and neighbourhood safety. Conversely, girls reported lower scores than boys for eleven individual and social variables. These were physical activity competence, physical activity self-efficacy, family co-participation in electronic games, sedentary behaviour discouragement from family, co-participation in electronic games and physical activity with friends or colleagues, social support for physical activity from friends or colleagues, and social network count.

The baseline distribution of most of the environmental independent variables and few of the individual and social variables differed by remoteness and area-level SEP (Appendix 4.5). Of the variables that differed significantly, the trend was that those who lived rurally or in the highest tertile of socio-economic area generally reported a higher proportion of favourable scores or counts.

	Max.			Gender		
	n	score	Overall	Girls	Boys	
Individual variables (scores	)					
PA enjoyment	1 019	15	11.7±2.6	11.7±2.6	$11.9 \pm 2.7$	
PA goal setting	1 0 2 2	18	$8.6 \pm 3.9$	8.7±3.7**	8.6±4.3	
PA competence	1 0 2 0	5	$3.5 \pm 1.2$	3.4±1.2***	3.8±1.2	
PA self-efficacy	1 0 2 2	25	$14.4 \pm 5.0$	13.8±4.8***	$15.9 \pm 5.2$	
TV avoidance self-efficacy	1 009	25	16.1±4.7	16.1±4.7	16.1±4.7	
Social variable						
Family (scores):						
E-games co-participation	995	5	$1.7{\pm}1.0$	1.6±1.0***	$1.9{\pm}1.2$	
PA co-participation	1 012	5	$2.7 \pm 1.3$	2.6±1.3	2.7±1.3	
TV/DVDs co-	1 016	5	$3.4{\pm}1.2$	3.4±1.2	$3.4{\pm}1.1$	
participation						
PA social support	1 012	15	9.5±3.3	9.4±3.2	9.8±3.3	
SB discouragement	1 015	5	3.1±1.3	3.0±1.3*	3.2±1.3	
Friends/colleagues (scores):						
E-games co-participation	1 0 2 0	5	$2.1 \pm 1.3$	1.8±1.1***	$2.8 \pm 1.4$	
PA co-participation	1 0 2 0	5	$2.9{\pm}1.4$	2.7±1.3***	3.3±1.4	
TV/DVDs co-	1 0 2 0	5	$2.8 \pm 1.2$	$2.8{\pm}1.2$	$2.7{\pm}1.2$	
participation						
PA social support	1 0 2 0	15	$7.5 \pm 3.2$	7.2±3.1***	$8.5 \pm 3.4$	
SB discouragement	1 0 2 0	5	$1.7{\pm}1.0$	1.7±0.9	$1.9{\pm}1.1$	
Social network count	1 017	$\infty$	$11.8 \pm 10.1$	11.4±9.4*	$12.9 \pm 11.6$	
Gym membership (yes)	1 021	1	28.4%	28.1%	29.2%	
Environmental variables:						
Home environment:						
E-devices, no. of	1 018	6	$4.5 \pm 1.1$	4.6±1.1	$4.5 \pm 1.1$	
PA equipment, no. of	1 019	5	$2.0{\pm}1.2$	2.1±1.2	$2.0{\pm}1.2$	
TVs, no. of	1 016	$\infty$	$2.7 \pm 2.3$	$2.6 \pm 2.5$	$2.7{\pm}1.4$	
Neighbourhood environment						
(scores):						
Noise	1 017	5	$2.5 \pm 1.0$	2.6±1.0	$2.5 \pm 1.0$	
Walking environment	1 015	20	$15.8 \pm 3.0$	$15.9 \pm 2.8$	$15.7 \pm 3.1$	
Safety	1 016	10	7.3±1.9	7.9±1.8***	7.1±1.9	
Social cohesion	1 016	10	$7.4{\pm}1.7$	$7.5 \pm 1.7$	7.3±1.6	
Land use mix diversity	1 017	8	$3.0\pm2.3$	3.0±2.3	$2.8 \pm 2.2$	
Recreation facilities	1 016	5	$2.4{\pm}1.5$	2.4±1.5	$2.4{\pm}1.5$	

**Table 4.5**Baseline distribution (mean±SD) for independent variables

\*p<0.05, \*\*p<0.01, \*\*\*p<0.001: Continuous variables used independent t-test and categorical variables used Pearson's  $\chi^2$  test of significance.

# 4.5 Discussion

This chapter described the methodology of ProjectADAPT and participant sociodemographic characteristics at baseline. The test-retest reliability of key survey items and the distribution of the independent variables at baseline were examined. The sample was diverse (Aim 1). Some participants were born overseas (15%) and a higher proportion of parents or carers were born overseas (39%), highlighting that about a quarter of the participants are first-generation Australians. Variations between demographic characteristics by subgroups were found. A higher proportion of rural participants than urban were recruited via schools than social media, which may reflect less internet use in rural 16-17-year-olds than urban counterparts. This contrasts with a study of 16-25-year-old women that noted success recruiting rural or regional participants via social media<sup>199</sup>. A higher proportion of those who lived in the highest tertile of socio-economic area reported income <\$150/wk compared to those who lived in lowest and mid tertiles, potentially due to more parental financial support, different social norms and varied expectations in that subgroup. A study reported that young adults expect continued parental support; however, parents of young adult boys expect more independence<sup>249</sup>. This may be important, as reduced parental financial assistance has been linked to lower self-esteem and higher depressive symptoms in young adults<sup>250</sup>.

Importantly, the test-retest reliability of the survey items ranged from substantial-tohigh; therefore, all were acceptable and appropriate to use in the remaining chapters of this thesis (Aim 2). Findings were consistent with previous research for watching television, DVDs and videos (high<sup>221</sup>). Findings were higher than previously reported for physical activity variables (substantial-to-high vs fair-to-moderate<sup>54</sup>); computer use (high vs substantial<sup>221</sup>); and sedentary behaviour (excluding transport; high vs substantial<sup>221</sup>). Inconsistencies with previous studies may be partially attributed to shorter follow-ups and adaptations to wording and response options<sup>54, 221</sup>. Further, previous studies had older<sup>54, 221</sup>, larger<sup>221</sup> samples and represented gender more equally<sup>54</sup>.

The initial cluster sampling via school recruitment resulted in a very low response proportion. Although this was a stratified random sampling approach, the sample disproportionately included girls who lived in urban neighbourhoods with the least disadvantage. The subsequent social media recruitment was more successful than school recruitment, although there are differences in demographic and personality characteristics between social media users and non-users<sup>251</sup> or the general population<sup>252</sup> (described more subsequently). The combination of sampling approaches in this study formed a cohort that was, overall, a convenience sample.

Recruitment strategies aiming at blanket coverage were also successful for a randomised controlled trial of 6-11-year-olds in the UK<sup>253</sup>. The trial found that passive recruitment strategies including flyers and media were more successful than active recruitment strategies, such as directly contacting eligible participants. The authors attributed the success of passive methods to the larger reach and multiple views. Further, passive methods may attract participants who are interested in the study topic and, therefore, more motivated to complete the study<sup>253</sup>. Due to the low response from boys, an additional sampling strategy was employed. Targeted social media advertisements were delivered to boys and resulted in a further n=103consents (n=99 completed surveys). Recruiting boys was challenging and may have been due to girls using social media more<sup>252, 254</sup>. Large studies of adolescents that partially or solely recruited via secondary schools achieved more equal gender representation<sup>255-258</sup>. This is inconsistent with this study and may be due to a larger number of additional recruitment strategies (cartoons, brochure, poster, slogan, food, website, television and local paper advertisement)<sup>255</sup>, larger recruitment budget<sup>255</sup>, less participant burden from a single data collection (cross-sectional survey)<sup>256</sup> or more convenience from administering surveys during class time<sup>257, 258</sup>.

The current study provides some insight for future research into difficulties recruiting older adolescents. Maximise participation is critical for studies to be adequately powered. Offering vouchers as compensation may have assisted in recruiting more participants and may be important for retaining participants in cohort studies. In this study, recruitment via social media appeared more successful than recruitment via schools for the age group and boosted the sample considerably. While the advertisements were paid (total cost >\$3 700AUD), recruitment via advertising on Facebook involved considerably less staff time than recruiting through schools, which involved extensive time contacting and liaising with schools, scheduling, travel and school visits, sometimes with two staff.

The independent variables that participants reported >50% of the maximum score or count at baseline were physical activity enjoyment, physical activity competence, social support for physical activity from family, the amount of physical activity

equipment at home, and neighbourhood walking environment (Aim 3). Future research is needed that tests whether this reflects a sample bias for higher physical activity. The five variables differed by remoteness, area-level SEP and gender. Gender differences were consistent with other studies of older adolescents<sup>259, 260</sup>, with boys reporting >30% higher than girls of the independent variables including physical activity competence, physical activity self-efficacy and social network count. However, gender differences were inconsistent with a small 10-year follow-up study of 12-year-olds (n=39) from Sweden<sup>261</sup>. This study found no difference by gender in self-worth or physical self-esteem<sup>261</sup>. The inconsistencies in findings compared to this study may be due to a wider age range, smaller sample size and different source country<sup>261</sup>.

#### **Strengths and limitations**

#### Study design and recruitment

A strength of this study was an adequate sample size of >1 000 participants. Despite the non-probability sample, participants were from a range of remoteness, area-level SEP, income profiles and countries of birth. Another strength of this study is that the data collection instrument was a cost-effective, quick and efficient method of assessing domain-specific information about physical activity and sedentary behaviour that, importantly, does not alter the behaviour assessed<sup>262</sup>. The survey was designed to be a short length (30 mins), limiting participant burden. An advantage of the online survey was that it appeared in manageable sections.

A further strength is the study design. Year 11 (second last year of secondary school) was chosen as baseline due to schools often limiting access to Year 12 (final year) students who are focusing on exam preparation. An advantage of this sampling restriction was that the annual assessments captured data on early school leavers who are an understudied minority group. The study design included three annual data collection points that allowed examination of yearly and small behavioural changes which studies with long follow-ups may have masked<sup>163, 164</sup>. The prospective cohort study design captured numerous post-school pathway and situational transitions that students took, not only university.

A strength of the study was the focus on late adolescence and the transition of leaving secondary school. Even though that transition is normative and most people experience it, it is understudied compared to other normative transitions, such as completeing primary school and commencing secondary school. Although adolescence is crucial in the development of behaviour that tracks throughout the lifespan, it is understudied compared to other life stages of adulthood and childhood. Overall, the data from this study will progress these underresearched and important areas within the field of behavioural epidemiology.

A key limitation of this study is that generalisability and external validity are reduced for two main reasons. Firstly, the sample had inflated proportions of girls (74%) compared to population average of 49% of 17-year-old Victorians<sup>263</sup>. Also, those living in a rural area (29%) in this sample was higher than the Victorian population average of 8%<sup>264</sup>. This overrepresentation may mean that unique findings may be masked, especially for boys. Secondly, the convenience sample lived in only one state within one country (Victoria, Australia). To increase generalisability, future research with these biases could consider weighting the sample or should ensure more equal gender representation and employ random sampling of multiple states or countries.

Social media recruitment, compared to school recruitment, contributed proportionally more participants who were living in the highest tertile of socioeconomic area, girls (consistent with studies about social media users<sup>252, 254</sup>), living in an urban area, or unemployed. There are likely differences between social media users who respond to advertisements and users who do not; however, the demographic characteristics of non-responders are unavailable to studies using the platform to recruit. A general limitation of social media recruitment is that, although social media use is prevalent, not all segments of populations have social media accounts. Compared to the general population, social media users are younger, female and are more educated<sup>252</sup>. More social media users than non-users have low conscientiousness, are shy, and have high scores for extraversion, family and social loneliness, narcissism, exhibitionism and leadership<sup>251</sup>. Participants recruited from social media may have addictions to social media, mobile phones and/or the internet<sup>265</sup>. The subsequent issues common to high technology use may also be associated with the primary outcomes variables of this study (high sedentary behaviour and low physical activity), such as high narcissism and low self-esteem<sup>266</sup>, or anxiety and insomnia<sup>267</sup>.

A study limitation may be the varied survey delivery. When comparing survey delivery mode between telephone and online, a study reported that telephonedelivery elicited higher social desirability, anxiety and stress, and that participants gave different responses depending on the delivery mode<sup>268</sup>. This suggests that multimodal responses may have impaired comparability<sup>268</sup>. However, a cross-sectional study of 24-year-olds from the USA found that completing online surveys on smartphones was comparable to computers<sup>269</sup>. This was dependent on survey characteristics (wide/long format), smartphone features (screen brightness, text size) and age of participants (adolescents are adept at reading smartphones due to high use)<sup>269</sup>. Modes of sampling and delivery including online surveys on a smartphone or computer<sup>269</sup> should be tested for interactions in future research and accounted for accordingly.

#### Dependent variables

A strength of this study is the inclusion of sedentary behaviour which is historically understudied compared to physical activity. This is because physical activity is well-established as beneficial for health (see Chapter 2) while the evidence base for sedentary behaviour's negative association with health is more recent but growing, especially for adolescents<sup>74, 270-272</sup>. Another strength of this study is the assessment of sedentary behaviour at work and during school breaks and total sedentary behaviour, as most previous research has focusessed only on watching television (as a proxy for total sedentary behaviour) and screen time<sup>273</sup>. An additional strength is that the measurement of sedentary behaviour in the survey only asked for time while sitting, which addressed how technology is changing and some devices can be used while standing or moving. This clarification was prior to each of the questions about total time sitting and engagement with television, DVDs, videos, computer, laptop, tablet and electronic games.

A limitation of the way sedentary behaviour was assessed was that motorised transport was included in total sedentary behaviour even though that may including standing (LPA) while on a tram or train. Another limitation was that the survey did not adequately address a common way sedentary behaviour is accumulated which often includes concurrent use, for example, watching a movie or series on a television or tablet while browsing social media on a mobile phone, or sitting to have a conversation while listening to music. Further, prolonged sedentary behaviour

(bouts  $\geq$ 30 mins) was not assessed by this survey even though it is detrimentally associated with several cardio-metabolic biomarkers (BMI, waist circumference, HDL-C and triglycerides)<sup>274</sup> regardless of the amount of total sedentary behaviour. Future research should reduce chances of double-reporting types of sedentary behaviour and should assess prolonged sedentary behaviour and its influences to translate findings into preventative measures and guidelines.

In general, self-report tools have poorer reliability and validity than device-based measurement of movement<sup>61</sup>. Disadvantages of self-report assessment tools include the inaccuracies due to human recall error<sup>262</sup>, bias and application of social norms that fit social desirability. This is speculated to result in under-reporting sedentary behaviour and over-reporting physical activity<sup>61, 275</sup>. However, a methodological study that administered a similar self-report survey found that Belgian adolescents (mean 15-years-old; n=20) over-estimated average daily total sedentary behaviour by 53%. This may be partially attributed to simultaneous behaviour such as watching television while using a mobile phone resulting in double-reporting<sup>276</sup>.

In addition to the test-retest reliability testing presented in this chapter, future research should employ validity testing in this age group of the dependent variables using devices such as an ActiGraph accelerometer (Pensacola, Florida, USA) for physical activity and the *activ*PAL<sup>TM</sup> inclinometer (PAL Technologies Limited, Glasgow, UK) for sedentary behaviour to support interpretation of the findings. Future research should also use the most reliable and valid measures for physical activity and sedentary behaviour. This is currently concurrent use of an accelerometer for physical activity, inclinometer for sedentary behaviour and log book to record context, wear time, sleep time and non-ambulatory activity<sup>61</sup>. Despite this combination approach being more expensive than self-report alone, the resulting data is of a higher quality (more sensitive) for monitoring behaviour change<sup>61, 275</sup>.

Another possible limitation is that sedentary behaviour survey items had multiple sources (IPAQ (www.ipaq.ki.se) and the occupational physical activity questionnaire<sup>218</sup>) and units that varied from time per weekday and weekend day, to total time spent on weekdays (Monday-Friday summed) and weekend days (Saturday-Sunday summed). This may result in mis-reporting, due to it being harder and more time consuming to reflect then estimate the sum of time spent across multiple days, rather than recalling time spent on one day. Completing an interviewer-administered telephone survey may have amplified the pressure for participants to respond quickly, rather than taking the necessary time to work out the answer. In contrast, the physical activity survey items were phrased simplistically, such as per weekday and weekend day. This discrepancy may have resulted in overreporting sedentary behaviour and may explain why the sedentary behaviour variables generally had lower test-retest reliability scores than physical activity. Future research should use assessment items with consistent phrasing of wording and response options. This may increase participant comprehension of the survey items, particularly for the sedentary behaviour items used, and potentially improve the testretest reliability scores.

The main outcome variable for physical activity represented non-utilitarian and discretionary behaviour, because LPA and the time spent active in school and work domains was not assessed by the survey. Not assessing LPA, school or work physical activity time may not have affected the compliance of meeting guidelines since that is are based on MVPA. However, LPA is beneficial for health<sup>277, 278</sup> and most of adolescent's daily time is spent at school<sup>279</sup>; therefore, future research should assess LPA and physical activity in these domains, as well as leisure-time and transport.

#### Independent variables

A strength of the study was in applying the ecological model<sup>20, 21</sup>. This model is a useful framework to organise correlates according to levels of individual, social and environmental. Applying this model aids understanding of health behaviours by suggesting multiple influences in numerous categories to consider<sup>143</sup>. The findings can inform a multi-setting efficacious intervention with tailored strategies<sup>20, 21</sup>. Further, the study assessed a large range of independent variables from three levels of the ecological model<sup>20, 21</sup> as correlates, determinants and moderators of physical activity and sedentary behaviour. Previous similar studies<sup>149-155, 166, 280</sup> have failed to assess variables from three different levels of the ecological model<sup>20, 21</sup>. The comprehensive collection of variables assessed may reveal novel information about underlying influences on physical activity and sedentary behaviour during the transition out of secondary school. The large range of independent variables measured was informed by the qualitative work presented in the previous chapter, including social support, internal attributes (enjoyment, competence and self-

efficacy), priorities and habits (goal setting), technology access (electronic devices and televisions) and facilities.

Given the breadth of independent variables included, there may be a high probability of finding results by chance due to a high number (26) assessed. Furthermore, 11 of the independent variables were comprised of single survey items. This increases measurement error and, thus, decreases the reliability of the survey item, because a single item only captures one source of variance<sup>281</sup>. It is recommended that the use of single-item variables be minimised, with multiple-item variables preferred when available<sup>281</sup>. Single-item independent variables should be cautiously interpreted if they are physical activity or sedentary behaviour cross-sectional correlates or longitudinal determinants, or moderators of associations between situational transitions and physical activity or sedentary behaviour.

A potential limitation of the independent variables was that the individual variables were limited to psychological and cognitive attributes, rather than also including demographic and biological characteristics. This limits the comparability of findings with previous literature. However, as demographic and biological characteristics are the most studied correlates and determinants of physical activity and sedentary behaviour, further studying these may not be informative<sup>147</sup>. Compared to static demographic and biological variables, psychological and cognitive features are likely to be modifiable and, therefore, relevant for informing design of future health behaviour change interventions<sup>147</sup>.

A probable limitation of the environmental independent variables is that all were self-reported and not measured objectively, such as via Geographic Information System (GIS) software. Future research should include objective assessment tools such as GIS to increase data reliability and validity. Supporting this is a large study of 42-year-old Australian women (n=1 540) found poor agreement between self-report and GIS for neighbourhood physical activity facilities<sup>282</sup>. The neighbourhood environment survey items asked about the local area but did not define it. The definition of local area and neighbourhood may differ for each participant, based on individual perceptions; therefore, findings may differ if a consistent definition and distance from home is used. Future research of neighbourhood environment could define the neighbourhood scale using consistent units such as postal boundaries or 800m or 2km pedestrian catchment areas from home, as seen in other studies<sup>283, 284</sup>.

A further limitation of the environmental independent variables was that no built environment variables were measured, such as school or work. Future research should assess built environment characteristics, mass media advertisements, policies related to parks, and government investment in public recreation<sup>20</sup>. The built environment at school may be particularly important to older adolescents. Examples of characteristics to assess are building and stair design, policies for being active or sedentary during breaks and physical education classes, availability of sit-to-stand workstations, access policies for gymnasium facilities, support for active transport programs, pedestrian and bicycle facilities, and public transport access and investments<sup>20, 21</sup>.

Because of its ease of use, Cronbach's alpha coefficient was used to identify the internal consistency of the multi-component independent variables<sup>236</sup>. However, it has well-documented limitations including not being recommended for skewed or asymmetrical data<sup>285</sup> and some multi-component independent variables were skewed. Physical activity enjoyment and neighbourhood walking environment were negatively skewed. Therefore, the internal reliability scores for these should be interpreted with due consideration (0.74 and 0.73, respectively).

#### Conclusion

ProjectADAPT was a longitudinal study that followed a cohort of students during the transition out of secondary school. The sample was in the second last year of secondary school and there was a higher proportion of girls from urban and the highest tertile of socio-economic area. The self-reported measures assessed physical activity, sedentary behaviour and numerous independent variables. The test-retest reliability of the measures was found to be acceptable. The following chapters of this thesis (Chapters 5-8) will use ProjectADAPT data to develop findings from the qualitative interviews presented in Chapter 3 on how and why physical activity and sedentary behaviour change during the transition out of secondary school. Specifically, the following chapters will cross-sectionally examine physical activity and sedentary behaviour in late adolescence, and will identify individual, social and environmental correlates.

Chapter 5

Physical activity in late adolescence

# 5.1 Introduction

hapter 2 highlighted that physical activity is important for health because it is a modifiable risk factor for a range of health outcomes<sup>71-73</sup>. During adolescence, physical activity is positively associated with cardiorespiratory fitness and skeletal and muscular health, and negatively associated with cardiovascular disease risk factors<sup>74, 286</sup> and depressive symptoms<sup>80</sup>. Additionally, inadequate physical activity in childhood has negative health implications that track into adulthood, such as the future development of cardiovascular disease and comorbidities<sup>81-84, 157</sup>. Chapter 2 summarised the previous studies that have described physical activity in late adolescence. Only 6% of Australian 15-17-year-olds complied with National Physical Activity Guidelines<sup>15</sup>. Previous literature has not described different domains of physical activity in this age group. This chapter aims to address these gaps and develop understanding of physical activity in late adolescence.

Cross-sectional research has identified SEP<sup>151</sup>, parental support<sup>147</sup> and high-quality neighbourhood sport facilities<sup>155</sup> as being positively correlated with physical activity in adolescence. However, most previous studies sampled a wide age range that included 10-18-year-olds<sup>147, 151</sup> and some<sup>147</sup> assessed only one or two levels of the ecological model<sup>20, 21</sup>. In Chapter 3, perceived influences on physical activity were explored in one-on-one interviews with recent school leavers. Major influences on perceived physical activity were time use and social support. This chapter aims to highlight consistent and new findings by identifying individual, social and environmental correlates of discretionary physical activity.

# 5.2 Chapter aims

The specific aims of this chapter are to:

- 1. Present the descriptive epidemiology of physical activity patterns in late adolescence; and
- 2. Identify individual, social and environmental correlates of discretionary physical activity in late adolescence.

# 5.3 Methods

This chapter used cross-sectional baseline data from ProjectADAPT. The types of physical activity described in this chapter include total duration of physical activity (mins/day), time spent in active transport, LTPA, school-related physical activity, occupational physical activity and adherence to National Guidelines<sup>109</sup>. A total of 26 independent variables were tested as correlates of physical activity. Chapter 4 described the data cleaning and management of the dependent and independent variables (section 4.3.4). The following section outlines aspects of data analysis relevant to this chapter.

#### 5.3.1 Data analysis

The continuous variables of discretionary physical activity, active transport and LTPA were described as means and SD (Aim 1). These were compared by subgroups using t-tests for binary variables (gender and remoteness) and one-way ANOVA for area-level SEP. Some t-test assumptions were violated since the cohort was not a simple random sample and one group (gender) had a size comparison ratio >1.5. Despite this, the outcome variables were not transformed because independent t-tests are valid for non-normally distributed data using large samples<sup>287</sup>. Categorical variables of school-related physical activity, occupational physical activity and adherence to National Guidelines<sup>109</sup> were described using tabulation and compared by subgroups using two-way tables of frequencies reporting Pearson's chi-squared statistics.

Linear regression analyses were used to address Aim 2. Discretionary physical activity was checked for normality using a histogram and data were positively skewed. However, it was not transformed because the residuals of the discretionary physical activity regression analyses were checked and normality assumptions were met<sup>288, 289</sup>. As some participants were recruited from secondary schools, clustering at the school level was accounted for a priori for all participants as a design factor in models using the cluster command. Confounders were identified using a method for purposeful selection of variables in regression models<sup>290</sup>, specifically, a change in estimates<sup>291</sup>. Potential confounders tested were age, gender (woman, man or other), maternal and paternal education (tertiary, technical or trade school certificate, high school, some high school, primary school, never attended school), English as the primary language spoken at home, birth country (Australia or other), remoteness and
area-level SEP. Firstly, a simple linear regression model was used that included the dependent variable (discretionary physical activity) and each independent variable, accounting for clustering by school. These models were then repeated, adding in each of the potential confounders in separate models, seeking a  $\pm 10\%$  change in the correlation coefficient from the simple linear regression models. Gender, maternal education, English as the primary language spoken at home and area-level SEP were determined to be confounders, and all subsequent regression models adjusted for these socio-demographic characteristics.

After identification of confounders, 26 separate linear regression models were run, one for each independent variable, adjusting for the four confounders. Significant correlates were added into a 'fully-adjusted' multivariable model to identify the key correlates. Multicollinearity of the fully-adjusted model was checked by reviewing the variance inflation factors. If high variance inflation factors were identified  $(>5.0)^{292}$ , correlation matrices were constructed to determine which variable/s to exclude and the fully-adjusted model was repeated excluding these variables.

# 5.4 Results

# **5.4.1** Descriptive epidemiology of physical activity patterns in late adolescence

This section presents and describes various aspects of physical activity patterns in late adolescence (Aim 1). Table 5.1 presents the average duration of LTPA, active transport and discretionary physical activity. Participants reported just over 1.25 h/day of discretionary physical activity. Girls reported 15 mins/day less discretionary physical activity than did boys (p<0.01). Almost half of the participants (45%) met the National Guidelines<sup>109</sup>.

On average, participants reported spending 51 mins/day in LTPA (Table 5.1), comprised of walking for 12 mins/day, other MPA for 16 mins/day, and VPA for 22 mins/day. Girls reported 3 mins/day less MPA, 10 mins/day less VPA and 13 mins/day less total LTPA than did boys (p<0.05). Those who lived in the mid tertile of socio-economic area reported 4 mins/day more leisure-time walking than those who lived in the highest tertile (14.3±23.9 mins/day vs 10.6±17.1; p=0.042).

Mean duration of active transport was 26 mins/day, comprised of 3 mins/day of cycling and 23 mins/day of walking for transport (Table 5.1). Girls reported 3 mins/day less cycling for transport than did boys (p<0.001). Those who lived in the mid tertile of socio-economic area reported more time in active transport (29.2±36.7 mins/day vs 23.5±25.6, p=0.031) and cycling for transport (4.0±15.6 mins/day vs 1.9±7.1, p=0.020) than those who lived in the highest tertile.

				Gender	
	n	Overall	Girls	Boys	р
Discretionary PA	1 005	76.6±72.3	72.6±68.5	88.1±81.1	0.003
LTPA	1 009	$50.8 \pm 55.9$	47.3±51.9	$60.8 \pm 65.0$	0.001
Walking	1 013	$12.4{\pm}20.8$	$12.4\pm20.5$	12.3±21.7	0.945
Other MPA	1 017	16.4±23.7	15.5±21.5	19.0±29.1	0.043
VPA	1 017	$21.8 \pm 31.1$	$19.2 \pm 28.7$	$29.4 \pm 36.1$	0.000
Active transport	1 016	$25.7 \pm 30.4$	25.2±30.3	27.3±30.7	0.331
Cycling for transport	1 019	$2.7{\pm}10.7$	$1.8 \pm 8.1$	$5.3 \pm 15.6$	0.000
Walking for transport	1 018	$23.0\pm27.9$	23.4±29.0	$21.9 \pm 24.9$	0.469

Table 5.1Duration of discretionary physical activity, LTPA and active transport(mean mins/day±SD) in late adolescence

Independent t-test by gender.

Participation in school-based physical activity in late adolescence is presented in Table 5.2. The main activity during recess and lunch breaks for girls was sitting. Boys reported almost even proportions of sitting, standing, walking and MPA during recess and lunch breaks. The gender differences were statistically significant. Almost three quarters of participants reported no physical education classes each week (these classes are not compulsory during the final two years of schooling in most secondary schools in Victoria, Australia). A greater proportion of girls compared to boys participated in school sport. Most participants were not on a school sport team. Arealevel SEP differences were the proportion who participated in school sport (lowest tertile 9.8% vs mid 10.6% vs highest 22.6%; p=0.001) and the main activity during morning recess and lunch break (mostly walk: lowest 21.0% vs mid 23.5% vs highest 13.2%; p=0.010).

			Gende	r
	Overall	Girls	Boys	
	%	%	%	р
Main activity level during school breaks, $n=1.022$				
Mostly sit/stand	72.7	80.9	49.4	
Mostly walk	18.1	17.4	20.2	
Mostly MPA	7.1	1.3	23.2	
Mostly VPA	2.2	0.4	7.1	
				0.000
PE classes/wk, $n=1$ 022				
0	73.2	72.7	74.5	
1	11.6	12.3	9.4	
2-3	10.5	10.2	11.2	
≥4	4.8	4.8	4.9	
				0.762
Participate in school sport, $n=1$ 022				
Yes		40.3	49.4	
	42.8			0.010
School sport teams count, n=439				
0	59.5	61.9	52.4	
1	19.3	19.0	20.2	
2-3	17.4	14.9	24.0	
≥4	3.8	4.1	3.0	
				0.348

Table 5.2	School-related	nhysical	activity	in late	adolescence
1 abic 3.4	School-Iclaicu	physical	activity	in rate	autorescence

*Pearson's*  $\chi^2$  *test of significance by gender between school activity levels.* 

Participation in occupational physical activity in late adolescence is presented in Table 5.3. Just over a third of participants reported mainly sitting or standing at work, while just under a fifth mainly walked at work. Few participants performed heavy labour at work and there was a much higher proportion of boys reporting physically demanding work than girls.

			Gender	r
	Overall	Girls	Boys	
	%	%	%	р
Main activity level at work, $n=1 012$				
Not working	42.0	40.8	45.5	
Mostly sit/stand	35.5	37.5	29.5	
Mostly walk	17.1	18.9	12.1	
Mostly heavy labour or physically demanding	5.4	2.8	12.9	
				0.000

#### **Table 5.3**Occupational physical activity in late adolescence

*Pearson's*  $\chi^2$  *test of significance by gender between employment activity levels.* 

#### 5.4.2 Correlates of discretionary physical activity in late adolescence

In the partially-adjusted linear regression models, most variables (15) were associated with discretionary physical activity (Table 5.4). Multicollinearity of the fully-adjusted model was checked and social support for physical activity from friends/colleagues was excluded due to a high variance inflation factor ( $>5^{292}$ ). The final model explained 34% of variance in discretionary physical activity (Table 5.4). Of the 14 variables included in the final model, five remained significant. These were physical activity enjoyment, goal setting and self-efficacy, sedentary behaviour discouragement from friends or colleagues, and social network (Aim 2). The directions of the associations were all positive.

Discretionary physical activity was 2.6 mins/day, 5.2 mins/day or 2.5 mins/day higher with each additional unit increase in physical activity enjoyment, physical activity goal setting or physical activity self-efficacy, respectively. Discretionary physical activity was 5.7 mins/day higher with each additional unit increase in friends or colleagues discouraging sedentary behaviour and 0.7 mins/day higher for each additional person in the social network count.

	Partially-adjusted	Fully-adjusted
	B (95%CI)	B (95% CI) n - 907
Individual variables (scores)	D ()5/0CI)	D ()5/0CI), <i>n=</i> 907
PA enjoyment	10.6 (8.6, 12.6)***	2.6 (0.5, 4.8)*
PA goal setting	8.5 (7.0, 10.0)***	5.2 (3.6, 6.8)***
PA competence	20.6 (16.0, 25.3)***	3.5 (-1.1, 8.1)
PA self-efficacy	6.0 (4.9, 7.2)***	2.5 (1.3, 3.8)***
TV avoidance self-efficacy	1.7 (0.7, 2.7)**	-0.5 (-1.6, 0.7)
Social variables		
Family (scores):		
E-games co-participation	-1.4 (-7.0, 7.3)	-
PA co-participation	8.0 (3.9, 12.1)***	0.7 (-3.7, 5.2)
TV/DVDs co-participation	-2.4 (-6.9, 2.2)	-
PA social support	5.1 (3.4, 6.8)***	-0.7 (-3.0, 1.6)
SB discouragement	1.8 (-2.5, 6.1)	-
Friends/colleagues (scores):		
E-games co-participation	-2.1 (-7.1, 2.8)	-
PA co-participation	13.4 (9.2, 17.7)***	-0.7 (-4.1, 2.7)
TV/DVDs co-participation	0.2 (-4.6, 4.9)	-
PA social support	<b>6.8</b> ( <b>4.8</b> , <b>8.7</b> )***	-
SB discouragement	11.2 (5.2, 17.1)***	5.7 (0.9, 10.4)*
Social network count	0.9 (0.2, 1.5)**	0.7 (0.1, 1.3)*
Gym membership	30.8 (20.2, 41.3)***	7.7 (-0.2, 15.6)
Environmental variables		
Home environment:		
E-devices, no. of	-0.3 (-4.3, 3.7)	-
PA equipment, no. of	9.8 (4.4, 15.2)***	2.0 (-1.7, 5.7)
TV, no. of	1.2 (-0.2, 2.6)	-
Neighbourhood environment (sco	ores):	
Noise	1.7 (-3.5, 7.0)	-
Walking environment	3.9 (2.4, 5.5)***	-0.2 (-1.3, 1.6)
Safety	2.2 (-0.4, 4.9)	-
Social cohesion	3.1 (-0.0, 6.2)	-
Land use mix diversity	2.0 (-0.4, 4.5)	-
Recreation facilities	6.9 (3.7, 10.2)***	2.0 (-0.9, 5.0)
$R^2 = 0.34$		

**Table 5.4**Partially- and fully-adjusted<sup>1</sup> associations between independentvariables and discretionary physical activity (mins/day) in late adolescence

<sup>1</sup> Adjusted for gender, maternal education, English as the primary language spoken at home and area-level SEP. Accounted for clustering by school.

\**p*<0.05, \*\**p*<0.01, \*\*\**p*<0.001: Linear regression models.

# 5.5 Discussion

This chapter described physical activity in late adolescence, and identified individual, social and environmental correlates of discretionary physical activity. On average, participants reported engaging in 77 mins/day of discretionary physical activity and 45% complied with guidelines<sup>109</sup>. Although most discretionary physical activity was comprised of LTPA, participants reported spending on average 26 mins walking or cycling for travel purposes. The key correlates of discretionary physical activity were physical activity enjoyment, goal setting and self-efficacy, friends or colleagues discouraging sedentary behaviour and social network count, each positively associated.

There appears to be much scope in this age group to increase physical activity during leisure-time and for transport, as more than half of the sample did not comply with guidelines. This is a concern as it is recommended to accumulate  $\geq 60 \text{ mins/day}$  of MVPA for optimal health benefits<sup>109</sup>. There also appears to be a need to increase physical activity at schools, given that only two-fifths of participants engaged in school sport and few were mostly active during school breaks. This appears to be a missed opportunity, especially considering adolescents sit 70% of school hours and 75% of class time<sup>279</sup>. Further, a longitudinal study found that Australian 14-year-olds (50% males) who were active during school recess had more device-assessed total physical activity over time<sup>293</sup>, suggesting that school breaks are critical.

Given variability in assessment tools between studies, it is important to compare findings of those using similar measures. Physical activity in this study was generally higher compared to other studies that used the IPAQ or IPAQ-A. What this study found was similar to those of a large study of Brazilian 14-18-year-olds (n=2 545; 51.5% active; 46.1% girls vs 58.6% boys; defined as  $\geq$ 300 mins/wk)<sup>294</sup>. However, this study's findings were higher for girls and similar for boys compared to Spanish 15-16-year-olds (40% girls active; 47.2% boys active; 'active' computed based on any LTPA)<sup>295</sup>. Lastly, this study's findings were higher than 15-17-year-old Czech boys and girls and Polish girls, and similar to Polish boys (Czech: 20.1% girls, 32.3% boys; Polish: 25.2% girls, 40.5% boys; based on 60 mins/day MVPA)<sup>296</sup>. Numerous studies of adolescents that use IPAQ have reported average duration in metabolic equivalent (MET)/mins/wk<sup>296-299</sup> as per the IPAQ protocol, impairing comparisons with this study.

Sufficient activity and meeting guidelines have been operationalised differently in studies which makes it difficult to compare. Both the average duration and compliance with guidelines was higher in this study than what has been reported for 15-17-year-olds in the Australian Health Survey<sup>15, 300</sup>. Key study differences include a measure that assessed activity on each of the previous seven days, whereas this survey measured activity in a usual week. Secondly, this study measured time spent walking during leisure-time and other MPA which may have contributed to more total MVPA being reported compared to the Australian Health Survey<sup>15, 300</sup>. Lastly, in the Australian Health Survey, some MVPA was mis-categorised to LPA due to participant confusion and excluded from total MVPA (which is used to generate guideline compliance)<sup>15, 300</sup>. When reviewing studies that used accelerometry, this study's findings were higher than adolescents in Australia  $(<6\%)^{301}$  and the USA  $(8\%)^{302}$ . This may reflect over-reporting in this self-report survey, especially since accelerometry measures most types and domains of physical activity, including LPA and time in physical activity at school and work, whereas this study only measured active transport and leisure-time MVPA. Regardless of how physical activity is measured, it remains clear that older adolescents are not engaging in enough physical activity.

Some differences by sub-groups were found for gender. Girls reported less time than did boys in all physical activity except for walking for transport and leisure, and also reported lower engagement in school sport and active school breaks. The finding that girls were less active than boys is consistent with recent studies of older adolescents using IPAQ-A or IPAQ<sup>294-296, 299</sup>, as well as across all ages internationally<sup>303</sup>. Numerous reasons have been proposed for these gender differences. Firstly, gender bias in sport and socialised gender roles contributes to the stereotypes that VPA is masculine and girls should maintain aesthetic appearance (including not sweating), resulting in adolescent girls feeling self-conscious, embarrassed and anxious to avoid being shamed and objectified while being active<sup>304, 305</sup>. Secondly, girls have lower proficiency in fundamental movement and object-control skills<sup>306, 307</sup>, and less interest (in sport), spectatorship, competitiveness and risk-taking<sup>308</sup>, and alternate time-use preferences at this age (more studying, socialising and chores) than boys<sup>309</sup>. In Chapter 3, some participants reflected on why they were not active during secondary school and some women proposed the reasons were the pressures and demands of a large study load and how physical education classes were no longer

compulsory. Although this current study combined boys and girls due to the lower proportion of boys in the sample, future studies could assess correlates of girls and boys separately.

When reviewing findings by sub-groups, adolescent girls are clearly an important target group for physical activity intervention. Although it is encouraging to see no disparity across geographic areas, findings for the components of discretionary physical activity were mixed. This highlights that older adolescents living in various socio-economic areas accumulate physical activity differently, with the clearest example being more school sport in highest tertile of socio-economic area than the lowest or mid. This may reflect better facilities and higher financial resources among these families, consistent with a study from the USA<sup>310</sup>. Other examples were more active transport in mid than highest tertile of socio-economic area, more leisure-time walking in mid than highest tertile of socio-economic area, and more walking during recess and lunch breaks (as opposed to sitting, standing, MPA or VPA) in lowest and mid than highest tertile of socio-economic area. One narrative review<sup>311</sup> of systematic and narrative reviews argued there is no clear pattern across early, mid and late adolescence that physical activity is associated with SEP. Meanwhile, recent studies from Australia and Lithuania have suggested there is a relationship between physical activity and SEP during late adolescence<sup>312, 313</sup>. Future research should explore variance in outcomes by additional groups, such as individual SEP based on maternal/paternal income, education and employment.

A strength of this current study is a wide range of individual, social and environmental correlates assessed. In general, most of the correlates of physical activity in the partially-adjusted models have been reported previously<sup>311</sup>, although, the age range in the previous studies was broader than this current study. In this study, five variables remained significant; physical activity enjoyment, goal setting and self-efficacy, friends or colleagues discouraging sedentary behaviour, and social network count. However, the fully-adjusted model did not find that the home or neighbourhood environments were associated with physical activity in late adolescence, despite the ecological model<sup>20, 21</sup> positing that individual, social and environmental levels are important. This is unsurprising, given that the individuallevel is proximal and reflects what adolescents control. The score for physical activity goal setting reflected planning ahead for physical activity, including amount of physical activity, and scheduling in general and with other people. Those with high goal setting scores are likely to prioritise physical activity, valuing it highly and regard it as important enough to make time for, and they might have people within their social networks who are highly active who they meet with often. Goal orientation is consistently identified as a correlate of physical activity during adolescence<sup>311</sup>. Similarly, self-efficacy has been consistently reported as positively correlated with physical activity during adolescence in five reviews since 2005<sup>311</sup>. High physical activity self-efficacy may represent being capable at overcoming challenges and continuing to pursue a goal despite setbacks, which has been linked to having high success markers including self control and grit<sup>314</sup>. There are other types of self-efficacy that should be explored in future studies.

One review found that enjoyment was not always identified as a correlate of physical activity among adolescents<sup>311</sup>; however, one review found a positive association for adolescent girls<sup>315</sup>. Therefore, gender proportion of this current study may have driven this finding. Future studies could examine the efficacy of tailoring strategies that boost enjoyment to older adolescents within physical activity interventions. Testing social network count as a correlate is novel and the finding may imply that those who are active are extraverted, as suggested by a meta-analysis and systematic review in adults<sup>316</sup>. Although the direction of association cannot be ascertained from correlation, another meta analysis and systematic review of adults argued the impact of personality traits on physical activity is not bidirectional<sup>317</sup>. Further research is needed as to whether physical activity interventions may benefit from tailoring to target groups at high-risk of low physical activity with personality-traits like introversion.

In adolescence, social support has been identified as an important correlate of physical activity partly due to relying on families for resources, transport and decision-making. Findings from this study suggest that responsibility for own time, becoming independent and some decision-making are emerging in late adolescence. Friends or colleagues discouraging excessive sedentary behaviour is a specific aspect of social support, with previous studies showing mixed findings as a correlate of physical activity during adolescence<sup>311</sup>. This suggests that seeking approval within peer relationships may be a particularly strong motivation and future research could

examine incorporating a buddy system to boost peer support into physical activity interventions of older adolescents.

The findings in this chapter are consistent with the qualitative research results presented in Chapter 3. When participants in the interviews reflected on what facilitated or inhibited physical activity and sedentary behaviour, poor time management was considered a major barrier. Furthermore, the recent school leavers also described self-efficacy, and spoke of strategies to overcome a range of challenges that were related to demands on time. Although enjoyment and having enough social support were similarly described as influential, so was having enough sleep which was not assessed in this current study. This suggests that while sleep is important, older adolescents are unlikely to be active without a supportive social environment and ability to manage time. Current findings imply that physical activity interventions for this age group need to include strategies to increase enjoyment, goal setting, supportive peer relationships and overcoming barriers such as poor time management.

Most of the five key correlates were related to physical activity (enjoyment, goal setting and self-efficacy), consistent with behavioural-specificity literature<sup>318</sup>. Future research needs to examine whether the five key correlates are also longitudinal determinants (Chapter 8) of physical activity or whether different influences impact behaviour over time. Furthermore, future research is needed to identify whether some of the distal independent variables assessed by this current study are moderators (Chapter 8) or mediators of physical activity in late adolescence.

#### **Strengths and limitations**

Strengths of this study include the description of a range of types of physical activity in an infrequently studied age group. Furthermore, school, sport and occupational physical activity are not commonly examined during late adolescence. Applying the ecological model<sup>20, 21</sup> meant a broad range of individual, social and environmental correlates were tested and highlighted that individual and social correlates seem to be more important to discretionary physical activity in late adolescence than environmental correlates. Future studies of late adolescence should also assess the school and work environments, including attributes such as pedestrian and bicycle facilities, support for active transport programs such as safe routes to school, public transport access and investments, and building and stair design<sup>21</sup>. Lastly, examining

sedentary behaviour-related individual, social and environmental correlates (such as television avoidance self-efficacy; number of electronic devices; number of televisions; family, friends or colleagues co-participating in electronic games or television, or discouraging sedentary behaviour) of physical activity was novel and resulted in a unique correlate finding (friends or colleagues discouraging sedentary behaviour).

Physical activity in late adolescence may be underestimated as time in LPA in any domain and time in physical activity at school and work were not included in the physical activity measure. Also not examined in this study were correlates of LTPA and active transport separately, even though other studies of older adolescents have<sup>193, 319-322</sup> and this may clarify what is driving the associations found. Active transport and LTPA were combined because the former made a substantial contribution to discretionary physical activity in this study. This suggests that discretionary physical activity would substantially increase due to more active transport, hence more promotion of the National Guidelines is needed due to its recommendations to walk, cycle or skateboard safely for short trips instead of travelling by car<sup>109</sup>.

Environmental variables were not found to be the most important correlates of discretionary physical activity in this study and not testing them as correlates of active transport may partially explain  $why^{318}$ . This stems from the suggestion that behavioural-specificity may be important for environmental variables<sup>318</sup>. For example, neighbourhood perceptions (land use mix diversity, parks, pedestrian footpaths, shops nearby and traffic speed) were associated with walking or cycling to school in a large cross-sectional study of Irish 15-17-year-olds  $(n=2 \ 159)^{319}$ . Findings from different aged samples (adults<sup>323</sup> and older adults<sup>324, 325</sup>) also suggest that the neighbourhood environment may have stronger associations with walking for transport rather than discretionary physical activity. Future surveys need to be designed specifically for domain-specific physical activity correlate analyses, for example, assessing independent variables specific to active transport. That new knowledge may inform the design of behaviour-relevant interventions and domainspecific guidelines for active transport in late adolescence. Additionally, future research is needed that examines the correlates of meeting guidelines, acknowledging the limitation of a loss of power from dichotimising the continuous outcome variable<sup>326</sup>.

The cross-sectional nature of this study means that it cannot infer causality or directionality. It may be that more physical activity leads to more enjoyment of being active, goal setting, the number of friends, and self-efficacy, for example. The study relied on self-reported measures and correlates, however the test-retest reliability of these were established and were acceptable (Chapter 4). Further, all participants were from Year 11 which, in the Victorian context, is a more studious time compared previous years since scores impact university-entrance. As described in Chapter 4, the sample comprised mostly girls which may have biased findings. Future research using a sample with more equal gender proportions may have different findings. Although an established measure of physical activity was used (IPAQ), a different measure may elicit alternate findings.

#### Conclusion

This study found 45% of 17-year-olds met National Guidelines which suggests that there is room for improvement. At-risk target groups and behaviours were identified for further interventions. Girls are a particularly important target group for intervention as they reported engaging in less than boys across most types of physical activity. Key correlates appear to be physical activity enjoyment, goal setting and self-efficacy, social network count, and friends or colleagues discouraging sedentary behaviour. The next chapter builds on what this study found by examining patterns and correlates of sedentary behaviour in late adolescence.

Chapter 6

# Sedentary behaviour in late adolescence

# 6.1 Introduction

S edentary behaviour is defined as an activity that involves sitting, reclining, or limited movement, and low energy expenditure during waking hours<sup>50</sup>, and is distinct from physical activity. As described in Chapter 2, different types of sedentary behaviour have associations with health in adolescence that are distinct from physical activity<sup>99, 101-104</sup>. Notably, cardiovascular disease risk factors were reported in some studies<sup>99</sup>, the presence of which during adolescence is a serious concern as it tracks into adulthood<sup>84, 105, 106</sup>. Recent compositional analyses consider the interdependence of sedentary behaviour and physical activity; for example, an international meta-analysis found that the threshold for a dose-response relationship with all-cause mortality risk from sitting starts at 7MET h/day or watching television starts at 3MET h/day without physical activity compensation<sup>14</sup>. That study argued that decreasing sitting time and increasing physical activity are equally important to offset all-cause mortality risk<sup>14</sup>.

There is a plethora of mobile electronic entertainment options<sup>272</sup>. This has resulted in an appealing and comfortable sedentary environment that also includes convenient labour-saving devices<sup>21</sup>. The accompanying increase in screen addiction, in the form of gaming, social media and texting<sup>327</sup>, has harmful effects on the still-developing adolescent brain, including poor academic achievement<sup>328</sup>, social and emotional impairment, sleep interference<sup>329</sup> and attention problems<sup>330-332</sup>. Subsequently, as Chapter 2 described, it is common internationally for few adolescents to meet recreational screen-time guidelines<sup>15, 116, 124</sup>. This age represents an important time for intervening as it may be a 'teachable moment' during the life course when individuals may implement health-promoting or negative habits that track into adulthood.

The qualitative findings from Chapter 3 suggest that sedentary behaviour during late secondary school is common. Recent school leavers described using mobile phones, studying and using a computer were popular types of sedentary behaviour, consistent with previous research<sup>15</sup>. This chapter aims to progress those findings by describing total and types of sedentary behaviour, including sedentary transport, sedentary behaviour (excluding transport), recreational screen time, school-related sedentary behaviour and occupational sedentary behaviour. As described in Chapter 2, few studies have examined correlates of total sedentary behaviour in late adolescence from three levels of the ecological model<sup>20, 21</sup>. As argued in Chapter 5, it is possible

to engage in both sedentary behaviour and MVPA during the day<sup>333</sup>. However, sedentary behaviour and LPA are strongly negatively associated (displace each other)<sup>223, 224</sup>. Therefore, correlates traditionally associated with physical activity may affect total sedentary behaviour and warrant testing.

# 6.2 Chapter aims

The specific aims of this chapter are to:

- 1. Present the descriptive epidemiology of sedentary behaviour patterns in late adolescence; and
- 2. Identify individual, social and environmental correlates of total sedentary behaviour in late adolescence.

## 6.3 Methods

Cross-sectional sedentary behaviour baseline data forms the basis of this chapter. Mean total sedentary behaviour (h/day) is described and types of sedentary behaviour including time spent sedentary at school, work, home, when traveling and during leisure-time. Also examined is recreational screen time (h/day), meeting the National Guidelines<sup>109</sup>, sedentary behaviour during breaks at school and occupational sedentary behaviour. The same independent variables examined in Chapter 5 were tested in this current chapter. Chapter 4 described the data cleaning and management of each of these variables (section 4.3.4). The following section details aspects of data analysis relevant to this chapter.

#### 6.3.1 Data analysis

To address Aim 1, continuous variables were described using means and SD. Prior to t-tests, normality of total sedentary behaviour was checked using a histogram and data were normally distributed. Continuous variables were compared by subgroups using t-tests for binary subgroups (gender and remoteness) or one-way ANOVA for area-level SEP. Categorical variables (school-related sedentary behaviour, occupational sedentary behaviour, adherence to National Guidelines<sup>109</sup> and non-active games use) were described using tabulation and compared by subgroups using Pearson's chi-squared statistics.

Linear regression was used to address Aim 2. Firstly, partially-adjusted models were initially run with total sedentary behaviour as the dependent variable and each separate independent variable (totalling 26 models). The residuals of the regression models were checked and were normally distributed, meeting an assumption of regression models<sup>289</sup>. Confounders were identified using the same process described in Chapter 5 (the purposeful selection of variables in regressions method, specifically, a change in estimates<sup>291</sup>). Clustering by school was accounted for a priori, as a design factor. Gender, paternal education, English as the primary language spoken at home, birth country and remoteness were determined to be confounders and partially- and fully-adjusted regression models adjusted for these.

Secondly, a corresponding fully-adjusted model was run, that included only the significant independent variables from the partially-adjusted linear regression models. Multicollinearity of the fully-adjusted model was checked by reviewing the variance inflation factors. If high variance inflation factors were identified ( $>5.0^{292}$ ), correlation matrices were constructed to determine variable/s to exclude. Lastly, the fully-adjusted model was repeated without the identified variables.

# 6.4 Results

# 6.4.1 Descriptive epidemiology of sedentary behaviour in late adolescence

In this section, sedentary behaviour patterns in late adolescence are presented and described (Aim 1). Table 6.1 presents information on sedentary behaviour at school and work. Just over half of participants reported mainly sitting during school breaks (recess and lunchtime), although that was driven by girls because a much lower proportion of boys than girls mainly sat during breaks. The proportion of participants who reported mainly sitting during school recess and lunch breaks differed by tertiles of area-level SEP (54% lowest vs 48% mid vs 57% highest; p=0.010). Very few of those employed reported mainly sitting at work. The proportion of boys who reported mainly sitting at work this was double that of girls.

Table 6.1	Proportion of participants reporting 'mainly sitting' during school
breaks and at v	vork in late adolescence

			Gende	r
	Overall	Girls	Boys	
	%	%	%	р
School breaks, $n=1$ 022	53.3	63.9	23.2	0.000
Main activity level at work, $n=1 012$				
Did not work	42.0	40.8	45.5	
Mostly sit	3.1	2.4	4.9	
Mostly stand/walk/heavy labour	54.9	56.8	49.6	
				0.000

*Pearson's*  $\chi^2$  *test of significance by gender.* 

Table 6.2 details the mean total sedentary behaviour, sedentary behaviour (excluding transport), sedentary transport, and recreational screen time in late adolescence. Overall, participants reported spending just under 9 h/day engaged in total sedentary behaviour, comprised of 8h at school, work, home, and during free time, and a little under 1h traveling in a motor vehicle. Total recreational screen time was around 3.5 h/day. This was comprised of 1.5h of watching television, DVDs or videos, almost 2h of using a computer, laptop or tablet, and 14 mins of playing electronic games. Correspondingly, a third (35%) of the overall sample complied with the National

Guidelines<sup>10</sup>, with more girls complying than boys (p<0.01). Non-active electronic games use was reported by 26% of the cohort (15% girls vs 57% boys; p<0.001).

Sedentary behaviour differed according to gender, remoteness and area-level SEP. Firstly, girls reported significantly more total sedentary behaviour, and more sitting at school, work, home and during free time than did boys (Table 6.2). Furthermore, girls reported significantly less total recreational screen time and electronic games than did boys. Sedentary transport or watching television did not differ by gender. Secondly, those who lived in an urban area reported significantly more total sedentary behaviour (9.0±2.9 h/day vs  $8.4\pm2.7$ ; p=0.003) and sedentary behaviour (excluding transport;  $8.1\pm2.8$  h/day vs  $7.5\pm2.6$ ; p=0.001) than those who lived rurally. Lastly, those who lived in the lowest tertile of socio-economic area watched significantly more television, DVDs or videos than those who lived in the highest tertile ( $1.7\pm1.8$  h/day vs  $1.3\pm1.2$ ; p<0.001).

				Gender	
	n	Overall	Girls	Boys	р
Total SB	1 003	$8.8 \pm 2.9$	$9.0{\pm}2.8$	8.3±3.0	0.001
SB (excluding transport)	1 005	$8.0{\pm}2.7$	8.1±2.7	$7.4 \pm 2.8$	0.001
Sedentary transport	1 0 2 0	$0.9\pm0.8$	$0.9\pm0.8$	$0.9\pm0.7$	0.478
					• • • <b>-</b>
Total recreational screen time	956	$3.5 \pm 2.6$	$3.3\pm2.5$	$3.8 \pm 2.7$	0.007
Watching TV, DVDs or videos	986	$1.5 \pm 1.4$	$1.5 \pm 1.4$	$1.4{\pm}1.4$	0.681
Computer, laptop or tablet	983	$1.8 \pm 1.6$	$1.8 \pm 1.5$	$2.0{\pm}1.7$	0.115
E-games	1 012	$0.2\pm0.7$	0.1±0.4	$0.6 \pm 1.2$	0.000
-					

Table 6.2Duration of total sedentary behaviour and recreational screen time $(mean h/day\pm SD)$  in late adolescence

Independent t-test by gender.

#### 6.4.2 Correlates of total sedentary behaviour in late adolescence

Results of the linear regression models examining associations between the independent variables and total sedentary behaviour in late adolescence are presented in Table 6.3. Most (17) of the 26 independent variables were significantly negatively associated with total sedentary behaviour in the partially-adjusted models. Multicollinearity of the fully-adjusted model was checked and social support for physical activity from friends/colleagues was excluded due to a high variance inflation factor ( $>5^{292}$ ). The fully-adjusted model explained 13% of the variance in total sedentary behaviour. Of the 16 variables included in the final model, one remained significantly associated with total sedentary behaviour, physical activity goal setting (Aim 2). This finding suggests that for each additional unit increase for physical activity goal setting, Year 11 students engaged in 5 mins/day less total sedentary behaviour.

	Partially-adjusted	Fully-adjusted
	model	model
	B (95%CI)	B (95%CI), <i>n</i> =901
Individual variables (scores)		
PA enjoyment	-13.5 (-18.2, -8.8)***	-3.1 (-9.5, 3.4)
PA goal setting	-11.0 (-14.3, -7.7)***	-5.1 (-8.6, -1.7)**
PA competence	-32.1 (-42.7, -21.6)***	-9.9 (-24.7, 4.8)
PA self-efficacy	<b>-6.9</b> ( <b>-9.2</b> , <b>-4.6</b> )***	-0.9 (-3.7, 1.9)
TV avoidance self-efficacy	-2.8 (-5.4, -0.3)*	-0.2 (-2.4, 2.1)
Social variables		
Family (scores):		
E-games co-participation	10.2 (-3.6, 23.9)	-
PA co-participation	-24.0 (-32.4, -15.6)***	-8.0 (-21.3, 5.3)
TV/DVDs co-participation	2.6 (-9.2, 14.3)	-
PA social support	-10.5 (-14.1, -6.9)***	0.2 (-6.9, 7.4)
SB discouragement	-12.9 (-21.6, -4.2)**	-2.4 (-10.9, 6.1)
Friends/colleagues (scores):		
E-games co-participation	6.8 (-4.6, 18.1)	-
PA co-participation	-21.5 (-31.1, -11.8)***	-2.1 (-13.0, 8.8)
TV/DVDs co-participation	-2.0 (-8.8, 12.9)	-
PA social support	-10.2 (-13.7, -6.8)***	-
SB discouragement	-23.5 (-34.7, -12.4)***	-8.7 (-20.8, 3.4)
Social network count	-1.2 (-2.1, -0.3)*	-0.2 (-1.0, 0.6)
Gym membership	-47.3 (-69.9, -24.7)***	-15.5 (-39.2, 8.1)
Environmental variables		
Home environment (counts):		
E-devices, no. of	1.4 (-9.3, 12.1)	-
PA equipment, no. of	-16.6 (-28.1, -5.2)**	-3.2 (-14.4, 7.9)
TVs, no. of	-2.6 (-6.5, 1.2)	-
Neighbourhood environment (sco	ores):	
Noise	-1.8 (-13.6, 9.9)	-
Walking environment	-8.3 (-12.2, -4.3)***	-2.8 (-7.1, 1.4)
Safety	-9.3 (-15.9, -2.8)**	-7.2 (-15.1, 0.7)
Social cohesion	-9.1 (-16.4, -1.8)*	5.8 (-3.8, 15.5)
Land use mix diversity	-3.9 (-9.4, 1.6)	-
Recreation facilities	-4.0 (-12.5, 4.6)	-
$R^2=0.13$		

**Table 6.3**Partially- and fully-adjusted associations<sup>1</sup> between independentvariables and total sedentary behaviour (mins/day) in late adolescence

<sup>1</sup> Adjusted for gender, paternal education, English as the primary language spoken at home, birth country and remoteness. Accounted for clustering by school.

\**p*<0.05, \*\**p*<0.01, \*\*\**p*<0.001: Linear regression models.

# 6.5 Discussion

The aims of this chapter were twofold. Firstly, to describe sedentary behaviour of a cohort of older adolescents. Secondly, to identify individual, social and environmental correlates of total sedentary behaviour. It is novel to examine a range of correlates of sedentary behaviour in late adolescence. This cohort reported accumulating approximately 9 h/day of total sedentary behaviour. Only 35% complied with National Guidelines<sup>109</sup>. Fully-adjusted linear regression models identified that physical activity goal setting was negatively correlated with total sedentary behaviour in late adolescence.

In this sample, adolescents near the end of secondary school spent a considerable amount of the day engaging in sedentary behaviour (almost 9h). An observational study of Australian 15-year-olds argued that most of this occurs during school time and class time<sup>279</sup>. Of this current cohort, >50% reported mainly sitting during recess and lunch breaks at school and particularly girls. This appears to signal a lot of scope for improvement. Notably, the proportion employed in sedentary occupations was only 4%.

Previous studies that also administered IPAQ reported less sedentary behaviour compared to this current study. In a study of 16-20-year-olds, high school students engaged in 7 h/day of sedentary behaviour in Spain or 4 h/day in Portugal<sup>334</sup>. Another study of Portuguese adolescents reported 13-18-year-olds' median sedentary behaviour was 7 h/day<sup>335</sup>. A large cluster randomised trial of French 14-18-year-olds (*n*=1 445) reported daily mean sedentary behaviour was 7h and 33% sat for  $\geq$ 7 h/day<sup>336</sup>. A reason for these differences may include region-specific time-use profiles from a lower proportion of sedentary lifestyles in Europe.

Previous studies that did not use IPAQ observed similar findings to this current study. A large cross-sectional study reported 13-18-year-olds from Europe (n=2 200) engaged in 9 h/day of device-assessed total sedentary behaviour which is very consistent with this current study<sup>124</sup>. That study also reported gender differences in total sedentary behaviour, consistent with this study<sup>124</sup>. Qualitative findings from Chapter 3 support what this chapter found and reflected that sedentary behaviour levels during secondary school were high, with common types being watching television, studying, and using a computer. This is consistent with previous research<sup>15, 337</sup>. In another Australian cross-sectional study of 15-17-year-olds, the

authors reported 3 h/day of screen-based leisure activity<sup>15</sup>. A large cross-sectional study found that Australian older adolescents (n=2 620) self-reported spending more time than younger children using computers recreationally than watching television, which is a typical finding from the research in this field<sup>337</sup>. The substitution of watching television in childhood to using computers in adolescence may be due to changing interests with age, increased school and study demands, and daily use of social media.

A third of this current study sample complied with guidelines. This was higher than compliance in other studies<sup>15, 116</sup>. A cross-sectional self-report study reported 19% of Australian 15-17-year-olds complied with guidelines<sup>15</sup> and a large cross-sectional study of 10-16-year-old Canadians (n=6 942) reported 14% of boys and 18% of girls met guidelines<sup>116</sup>. Explanations for differences in findings are that the other Australian study surveyed a national sample rather than a single state<sup>15</sup> and the sample in the Canadian study had a broader age range<sup>116</sup>. Regardless, intervening is warranted as older adolescents are engaging in excessive total sedentary behaviour and recreational screen time. Future research should focus efforts on decreasing recreational screen time, as it is modifiable and was a considerable portion of total sedentary behaviour in this current study. Further, recreational screen time should be targeted since it is more discretionary than sedentary transport, which may be the only or best option for commuting between school, work and home.

A few target groups emerged when reviewing sub-group variance in sedentary behaviour. Firstly, girls reported more total sedentary behaviour than boys, and boys reported more recreational screen time. Girls sit more to socialize at school, while boys spent more time using screens and electronic games. Strategies to reduce this may be more generic for girls such as 'Sit Less, Move More', while for boys a type-specific message could be 'Less Gaming'. Secondly, there were few differences by remoteness. Those living in urban areas consistently reported more of total and specific types of sedentary behaviour than the rural counterparts. Lastly, those who lived in the lowest tertile of socio-economic area reported watching television, DVDs or videos more. Future research is needed to explain mechanisms behind the remoteness and SEP differences. A cross-sectional study of Nepalese 17-year-olds (44% girls) that also administered the IPAQ similarly found girls engaged in more total sedentary behaviour than urban students, and that boys at private schools,

arguably a proxy for SEP, were more likely to sit for >6 h/day than boys at public schools<sup>338</sup>. The discrepancies may be due to country-specific differences in time-use between Australia and Nepal.

When reviewing studies that did not administer the IPAQ, the differences by gender, remoteness and SEP are similar. Firstly, a large cross-sectional study (n=2071) found Australian 12-16-year-olds boys, particularly from urban areas, self-reported more screen time than girls and boys from inner regional, outer regional and remote areas<sup>339</sup>. Similarly, a study of Australian 10-18-year-olds<sup>340</sup> and an international study of 11-15-year-olds<sup>341</sup> both concluded adolescent boys engaged in more recreational screen time than did girls. Secondly, the differences according to SEP are consistent with a systematic review of studies of 13-18-year-olds<sup>149</sup> and a narrative review<sup>148</sup>. Specifically, a cross-sectional study of European 13-18-year-old girls found self-reported television watching was associated with SEP and may reflect low maternal education and parent occupation status<sup>342</sup>. In summary, this study's findings imply that the key target groups during late adolescence are girls and those living in urban and the lowest tertile of socio-economic area. These groups need support, perhaps in the form of tailored strategies designed within sedentary behaviour reduction interventions and guidelines. Future research during late adolescence that uses device-based measurement of movement such as inclinometers would strengthen support for this implication.

A strength of this study is that a wide range of correlates were tested from three levels of the ecological model<sup>20, 21</sup>. The key correlate of sedentary behaviour in late adolescence was physical activity goal setting. Similar to Chapter 5, most independent variables were associated in the partially-adjusted models but less remained significant in the fully-adjusted model. Safety and social cohesion were significant in partially-adjusted models for sedentary behaviour but not for physical activity. This may be explained by better social conditions in a neighbourhood resulting in a lower likelihood of being sedentary inside the home and vice versa. Future research needs to examine mechanisms to explain why physical activity goal setting may be important to sedentary behaviour in late adolescence, perhaps using moderation or mediation analyses<sup>5</sup>. The findings of previous studies that administered the IPAQ to adolescents provided mixed support to what this study found. Firstly, students in the final two years of secondary school in Portugal and Spain reported physical self-concept (competence and body-image) was negatively associated with sedentary behaviour during high school<sup>334</sup>. Body-image was not assessed in this current study. Secondly, Nepalese students reported individual correlates of sitting >6 h/day were sedentary transport to school (although arguably on the causal pathway and should be excluded), cycling to school (in girls only), no school playgrounds and no extra-curricular activities provided at school<sup>338</sup>. The latter environmental correlates are similar to this current study which found physical activity equipment at home is associated with total sedentary behaviour in a partially-adjusted model, albeit applied to a different domain. Thirdly, social correlates of sedentary behaviour include the amount of sitting that the best friend engaged in (Portuguese 13-18-year-olds)<sup>335</sup> and the number of children and having a partner (European 15-25-year-old women; *n*=1 218)<sup>343</sup>. The current study did not test these but found eight family and peer environment characteristics were social correlates of sedentary behaviour in partially-adjusted models.

Some of the findings of this current study are supported by the findings of reviews of studies that administered measures other than IPAQ-A or IPAQ. Firstly, a narrative and systematic review reported key individual correlates of sedentary behaviour in adolescence were age (positive)<sup>148</sup>, ethnicity (positive; non-white)<sup>148, 149</sup> and parental education (negative)<sup>149</sup>. Secondly, the social correlates from the partially-adjusted models in this current study were consistent with a narrative review which reported parental modelling (positive), avoidance of risk (positive; outdoor safety concerns), restriction (negative; enforcement of rules and limits), and discouragement (negative) are key social correlates of sedentary behaviour in childhood and adolescence<sup>148</sup>. However, narrative reviews reported the number of televisions and computers per household (positive)<sup>148</sup> and having a television in the bedroom (positive)<sup>154</sup> were environmental correlates of sedentary behaviour. Differences may be due to a broader age range (2-18-year-olds)<sup>148, 154</sup> and this current study not assessing demographic characteristics as correlates, instead focussing on modifiable attributes to inform interventions. These previous studies have not tested physical activity goal setting as a correlate of total sedentary behaviour in adolescence, suggesting future research needs to.

Only one of the variables from the partially-adjusted models remained associated with total sedentary behaviour in late adolescence in the fully-adjusted model and future research is needed to clarify why. An individual correlate may have been found to be most important in this study because the current outcome variable of total sedentary behaviour captured other behaviours in various domains and at numerous times of day. As sedentary behaviour is pervasive throughout the day, there may be another explanation regarding the other types of sedentary behaviour that the outcome variable included. Future research needs to study correlates of specific types of sedentary behaviour.

Individuals with higher scores may be people who are likely to be highly active, health conscious and mindful of time use. Chapter 3 provided some insights into why physical activity goal setting may be important for sedentary behaviour in late adolescence, as recent school leavers described routines and regimented/structured days were influences on sedentary behaviour, as was personal preference. An observational study of Australia 15-year-olds identified the typologies of physical activity and sedentary behaviour and found a distinct cluster of MPA and high screen-time<sup>313</sup>. Included in the cluster was device-assessed MVPA, active travel, leisure-time sport or physical activity, watching television, electronic games and media, and sedentary homework<sup>313</sup>. Future studies need to study the correlates of these typologies in late adolescence to identify explaining mechanisms of these interdependent behaviours and test if physical activity goal setting may be important for certain clusters.

Notably, although television avoidance self-efficacy and sedentary behaviour discouragement from family, friends or colleagues were associated in partiallyadjusted models, this study suggests these are less important to total sedentary behaviour than physical activity goal setting. Another unexpected finding was that sedentary behaviour was not associated with co-participation in electronic games, television or DVDs with family, friends or colleagues or the number of electronic devices or televisions at home. An explanation might be that some of these correlates tested are domain-specific, whereas the outcome variable was across the whole day. Further, none of the social or environmental associations were retained in the final model, perhaps signifying that sedentary behaviour may be driven by intrinsic attributes in the later years of secondary school. Qualitative methods explore responses to gain understanding of this knowledge-gap and Chapter 3 reported that

recent school leavers described social support as a major influence on sedentary behaviour during the transition out of secondary school. Further, interviewees described other influences on sedentary behaviour were school (size and quality of grounds, play equipment and facilities, and exams and homework), transport (car ownership, commute duration and distance), time use (holidays and free time), internal (maturity, embarrassed feelings and mental health) and occupation type. Occupation type was also found to be an influential factor linked to sedentary behaviour for European 15-25-year-olds<sup>343</sup>. Future studies are needed to confirm what this study found using device-based measurement of sedentary behaviour such as a wearable inclinometer without the biases of self-report<sup>61</sup>.

The key correlate this current study found is characterised as traditionally behaviourally-specific to physical activity, similar to a 2016 systematic review<sup>344</sup>. That review reported proximity and density of green space, neighbourhood walkability, workplace showers, lockers and bicycle storage were environmental correlates of adult sedentary behaviour. The mix of behavioural specificity between correlates and outcomes in that review and in the current findings support the interdependency of sedentary behaviour and physical activity. Further, the findings advocate for the importance of testing variables that have been traditionally behaviourally-specific to physical activity as correlates of sedentary behaviour to better understand the complex relationship and inform sedentary behaviour recommendations and interventions.

#### **Strengths and limitations**

Some strengths of the study have been previously outlined in Chapter 4, such as assessing sedentary behaviour as distinct from physical activity and not simply a lack of physical activity. Also, assessing total sedentary behaviour, since previous research primarily assessed screen-based sedentary behaviour<sup>273</sup> and behaviour in the work and school domains. Further, this study acknowledged the interdependence of sedentary behaviour and physical activity by testing physical activity enjoyment, goal setting, competence, self-efficacy, co-participation, and equipment at home as correlates of sedentary behaviour. This unique approach revealed a novel key correlate of sedentary behaviour in late adolescence, adding to the sparse knowledge base of correlates of sedentary behaviour in late adolescence. Physical activity goal setting is known to be important for physical activity in adulthood but its role in

sedentary behaviour and late adolescence overall is less clear<sup>345</sup>. Future research is needed that similarly tests a wide variety of potentially influential attributes to continue to explore the interdependency of sedentary behaviour and physical activity.

Applying the ecological model<sup>20, 21</sup> emphasized that an individual-level correlate seemed to be most important for total sedentary behaviour in late adolescence, as opposed to social or environmental attributes. Qualitative research may be needed to explore the reasons for this and if there are social and environmental influences on sedentary behaviour important to this age group that were not assessed by this current study. The ecological model suggests some social and environmental chacteristics that may be important and should be tested in future research, including prompts to sit, awkwardness of standing, modelling, mass media, relevant government policies and investments, and schools and workplaces that have sitting requirements and only provide furniture designed for sitting<sup>21</sup>. Future research with alternate analyses is needed to identify the correlates and odds ratios for the likelihood of complying guidelines if older adolescents have high or low levels of individual, social and environmental attributes.

A limitation of this current study is that only correlates of total sedentary behaviour and not recreational screen time or sedentary transport were analysed. The main reason for this was that the survey did not ask in what domain the dependent variable took place in, such as recreational screen time at a friend's home, school, neighbourhood library or during transit. Correspondingly, some of the independent variables were domain-specific (the number of electronic devices and televisions at home) while others were not (television avoidance self-efficacy, electronic games coparticipation and sedentary behaviour discouragement). This may have led to underestimated associations<sup>318, 346</sup>. Meanwhile, the independent variables did not include the school environment, despite total sedentary behaviour including sitting at school and it is known by previous literature that most sitting of older adolescents occurs at school and during class time<sup>279</sup>. Future research is needed to identify correlates of multiple types of sedentary behaviour, and to test school environment correlates of sedentary behaviour in late adolescence, as findings may inform domain-specific strategies to use in interventions and guidelines. As mentioned in Chapter 5, a limitation of cross-sectional research is that it cannot confirm causality or directionality; consequently, it may be that more sedentary behaviour leads to less physical activity goal setting and longitudinal research is needed to clarify. Further, prospective studies are needed to examine whether physical activity goal setting is also a determinant of sedentary behaviour during the transition out of secondary school or whether different mechanisms partially explain behaviour over time. Other study-specific limitations were previously described in Chapter 4 such as sample bias due to the very high proportion of girls, or how the survey did not adequately address a common way of accumulating sedentary behaviour via concurrent use of multiple types, for example, watching movies or series while using a mobile phone, or measure prolonged sedentary behaviour which has the strongest associations with negative health outcomes.

#### Conclusion

Only 35% of 17-year-olds met guidelines, indicating more needs to be done to decrease the amount of recreational screen time during late adolescence. Compared to girls, less boys met the guidelines and more boys had sedentary occupations. Those living in an urban area reported higher total sedentary behaviour and those living in the lowest or mid tertiles of socio-economic area reported higher recreational screen time. The key correlate of sedentary behaviour in late adolescence appears to be physical activity goal setting. As this was also a correlate of physical activity and is modifiable, it warrants further research attention in behaviour change interventions. The next chapter builds on what this chapter found by examining changes in physical activity and sedentary behaviour during the transition out of secondary school and whether those differences are unique to situational transitions experienced post-school.

Chapter 7

Changes in physical activity and sedentary behaviour during the transition out of secondary school

## 7.1 Introduction

hapter 2 summarised how most cross-sectional and longitudinal studies during late adolescence report that as age increases, physical activity is lower<sup>130</sup>; however, evidence is mixed for changes in sedentary behaviour<sup>15</sup>, <sup>130, 139</sup>. These studies also have various limitations including heterogeneous data collection instruments that lacked validation and assessed only one component of physical activity and sedentary behaviour such as LTPA or television-viewing. Further, longitudinal studies with large follow-up intervals may mask the effects of transitioning out of secondary school and include a range of additional life changes, such as marital status or becoming a parent, that typically occur later in adulthood. In the qualitative study described in Chapter 3, most recent school leavers described changing physical activity and sedentary behaviour after leaving secondary school. The direction of the changes had no clear pattern (physical activity: *n*=13 increased and n=11 decreased; sedentary behaviour: n=13 increased and n=12 decreased) and few recent school leavers described maintaining physical activity (n=5) or sedentary behaviour (n=4). This current chapter aims to progress the understanding of those results by highlighting consistent or new findings about the changes in the descriptive epidemiology of physical activity and sedentary behaviour patterns during the transition out of secondary school.

Prospective studies have not adequately examined whether changes in physical activity and sedentary behaviour during the transition out of secondary school differ by gender, remoteness or area-level SEP or are impacted by a broad range of situational transitions. Situational transitions may require life adjustments that disrupt routines and habits around physical activity and sedentary behaviour<sup>8</sup>. Further, Chapter 3 reported a range of situational transitions were experienced immediately post-school by interviewees, including commencing full-time work or tertiary education at a university or vocational training, and moving out of the family home. Notably, most recent school leavers believed tertiary education was a major reason for changes in physical activity and sedentary behaviour.

Previous literature reported physical activity after leaving secondary school differed according to work status<sup>9, 10</sup> and tertiary student status<sup>118, 125, 170-173</sup>; however, physical activity was not assessed while still attending secondary school, leaving the impact of the transition unknown. Further, there are mixed findings according to living with parents or independently for changes in physical activity after leaving

secondary school<sup>10, 347</sup> and for changes in sedentary behaviour according to tertiary student status<sup>170, 171</sup>. Therefore, the impact of tertiary student status, work status and living independently/with parents on changes in physical activity and sedentary behaviour immediately post-school have not been quantitatively investigated in a prospective cohort recruited while still attending secondary school. This knowledge is crucial for understanding changes in physical activity and sedentary behaviour during this normative transition, so adolescents are equipped with resilience skills to avoid decreases in physical activity or increases in sedentary behaviour during early adulthood and beyond.

# 7.2 Chapter aims

The specific aims of this chapter are to:

- 1. Describe the changes in the descriptive epidemiology of physical activity and sedentary behaviour patterns during the transition out of secondary school;
- 2. Describe situational transitions experienced after leaving secondary school; and
- 3. Examine associations between situational transitions and changes in discretionary physical activity and total sedentary behaviour during the transition out of secondary school.

# 7.3 Methods

Chapter 4 described the full methodology of ProjectADAPT. To recapitulate, this study was a prospective cohort study of secondary school students in Victoria, Australia and took place between August, 2013 and June, 2017. Data collection occurred on three annual occasions at the same time of year as baseline for each participant (Figure 7.1). Participants recruited in 2013 completed all surveys by telephone (interviewer-administered). Those recruited in 2014-2015 opted at the time of consent to complete the survey via telephone or online (self-administered) and continued to be administered the survey via that mode for the two follow-up surveys. Survey items included at baseline were repeated at the one- and two-year follow-ups (Appendix 4.1), except for some school-based items in the final survey.

Figure 7.1 Study design and timeline



#### 7.3.1 Data management

Cleaning and management of the socio-demographic, physical activity and sedentary behaviour data was explained within Chapter 4.

#### Physical activity and sedentary behaviour

How the physical activity and sedentary behaviour variables were generated is described in Chapter 4. The outcome variables include discretionary physical activity (leisure and transport domains), total sedentary behaviour (all domains), physical activity and sedentary behaviour during school breaks, occupational physical activity and sedentary behaviour, and recreational screen time (watching television, DVDs or videos, using a computer, laptop or tablet, and playing electronic games) (Aim 1).

#### Situational transitions experienced

Situational transition variables related to study, employment and household composition were generated using responses from relevant single or multiple survey items at the two-year follow-up (Aims 2 and 3). To summarise, the first three situational transitions were: 1) tertiary student (yes, no); 2) substantial weekly work hours ( $\geq$ 20, <20, none); and 3) living with parents (yes, no). The next three situational transitions were more complex: 4) combinations of study and work (concurrently studied and worked, exclusively studied, exclusively worked, neither studied nor worked); 5) combinations of study and living with parents (studying while living with/without parents, not studying); and 6) combinations of work and living with parents (working while living with/without parents, not working).

The situational transitions were identified from literature or by Chapter 3. Firstly, there has been some literature on associations between changes in physical activity and tertiary student status<sup>125, 170-173</sup> or living with parents status<sup>10, 347</sup>. Secondly, Chapter 3 reported that interviewees perceived combinations of work/study and household composition were important influences on physical activity and sedentary

behaviour. Lastly, weekly work hours were the main employment-related situational transition variable, since hours of casual or part-time work varied greatly in the cohort from a few hours to full-time equivalent (>35 h/wk). Further, the cut-point applied of 20 h/wk of work is consistent with the average part-time weekly paid hours amongst Australian 18-21-year-olds<sup>348</sup>. When generating the variable for living with parents status, one participant consistently lived with a cousin, Aunt and Uncle at each time-point, hence was categorised as living with parents due to maintaining household composition.

#### 7.3.2 Data analysis

Baseline participant characteristics were compared between the study noncompleters (completed one or two of the surveys) and completers (completed all three surveys) using Pearson's chi-square analyses (categorical data) or a two-sample t-test using groups (continuous data). Further analyses used a restricted sample that only included study completers and those not attending secondary school at the twoyear follow-up to examine a cohort who have experienced the normative situational transition of leaving secondary school. The outcome variables of discretionary physical activity and total sedentary behaviour were checked for normality using a histogram. Discretionary physical activity data were positively skewed at all timepoints, while total sedentary behaviour data were normally distributed at all timepoints. The residuals for the discretionary physical activity and total sedentary behaviour regression analyses were checked and were normally distributed, in accordance with relevant assumptions<sup>288, 289</sup>.

To address Aim 1, changes in discretionary physical activity and total sedentary behaviour were identified using multilevel mixed-effects linear regression models. Marginal means were extracted for each model. Data was used from all time-points (long-form), adjusting for age and mode of survey administration a priori, and for confounders. The same confounders were used for consistency across the thesis, and because it is arguably more important to include potentially important confounders than to exclude weak or non-confounders<sup>291</sup>. Confounders in the physical activity models were gender, maternal education, English as the primary language at home and area-level SEP. Confounders in the sedentary behaviour models were gender, paternal education, English as the primary language at home, birth country and remoteness. As some participants were clustered within secondary schools due to the

sampling methods, multilevel modelling was required, and a two-level model was used that accounted for individuals nested within schools.

Additional models were run with an interaction term added between time and gender, remoteness or area-level SEP. If significant, marginal means of the within-group changes were extracted. Changes in secondary outcome variables from baseline were examined using the same procedures as above. Secondary outcome variables included LTPA, active transport, sedentary behaviour (excluding transport), sedentary transport and recreational screen time. Lastly, changes in guideline compliance were examined over time using a multilevel mixed effects logistic regression accounting for individuals nested within schools.

To address Aim 2, characteristics of the cohort at the two-year follow-up were described using tabulation or means and standard deviations. Firstly, study-related variables included secondary school status (early school leaver, early school leaver but returned the following year, still at secondary school, finished secondary school), tertiary student status (yes - higher education or vocational training, no - including deferred), tertiary study mode (full-time, part-time, deferred, no study), duration studying (h/wk), and contact hours since secondary school (increased, decreased, no change, no study). Secondly, employment-related variables included employment status (yes, no), employment type (volunteering, paid, did not work), employment mode (full-time, part-time, casual, did not work), number of jobs, hours worked/wk and changed work hours since secondary school (increased, decreased, no change). Thirdly, living situation variables were parent's home, rental or share house, residence hall, other relative's home or other, and many variables were created for the combinations of situations based on studying and working, studying and living with parents, and working and living with parents. Lastly, predominant time-use variables were vocational training, university, part-time work, full-time work, job seeking, apprenticeship, volunteering, traineeship, defence forces or other (participants were asked to specify). Categorical data were compared by subgroups using Pearson's chi-square analyses and continuous data were compared by subgroups using two-sample t-tests (binary groups; gender and remoteness) or oneway ANOVAs (area-level SEP).

To address Aim 3, time by transition interactions were examined to determine whether there were interactions between the main situational transitions experienced at the two-year follow-up and time-point for discretionary physical activity and total sedentary behaviour. The main situational transitions were each added as an interaction term with time-point into separate multilevel mixed-effects linear regression models for both outcome variables (totalling 12 models) and *p*-values were reviewed. To determine if interactions were meaningful, Akaike's information criteria (AIC) was compared between main effect models that included each situational transition as an independent variable and the models that included each situational transition as an interaction term (totalling 24 models) with time-point. If a significant interaction was confirmed based on an AIC improvement (i.e., decrease) of  $\geq$ 2 points<sup>349, 350</sup>, within-group change for the situational transition categories at the three time-points were examined, and marginal means were extracted.

## 7.4 Results

#### 7.4.1 Participant characteristics

Figure 7.2 presents retention proportions. Of the 1 022 participants at baseline, 7% did not respond to correspondence to participate in the one-year follow-up survey, and an additional 9% did not respond to correspondence or declined to complete the survey at the two-year follow-up, resulting in an 84% retention proportion.
**Figure 7.2** Retention proportions (RP) among participants



Table 7.1 presents the baseline characteristics of study completers and noncompleters. A higher proportion of completers were recruited via social media, took the survey online and identified as a woman than non-completers. Completers reported lower discretionary physical activity and higher total sedentary behaviour at baseline than non-completers.

	Participants				
	Non-				
	completers	Completers			
	(%)	(%)	р		
_( <i>n</i> )	(170)	(852)			
Recruitment method <sup>1</sup>			0.004		
Secondary schools	47.1	35.5			
Social media	52.9	64.6			
Survey mode <sup>1</sup>			0.000		
Online	78.2	95.4			
Telephone	21.8	4.6			
Demographic characteristics <sup>2</sup>					
Age (mean years $\pm SD$ ) <sup>3</sup>	16.8±0.4	16.9±0.4	0.073		
Gender <sup>1</sup>			0.038		
Girl	67.5	75.1			
Boy	32.5	24.9			
<i>Remoteness</i> <sup>1</sup>			0.363		
Urban	67.7	71.1			
Rural	32.4	28.9			
$SEP^{1}$			0.214		
Lowest tertile (area-level)	26.5	23.3			
Mid tertile (area-level)	33.5	29.4			
Highest tertile (area-level)	40.0	47.4			
Mother tertiary-educated	51.8	55.2	0.427		
Father tertiary-educated	42.2	44.4	0.593		
Birth country <sup>1</sup>					
Participant - Australia	81.2	85.1	0.455		
Mother - Australia	62.9	61.4	0.533		
Father - Australia	58.2	61.7	0.663		
English as the primary language spoken at			0.473		
home <sup>1</sup>	85.3	85.2			
Employment (types not mutually					
exclusive) <sup>1</sup>					
Not employed	44.1	41.1	0.269		
Paid employment	50.6	56.2	0.372		
Casual employment	37.1	43.4	0.527		
Part-time employment	18.2	21.4	0.777		
>1 job	8.2	8.3	0.747		
Income > \$150/wk	8.2	10.8	0.403		
Discretionary PA (mean h/day $\pm$ SD) <sup>3</sup>	1.6±1.5	$1.2\pm1.1$	0.000		
Total SB (mean h/day±SD) <sup>3</sup>	8.4±2.7	8.9±2.9	0.047		

**Table 7.1** Baseline characteristics of study completers and non-completers

<sup>1</sup> Pearson's  $\chi^2$  test of significance by completion status.

<sup>2</sup> Most demographic characteristics had a small amount missing (n=1-9; age, gender, remoteness, area-level SEP, maternal education, paternal education, birth country and employment variables). Outcomes variables also had a small amount missing (PA: n=17; SB: n=19).

<sup>3</sup> *Two-sample t-test using completion status.* 

#### 7.4.2 Changes in physical activity and sedentary behaviour

#### Changes in physical activity

Table 7.2 presents the changes in physical activity over time (Aim 1). Discretionary physical activity decreased from baseline to the two-year follow-up by an average of almost 10 mins/day. The proportion who met the National Physical Activity Guidelines<sup>109</sup> declined from baseline to the one-year follow-up (45% vs 37% (p<0.001) vs 41%). Total LTPA, MPA and VPA decreased over time, and total active transport and walking for transport decreased between baseline and the one-year follow-up (Table 7.2). There were no significant interactions between time and gender or remoteness and one significant interaction between time and area-level SEP (Appendix 7.1). Within-group change in discretionary physical activity (B:-7.1; 95% CI:-11.8, -2.3; p<0.01) between baseline and the one-year follow-up for those living in the highest tertile of socio-economic area differed significantly to the within-group changes for those living in the lowest tertile.

For community sport, the mean number of non-school sport teams decreased over time  $(1.0\pm1.3 \text{ vs } 0.8\pm1.1 \text{ vs } 0.7\pm1.2; p<0.001)$ . Of those who reported working, occupational physical activity decreased from baseline to two-year follow-up (p<0.001). Firstly, the proportion who reported mostly standing at work decreased (56% vs 55% vs 53%). Secondly, the proportion who reported mostly walking at work decreased over the transition (29% vs 31% vs 26%). Lastly, the proportion who reported mostly heavy labour or physically demanding work was maintained (9% vs 10% vs 9%).

	Baseline			1-year fo	llow-up	2-year follow-up			
	n	<b>Marginal mean</b> mins/day (95% CI)	n	<b>Marginal mean</b> mins/day (95% CI)	Change from baseline mins/day (95% CI)	n	<b>Marginal mean</b> mins/day (95% CI)	Change from baseline mins/day (95% CI)	
Discretionary PA	809	77.1 (72.3, 81.9)	819	66.1 (61.2, 71.0)	-11.0 (-15.7, -6.3)***	809	67.8 (62.7, 72.8)	-9.3 (-14.2, -4.5)***	
<i>LTPA</i> Walking Other MPA VPA	811 815 819 819	51.3 (47.8, 54.9) 12.5 (11.2, 13.9) 16.5 (15.1, 17.9) 22.0 (20.1, 23.9)	820 821 820 821	43.3 (39.7, 46.9) 11.9 (10.5, 13.2) 12.6 (11.2, 14.1) 18.6 (16.7, 20.6)	-8.0 (-11.5, -4.5)*** -0.7 (-2.3, 0.9) -3.9 (-5.5, -2.3)*** -3.4 (-5.2, -1.5)***	811 811 812 811	42.4 (38.7, 46.2) 11.0 (9.5, 12.5) 12.9 (11.4, 14.4) 18.2 (16.2, 20.2)	-8.9 (-12.5, -5.2)*** -1.5 (-3.2, 0.1) -3.6 (-5.2, -2.0)*** -3.8 (-5.7, -1.9)***	
Active transport Cycling for transport Walking for transport	820 822 820	25.6 (23.4, 27.8) 2.6 (1.9, 3.3) 22.9 (21.0, 24.9)	821 821 821	23.0 (20.8, 25.3) 2.8 (2.1, 3.5) 20.2 (18.2, 22.2)	-2.6 (-4.9, -0.2)* 0.2 (-0.6, 1.0) -2.7 (-4.9, -0.5)*	812 814 812	25.4 (23.0, 27.7) 3.0 (2.2, 3.7) 22.3 (20.3, 24.4)	-0.2 (-2.7, 2.2) 0.3 (-0.5, 1.2) -0.6 (-2.8, 1.7)	

**Table 7.2** Changes in physical activity during the transition out of secondary school

p<0.05, p<0.001: multilevel mixed-effects linear regression models between 1- or 2-year follow-up and baseline. Adjusted for age, survey mode and confounders (gender, maternal education, English as the primary language spoken at home, and area-level SEP), with repeated measurements for individuals nested within schools.

#### Changes in sedentary behaviour

Table 7.3 presents the changes in types of sedentary behaviour over time (Aim 1). Total sedentary behaviour increased from baseline to the one-year follow-up by 16 mins/day (p<0.05). The increase was not sustained, and total sedentary behaviour declined at the two-year follow-up compared to baseline by 44 mins/day (p<0.001). Sedentary behaviour (excluding transport) decreased from baseline to the two-year follow-up, sedentary transport increased from baseline to the two-year follow-up, and total recreational screen time increased from baseline to the two-year follow-up. The proportion who met the National Sedentary Behaviour Guidelines<sup>109</sup> over time was 35% vs 38% vs 30%, with a decline from the one- to two-year follow-up (p<0.01).

There were four interactions between time and remoteness for sedentary behaviour (Appendix 7.1). Firstly, those who lived in a rural area had a larger within-group change in total sedentary behaviour of 0.7 h/day (95%CI:-0.9, -0.5; p<0.001) over time compared to those living in an urban area. Secondly, those who lived in an urban area had a larger within-group change in sedentary transport of -0.3 h/day (95%CI:0.2, 0.3; p<0.001), in total recreational screen time of 0.2 h/day (95%CI:0.1, 0.3; p < 0.01) and in electronic games of -0.0 h/day (95%CI:-0.1, -0.0; p < 0.01) over time compared to those living in a rural area. There were three interactions between time and area-level SEP (Appendix 7.1). Those who lived in the highest tertile of socio-economic area had a larger within-group change in 1) total recreational screen time of 0.6 h/day (95%CI:0.1, 1.2; p < 0.05) over time compared to those living in the lowest tertile, and 2) watching televisions, DVDs or videos of 0.5 h/day (95%CI:0.2, 0.8; p<0.01) over time compared to those living in the lowest tertile. Lastly, those who lived in the mid tertile of socio-economic area had a larger within-group change in watching televisions, DVDs or videos of 0.3 hours/day (95%CI:0.0, 0.7; p<0.05) over time compared to those living in the lowest tertile. There were no interactions between time and gender for any sedentary behaviour variable (Appendix 7.1).

	Baseline			1-year follow-up			2-year follow-up		
					Change from			Change from	
		Marginal mean		Marginal mean	baseline		Marginal mean	baseline	
	п	h/day (95% CI)	п	h/day (95% CI)	h/day (95% CI)	n	h/day (95% CI)	h/day (95% CI)	
<i>Total SB</i> SB (excluding	808	8.8 (8.6, 9.0)	803	9.1 (8.8, 9.3)	0.3 (0.0, 0.6)*	805	8.1 (7.8, 8.3)	-0.8 (-1.0, -0.5)***	
transport)	810	7.9 (7.7, 8.1)	805	8.1 (7.9, 8.4)	0.2 (0.0, 0.5)*	805	6.8 (6.6, 7.1)	-1.1 (-1.4, -0.9)***	
Sedentary transport	821	0.9 (0.8, 0.9)	820	0.9 (0.9, 1.0)	0.0 (-0.0, 0.1)	814	1.2 (1.1, 1.3)	0.3 (0.3, 0.4)***	
<i>Total recreational</i> <i>screen time</i> Watching TV, DVDs or videos Computer, laptop or tablet E-games	774 798 796 816	3.4 (3.2, 3.6) 1.5 (1.4, 1.6) 1.7 (1.6, 1.9) 0.2 (0.2, 0.2)	778 798 796 816	3.3 (3.1, 3.5) 1.4 (1.3, 1.5) 1.8 (1.6, 1.9) 0.2 (0.1, 0.2)	-0.1 (-0.3, 0.1) -0.1 (-0.2, 0.0) -0.0 (-0.1, 0.1) -0.0 (-0.1, 0.0)	724 798 784 764	3.7 (3.4, 3.9) 1.6 (1.5, 1.7) 1.9 (1.7, 2.0) 0.2 (0.1, 0.2)	<b>0.2 (0.0, 0.5)*</b> 0.1 (-0.0, 0.2) 0.1 (-0.0, 0.2) -0 0 (-0 1, 0 0)	
L Sumos	010	(0.2, 0.2)	010	0.2 (0.1, 0.2)	0.0 ( 0.1, 0.0)	/01	0.2 (0.1, 0.2)	0.0 ( 0.1, 0.0)	

**Table 7.3** Changes in sedentary behaviour during the transition out of secondary school

p<0.05, p<0.01, p<0.01; multilevel mixed-effects linear regression models between 1- or 2-year follow-up and baseline. Adjusted for age, survey mode and confounders (gender, paternal education, English as the primary language at home, birth country and remoteness), with repeated measurements for individuals nested within schools.

# **7.4.3** Are situational transitions associated with changes in physical activity and sedentary behaviour?

## Describing the situational transitions experienced

Few (n=29; 3.5%) of the cohort reported that they were still attending secondary school at the two-year follow-up. Very few (n=7; 0.8%) were early school leavers, i.e., reported not attending secondary school at the one- and two-year follow-ups. Also, very few (n=3; 0.4%) were early school leavers who returned, i.e., were attending school at baseline and the two-year follow-up but not at the one-year follow-up. Only those who reported not attending secondary school at the two-year follow-up were included in further analyses.

Tables 7.4 presents the study-related characteristics of the cohort at the two-year follow-up (Aim 2). Most of the cohort studied at a tertiary institute, studied full-time, reported 11-15 h/wk contact hours and had fewer contact hours compared to secondary school. A greater proportion of men reported not studying compared to women, while women studied off-campus more than men.

			Gender	
	Overall	Women	Men	
	(%)	(%)	(%)	р
Tertiary student status, $n=796$				
Not studying (including deferred)	20.7	20.1	22.5	
Studying	79.3	79.9	77.5	
				0.475
Study mode, $n=796$				
Not studying	9.2	7.9	13.0	
Full-time	75.1	75.0	75.5	
Part-time	4.2	4.9	2.0	
Deferred study	11.6	12.2	9.5	
				0.042
In those who study, changed				
contact hours at place of study				
since baseline, n=796				
Not studying	20.7	20.1	22.5	
Increased	10.3	10.7	9.0	
Decreased	57.2	58.4	53.5	
No change	11.8	10.7	15.0	
				0.287
H/wk in class and studying on				
campus, n=782				
0 or off-campus student	20.6	20.4	21.4	
1-5	3.3	3.7	2.0	
6-10	12.4	12.7	11.2	
11-15	22.7	23.9	18.9	
16-20	18.3	17.1	21.9	
21-25	12.2	11.2	15.3	
26-30	6.9	7.1	6.1	
>30	3.6	3.7	3.1	0.054
		,	,	0.374
	(mean	(mean	(mean	
	h/wk	h/wk	$h/wk \pm SD$ )	
	$\pm SD)$	$\pm SD)$		
Study time at home if enrolled on- campus, $n=618$	10.0±8.4	10.2±8.5	9.6±8.2	0.226
Study time if enrolled off-campus, n=153	1.3±5.3	1.7±6.2	0.1±0.8	0.049

**Table 7.4**Study characteristics of participants at the two-year follow-up

*Pearson's*  $\chi^2$  *test of significance or two-sample t-test by gender.* 

Table 7.5 presents the employment-related characteristics of the cohort at the twoyear follow-up (Aim 2). A higher proportion of participants reported working, mainly on a casual basis, for <20 h/wk, and had increased weekly work hours since baseline. Most of the employment characteristics changed over time. For employment status, the proportion of those not employed (paid or voluntary) decreased (40% vs 40% vs 23%; *p*<0.001). Of those employed, part-time employment was lower at the one-year follow-up coinciding with the final year of school (36% vs 30% vs 35%; *p*<0.001), whereas casual employment decreased overtime (74% vs 76% vs 60%; *p*<0.001). Work hours increased for those with one job (8±5 h/wk vs 8±5 vs 16±12; *p*<0.001). The proportion with >1 job was higher over time (14% vs 15% vs 23%; *p*<0.05) and hours worked increased over time (11±7 h/wk vs 11±8 vs 23±15; *p*<0.001). Consequently, reported earnings >\$150/wk increased over time (14% vs 21% vs 60%; *p*<0.01).

Table 7.6 presents other characteristics of the cohort at the two-year follow-up (Aim 2). A higher proportion of participants concurrently studied and worked, were living with parents, simultaneously studied while living with parents, and reported that time was mostly occupied by university study. A higher proportion of women compared to men reported time was mostly occupied by part-time work. Meanwhile, a higher proportion of men compared to women reported time was mostly occupied by an apprenticeship.

			Gender	
	Overall	Women	Men	
	(%)	(%)	(%)	р
<i>Employment status</i> <sup>1</sup> , $n=804$				
Not working	22.9	22.0	25.6	
Working	77.1	78.0	74.4	
				0.288
<i>Employment type, n=804</i>				
Not working	22.6	21.8	25.1	
Volunteering	8.2	7.9	9.0	
Paid work	69.2	70.2	65.8	
				0.504
<i>Employment mode</i> <sup>1</sup> , $n=799$				
Not working	23.0	22.2	25.5	
Casual	46.3	46.3	46.0	
Part-time	22.8	18.9	19.0	
Full-time	7.9	7.3	9.5	
				0.352
Number of jobs, $n=804$				
0	22.9	22.0	25.5	
1	59.2	59.8	57.5	
≥2	17.9	18.2	17.0	
				0.592
Substantial work hours <sup>1</sup> , $n=804$				
Not working	22.9	22.0	25.5	
Working <20h/wk	49.9	50.2	49.0	
Working $\geq 20h/wk$	27.2	27.8	25.5	
(Working $\geq$ 35h/wk)	(14.2)	(12.8)	(18.4)	
	~ /	× ,		$0.568^{2}$
Changed work <sup>1</sup> hours since				
baseline, n=681				
Not working	18.4	16.8	23.2	
Increased	61.2	62.6	57.1	
Decreased	17.0	17.9	14.3	
No change	3.4	2.7	5.4	
C				0.070

Table 7.5	Employment c	haracteristics of	participants at	t the two-year follow	-up
-----------	--------------	-------------------	-----------------	-----------------------	-----

<sup>1</sup> Paid and/or voluntary work.

<sup>2</sup> Does not include 'Working  $\geq$  35h/wk'

*Pearson's*  $\chi^2$  *test of significance by gender.* 

		Gender		
	Overall	Women	Men	
	(%)	(%)	(%)	р
Combinations of study and work <sup>1</sup> , $n=795$				
Concurrently studied and worked	58.9	60.7	53.3	
Exclusively studied	20.5	19.1	24.6	
Exclusively worked	18.2	17.3	21.1	
Neither studied nor worked	2.4	2.9	1.0	
				0.074
Household type, n=805				
Parent's home	79.3	78.8	80.5	
Rental or share house	8.1	7.8	9.0	
Residence hall	9.3	9.8	8.0	
Another relative's home	1.7	1.5	2.5	
Other	1.6	2.1	0.0	
				0.198
Combinations of study and household				
composition, $n=785$				
Studied and lived with parents	62.8	63.5	60.7	
Studied and lived independently	16.2	16.1	16.3	
Did not study	21.0	20.4	23.0	
				0.722
<i>Combinations of work<sup>1</sup> and household</i>				
composition, $n=795$				
Worked and lived with parents	64.8	65.4	62.9	
Worked and lived independently	12.1	12.4	11.2	
Did not work	23.1	22.2	25.9	
				0.558
'Time mostly occupied by'. $n=805$				
University	72.8	73.6	70.5	0.400
Part-time work	35.0	37.0	29.0	0.039
Job seeking	12.8	13.1	12.0	0.698
Full-time work	9.1	8.8	10.0	0.597
Vocational training	6.7	6.9	6.0	0.644
Volunteering	6.5	7.3	4.0	0.103
Traineeship	1.7	2.0	1.0	0.356
Apprenticeship	0.9	0.3	2.5	0.004
Defence	0.4	0.5	0.0	0.318
Other <sup>2</sup>	6.0	5.6	7.0	0.475

**Table 7.6**Other characteristics of the participants at the two-year follow-up

<sup>1</sup> Paid and/or voluntary work.

<sup>2</sup> Time was mostly occupied by (in order of most common) casual work, travelling/worked aboard for the gap year, gap year, sport, musician, blogging, or undergoing cancer treatment.

*Pearson's*  $\chi^2$  *test of significance by gender.* 

#### Changes in behaviour according to situational transitions

The interactions between the main situational transitions experienced and changes in discretionary physical activity and total sedentary behaviour during the transition out of secondary school are presented in Appendix 7.2 (Aim 3). There were significant interactions between time and the situational transitions for total sedentary behaviour but not for discretionary physical activity. Overall, interactions were found for tertiary student status, substantial weekly work hours status, living with parents status, study and work combinations, studying while living with or without parents, and working while living with or without parents. These interactions are explored graphically below.

Those not studying at a tertiary institute at the two-year follow-up decreased total sedentary behaviour more over time compared to tertiary students (Figure 7.3).

**Figure 7.3** Change in marginal means (95% confidence intervals) for total sedentary behaviour according to tertiary student status at two-year follow-up (n=796)



*….. tertiary student; — not a tertiary student.* 

Those working  $\geq 20$  h/wk at the two-year follow-up had a greater decrease in sedentary behaviour over time, compared to those not working (Figure 7.4).

**Figure 7.4** Change in marginal means (95% confidence intervals) for total sedentary behaviour according to substantial weekly work hours at two-year follow-up (n=804)



---- working  $\geq 20$  h/wk; .... working < 20 h/wk; — not working.

After leaving secondary school, those living independently decreased sedentary behaviour significantly more over time than those living with parents (Figure 7.5).

**Figure 7.5** Change in marginal means (95% confidence intervals) for total sedentary behaviour according to living with parents at the two-year follow-up (n=805)



.... living with parents; —·living independently.

After leaving school, those exclusively working reported engaging in the least sedentary behaviour, while those exclusively studying reported the most. Further, those exclusively studying after leaving secondary school changed sedentary behaviour the least amount over time, compared to those exclusively working, studying while working, and neither studying nor working (Figure 7.6).

**Figure 7.6** Change in marginal means (95% confidence intervals) for total sedentary behaviour according to study and work combinations at the two-year follow-up (n=795)



---- exclusively working; ----- exclusively studying; ---- concurrently studying while working; ---- neither working nor studying.

Those not studying at a tertiary institute after leaving secondary school decreased total sedentary behaviour significantly more over time compared to those studying while living with parents (Figure 7.7).

**Figure 7.7** Change in marginal means (95% confidence intervals) for total sedentary behaviour according to whether participants were studying while living with or without parents at the two-year follow-up (n=785)



---- living with parents while studying; … living independently while study; — not studying.

Irrespective of living with or without parents, those working after leaving secondary school decreased total sedentary behaviour significantly more over time compared to those not working (Figure 7.8).

**Figure 7.8** Change in marginal means (95% confidence intervals) for total sedentary behaviour according to whether participants were working while living with or without parents at the two-year follow-up (n=795)



----- living with parents and worked; ----- living independently and worked; --- not working.

#### 7.5 Discussion

In summary, this study aimed to describe changes in physical activity and sedentary behaviour during the transition out of secondary school, describe situational transitions post-school, and examine interactions between these. This study found that both discretionary physical activity and total sedentary behaviour declined, consistent with previous studies of older adolescents<sup>15, 85, 118, 122, 123, 134, 135, 139</sup>. After leaving secondary school, most of the cohort lived with parents and concurrently studied and worked. The three most common activities that occupied time were university studies, part-time work or job-seeking. Situational transitions associated with a decrease in sedentary behaviour were not studying, working substantial hours, exclusively working, living independently, and studying while living independently.

This study found the magnitude of the discretionary physical activity changes were small with an average decline of almost 10 mins/day over time and an 8% reduction in compliance with guidelines in Year 12. This is a concern due to the associated health outcomes outlined in Chapter 2. The physical activity declines may be partially attributed to less structure and routine on weekdays compared to secondary school classes but future research is needed to confirm determinants. Regarding changes in types of physical activity, school leavers decreased MPA other than walking and VPA over time. The physical activity declines observed in this study signal that preventative initiatives are needed that prepare adolescents for leaving secondary school. Active transport comprised almost a third of discretionary physical activity and was therefore important. However, at the one-year follow-up, it declined. Future interventions need to encourage older adolescents to use active transport by promoting walking and cycling for short trips and tailor messages to the proenvironmental interest and awareness in this age group by highlighting the sustainability benefits such as climate crisis mitigation and reducing traffic congestion<sup>351, 352</sup>.

While national statistics suggested sedentary behaviour was higher as age increased (3 h/day in 15-17-year-olds and 5.5 h/day in 18-24-year-olds)<sup>15</sup>, this current study found that total sedentary behaviour decreased by 44 mins/day over time, which is promising. Differences may be due to different measures and samples, as the current study included proportionally more women, tertiary students and those living in the highest tertile of socio-economic area. Sedentary behaviour may have declined post-school due to less pressure and demand to study compared to the final years of

#### Chapter 7: Changes in physical activity and sedentary behaviour

secondary school, but future research is needed that identifies determinants. The sedentary behaviour declines could imply that the secondary school environment may be conducive to an extra 44 mins/day of sedentary behaviour that interventions could target with strategies to reduce prolonged sitting in secondary school students such as encouraging active breaks during classes and when studying at home. Regarding changes in specific types of sedentary behaviour, school leavers increased their recreational screen time which caused an 8% decline in compliance with National Sedentary Behaviour Guidelines. Sedentary transport also increased, while sedentary behaviour in domestic and leisure-time domains decreased. Future research is needed to clarify whether the sedentary transport increase was based on private car use, which is negative, or public transport use, which is positive since it may include standing, walking or cycling as part of the journey. Older adolescents living in an urban area or the highest or mid tertiles of socio-economic area were identified as target groups for intervention because they increased recreational screen time over time more than those living in an urban area or in the lowest tertile, respectively.

This current study found that participants experienced a range of disruptive changes during late adolescence that included major life adjustments, such as living independently, substantial weekly work hours, commencing tertiary education, and, importantly, simultaneous combinations of these. This implies that adolescents are not a homogenous group but instead experience a range of situational transitions. Following a cohort from secondary school meant that it was possible to explore a range of situational transitions after leaving school and addressed a literature gap as similar studies mainly assessed tertiary students who had moved away from home onto student residences on campus<sup>118, 125, 170-173</sup>. The proportions who experienced specific situational transitions in this study were mostly inconsistent with national data among 15-20-year-olds<sup>353</sup>. Compared to this current study, national data showed that a lower proportion of Australian recent (<1-year) school leavers were studying at university or further education (63% vs 79%), or concurrently studied and worked (39% vs 59%); a higher proportion exclusively worked (24 % vs 18%), or neither studied nor worked (14% vs 2%); and a similar proportion exclusively studied (24% vs 21%)<sup>353</sup>. The dissimilar findings may be partially explained by the wide age range included in the national data, which was representative of 15-20-year-olds.

Although many previous studies found associations between changes in physical activity and situational transitions experienced after leaving secondary school<sup>9, 118, 125,</sup>

<sup>170, 171, 173, 347, 354, 355</sup>, this study did not. The inconsistent findings may be due to methodological differences, in particular, this current study had a short, two-year timeframe. Future research needs to continue examining annual changes in physical activity by situational transition over a longer period of time, assess physical activity using devices and large random samples to capture a broader range of situational transitions and combinations of these. Further, future research needs to examine whether there is an accumulative impact of multiple simultaneous situational transitions on changes in behaviour.

In contrast, changes in sedentary behaviour were associated with situational transitions in this current study. This is a novel finding, as no other studies have been found that focus on the role of situational transitions on explaining sedentary behaviour changes post-secondary school. In all of the associations, behaviour was maintained at the one-year follow-up and the declines were observed only after leaving secondary school. The finding supports the life transition model<sup>18</sup> that disruptions and instability can change the trajectory of health behaviour. This study found a cluster of helpful (sedentary behaviour declined) situational transitions including not studying, working  $\geq$ 20 h/wk, living independently and exclusively working (without concurrent study). Future research needs to examine why these groups are helpful.

More work hours may be a proxy for having stable and consistent employment that leads to a structured lifestyle, being financially independent and more time in a different social environment with work colleagues. Also, occupation type may be an explaining mechanism for declines in sedentary behaviour, since there was a low proportion of sedentary occupations in this age group. As supportive evidence, the most common (n>20) occupation types of the cohort were retail assistant, fast food team member, cashier, waitress and customer service officer, which traditionally involve a lot of standing and/or walking. Chapter 3 supported this finding as recent school leavers perceived that their work influenced changes in sedentary behaviour and the direction of influence was based on their job type. Studies have found sedentary behaviour associated with employment status<sup>280</sup> and, specifically, unemployment associated with more television viewing, non-full-time employment associated with more leisure-time sedentary behaviour and mid-income range associated with more home computer use<sup>356</sup>.

The impact of living independently on changes in sedentary behaviour may be partially attributed to improved access to neighbourhood facilities, doing more chores and grocery shopping as part of changes in home-life routines, and being actively involved with new flatmates and a new social support base in a new neighbourhood. An insight from Chapter 3 is that numerous recent school leavers reported a shift in friendship groups away from secondary school peers. Also, the qualitative study participants similarly reported that home life was an influence on changes in sedentary behaviour, as well as independence and pet ownership. A study found living alone was associated with more sedentary behaviour (watching television, using a home computer and leisure-time sedentary behaviour)<sup>356</sup>.

This study found a cluster of unhelpful (sedentary behaviour was maintained) situational transitions for intervention including living with parents, not working, working <20 h/wk, exclusively studying and the combinations of studying while living with parents, neither working nor studying, and working while studying. Although it was positive that sedentary behaviour did not increase in these target groups, future research is needed to examine how to decrease sedentary behaviour in these populations. Tertiary students may have maintained sedentary behaviour because of continued studious habits from school; therefore, future research needs to examine how to decrease sedentary behaviour in students, such as introducing strategies that break up prolonged sitting (sit-stand desks and frequent active study breaks). As mentioned earlier, a study found unemployment associated with more television viewing and non-full-time employment associated with more leisure-time sedentary behaviour<sup>356</sup>.

#### **Strengths and limitations**

Strengths of this study include an adequate sample size and high retention. There were three data collection points and the annual follow-ups were conducted at the same time of year as baseline to reduce intra-individual seasonal variation in behaviour. Future studies could consider collecting data within only one season as Chapter 3 findings suggest that Winter and the school holidays influenced less socialising, more sedentary behaviour, and less physical activity. The annual assessment intervals addressed a literature gap because large intervals, such as four years, had been previously used<sup>9, 354, 355</sup> and may have masked changes in behaviour due to the unique transition out of secondary school that had not been previously

exposed. Another literature gap addressed was assessing physical activity and sedentary behaviour prior to the transition out of secondary school which gave this study robust baseline data, since previous studies on the transition post-school only assessed behaviour after leaving secondary school<sup>9, 125, 170, 172, 354, 355</sup> or relied on recall of secondary school behaviour once the participant had already left school<sup>118, 171, 173</sup>. Identifying an impact of situational transitions on changes in sedentary behaviour during the transition out of secondary school was novel. Future research needs to examine the effect of single and cumulative situational transitions experienced after leaving secondary school on changes in other health behaviours, such as meal patterns and smoking, to identify explaining mechanisms of change that are common across multiple health behaviours.

The findings may largely apply to women who are tertiary students living in the mid or highest tertile of socio-economic area and would benefit from verification in studies that use random sampling methods and sample an even representation of gender to increase generalisability. Further, study completers had lower baseline discretionary physical activity, higher baseline total sedentary behaviour, more identified as girls, were recruited via social media and completed the surveys online than non-completers. The latter two factors possibly reflect being highly active on social media and tech-savvy, which may have contributed to higher total sedentary behaviour in those participants.

Changes in physical activity in late adolescence may be underestimated in the findings as time in LPA in any domain and time in physical activity at school and work were not assessed in the physical activity measure. Although the test-retest reliability of the measure is acceptable, the magnitude difference between declines in physical activity and sedentary behaviour may be due to measurement error and a more robust measure is needed to clarify findings. As mentioned in Chapter 4's discussion, although at a larger cost, device-based measures with concurrent log book use has more sensitivity and would provide a more reliable and valid estimate of health behaviours<sup>61, 275</sup>. The use of a self-reported measure is associated with a number of limitations including human recall error<sup>61, 275</sup> and applying social norms by under-reporting sedentary behaviour. Self-report measures are a cost-effective method, however, for assessing physical activity and sedentary behaviour, are unlikely to alter the behaviour<sup>262</sup> and are one of the only ways to collect domain-specific information. Lastly, relying on mean changes in behaviour at the group-level

can mask the behavioural trajectories of sub-groups. To develop what this study found, future research is needed that uses a different longitudinal analysis approach such as trajectory, path or latent class growth analyses to identify clusters of behaviour patterns and identify the unique influences on behaviour.

### Conclusion

This chapter showed that total sedentary behaviour declined during the transition out of secondary school; however, recreational screen time increased and discretionary physical activity and compliance with National Physical Activity and Sedentary Behaviour Guidelines declined. This highlights the need to advocacy for older adolescents and development of preventive initiatives prior to the transition out of school. Various situational transitions occurred post-school, mostly simultaneously. Living independently, not studying, and working were associated with declines in total sedentary behaviour over time. Future research could identify insulating characteristics and coping skills that need to be in place during secondary school associated with higher physical activity and lower sedentary behaviour. This could develop what this study found by revealing recommendations for protective attributes important for the various situational transitions that older adolescents commonly experience after leaving secondary school. **Chapter 8** 

Determinants and moderators of physical activity and sedentary behaviour during the transition out of secondary school

#### 8.1 Introduction

In Chapter 3, recent school leavers suggested that the main influences on physical activity during the transition out of secondary school were time use and social support, whereas tertiary study and social support were the main influences on sedentary behaviour. Chapters 5-6 found correlates of physical activity in late adolescence were physical activity enjoyment, goal setting and self-efficacy, friends or colleagues discouraging sedentary behaviour and social network count (all positive), and the correlate of sedentary behaviour in late adolescence was physical activity goal setting (negative). Therefore, this current chapter progresses the understanding of those findings by identifying individual, social and environmental longitudinal determinants of physical activity and sedentary behaviour during the transition out of secondary school, highlighting consistent or new findings.

The current study addresses the various limitations of previous studies that reported longitudinal determinants of physical activity and sedentary behaviour during the transition from adolescence into adulthood. Limitations include long follow-up periods<sup>163, 164</sup> and broad age ranges<sup>160, 162, 164, 165</sup>. A study assessed the transition out of secondary school and found that perceptions of peers' physical activity, family support and planning for VPA were determinants of physical activity<sup>159</sup>. Some studies assessed a limited scope of types of sedentary behaviour (only television, videos or video games on the computer on school days<sup>160</sup>) and physical activity (only LTPA frequency)<sup>164</sup>. Few studies assessed both physical activity and sedentary behaviour<sup>160, 163</sup>. Elucidating longitudinal determinants of physical activity and sedentary behaviour during the transition out of secondary school will provide important information on which to base intervention strategies.

Chapter 7 identified that the magnitude of change in sedentary behaviour differed according to which situational transition adolescents experienced after leaving school. School leavers who did not study at a tertiary institute, worked  $\geq 20$  h/wk or did not live with parents decreased sedentary behaviour after leaving school significantly more than respective comparison groups. It is possible that individual, social and environmental attributes during secondary school may moderate associations between situational transitions and physical activity and sedentary behaviour during the transition out of secondary school, helping to promote resilience to detrimental changes associated with specific situational transitions. However, to the author's knowledge, only one previous prospective study assessed

moderators of associations between situational transitions and changes in physical activity during the specific transition out of secondary school<sup>22</sup>. That study reported that belonging to a sporting club during school moderated the association between working full-time and decreasing LTPA post-school<sup>22</sup>. That is, belonging to a sports club during school helped ameliorate the decline in physical activity associated with full-time work after school. Sedentary behaviour and potential environmental moderators were not assessed in that study.

## 8.2 Chapter aims

The specific aims of this chapter are to:

- Identify individual, social and environmental longitudinal determinants of discretionary physical activity and total sedentary behaviour during the transition out of secondary school; and
- 2. Identify individual, social and environmental baseline moderators of associations between situational transitions experienced and discretionary physical activity and total sedentary behaviour during the transition out of secondary school.

#### 8.3 Methods

Data from the baseline and two-year follow-up of ProjectADAPT forms the basis of this chapter. The dataset used in this chapter excludes those participants studying at secondary school at the two-year follow-up and those who did not complete all three surveys. The analytical sample, therefore, comprises n=823 participants. Situational transitions examined in this chapter are tertiary student (yes, no), substantial weekly work hours ( $\geq 20$ , <20) and living with parents (yes, no) and were selected due to four reasons. The situational transitions chosen were themes that were discussed by interviewees in Chapter 3, are post-school pathways that could be targeted in interventions, represent the main situational changes that occur during the transition out of secondary school presented in Chapter 7, and were associated with changes in sedentary behaviour in Chapter 7.

#### 8.3.1 Data management

Outcome variables of this chapter (discretionary physical activity and total sedentary behaviour) and socio-demographic variables were described in Chapter 4 and are consistent throughout Chapters 4-7. Situational transition variables were mostly consistent with Chapter 7, except for substantial weekly work hours since 'not working' was coded as <20 h/wk.

In preparation for moderation analyses, baseline independent variables (previously described in detail in Table 4.1) were dichotomised to increase the interpretability of the findings. Meaningful cut-points were based on either the response options (if the scale had clear high and low options such as agree or disagree; 20/26 variables) or were determined from examining the mean distribution (6/26 variables; social network count, the number of electronic devices, physical activity equipment and televisions at home, and land use mix diversity and recreation facilities scores). Cutpoints applied were "0" to represent a low level of the scores and counts or "1" to represent a high level (Table 8.1). Some (10) variables had fairly even (40-60%) proportions of participants with low and high levels, whereas other variables were skewed. Firstly, >80% of the cohort reported high scores of physical activity enjoyment at baseline. Lastly, <20% of the cohort reported high scores of electronic games coparticipation with family and friends or colleagues, discouragement of sedentary behaviour from friends or colleagues, and neighbourhood noise at baseline.

**Table 8.1**Additional treatment of baseline independent variables in preparationfor moderation analyses and the proportion of the cohort who had high baselinelevels

	Score cut-points		High level
	(low vs high)	n	(%)
Individual variables (scores)			
PA enjoyment (3 items summed)	<9:>9	822	80.1
PA goal setting (3 items summed)		823	38.8
PA competence (1 item)	≤3;>3	823	54.3
PA self-efficacy (5 items summed)		823	40.7
TV avoidance self-efficacy (5 items	≤15;>15	814	54 7
summed)		011	0 117
Social variables			
Family (scores):		001	7.0
E-games co-participation (1 item)	$\sim 2$	801	/.0
PA co-participation (1 item)	<u>&lt;</u> 3; >3	81/	25.7
TV/DVDs co-participation (1 item)	10 0	819	50.2
PA social support (3 items summed)	<u>&lt;9;&gt;9</u>	817	51.2
SB discouragement (1 item)	≤3;>3	819	39.9
Friends/colleagues (scores):			
E-games co-participation (1 item)		823	16.0
PA co-participation (1 item)	<i>≤</i> 3; <i>&gt;</i> 3	823	34.9
TV/DVDs co-participation (1 item)		823	30.0
PA social support (3 items summed)	<i>≤</i> 9; >9	823	25.5
SB discouragement (1 item)	<u>≤</u> 3; >3	823	5.8
Social network count (3 items	<0.>0 neonle	877	517
summed)		022	51.7
Gym membership (1 item)	=0; =1	821	28.5
Environmental variables			
Home environment (counts):		0.01	
E-devices, no. of (count of 6 items)	$\leq 3; >3$ items	821	82.2
PA equipment, no. of (count of 5		822	33.5
items)	$\leq 2; >2$ items		
TVs, no. of (1 item)		819	44.1
Neighbourhood environment (scores):			
Noise (1 item)	≤3;>3	822	19.5
Walking environment (4 items	<12.>12	820	86 5
summed)	_12,* 12	020	
Safety (2 items summed)	<6.>6	821	68.1
Social cohesion (2 items summed)	_0,* 0	821	70.8
Land use mix diversity (count of 8		877	<u>49 0</u>
destinations within 10 mins)	· ?	022	T7.0
Recreation facilities (count of 5	_2, ~ 2	821	<i>4</i> 7 6
destinations within 10 mins)		021	+/.0

#### 8.3.2 Data analysis

T-tests were used to compare the distribution of independent variables at baseline between the cohort who completed all three surveys and were no longer at secondary school at the two-year follow-up (n=823), and non-completers or those at secondary school at the two-year follow-up (n=199). Baseline independent variables that were previously described in Chapter 4 (Table 4.1) and used in Chapters 5-6, were examined as potential determinants (Aim 1) and moderators (Aim 2). Physical activity and sedentary behaviour data from baseline and two-year follow-up were used. Excluding the one-year follow-up physical activity and sedentary behaviour data (Year 12, for most participants) did not limit the ability to address this chapter's aims which focussed on behaviour after leaving secondary school.

To address Aim 1, mixed-effects linear regression models were used. Discretionary physical activity or total sedentary behaviour at the two-year follow-up was the dependent variable. Each potential determinant (baseline value) was the independent variable in separate mixed-effects linear regression models (partially-adjusted models; 26 potential determinants resulted in 52 models). Two fully-adjusted models were then run to confirm determinants that included the significant independent variables from the partially-adjusted models (one each for physical activity and sedentary behaviour). Multicollinearity of the fully-adjusted models was checked but no variable had a high variance inflation factor (>5.0)<sup>292</sup>.

To address Aim 2, generalised estimating equation models were used. This was chosen because generalised estimating equation models fit a marginal distribution and a population average effect is sought, whereas linear mixed effects models often are interested in a conditional approach and the individual specific effect<sup>357</sup>. Within the models, the dependent variable was discretionary physical activity or total sedentary behaviour at the two-year follow-up and the independent variable was the situational transition. A series of models were used, adding an interaction term for each potential moderator (baseline value; three situational transitions and 26 potential moderators resulted in 78 models for discretionary physical activity and total sedentary behaviour, totalling 156 models; Appendices 8.1-8.2). If there was a significant interaction of the associations of the low and high levels of the moderator with discretionary physical activity or total sedentary behaviour (Appendix 8.3).

Graphing the adjusted means (margins) from models in a series of figures provided a visual representation of the moderation effect.

Mixed-effects linear regression models (Aim 1) and generalised estimating equations (Aim 2) were chosen to account for clustering within schools, as the data had nonindependence rising from its hierarchical structure<sup>358</sup>. The data was used in wideform over long-form for two main reasons. Firstly, this avoided a three-level model needed to analyse long-form data such as time-points nested within individuals nested within schools. Lastly, this avoided a three-way interaction between changes in behaviour over time, changes in the determinants over time, and, for Aim 2 only, whether situational transition were experienced.

For both aims, models adjusted for baseline discretionary physical activity or total sedentary behaviour, age, survey mode of administration and confounders. Confounders were consistent with Chapter 5-7. Confounders for the physical activity models were gender, maternal education, English as the primary language spoken at home and area-level SEP, and confounders in the sedentary behaviour models were gender, paternal education, English as the primary language at home, birth country (Australian, other) and remoteness (urban, rural). Further, as some participants were recruited from secondary schools, multilevel modelling was required so models had two-levels that included individuals nested within schools.

#### 8.4 **Results**

#### 8.4.1 Impact of study attrition on distribution of independent variables

Table 8.2 describes the baseline distribution of the independent variables among those included in the analytical sample (n=823), compared to those excluded from further analyses (n=199). Study completers who had left secondary school by the two-year follow-up reported lower scores for five variables compared to non-completers, specifically, co-participation in electronic games with family, friends or colleagues, social support for physical activity from friends or colleagues, and neighbourhood land use mix diversity.

**Table 8.2**Comparison of the distribution of independent variables at baselinebetween study completers and non-completers

			Pa	Participants			
				-	Non-		
	Max.	Co	mpleters	co	mpleters		
	score	п	mean±SD	п	mean±SD	р	
						-	
Individual variables (scores)							
PA enjoyment	15	822	11.7±2.6	197	$11.8 \pm 2.7$	0.807	
PA goal setting	18	823	8.6±3.8	199	$8.8 \pm 3.9$	0.504	
PA competence	5	823	3.5±1.2	197	3.6±1.2	0.142	
PA self-efficacy	25	823	14.3±4.9	199	$14.9 \pm 5.2$	0.127	
TV avoidance self-efficacy	25	814	16.1±4.7	195	16.0±4.9	0.768	
Social variables							
Family (scores):							
E-games co-participation	5	801	$1.6 \pm 1.0$	194	$1.9 \pm 1.3$	0.006	
PA co-participation	5	817	2.7±1.3	195	$2.7{\pm}1.4$	0.812	
TV/DVDs co-participation	5	819	3.4±1.2	197	$3.4{\pm}1.2$	0.582	
PA social support	15	817	9.5±3.2	195	$9.6 \pm 3.5$	0.716	
SB discouragement	5	819	3.0±1.3	196	3.1±1.4	0.858	
Friends/colleagues (scores):							
E-games co-participation	5	823	$2.0{\pm}1.2$	197	2.3±1.3	0.007	
PA co-participation	5	823	$2.9{\pm}1.4$	197	$2.9{\pm}1.4$	0.624	
TV/DVDs co-participation	5	823	$2.8 \pm 1.2$	197	$2.8 \pm 1.3$	0.787	
PA social support	15	823	$7.4 \pm 3.2$	197	$7.9 \pm 3.4$	0.041	
SB discouragement	5	823	$1.7\pm0.9$	197	$1.9 \pm 1.2$	0.017	
Social network count	$\infty$	822	11.6±10.1	195	12.6±10.0	0.229	
Gym membership (yes)	1	822	28.5%	199	28.1%	0.927	
Environmental variables							
Home environment:							
E-devices, no. of	6	821	4.6±1.1	197	4.5±1.3	0.470	
PA equipment, no. of	5	822	$2.0{\pm}1.2$	197	2.1±1.2	0.331	
TVs, no. of	8	819	$2.7 \pm 2.4$	197	$2.5 \pm 1.5$	0.340	
Neighbourhood environment							
(scores):							
Noise	5	822	$2.5 \pm 1.0$	195	$2.5 \pm 1.0$	0.938	
Walking environment	20	820	15.7±3.1	195	$15.9 \pm 2.7$	0.400	
Safety	10	821	7.3±1.9	195	$7.5 \pm 2.0$	0.068	
Social cohesion	10	821	7.3±1.7	195	$7.5 \pm 1.7$	0.115	
Land use mix diversity	8	822	$2.9 \pm 2.2$	195	$3.3 \pm 2.4$	0.023	
Recreation facilities	5	821	$2.4{\pm}1.4$	195	2.4±1.6	0.535	

T-test or Pearson's  $\chi 2$  test of significance by completed study status.

# **8.4.2** Longitudinal determinants of physical activity and sedentary behaviour during the transition out of secondary school

Table 8.3 presents the longitudinal determinants of discretionary physical activity and total sedentary behaviour during the transition out of secondary school. In the partially-adjusted models, discretionary physical activity after leaving secondary school was associated with eight baseline independent variables. In the fully-adjusted model, two variables remained significant. Every unit increase in physical activity enjoyment and goal setting scores in Year 11 was associated with 2.6 mins/day (95%CI:0.3, 5.0) or 1.6 mins/day (95%CI:0.1, 3.1) higher discretionary physical activity after leaving secondary school, respectively, independent of baseline physical activity. For sedentary behaviour, one of the seven associations from the partially-adjusted models remained significant in the fully-adjusted model. Every unit increase in television avoidance self-efficacy score (25 units maximum) in Year 11 was associated with 3.9 mins/day (95%CI:-7.1, -0.7; p=0.016) lower total sedentary behaviour after leaving secondary school, respectively, independent of baseline sedentary behaviour.

	Discretionary PA mins/c	lay at 2-year f/up	Total SB mins/day at 2-year f/up B (95%CI)			
	B (95%C	I)				
		Fully-adjusted		Fully-adjusted		
	Partially-adjusted model	model, <i>n</i> =733	Partially-adjusted model	model, <i>n</i> =719		
Individual variables (scores)						
$\mathbf{P}$ $\mathbf{P}$ $\mathbf{P}$	37(20 55)***	26(0350)*	-11 3 (-16 0 -5 6)***	(133,18)		
DA goal actting	$3.7 (2.0, 3.3)^{+++}$	$2.0(0.3, 3.0)^{+}$ 1 6 (0 1 2 1)*	64(10522)**	-4.2(-15.5, 4.0)		
PA goal setting	2.0(1.4, 4.2)	$1.0(0.1, 3.1)^{1}$	$-0.4 (-10.5, -2.2)^{++}$	-1.5(-0.0, 4.0)		
PA competence	4.0 (0.9, 8.3)*	-5.1(-6.1, 1.9)	$-23.1(-34.6, -11.3)^{+++}$	-0.9(-23.0, 11.3)		
PA self-efficacy	$1.5(0.0, 2.5)^{**}$	0.4 (-0.0, 1.3)	-5.5 (-8.9, -2.1)**	-1.5(-0.1, 5.1)		
I v avoidance self-efficacy	0.7 (-0.1, 1.6)	-	-4.9 (-8.0, -1.9)**	-3.6 (-6.9, -0.3)*		
Social variables						
Family (scores):						
E-games co-participation	-1.3 (-5.7, 3.1)	-	13.8 (-2.2, 29.7)	-		
PA co-participation	6.7 (3.2, 10.2)***	3.6 (-1.7, 9.0)	-7.9 (-18.5, 2.7)	-		
TV/DVDs co-participation	-1.9 (-6.0, 2.1)	-	8.5 (-3.6, 20.7)	-		
PA social support	2.8 (1.5, 4.2)***	0.4 (-1.8, 2.6)	-4.3 (-9.0, 0.3)	-		
SB discouragement	2.0 (-1.6, 5.6)	-	5.2 (-6.1, 16.4)	-		
Friends/colleagues (scores):						
E-games co-participation	3.4 (-1.5, 8.3)	-	7.5 (-5.4, 20.5)	-		
PA co-participation	2.8 (-0.3, 6.0)	-	-8.4 (-19.2, 2.3)	-		
TV/DVDs co-participation	0.2(-3.8, 4.1)	-	2.1 (-11.6, 15.8)	-		
PA social support	1.6 (0.1, 3.1)*	-0.2 (-2.1, 1.7)	-2.7 (-7.2, 1.8)	-		
SB discouragement	5.4(-0.3, 11.1)	-	-3.3 (-20.6, 14.0)	-		
Social network count	0.5 (-0.0, 1.0)	-	-0.8 (-1.7, 0.2)	-		
Gym membership	9.9 (-0.3, 20.0)	-	-40.9 (-75.3, -6.5)*	-29.7 (-64.8, 5.5)		

**Table 8.3**Baseline determinants of discretionary physical activity and total sedentary behaviour after leaving secondary school<sup>1</sup>

#### Table 8.3Continued

	Discretionary PA mins/d B (95%C	lay at 2-year f/up I)	<b>Total SB mins/day at 2-year f/up</b> B (95%CI)		
	Partially-adjusted model	Fully-adjusted	Partially-adjusted model	Fully-adjusted	
	T at trany-augusteu mouer	mouel, <i>n</i> =755	Tartiany-aujusteu mouer	mouci, <i>n</i> =/1)	
Environmental variables					
Home environment:					
E-devices, no. of	2.4 (-1.0, 5.8)	-	-0.5 (-13.1, 12.2)	-	
PA equipment, no. of	5.3 (1.6, 9.0)**	3.1 (-0.6, 6.9)	-13.0 (-25.1, -0.9)*	-6.9 (-19.4, 5.6)	
TVs, no. of	-0.5 (-2.5, 1.2)	-	1.7 (-4.3, 7.7)	-	
Neighbourhood environment (scores):					
Noise	2.0 (-1.9, 5.8)	-	1.0 (-13.2, 15.3)	-	
Walking environment	0.8 (-0.6, 2.3)	-	-0.5 (-5.3, 4.4)	-	
Safety	2.1 (-0.5, 4.8)	-	-0.3 (-9.9, 9.4)	-	
Social cohesion	2.2 (-0.9, 5.4)	-	-1.6 (-12.2, 9.0)	-	
Land use mix diversity	-0.8 (-2.7, 1.0)	-	-2.9 (-10.3, 4.5)	-	
Recreation facilities	1.6 (-1.9, 5.1)	-	0.8 (-9.6, 11.3)	-	

\*p<0.05, \*\*p<0.01, \*\*\*p<0.001: Mixed-effects linear regression models.

<sup>1</sup> Partially-adjusted models adjusted for baseline PA/SB, age, survey mode and confounders (PA: gender, maternal education, English as the primary language spoken at home, and area-level SEP; SB: gender, paternal education, English as the primary language spoken at home, birth country and remoteness), with repeated measurements for individuals nested within schools. Fully-adjusted models additionally adjusted for significant variables from the partially-adjusted models.

# **8.4.3** Moderators of associations between situational transitions and physical activity and sedentary behaviour during the transition out of secondary school

#### Situational transition: Tertiary student status at two-year follow-up

There was no significant association between tertiary student status after leaving school and changes in discretionary physical activity (p=0.087; Akaike's information criteria (AIC) change <2) at the two-year follow-up compared to baseline but there was for changes in total sedentary behaviour (p=0.000; AIC change >2; Appendix 7.2). For sedentary behaviour, there were no significant interactions between tertiary student status and the potential moderators (Appendix 8.2). However, receiving discouragement for sedentary behaviour from family (B:-27.0; 95%CI:-49.6, -4.5; p=0.019) and the number of televisions at the home (B:-23.5; 95%CI:-45.2, -1.8; p=0.034) at baseline moderated the associations between tertiary student status and discretionary physical activity at the two-year follow-up (Appendix 8.1; Figure 8.1). Stratified analyses showed that the low and high levels of the two moderators produced associations that differed in direction (Appendix 8.3).

In summary, those not studying at a tertiary institute reported higher discretionary physical activity compared to those studying, only among those who reported higher sedentary behaviour discouragement from family and televisions at home in Year 11.





•: not studying; ▲: studying; \*p<0.05; error bars: 95% confidence intervals.

#### Situational transition: Working hours at two-year follow-up

There was a significant association between working hours after leaving school and changes in total sedentary behaviour at the two-year follow-up compared to baseline (p=0.000; AIC change >2; Appendix 7.2). Baseline sedentary behaviour discouragement from friends or colleagues (B:-3.5; 95%CI:-5.6, -1.3; p=0.002) moderated the associations between working hours and total sedentary behaviour at the two-year follow-up (Appendix 8.2; Figure 8.2). Stratified analyses showed that the low and high levels of the moderator produced an association that differed in strength (Appendix 8.3).

To summarise, those who worked  $\geq 20$  h/wk post-school reported lower total sedentary behaviour compared to those who worked less, and this was regardless of how much discouragement of sedentary behaviour from family in Year 11. However, the difference in total sedentary behaviour was more marked among those with high discouragement of sedentary behaviour from family in Year 11.

**Figure 8.2** Moderation of associations between substantial weekly work hours and total sedentary behaviour at the two-year follow-up, adjusted marginal means, n=804



•: worked <20 h/wk; ▲: worked ≥20 h/wk; \*\*\*p<0.001; error bars: 95% confidence intervals. There was no significant association between working hours after leaving school and changes in discretionary physical activity at the two-year follow-up compared to baseline (p=0.584; AIC change <2). Despite this, there were five baseline moderators of the association between working hours and discretionary physical activity at the two-year follow-up (Appendix 8.1; Figure 8.3). Firstly, the individual moderator was physical activity self-efficacy (B:27.4; 95%CI:8.1, 46.6; p=0.005). Secondly, social moderators were co-participation with family in physical activity (B:31.3; 95%CI:9.3, 53.4; p=0.005) and sedentary behaviour discouragement from friends or colleagues (B:62.6; 95%CI:21.8, 103.4; p=0.003). Lastly, environmental moderators were electronic devices at home (B:26.2; 95%CI:1.7, 50.8; p=0.036) and neighbourhood land use mix diversity (B:-20.1; 95%CI:-39.4, -0.8; p=0.041). Stratified analyses showed that the low and high levels of the moderators produced associations that differed mainly in strength, except for electronic devices at home which differed in direction (Appendix 8.3).

In summary, those working  $\geq 20$  h/wk reported markedly higher discretionary physical activity compared to those working less, only among those who had high physical activity self-efficacy, co-participation with family in physical activity, sedentary behaviour discouragement from friends or colleagues, and electronic devices at home in Year 11. Whereas the opposite pattern was evident for neighbourhood land use mix diversity (higher physical activity only among those with low scores in Year 11).

**Figure 8.3** Moderation of associations between substantial weekly work hours and discretionary physical activity at the two-year follow-up, adjusted marginal means, n=804



•: worked <20 h/wk; ▲: worked ≥20 h/wk; \*p<0.05, \*\*p<0.01, \*\*\*p<0.001; error bars: 95% confidence intervals.
## Situational transition: Living with parents status at two-year follow-up

Living with parents status after leaving school was not associated with changes in discretionary physical activity (p=0.662; AIC change <2; Appendix 7.2) at the twoyear follow-up compared to baseline but was associated with changes in total sedentary behaviour (p=0.009; AIC change >2; Appendix 7.2). Family and peer environments during Year 11 moderated associations between living with parents status and discretionary physical activity and total sedentary behaviour after leaving school (Appendices 8.1-8.2; Figure 8.4). Four moderators of physical activity were discouragement of sedentary behaviour from family (B:-22.7; 95%CI:-45.2, -0.3; p=0.047), co-participation in electronic games with friends or colleagues (B:40.3; 95%CI:5.1, 75.4; p=0.025), co-participation in television or DVDs with friends or colleagues (B:24.8; 95% CI:0.9, 48.7; p=0.042), and physical activity equipment at home (B:22.4; 95%CI:0.2, 44.6; p=0.048). One moderator of sedentary behaviour was co-participation in physical activity with family (B:1.4; 95%CI:0.1, 2.8; p=0.037). Stratified analyses showed that the low and high levels of the five moderators produced associations that differed in direction, not strength (Appendix 8.3).

To summarise, among those living independently post-school, there was a pattern for higher physical activity if they received frequent sedentary behaviour discouragement from family in Year 11, lower physical activity if they had high co-participation in electronic games and in television/DVD viewing with friends or colleagues in Year 11, and lower sedentary behaviour if they had high co-participation in physical activity with family in Year 11. Whereas, those living with parents post-school had higher physical activity if they reported high levels of physical activity equipment at home in Year 11.

**Figure 8.4** Moderation of associations between living with parents and discretionary physical activity or sedentary behaviour at the two-year follow-up, adjusted marginal means, n=805



•: lived independently; ▲: lived with parents; \*p<0.05, \*\*p<0.01; error bars: 95% confidence intervals.

# 8.5 Discussion

This is one of the first prospective studies to examine individual, social and environmental determinants of physical activity and sedentary behaviour during the transition out of secondary school<sup>159</sup>. Although numerous independent variables were significantly associated in the partially-adjusted models, key determinants of physical activity were physical activity enjoyment and goal setting, while the key determinant of sedentary behaviour was television avoidance self-efficacy. This study is also one of the first to examine individual, social and environmental moderators of physical activity and sedentary behaviour in those who experience multiple situational transitions post-school<sup>22</sup>. Three key situational transitions post-school were tertiary study status, weekly working hours, and household composition. Moderators for the three key situational transition were mostly unique; however, discouragement of sedentary behaviour and co-participation with family, friends or colleagues in recreational screen time emerged as reoccurring moderators.

The determinants are attributes needed as a minimum for higher physical activity and lower sedentary behaviour and were individual and internalised characteristics. In contrast, the moderators found are needed during times of change and instability. Further, the moderators represent support mechanisms, were external and were characteristics that tend to change with age, such as the family or home environment or social attributes. Moderators are important in the presence of a disruptor, such as age or circumstance. Although changes in independent variables were not analysed, since the focus was to elucidate what to have in place during school that is protective, it is likely that some attributes changed over time. Nonetheless, the determinant and moderator findings highlight the high or low baseline levels of attributes that set individuals up for future health behaviours. As an example, for those who go on to live independently post-school to have higher physical activity and lower sedentary behaviour, it is recommended that they maintain physical activity equipment levels and that families co-participate in physical activity and discourage sedentary behaviour, particularly recreational screen time with friends.

The moderators were mostly social and external attributes. In contrast, the key determinants reflected internal attributes such as preferences and prioritising physical activity. These are, therefore, critical to develop during adolescence since they impact later physical activity. Three key points of interest about the moderators and key determinants are, firstly, discouragement of sedentary behaviour appears to be

important regardless of situational transition experience post-school. A pattern emerged suggesting family, friends and colleagues have an important role to play in communicating messages about reducing excessive sedentary behaviour during this transition. Secondly, healthy behaviours in place in Year 11 may help individuals cope with changes during the transition. As an example, whose who have established higher physical activity self-efficacy during school may be able to overcome barriers post-school in order to be active, such as time constraints from working substantial weekly hours. Lastly, a recommendation is that families are encouraged to coparticipate in physical activity with adolescents because those who with established high levels and went onto live independently or work substantially post-school had higher physical activity.

There are multiple potential reasons for the importance of sedentary behaviour discouragement as a reoccurring moderator for physical activity and sedentary behaviour. Firstly, it may be a marker of having a social network with high health literacy which mediated the resulting high physical activity and low sedentary behaviour. Secondly, it may reflect the high dependency on peers for cultural influences during adolescence<sup>359</sup>. Appeasing social networks, wanting to fit in and seeking approval of peers are commonly important social goals during adolescence<sup>360</sup>. There is a progression from dependence on parents in childhood to peers in adolescence then self-authority in adulthood<sup>359</sup> which is a reason peer pressure is common during adolescence. The current data does not provide insight into whether individuals have maintained or changed their social networks compared to their friends or colleagues who delivered these messages, and this may help explain this finding. Thirdly, having family members verbalise their expectations by reminding to avoid excessive sedentary behaviour may represent parents who provide good leadership for their children's health by actively engaging with choices. These parents may have an authoritative parenting style<sup>361</sup> which is associated with other health benefits such as mental health, dietary behaviours and weight status, academic success and grit<sup>362-365</sup>. Future research could examine if study parenting styles C and D (permissive, lenient, uninvolved and negligent parenting<sup>361</sup>) are associated with physical activity and sedentary behaviour after leaving school and if they moderate associations between living independently or with parents post-school and health behaviours.

This study had some null findings. Firstly, it is unclear why low land use mix diversity and electronic devices are associated with higher physical activity among those who work substantial weekly hours, since literature says the opposite is helpful for physical activity during adolescence<sup>366, 367</sup>. The role of employment needs to be explored in future research. A potential explanation is that more disposable income allows participants access to more opportunities regardless of distance to home, compared to baseline; whereas, those with high land use mix diversity did not change physical activity as much. This is consistent with a previous study that reported that belonging to a sporting club at baseline moderated the effect size of the association between working full-time and LTPA a year post-baseline<sup>22</sup>. Secondly, it is unexpected that established high levels of recreational screen time with friends or colleagues in Year 11 are associated with lower physical activity after school if the individuals lived independently, as the importance of household composition over employment or study is unclear. Future research is needed to understand why the association was found for physical activity and not sedentary behaviour and whether this supports the interdependence between physical activity and sedentary behaviour. Lastly, some expected associations were not found. Social support for physical activity from family did not moderate physical activity post-school among those living independently, there were no key social and environmental determinants of physical activity and sedentary behaviour, and most of the significant determinants in the partially-adjusted models were not moderators.

Some moderation findings were similar to the perceived influences on physical activity and sedentary behaviour specific to situational transitions experienced postschool suggested in Chapter 3 by interviewees. Firstly, if responders commenced tertiary education, influences were choice of active or sedentary tertiary course, increased free time and increased hours studying; however, this current study did not assess or analyse those characteristics. Secondly, if participants commenced full-time employment, influences were increased working hours and technology access and use. This finding was similar to the number of electronic devices at home moderating the association between working hours and physical activity during the transition out of secondary school. Lastly, if interviewees moved out of the family home, influences were increased independence and less rules and boundaries from parents. This current study similarly found high sedentary behaviour discouragement from family in Year 11, potentially operationalised as a house rule or boundary, was associated with high physical activity post-school, among those living independently post-school.

## **Strengths and limitations**

A key strength of this chapter is that it assessed determinants and moderators of physical activity and sedentary behaviour in late adolescence during the transition out of secondary school from three levels of the ecological model<sup>20, 21</sup> which have not been previously revealed or adequately described. Sampling Year 11 students annually captured data on the understudied minority group of early school leavers, as well as numerous post-school pathways. However, generalisability of findings is reduced from a cohort not randomly sampled and predominantly women. An ecological model<sup>20, 21</sup> proved to be appropriate and applicable since individual, social and environmental moderators were found for the associations between situational transitions and physical activity or sedentary behaviour after leaving secondary school. Specific to the instrument used, self-reported physical activity and sedentary behaviour data includes bias and the high number (26) of independent variables tested may have increased the probability of finding results by chance. Additionally, two-thirds of the moderators were variables comprised of a single item from the survey, the implications of which are higher measurement error and lower instrument reliability since only one source of variance is captured<sup>281</sup>.

Importantly, some moderators were newly discovered and were not determinants or correlates reported in Chapters 5-6. These were sedentary behaviour discouragement, the number of televisions, electronic devices and physical activity equipment at home, and land use mix diversity. However, the moderator findings should be interpreted with consideration of how situational transitions are not mutually exclusive, as evidenced by the large overlap between those who are tertiary students, working and living with parents post-school. Future research needs to assess the cumulative effect of simultaneous situational transitions on changes in physical activity and sedentary behaviour during the transition out of secondary school, and what the key individual, social and environmental attributes are that play a role in those associations.

Positively, most baseline attributes did not vary significantly between study completers and non-completers, signalling that an adjustment method did not need to be applied to weight the data from the analytical sample<sup>368</sup>. However, the final cohort had lower baseline levels of five independent variables and, notably, three of those were moderators (co-participation in electronic games with friends or colleagues, discouragement of sedentary behaviour from friends or colleagues and neighbourhood land use mix diversity). Therefore, those three moderation associations may have been weakened due to attrition bias. Further, while interpretation of moderation analyses was aided by stratifying the cohort by low and high levels of the baseline independent variables, a cost of dichotomising is a loss of power. In some instances, the mean or median of the cohort was the basis for the cutpoint of the dichotomised variable rather than an external reference point, which may limit the specificity of the findings to this cohort and decrease the generalisability of the moderation findings to the broader population.

Alternative analytical approaches may help explain this study's findings. Firstly, this chapter analysed baseline attributes to establish what protective attributes need to be in place during secondary school to inform interventions. Future research is needed to analyse changes in independent variables and whether they mediate the association between situational transitions and changes in physical activity and sedentary behaviour during the transition out of secondary school. Secondly, similar to the correlation findings in Chapters 5-6, there were numerous significant determinants in the partially-adjusted models but most dropped out of the fully-adjusted model despite no multi-collinearity. This suggests that physical activity competence, selfefficacy and equipment are still important determinants of physical activity and sedentary behaviour during the transition out of secondary school but are not as much of a driver as physical activity enjoyment, goal setting and television avoidance self-efficacy. Mediation models may help understand these findings. Lastly, an alternative analytical approach to moderation and interaction analyses and instead of relying on means, trajectory/path analysis and latent class growth analysis (LCA)/modelling may provide more insight into the explaining mechanisms of physical activity and sedentary behaviour during the transition out secondary school.

200

The findings of this study have possible implications in additional areas of study, in particular, other key health behaviours during late adolescence of improving diet quality, decreasing smoking and decreasing excess alcohol consumption. Examining the individual, social and environmental moderators of associations between situational transitions experienced after leaving secondary school and other health behaviours may confirm or elucidate additional population groups to target and protective attributes that inform interventions and curriculums. For example, future research could examine longitudinal changes in meal patterns during the transition out of secondary school, especially as adolescents move out of the family home. Further, any protective attributes that, if in place during the school years, may promote resilience to poor eating behaviours in adulthood could be identified. Identifying tailored strategies to different target groups or an individual risk profile<sup>176</sup> is important for informing interventions and has been implemented in studies aimed at changing health behaviours of diet, smoking, alcohol consumption and breast cancer screening<sup>177, 178</sup>.

## Conclusion

Physical activity enjoyment, goal setting and television avoidance self-efficacy emerged as key determinants of later physical activity and sedentary behaviour. Further, sedentary behaviour discouragement and recreational screen time coparticipation with friends, family and colleagues emerged as reoccurring moderators of physical activity and sedentary behaviour for the situational transitions of weekly work hours, household composition and tertiary study status. The determinant and moderator findings highlight what attributes are protective against inadequate physical activity and excessive sedentary behaviour during the transition out of secondary school. The findings also highlight whether high or low levels of these attributes need to be established during secondary school. Future research needs to identify efficacious and feasible strategies that influence these insulating attributes prior to the transition out of secondary school, followed by a translational approach of embedding scalable strategies into curriculums, interventions and community health promotion initiatives.

Chapter 9

Conclusions

# 9.1 Introduction

his thesis was designed to contribute to the field of behavioural epidemiology<sup>19</sup>. Specifically, this thesis aimed to:

- 1. Examine physical activity and sedentary behaviour changes during the transition out of secondary school; and
- 2. Examine individual, social and environmental influences on physical activity and sedentary behaviour during the transition out of secondary school.

This thesis utilised a mixed methods approach and used data from three research projects to address the thesis aims. Firstly, a qualitative investigation was conducted using semi-structured interviews. The second and third projects were part of a larger observational longitudinal study (ProjectADAPT) and included a test-retest reliability study. This chapter discusses the overall findings of the thesis in relation to the life transition model<sup>18</sup> and the ecological model<sup>20, 21</sup>, summarises key strengths and limitations, and proposes implications for future research directions and practice.

# 9.2 Overview of findings

This section synthesises the main findings around two themes. The first concerns how much physical activity and sedentary behaviour older adolescents engaged in during the transition, the second concerns the common influences underlying these health behaviours.

# 9.2.1 Behaviour pattern changes

This thesis found that physical activity and sedentary behaviour tend to change as older adolescents transition out of secondary school, with few maintaining these behaviours. This was apparent in both the qualitative and quantitative studies and supports the hypothesis of the life transition model<sup>18</sup> and the findings from two systematic reviews<sup>8, 10</sup> that transitions are times of instability and variability that may disrupt health behaviours. The qualitative study indicated that not all recent school leavers change physical activity and sedentary behaviour in the same way, with ~84% increasing or decreasing. The quantitative study found mean discretionary physical activity and total sedentary behaviour declined. Despite the latter, recreational screen time increased and participants suggested this was due to more

free time post-school and less boundaries after moving out of the family home. The situational transitions post-school of working  $\geq 20$  h/wk, exclusively working, not studying, and living independently resulted in declines in total sedentary behaviour over transition out of secondary school (in order of greatest declines), whereas there were no interactions between time and situational transitions for discretionary physical activity.

#### 9.2.2 Interventions leverage points

The influences underlying physical activity and sedentary behaviour had a number of commonalities. In the quantitative study, physical activity goal setting and enjoyment were associated with physical activity during secondary school and during the transition. This suggests that physical activity may be lower and decline if it is not prioritised and liked because of late adolescence being a busy life stage with competing demands on time from socialising, study and work. For both physical activity and sedentary behaviour, physical activity goal setting was a correlate during secondary school, and sedentary behaviour discouragement and co-participation with social networks were moderators for multiple situational transitions. Lastly, in the qualitative study, social support was considered to be a major influence on physical activity and sedentary behaviour changes during the transition out of secondary school. Therefore, physical activity goal setting and enjoyment, sedentary behaviour discouragement, co-participation and social support are key leverage points to focus strategies on within interventions for older adolescents, irrespective of students' demographic characteristics or the situational transitions experienced post-school. Future research needs to examine the practicalities of increasing these influences.

Physical activity and sedentary behaviour were cross-sectionally correlated with more individual attributes than social or environmental. This contrasted the longitudinal findings to an extent, with more social than individual or environmental attributes associated with physical activity and sedentary behaviour post-school. Further, most of the helpful (i.e., associated with higher physical activity or lower sedentary behaviour) moderators in those who were working  $\geq 20$  h/wk, not studying or living independently post-school were social attributes. These findings inform the focus of interventions during secondary school, support the sociable natures and social needs of adolescents<sup>23</sup>, and support the use of the ecological model<sup>20, 21</sup> to understand the influences on older adolescents' health behaviours.

205

## 9.3 Strengths and limitations

#### 9.3.1 Strengths

A key strength of this thesis was its novel focus on the impact of the transition period of leaving secondary school on physical activity and sedentary behaviour, since there has been limited research on this issue. This thesis has contributed to the field of behavioural epidemiology by addressing this gap. Most previous studies have included predominantly retrospective quantitative designs or did not survey students while at secondary school<sup>118, 170, 171, 173</sup>, or were based on a limited range of situational transitions<sup>118, 125, 170-173</sup>. This thesis utilised prospective quantitative data during the transition, which few other studies have<sup>159</sup>, as well as retrospective qualitative data. Another key strength of this thesis is that it measured physical activity only<sup>118, 173</sup>. Sedentary behaviour is important to assess as it is increasingly being recognised as a threat to health and wellbeing, especially excessive prolonged (uninterrupted) sedentary behaviour<sup>109</sup>.

An important strength of this thesis was that it identified moderators, which are under-researched when studying explanations for why older adolescents engage in higher or lower physical activity and sedentary behaviour during the transition out of secondary school. Also, it provides insights into modifiable attributes to inform interventions, in contrast to commonly tested biological and demographic characteristics<sup>369</sup>. Further, the large range of modifiable attributes measured were from across the individual, social and environmental levels of the ecological model<sup>20, 21</sup>, whereas many previous studies about influences on physical activity and sedentary behaviour in adolescence assessed attributes from only one level (mainly social)<sup>160, 162-164</sup>.

Another key strength of this thesis is that is utilised a mixed methods approach. The inductive qualitative study informed the quantitative study. Although baseline had already commenced for ProjectADAPT, the qualitative findings informed future waves of ProjectADAPT and were critical for interpreting survey findings. The test-retest reliability performed on the ProjectADAPT survey added further rigour to this thesis. This methodological study provided insight into the internal validity of ProjectADAPT and also informed interpretation of findings. This was important since some survey items were amended from original sources and others were newly

developed. The key strengths of the ProjectADAPT study specifically include an adequate cohort of >1 000 older adolescents at baseline, drawn from six areas (lowest, mid and highest tertiles of socio-economic areas from urban or rural remoteness), and high retention.

#### 9.3.2 Limitations

There are two main limitations of this thesis. The first is that this thesis employed self-reported measures to assess physical activity and sedentary behaviour and underlying influences. Self-reported data are limited by human recall errors and social desirability response bias<sup>370</sup>, suggesting that physical activity may be overestimated and sedentary behaviour may be underestimated. Although, the ProjectADAPT survey had acceptable test-retest reliability and internal validity, and the associations found were logical and reasonable. However, criterion validity of the behavioural items (physical activity and sedentary behaviour measurement; dependent variables) was not determined, nor was construct and content validity (face validity) of the influence items (independent variables). Previous validity testing may not apply to the newly developed items, amended items or to the population of older adolescents<sup>54, 68, 220, 371</sup>; therefore, a validity study using this age group is needed that addresses these limitations, particularly for the dependent measures. Future research about changes in physical activity and sedentary behaviour should use device-based measurement tools concurrently with a log book to reduce bias, improve reliability and validity, and provide more sensitive data on prolonged sitting, short and long bouts of behaviour, and the timing of these behaviours.

The second main limitation relates to the generalisability of findings to the broader youth population. Due to a range of recruitment difficulties during the ProjectADAPT study, two recruitment strategies were adopted (school recruitment and social media). However, the non-random sample comprised mostly of young women (74% vs 49% of 17-year-old Victorians<sup>263</sup>). Because of the disproportionate gender distribution potentially causing a lack of power for boys, gender was adjusted for in many analyses, as opposed to other studies on physical activity in adolescence stratifying influences by gender<sup>372</sup>. Additionally, there was a higher proportion of the cohort living in a rural area than population averages (29% of the baseline cohort vs 9% of Victorians<sup>264</sup>). Further, study completers differed to non-completers such that their baseline discretionary physical activity was lower, total sedentary behaviour

was higher and more were recruited via social media, completed the surveys online and identified as a woman. Future research would benefit from a random and representative sample.

# 9.4 Implications

# 9.4.1 Future research implications

There are two key areas for future research arising from this thesis. The first concerns the need for additional studies to understand, explain and expand some of this thesis' findings, especially null and unexpected findings on how and why physical activity and sedentary behaviour change during the transition out of secondary school. The second area concerns development and testing of strategies based on this thesis' findings to implement within interventions and translational research.

# Understanding how and why physical activity and sedentary behaviour change during the transition out of secondary school

This thesis examined influences on physical activity and sedentary behaviour. Future research could also consider influences on complying with National Guidelines and domain-specific behaviours to inform more targeted interventions. In addition, there is emerging literature examining how various types of physical activity and sedentary behaviour cluster in older adolescents<sup>313, 373, 374</sup>. Future research could examine clusters of these behaviours and how these typologies change during the transition out of secondary school, exposing new information on resilient and at-risk older adolescents, which follows a resilience model of health research<sup>174</sup>. A range of modifiable health behaviours, such as meal patterns, alcohol consumption, cigarette smoking and illicit drug use, could also be included in such research to identify health-promoting and health-adverse clusters, or different combinations, and their attributing factors. This is similar to recent studies of tertiary students in Ireland<sup>375, 376</sup>.

An analytical method that may further advance understanding of why behaviour changes during the transition out of secondary school is examining mediators of associations between situational transitions and changes in physical activity and sedentary behaviour. Mediation analyses could also be used in future research using larger samples to assess whether physical activity and sedentary behaviour mediate changes in health outcomes over time. Another analytical method for use in future research is examining how changes in influences over time may impact physical activity and sedentary behaviour. For example, the provision of parental support may decline as adolescents transition out of secondary school and progress to becoming fully independent, and this may partially explain changes in physical activity and sedentary behaviour during the transition period. Studying changes associated with relocation may also be helpful. Among those who relocate, the physical environment is likely to change and the environment during secondary school may be less important for changes in physical activity and sedentary behaviour than this shift in real or perceived physical activity supportiveness of the new environment.

The environments that the ProjectADAPT study assessed were neighbourhood, home and family; however, adolescents are also exposed to other environments. To address the lack of environmental influences identified in this thesis, future research should examine time in LPA and physical activity during school and work hours, as well as how school and workplace conditions during secondary school influence physical activity and sedentary behaviour changes during the transition into adulthood. For example, facility access and prompts and cues to sit at school that create social norms may be influential<sup>21</sup>. A future research question could be whether students exposed to activity-permissive classroom environments during school engage in less prolonged sitting over time. Another consideration for future research on environmental influences is using objective measures, although limitations include imposed and minimally relevant buffers. Attributes such as walkability characteristics, walking and cycling infrastructure, and employment density could be assessed via a Geographical Information System (GIS) software package<sup>377</sup> or auditing of streetscapes via desk-top or in-person<sup>377-379</sup>. Objective measures are more specific and sensitive to changes, whereas most self-reported environment measures are usually very general and related to awareness of surroundings. Lastly, to identify environmental influences, particularly on sedentary behaviour, qualitative studies

209

need to be conducted to uncover important and modifiable characteristics in the built, physical, school, work, home and neighbourhood environments.

Changes in physical activity were not associated with situational transitions in this thesis. Although this may reflect that adolescents tend to similarly change physical activity regardless of post-school pathway, studies with different methodological approaches are needed. Group-based trajectory analyses could explore whether everyone tends to change physical activity in the same way or if there are groups of people who might be more resilient to change than others<sup>380</sup>. Additionally, studies with larger samples would allow a broader range of situational transitions to be analysed, including multiple combinations of simultaneous post-school pathways and their cumulative effects on physical activity and sedentary behaviour.

#### Future interventions

The findings of this thesis highlight a need for interventions to boost the influences identified so adolescents are resilient to a decrease in physical activity and an increase in sedentary behaviour during the transition out of secondary school. Such preventative interventions would lead to the final stage of the Behavioural Epidemiology Framework of implementing translational research into practise<sup>19</sup>. An important next step would be to identify efficacious, feasible and scalable strategies that target older adolescents and their physical activity goal setting, enjoyment, selfefficacy and equipment at home, television avoidance self-efficacy, sedentary behaviour discouragement from family, friends and colleagues, co-participation in physical activity with family, and co-participation in recreational screen time with friends and colleagues. Further, future research is needed to confirm optimal timing of interventions and whether late adolescence is too late to increase some attributes such as physical activity enjoyment and if interventions are needed that commence in mid or early adolescence, or childhood. This is evidenced by declines in physical activity throughout adolescence; for example, sport participation reduced in Australian children from 66% for 12-14-year-olds to 60% for 9-11-year-olds<sup>381</sup>. Lastly, pinpointing settings to deliver strategies based on this thesis' findings is critically important. For adolescents about to transition out of school, the secondary school and tertiary settings, and families and communities are likely to be the most important settings to intervene. Implications for these settings are outlined in the following section.

## 9.4.2 Practice implications

## Secondary schools

This thesis' findings pointed to the role of student empowerment, voice and engagement which suggests employing school-based positive psychology interventions or evidence-based coaching. These student well-being programs could have a co-design process and be embedded into pastoral care or the curriculum in established subjects such as Arts, Health Development, or Religion. Targeted information delivered could focus on influencing the key correlates, determinants and moderators identified in this thesis, such as each individual's physical activity goal setting, self-efficacy and enjoyment, and promoting the key role students can play in discouraging their friends, family and colleagues from engaging in sedentary behaviour.

This thesis found a low proportion of physical activity and high proportion of sedentary behaviour during school breaks. Therefore, schools should offer more options and modes of physical activity for older adolescents during recess and lunch times and implement policies that discourage sedentary behaviour to create a supportive culture where movement is the norm. Physical activity interventions during school breaks have had some success for adolescents<sup>382-384</sup>, despite challenges such as inactive norms, social anxieties and peer judgement<sup>385</sup>. An after-school intervention also had success for adolescents in increasing physical activity and decreasing recreational sedentary behaviour, as well as improving health outcomes of lower waist circumference, waist to height ratio, BMI, BMI-z-score and scores on an unhealthy foods index, and higher self-esteem)<sup>386</sup>. Schools should promote active transport to school, especially to older adolescents living within 4km of school<sup>387</sup>, as well as sports to decrease overweight/obesity<sup>78</sup>.

Participants' recommendations could be explored, including introducing policies and resources that support active breaks during class and less sedentary lessons (e.g. using sit-stand desks), walking classrooms and linking students to community sport. Early work on the efficacy of these strategies in secondary schools appears promising, such as sit-stand desks and prompts<sup>388</sup>, standing classrooms<sup>389</sup>, flexible learning spaces<sup>390</sup> and links to community sport<sup>146</sup>.

## **Tertiary institutions**

This thesis' findings have broad implications for tertiary institutions. Participants suggested policies and resources are needed in tertiary institutions that support spending time outside in between classes, sit-stand desks in rooms, active breaks during lessons, and outdoor learning. Sit-stand desk interventions in tertiary institutions are efficacious<sup>391, 392</sup> and, broadly, systematic reviews and a meta-analysis found mixed findings for the efficacy of physical activity interventions in tertiary institutions<sup>393, 394</sup>. Students with minimal or nil on-campus contact hours can receive interventions via electronic-mail, mobile phone and web-based platforms, as these delivery methods were efficacious, feasible and acceptable for tertiary students in physical activity interventions<sup>395-398</sup>. Policies for increasing physical activity and decreasing sedentary behaviour at tertiary institutions could be based on the Okanagan international charter for health-promoting universities<sup>399, 400</sup>. Orientation week is an opportune time to provide information that aims to increase physical activity and tertiary behaviour and campuses typically have recreational facilities on-site available to be utilised<sup>401</sup>.

Current findings suggest that improved active and public transport around campuses are needed, especially if the increase in sedentary transport during the transition out of secondary school was based on private car use. An active transport intervention delivered to tertiary students via a smartphone application, social media and social marketing was efficacious<sup>402</sup>. At the on-campus student residences, different built environment factors may impact physical activity<sup>403</sup> and all have experienced a change in place of residence and do not live with parents. Therefore, tailored information could be delivered to these students with at-risk levels of moderators for those living independently (high co-participation in electronic games, television and DVD viewing with friends or colleagues) that are associated with lower physical activity and higher sedentary behaviour during the transition out of secondary school. For this target group, there has been an efficacious physical activity intervention consisting of peer-delivered information, structured activities, group challenges, a guidebook and behaviour change techniques<sup>404</sup>.

## **Workplaces**

Participants suggested some recommendations for workplaces to increase physical activity and decrease sedentary behaviour. These included employers supporting employees to exercise and participate in fitness classes, creating physical activity challenges between employees, and providing opportunities for employees to stand more during meetings and via sit-stand furniture in offices.

## Family and community settings

Families should encourage adolescent children to move more and sit less since there was much scope to increase physical activity and decrease sedentary behaviour in late adolescence, especially considering the 8% decline in compliance with National Physical Activity and Sedentary Behaviour Guidelines observed during this study. This thesis found suggestive evidence on how to do this, including parents participating in physical activity with adolescent children, discouraging sedentary behaviour, especially excessive prolonged sitting, and providing access to physical activity equipment in the home. Participants recommended that families walk their dog together frequently and create limits for television watching. Future research is needed that identifies efficacious, feasible and scalable strategies that promote parents providing consistent supportive messages to adolescent children in relation to increasing physical activity and decreasing sedentary behaviour. This is especially true as the dependency on parents that is prominent during childhood decreases and partially shifts to peers in adolescence<sup>359</sup>.

Physical activity enjoyment was associated with higher physical activity crosssectionally and longitudinally. To target this influence, families should encourage participation in a range of physical activities during childhood and early adolescence to identity preferred physical activities before late adolescence. The moderation findings suggest that families should create a home environment where norms encourage physical activity and discourage sedentary behaviour. This is to equip adolescents with coping and resilience skills prior to leaving secondary school, leaving home, starting tertiary study and commencing employment to resist the impact of these situational transitions. Despite the importance of parental concern, rules and regulation on adolescents' physical activity and sedentary behaviour<sup>405-408</sup>, the above recommendations for families should be delivered earlier than adolescence-onset and prior to the common decline in physical activity after late

213

childhood<sup>17, 381</sup>. This is because parents may be more receptive to messages when their children are younger<sup>409, 410</sup> and may be able to have more impact on preventing inadequate physical activity and excessive sedentary behaviour.

The aforementioned recommendations should be reflected in parenting advice services delivered in settings such as schools and communities. An efficacious example is a transdisciplinary study that increased Icelandic youth engagement and improved health behaviours<sup>411</sup>. This study delivered specific resources and strategies to parents in a school setting, as well as free community sport memberships, ideas for schools, and formed partnerships between community youth organisations and public health policy-makers within local governments. The study was successful in strengthening the evidence that youth engagement is a protective attribute and mechanism, that may also decrease other harmful health behaviours such as inadequate physical activity and excessive sedentary behaviour.

# 9.5 Conclusion

This thesis contributes to the field of behavioural epidemiology by identifying how physical activity and sedentary behaviour change during the transition out of secondary school and modifiable individual, social and environmental influences underlying these two health behaviours. Physical activity and sedentary behaviour changed in mixed directions in the qualitative study and, in the quantitative study, physical activity declined after Year 11 and sedentary behaviour declined after Year 12; the second last and last year of secondary schooling, respectively. The situational transitions of working  $\geq 20$  h/wk and/or exclusively working post-school resulted in the greatest declines in total sedentary behaviour over time. Correlates and determinants of physical activity and sedentary behaviour were mostly individual attributes, including physical activity goal setting, self-efficacy and enjoyment. In contrast, predominantly social attributes such as sedentary behaviour discouragement and co-participation with family, friends or colleagues moderated associations between situational transitions and physical activity and sedentary behaviour. Effective, feasible and scalable strategies are needed to target the influences found in multiple settings, including secondary schools, tertiary institutions, families and communities. This may insulate older adolescents to the impact of the transition out of secondary school and result in the adoption and maintenance of higher physical activity and lower sedentary behaviour over time which are associated with positive health outcomes in adolescence and later adulthood.

References

- 1. Kraemer HC, Stice E, Kazdin A, Offord D, Kupfer D. How do risk factors work together? Mediators, moderators, and independent, overlapping, and proxy risk factors. American Journal of Psychiatry. 2001;158(6):848-856.
- 2. Bauman AE, Sallis JF, Dzewaltowski DA, Owen N. Toward a better understanding of the influences on physical activity: The role of determinants, correlates, causal variables, mediators, moderators, and confounders. American Journal of Preventive Medicine. 2002;23(2):5-14.
- World Health Organization. Adolescent development. [Online]. Geneva: WHO Press; 2015. Available from: http://www.who.int/maternal\_child\_adolescent/topics/adolescence/dev/en/ [Accessed Mar 19 2015].
- 4. Australian Bureau of Statistics. Education across Australia. Australian Social Trends, 2008. Catalogue no. 4102.0. Canberra; 2008.
- 5. Baron RM, Kenny DA. The moderator–mediator variable distinction in social psychological research: Conceptual, strategic, and statistical considerations. Journal of Personality and Social Psychology. 1986;51(6):1173-1182.
- 6. Caspersen CJ, Powell KE, Christenson GM. Physical activity, exercise, and physical fitness: Definitions and distinctions for health-related research. Public Health Reports. 1985;100(2):126-131.
- Tremblay MS, Aubert S, Barnes JD, Saunders TJ, Carson V, Latimer-Cheung AE, Chastin SF, Altenburg TM, Chinapaw MJ. Sedentary Behavior Research Network (SBRN)–Terminology Consensus Project process and outcome. International Journal of Behavioral Nutrition and Physical Activity. 2017;14(1):75.
- 8. Engberg E, Alen M, Kukkonen-Harjula K, Peltonen JE, Tikkanen HO, Pekkarinen H. Life events and change in leisure time physical activity: A systematic review. Sports Medicine. 2012;42(5):433-447.
- 9. Brown WJ, Trost SG. Life transitions and changing physical activity patterns in young women. American Journal of Preventive Medicine. 2003;25(2):140-143.
- 10. Allender S, Hutchinson L, Foster C. Life-change events and participation in physical activity: A systematic review. Health Promotion International. 2008;23(2):160-172.
- 11. Australian Bureau of Statistics. Australian Social Trends, March Quarter. Catalogue no. 4102.0 Canberra; 2012.
- 12. Lee I-M, Shiroma EJ, Lobelo F, Puska P, Blair SN, Katzmarzyk PT. Effect of physical inactivity on major non-communicable diseases worldwide: An analysis of burden of disease and life expectancy. The Lancet. 2012;380(9838):219-229.
- 13. Homer AR, Owen N, Dunstan DW. Too much sitting and dysglycemia: Mechanistic links and implications for obesity. Current Opinion in Endocrine and Metabolic Research. 2019;4:42-49.
- Xu C, Furuya-Kanamori L, Liu Y, Færch K, Aadahl M, A. Seguin R, LaCroix A, Basterra-Gortari FJ, Dunstan DW, Owen N, Doi SAR. Sedentary behavior, physical activity, and all-cause mortality: Dose-response and intensity weighted time-use meta-analysis. Journal of the American Medical Directors Association. 2019;20(10):1206-1212.e1203.
- 15. Australian Bureau of Statistics. Australian Health Survey: Physical Activity, 2011-12. Catalogue no. 4364.0.55.004. Canberra; 2013.
- Power C, Kuh D, Morton S. From developmental origins of adult disease to life course research on adult disease and aging: Insights from birth cohort studies. Annual Review of Public Health. 2013;34:7-28.

- 17. Dumith SC, Gigante DP, Domingues MR, Kohl III HW. Physical activity change during adolescence: A systematic review and a pooled analysis. International Journal of Epidemiology. 2011;40(3):685-698.
- 18. Kaiser MM, Kaiser KL, Barry TL. Health effects of life transitions for women and children: A research model for public and community health nursing. Public Health Nursing. 2009;26(4):370-379.
- 19. Sallis JF, Owen N, Fotheringham MJ. Behavioral epidemiology: A systematic framework to classify phases of research on health promotion and disease prevention. Annals of Behavioral Medicine. 2000;22(4):294-298.
- Sallis JF, Owen N, Fisher EB. Ecological models of health behavior. In: Glanz K, Rimer BK, Viswanath K, editors. Health behavior and health education: Theory, research, and practice. 4th ed. San Francisco: Jossey-Bass; 2008. p. 465-486.
- 21. Owen N, Sugiyama T, Eakin EE, Gardiner PA, Tremblay MS, Sallis JF. Adults' sedentary behavior: Determinants and interventions. American Journal of Preventive Medicine. 2011;41(2):189-196.
- 22. Simons D, Rosenberg M, Salmon J, Knuiman M, Granich J, Deforche B, Timperio A. Psychosocial moderators of associations between life events and changes in physical activity after leaving high school. Preventive Medicine. 2015;72:30-33.
- 23. Patton GC, Sawyer SM, Santelli JS, Ross DA, Afifi R, Allen NB, Arora M, Azzopardi P, Baldwin W, Bonell C. Our future: A Lancet commission on adolescent health and wellbeing. The Lancet. 2016;387(10036):2423-2478.
- 24. Coeur TD, Rawes C, Warecki P. Challenge and change: Patterns, trends, and shifts in society. Canada: McGraw-Hill Ryerson; 2012.
- 25. Spear LP. The adolescent brain and age-related behavioral manifestations. Neuroscience and Biobehavioral Reviews. 2000;24(4):417-463.
- 26. Videnović M, Pešić J, Plut D. Young people's leisure time: Gender differences. Psihologija. 2010;43(2):199-214.
- 27. Cardoso AR, Fontainha E, Monfardini C. Children's and parents' time use: Empirical evidence on investment in human capital in France, Germany and Italy. Review of Economics of the Household. 2010;8(4):479-504.
- 28. Brown BB. Adolescents' relationships with peers. In: Lerner RM, Steinberg L, editors. Handbook of adolescent psychology. 2nd ed. Hoboken (New Jersey): John Wiley & Sons; 2004. p. 331-361.
- 29. Krizek KJ, Birnbaum AS, Levinson DM. A schematic for focusing on youth in investigations of community design and physical activity. American Journal of Health Promotion. 2004;19(1):33-38.
- Victorian Government. Liquor Control Reform Act 1998 [Internet]. Melbourne: 1998. Available from: http://www.legislation.vic.gov.au/ [Accessed Feb 10 2015].
- 31. Australian Electoral Commission. Voting [Internet]. 2014 [Updated May 30 2013]. Available from: http://www.aec.gov.au/voting/ [Accessed Apr 6 2014].
- 32. Gilson ND, Burton NW, van Uffelen JG, Brown W. Occupational sitting time: Employees' perceptions of health risks and intervention strategies. Health Promotion Journal of Australia. 2011;22(1):38-41.
- 33. Castillo-Retamal M, Hinckson EA. Measuring physical activity and sedentary behaviour at work: A review. Work. 2011;40(4):345-357.
- 34. Thorp AA, Healy GN, Winkler E, Clark BK, Gardiner PA, Owen N, Dunstan DW. Prolonged sedentary time and physical activity in workplace and non-work contexts: A cross-sectional study of office, customer service and call centre

employees. International Journal of Behavioral Nutrition and Physical Activity. 2012;9(1):128.

- 35. Australian Bureau of Statistics. Young people in employment. Year Book Australia, 2005. Catalogue no. 1301.0. Canberra; 2005.
- Fair Work Ombudsman. Young workers [Internet]. Canberra: Australian Government; 2013. Available from: http://www.fairwork.gov.au/employment/young-workers/pages/what-age-can-istart-work [Accessed May 16 2014].
- 37. Australian Bureau of Statistics. Labour Force, Australia, Feb 2014. Catalogue no. 6202.0. Canberra; 2014.
- 38. Blakemore SJ, Burnett S, Dahl RE. The role of puberty in the developing adolescent brain. Human Brain Mapping. 2010;31(6):926-933.
- 39. Blakemore SJ, Robbins TW. Decision-making in the adolescent brain. Nature Neuroscience. 2012;15(9):1184-1191.
- 40. Casey BJ, Getz S, Galvan A. The adolescent brain. Developmental Review. 2008;28(1):62-77.
- 41. Prokhorov AV, Hudmon KS, Stancic N. Adolescent smoking: Epidemiology and approaches for achieving cessation. Paediatric Drugs. 2003;5(1):1-10.
- 42. Shulman EP, Harden KP, Chein JM, Steinberg L. Sex differences in the developmental trajectories of impulse control and sensation-seeking from early adolescence to early adulthood. Journal of Youth and Adolescence. 2014;44(1):1-17.
- 43. Pearson N, Atkin AJ, Biddle SJH, Gorely T, Edwardson C. Patterns of adolescent physical activity and dietary behaviours. International Journal of Behavioral Nutrition and Physical Activity. 2009;6:45.
- 44. Sanchez A, Norman GJ, Sallis JF, Calfas KJ, Cella J, Patrick K. Patterns and correlates of physical activity and nutrition behaviors in adolescents. American Journal of Preventive Medicine. 2007;32(2):124-130.
- 45. Lake AA, Mathers JC, Rugg-Gunn AJ, Adamson AJ. Longitudinal change in food habits between adolescence (11-12 years) and adulthood (32-33 years): The ASH30 study. Journal of Public Health. 2006;28(1):10-16.
- 46. Hamilton MT, Healy GN, Dunstan DW, Zderic TW, Owen N. Too little exercise and too much sitting: Inactivity physiology and the need for new recommendations on sedentary behavior. Current Cardiovascular Risk Reports. 2008;2(4):292-298.
- 47. Centers for Disease Control and Prevention. Physical activity and health: A report of the Surgeon General. Report no. 0763706361. Atlanta (Georgia): United States Department of Health and Human Services; 1996.
- 48. United States Department of Health and Human Services. Physical activity guidelines for Americans. [Online]. United States of America 2008. Available from: http://www.health.gov/paguidelines [Accessed Apr 11 2012].
- 49. Tremblay MS, Colley RC, Saunders TJ, Healy GN, Owen N. Physiological and health implications of a sedentary lifestyle. Applied Physiology, Nutrition, and Metabolism. 2010;35(6):725-740.
- 50. Sedentary Behaviour Research Network. Letter to the editor: Standardized use of the terms "sedentary" and "sedentary behaviours". Applied Physiology, Nutrition, and Metabolism. 2012;37(3):540-542.
- 51. Pratt M, Macera CA, Sallis JF, O'Donnell M, Frank LD. Economic interventions to promote physical activity: Application of the SLOTH model. American Journal of Preventive Medicine. 2004;27(3):136-145.
- 52. Ainsworth BE, Haskell WL, Whitt MC, Irwin ML, Swartz AM, Strath SJ, O'Brien WL, Bassett DR, Jr., Schmitz KH, Emplaincourt PO, Jacobs DR, Jr.,

Leon AS. Compendium of physical activities: An update of activity codes and MET intensities. Medicine and Science in Sports and Exercise. 2000;32(9 Suppl):S498-504.

- 53. Norton K, Norton L, Sadgrove D. Position statement on physical activity and exercise intensity terminology. Journal of Science and Medicine in Sport. 2010;13(5):496-502.
- 54. Craig CL, Marshall AL, Sjorstrom M, Bauman AE, Booth ML, Ainsworth BE, Pratt M, Ekelund U, Yngve A, Sallis JF. International physical activity questionnaire: 12-country reliability and validity. Medicine and Science in Sports and Exercise. 2003;35(8):1381-1395.
- 55. Henderson S, Yeow J, editors. IPad in education: A case study of iPad adoption and use in a primary school. Proceedings of the Annual Hawaii International Conference on System Sciences; 2011.
- 56. Fuller R, Joynes V. Should mobile learning be compulsory for preparing students for learning in the workplace? British Journal of Educational Technology. 2015;46(1):153-158.
- 57. Healy GN, Dunstan DW, Salmon J, Cerin E, Shaw JE, Zimmet PZ, Owen N. Breaks in sedentary time: Beneficial associations with metabolic risk. Diabetes Care. 2008;31(4):661-666.
- Dunstan DW, Kingwell BA, Larsen R, Healy GN, Cerin E, Hamilton MT, Shaw JE, Bertovic DA, Zimmet PZ, Salmon J, Owen N. Breaking up prolonged sitting reduces postprandial glucose and insulin responses. Diabetes Care. 2012;35(5):976-983.
- 59. Dollman J, Okely AD, Hardy L, Timperio A, Salmon J, Hills AP. A hitchhiker's guide to assessing young people's physical activity: Deciding what method to use. Journal of Science and Medicine in Sport. 2009;12(5):518-525.
- 60. Nusser SM, Beyler NK, Welk GJ, Carriquiry AL, Fuller WA, King BMN. Modeling errors in physical activity recall data. Journal of Physical Activity and Health. 2012;9(Supp 1):S56-S67.
- 61. Troiano RP, Pettee GKK, Welk GJ, Owen N, Sternfeld B. Reported physical activity and sedentary behavior: Why do you ask? Journal of Physical Activity and Health. 2012;9(Supp 1):S68-S75.
- 62. Mâsse LC, de Niet JE. Sources of validity evidence needed with self-report measures of physical activity. Journal of Physical Activity and Health. 2012;9(Supp 1):S44-S55.
- 63. Lubans DR, Hesketh K, Cliff DP, Barnett LM, Salmon J, Dollman J, Morgan PJ, Hills AP, Hardy LL. A systematic review of the validity and reliability of sedentary behaviour measures used with children and adolescents. Obesity Reviews. 2011;12(10):781-799.
- 64. Haskell WL. Physical activity by self-report: A brief history and future issues. Journal of Physical Activity and Health. 2012;9(Supp 1):S5-S10.
- 65. Bowles HR. Measurement of active and sedentary behaviors: Closing the gaps in self-report methods. Journal of Physical Activity and Health. 2012;9(Supp 1):S1-S4.
- 66. McCormack G, Giles-Corti B, Milligan R. The test-retest reliability of habitual incidental physical activity. Australian and New Zealand Journal of Public Health. 2003;27(4):428-433.
- 67. Gilson ND, Puig-Ribera A, McKenna J, Brown WJ, Burton NW, Cooke CB. Do walking strategies to increase physical activity reduce reported sitting in workplaces: A randomized control trial. International Journal of Behavioral Nutrition and Physical Activity. 2009;6(1):43.

- 68. Timperio A, Salmon J, Bull F, Rosenberg M. Validation of adult physical activity questions for use in Australian population surveys. Canberra: Commonwealth Department of Health and Ageing; 2002
- 69. Hagstromer M, Bergman P, De Bourdeaudhuij I, Ortega FB, Ruiz JR, Manios Y, Rey-Lopez JP, Phillipp K, von Berlepsch J, Sjostrom M, Group HS. Concurrent validity of a modified version of the International Physical Activity Questionnaire (IPAQ-A) in European adolescents: The HELENA Study. International Journal of Obesity. 2008;32 Suppl 5:S42-S48.
- Celis-Morales CA, Perez-Bravo F, Ibañez L, Salas C, Bailey MES, Gill JMR. Objective vs self-reported physical activity and sedentary time: Effects of measurement method on relationships with risk biomarkers. PLOS ONE. 2012;7(5):e36345.
- 71. Commonwealth Scientific Industrial Research Organisation, University of South Australia. 2007 Australian National Children's Nutrition and Physical Activity survey: Main findings. [Online]. Canberra: Commonwealth of Australia; 2008. Available from:

https://www.health.gov.au/internet/main/publishing.nsf/Content/phd-nutrition-childrens-survey [Accessed Apr 3 2014].

- 72. Lobstein T, Baur L, Uauy R. Obesity in children and young people: A crisis in public health. Obesity Reviews. 2004;5(s1):4-85.
- 73. Gill JM, Cooper AR. Physical activity and prevention of type 2 diabetes mellitus. Sports Medicine. 2008;38(10):807-824.
- 74. Ekelund U, Luan J, Sherar LB, Esliger DW, Griew P, Cooper A. Moderate to vigorous physical activity and sedentary time and cardiometabolic risk factors in children and adolescents. Journal of the American Medical Association. 2012;307(7):704-712.
- 75. Lo KY, Wu MC, Tung SC, Hsieh CC, Yao HH, Ho CC. Association of school environment and after-school physical activity with health-related physical fitness among junior high school students in Taiwan. International Journal of Environmental Research and Public Health. 2017;14(1).
- 76. Martinez-Gomez D, Mielke GI, Menezes AM, Goncalves H, Barros FC, Hallal PC. Active commuting throughout adolescence and central fatness before adulthood: Prospective birth cohort study. PLOS ONE. 2014;9(5).
- 77. Chillón P, Ortega FB, Ruiz JR, Sjöström M, Veidebaum T, Oja L, Mäestu J. Active commuting to school in children and adolescents: An opportunity to increase physical activity and fitness. Scandinavian Journal of Public Health. 2010;38(8):873-879.
- 78. Woods CB. Adolescents who take part in team sports, or who actively commute to school, are less likely to be obese. Evidence-Based Nursing. 2013;16(3):87-88.
- 79. Eime RM, Young JA, Harvey JT, Charity MJ, Payne WR. A systematic review of the psychological and social benefits of participation in sport for children and adolescents: Informing development of a conceptual model of health through sport. International Journal of Behavioral Nutrition and Physical Activity. 2013;10(1):98.
- Baldursdottir B, Valdimarsdottir HB, Krettek A, Gylfason HF, Sigfusdottir ID. Age-related differences in physical activity and depressive symptoms among 10–19-year-old adolescents: A population based study. Psychology of Sport and Exercise. 2017;28:91-99.
- 81. Berenson GS, Srinivasan SR, Bao W, Newman WP, Tracy RE, Wattigney WA. Association between multiple cardiovascular risk factors and atherosclerosis in

children and young adults. New England Journal of Medicine. 1998;338(23):1650-1656.

- 82. Venn AJ, Thomson RJ, Schmidt MD, Cleland VJ, Curry BA, Gennat HC, Dwyer T. Overweight and obesity from childhood to adulthood: A follow-up of participants in the 1985 Australian Schools Health and Fitness Survey. Medical Journal of Australia. 2007;186(9):458-460.
- 83. Juhola J, Magnussen CG, Berenson GS, Venn A, Burns TL, Sabin MA, Srinivasan SR, Daniels SR, Davis PH, Chen W. Combined effects of child and adult elevated blood pressure on subclinical atherosclerosis: The international childhood cardiovascular cohort consortium. Circulation. 2013;128(3):217-224.
- 84. Juonala M, Magnussen CG, Venn A, Dwyer T, Burns TL, Davis PH, Chen W, Srinivasan SR, Daniels SR, Kähönen M. Influence of age on associations between childhood risk factors and carotid intima-media thickness in adulthood: Clinical perspective: The cardiovascular risk in young Finns study, the Childhood Determinants of Adult Health study, the Bogalusa Heart study, and the Muscatine study for the International Childhood Cardiovascular Cohort (i3C) Consortium. Circulation. 2010;122(24):2514-2520.
- 85. Twisk JWR, Kemper HCG, Van Mechelen W. Tracking of activity and fitness and the relationship with cardiovascular disease risk factors. Medicine and Science in Sports and Exercise. 2000;32(8):1455-1461.
- 86. Yang X, Telama R, Leskinen E, Mansikkaniemi K, Viikari J, Raitakari OT. Testing a model of physical activity and obesity tracking from youth to adulthood: The cardiovascular risk in young Finns study. International Journal of Obesity and Related Metabolic Disorders. 2007;31(3):521-527.
- 87. Australian Institute of Health and Welfare. Impact of physical inactivity as a risk factor for chronic conditions: Australian Burden of Disease study. Canberra: Australian Government; 2017.
- Begg S, Vos T, Barker B, Stevenson C, Stanley L, Lopez AD. The burden of disease and injury in Australia 2003. Catalogue no. PHE 82. Canberra: Australian Institute of Health and Welfare; 2007.
- 89. Ding D, Lawson KD, Kolbe-Alexander TL, Finkelstein EA, Katzmarzyk PT, Van Mechelen W, Pratt M, Committee LPASE. The economic burden of physical inactivity: A global analysis of major non-communicable diseases. The Lancet. 2016;388(10051):1311-1324.
- 90. Thorp AA, Kingwell BA, Sethi P, Hammond L, Owen N, Dunstan DW. Alternating bouts of sitting and standing attenuates postprandial glucose responses. Medicine and Science in Sports and Exercise. 2014;46(11):2056-2061.
- 91. Pettapiece-Phillips R, Kotlyar M, Chehade R, Salmena L, Narod SA, Akbari M, Jurisica I, Kotsopoulos J. Uninterrupted sedentary behavior downregulates BRCA1 gene expression. Cancer Prevention Research. 2016;9(1):83-88.
- 92. Vriens A, Provost EB, Saenen ND, De Boever P, Vrijens K, De Wever O, Plusquin M, Nawrot TS. Children's screen time alters the expression of saliva extracellular miR-222 and miR-146a. Scientific Reports. 2018;8(1):8209.
- 93. Schnurr TM, Viitasalo A, Eloranta A-M, Damsgaard CT, Mahendran Y, Have CT, Väistö J, Hjorth MF, Christensen LB, Brage S. Genetic predisposition to adiposity is associated with increased objectively assessed sedentary time in young children. International Journal of Obesity and Related Metabolic Disorders. 2018;42(1):111.
- 94. De Geus EJC, Bartels M, Kaprio J, Lightfoot JT, Thomis M. Genetics of regular exercise and sedentary behaviors. Twin Research and Human Genetics. 2014;17(4):262-271.

- 95. Wang S, Song J, Shang X, Chawla N, Yang Y, Meng X, Wang H, Ma J. Physical activity and sedentary behavior can modulate the effect of the PNPLA3 variant on childhood NAFLD: A case-control study in a Chinese population. BMC Medical Genetics. 2016;17(1):90.
- 96. Diego VP, de Chaves RN, Blangero J, de Souza MC, Santos D, Gomes TN, dos Santos FK, Garganta R, Katzmarzyk PT, Maia JA. Sex-specific genetic effects in physical activity: Results from a quantitative genetic analysis. BMC Medical Genetics. 2015;16(1):58.
- 97. Owen N, Bauman A, Brown WJ. Too much sitting: A novel and important predictor of chronic disease risk? British Journal of Sports Medicine. 2009;43:81-83.
- 98. Dunstan DW, Howard B, Healy GN, Owen N. Too much sitting A health hazard. Diabetes Research and Clinical Practice. 2012;97(3):368-376.
- 99. Martinez-Gómez D, Gomez-Martinez S, Ruiz JR, Ortega FB, Marcos A, Veiga OL. Video game playing time and cardiometabolic risk in adolescents: The AFINOS study. Medicina Clinica. 2012;139(7):290-292.
- 100. Kuzik N, Carson V, Andersen LB, Sardinha LB, Grøntved A, Hansen BH, Ekelund U, Collaborators ICsAD, Atkin A, Ekelund U. Physical activity and sedentary time associations with metabolic health across weight statuses in children and adolescents. Obesity. 2017;25(10):1762-1769.
- 101. Iannotti RJ, Kogan MD, Janssen I, Boyce WF. Patterns of adolescent physical activity, screen-based media use, and positive and negative health indicators in the US and Canada. Journal of Adolescent Health. 2009;44(5):493-499.
- 102. Iannotti RJ, Janssen I, Haug E, Kololo H, Annaheim B, Borraccino A. Interrelationships of adolescent physical activity, screen-based sedentary behaviour, and social and psychological health. International Journal Of Public Health. 2009;54(2):191-198.
- 103. Sharma B, Chavez RC, Jeong AS, Nam EW. Television viewing and its association with sedentary behaviors, self-rated heath and academic performance among secondary school students in Peru. International Journal of Environmental Research and Public Health. 2017;14(4):383.
- 104. Hrafnkelsdottir SM, Brychta RJ, Rognvaldsdottir V, Gestsdottir S, Chen KY, Johannsson E, Guðmundsdottir SL, Arngrimsson SA. Less screen time and more frequent vigorous physical activity is associated with lower risk of reporting negative mental health symptoms among Icelandic adolescents. PLOS ONE. 2018;13(4):e0196286.
- 105. Kemper HCG, Post GB, Twisk JWR, van Mechelen W. Lifestyle and obesity in adolescence and young adulthood: Results from the Amsterdam Growth And Health Longitudinal Study (AGAHLS). International Journal of Obesity. 1999;23:S34-S40.
- 106. Boreham C, Robson P, Gallagher A, Cran G, Savage MJ, Murray L. Tracking of physical activity, fitness, body composition and diet from adolescence to young adulthood: The Young Hearts Project, Northern Ireland. International Journal of Behavioral Nutrition and Physical Activity. 2004;1:14.
- 107. Costigan SA, Barnett L, Plotnikoff RC, Lubans DR. The health indicators associated with screen-based sedentary behavior among adolescent girls: A systematic review. Journal of Adolescent Health. 2013;52(4):382-392.
- 108. Durkin K, Barber B. Not so doomed: Computer game play and positive adolescent development. Journal of Applied Developmental Psychology. 2002;23(4):373-392.
- 109. Commonwealth of Australia. Australian 24-Hour Movement Guidelines for Children and Young People (5-17 years). [Online]. Canberra: Department of

Health; 2019. Available from:

https://www.health.gov.au/internet/main/publishing.nsf/Content/health-24-hours-phys-act-guidelines [Accessed Jul 16 2019].

- 110. World Health Organization. Global recommendations on physical activity for health. [Online]. Geneva: WHO Press; 2010. Available from: http://www.who.int/dietphysicalactivity/factsheet\_recommendations/en/ [Accessed Dec 21 2012].
- 111. Department of Health. Physical activity guidelines: Children and young people (5-18 years). [Online]. United Kingdom 2011. Available from: https://www.gov.uk/government/publications/uk-physical-activity-guidelines [Accessed Apr 12 2011].
- 112. Department of Health. Physical activity guidelines: Adults (19-64 years).
  [Online]. United Kingdom 2011. Available from: https://www.gov.uk/government/publications/uk-physical-activity-guidelines [Accessed Apr 12 2011].
- 113. Tremblay MS, LeBlanc AG, Janssen I, Kho ME, Hicks A, Murumets K, Colley RC, Duggan M. Canadian sedentary behaviour guidelines for children and youth. Applied Physiology, Nutrition, and Metabolism. 2011;36(1):59-64.
- 114. Tremblay MS, Warburton DE, Janssen I, Paterson DH, Latimer AE, Rhodes RE, Kho ME, Hicks A, LeBlanc AG, Zehr L. New Canadian physical activity guidelines. Applied Physiology, Nutrition, and Metabolism. 2011;36(1):36-46.
- 115. Strasburger VC, Hogan MJ, Mulligan DA, Ameenuddin N, Christakis DA, Cross C, Fagbuyi DB, Hill DL, Levine AE, McCarthy C. Children, adolescents, and the media. Pediatrics. 2013;132(5):958-961.
- 116. Mark AE, Boyce WF, Janssen I. Television viewing, computer use and total screen time in Canadian youth. Paediatrics and Child Health. 2006;11(9):595-599.
- 117. Bryan SN, Katzmarzyk PT. Are Canadians meeting the guidelines for moderate and vigorous leisure-time physical activity? Applied Physiology, Nutrition and Metabolism. 2009;34(4):707-715.
- 118. Bray SR, Born HA. Transition to university and vigorous physical activity: Implications for health and psychological well-being. Journal of American College Health. 2004;52(4):181-188.
- 119. Ekelund U, Tomkinson GR, Armstrong N. What proportion of youth are physically active? Measurement issues, levels and recent time trends. British Journal of Sports Medicine. 2011;45(11):859-865.
- 120. Song M, Carroll DD, Fulton JE. Meeting the 2008 Physical Activity Guidelines for Americans among U.S. youth. American Journal of Preventive Medicine. 2013;44(3):216-222.
- 121. Carlson SA, Fulton JE, Schoenborn CA, Loustalot F. Trend and prevalence estimates based on the 2008 Physical Activity Guidelines for Americans. American Journal of Preventive Medicine. 2010;39(4):305-313.
- 122. Sallis JF, Prochaska JJ, Taylor WC. A review of correlates of physical activity of children and adolescents. Medicine and Science in Sports and Exercise. 2000;32(5):963-975.
- 123. Caspersen CJ, Pereira MA, Curran KM. Changes in physical activity patterns in the United States, by sex and cross-sectional age. Medicine and Science in Sports and Exercise. 2000;32(9):1601-1609.
- 124. Ruiz JR, Ortega FB, Martínez-Gómez D, Labayen I, Moreno LA, De Bourdeaudhuij I, Manios Y, Gonzalez-Gross M, Mauro B, Molnar D, Widhalm K, Marcos A, Beghin L, Castillo MJ, Sjöström M. Objectively measured

physical activity and sedentary time in European adolescents: The HELENA study. American Journal of Epidemiology. 2011;174(2):173-184.

- 125. Cluskey M, Grobe D. College weight gain and behavior transitions: Male and female differences. Journal of the American Dietetic Association. 2009;109(2):325-329.
- 126. Health behavior and health education: Theory, research, and practice. 4th ed. San Franciso: Jossey-Bass; 2008.
- 127. Biddle SJ, Pearson N, Ross GM, Braithwaite R. Tracking of sedentary behaviours of young people: A systematic review. Preventive Medicine. 2010;51(5):345-351.
- 128. Telama R. Tracking of physical activity from childhood to adulthood: A review. Obesity Facts. 2009;2(3):187-195.
- 129. Cohen J. Statistical power analysis for the behavioral sciences. 2nd ed. Hillsdale (New Jersey): Lawrence Erlbaum Associates; 1988.
- 130. Hayes G, Dowd KP, MacDonncha C, Donnelly AE. Tracking of physical activity and sedentary behavior from adolescence to young adulthood: A systematic literature review. Journal of Adolescent Health. 2019;65(4):446-454.
- 131. Telama R, Yang X, Viikari J, Välimäki I, Wanne O, Raitakari O. Physical activity from childhood to adulthood: A 21-year tracking study. American Journal of Preventive Medicine. 2005;28(3):267-273.
- 132. Anderssen N, Wold B, Torsheim T. Tracking of physical activity in adolescence. Research Quarterly for Exercise and Sport. 2005;76(2):119-129.
- 133. Kjønniksen L, Torsheim T, Wold B. Tracking of leisure-time physical activity during adolescence and young adulthood: A 10-year longitudinal study. International Journal of Behavioral Nutrition and Physical Activity. 2008;5:69.
- 134. Tammelin T, Näyhä S, Laitinen J, Rintamäki H, Järvelin MR. Physical activity and social status in adolescence as predictors of physical inactivity in adulthood. Preventive Medicine. 2003;37(4):375-381.
- 135. Kwan MY, Cairney J, Faulkner GE, Pullenayegum EE. Physical activity and other health-risk behaviors during the transition into early adulthood: A longitudinal cohort study. American Journal of Preventive Medicine. 2012;42(1):14-20.
- 136. Cleland V, Dwyer T, Venn A. Which domains of childhood physical activity predict physical activity in adulthood? A 20-year prospective tracking study. British Journal of Sports Medicine. 2011;46(8):595-602.
- 137. Telama R, Yang X, Hirvensalo M, Raitakari O. Participation in organized youth sport as a predictor of adult physical activity: A 21-year longitudinal study. Pediatric Exercise Science. 2006;18(1):76-88.
- 138. Ortega FB, Konstabel K, Pasquali E, Ruiz JR, Hurtig-Wennlöf A, Mäestu J, Löf M, Harro J, Bellocco R, Labayen I. Objectively measured physical activity and sedentary time during childhood, adolescence and young adulthood: A cohort study. PLOS ONE. 2013;8(4):e60871.
- 139. Hancox RJ, Milne BJ, Poulton R. Association between child and adolescent television viewing and adult health: A longitudinal birth cohort study. The Lancet. 2004;364(9430):257-262.
- 140. Landhuis CE, Poulton R, Welch D, Hancox RJ. Programming obesity and poor fitness: The long-term impact of childhood television. Obesity. 2008;16(6):1457-1459.
- 141. Gordon-Larsen P, Nelson MC, Popkin BM. Longitudinal physical activity and sedentary behavior trends: Adolescence to adulthood. American Journal of Preventive Medicine. 2004;27(4):277-283.

- 142. Busschaert C, Cardon G, Van Cauwenberg J, Maes L, Van Damme J, Hublet A, De Bourdeaudhuij I, De Cocker K. Tracking and predictors of screen time from early adolescence to early adulthood: A 10-year follow-up study. Journal of Adolescent Health. 2015;56(4):440-448.
- 143. Stokols D, Allen J, Bellingham R. The social ecology of health promotion: Implications for research and practice. American Journal of Health Promotion. 1996;10(4):247-251.
- 144. King AC, Stokols D, Talen E, Brassington GS, Killingsworth R. Theoretical approaches to the promotion of physical activity: Forging a transdisciplinary paradigm. American Journal of Preventive Medicine. 2002;23(2):15-25.
- 145. Dwyer JJM, Allison KR, Goldenberg ER, Fein AJ, et al. Adolescent girls' perceived barriers to participation in physical activity. Adolescence. 2006;41(161):75-89.
- 146. Eime RM, Payne WR, Casey MM, Harvey JT. Transition in participation in sport and unstructured physical activity for rural living adolescent girls. Health Education Research. 2010;25(2):282-293.
- 147. Bauman AE, Reis RS, Sallis JF, Wells JC, Loos RJ, Martin BW. Correlates of physical activity: Why are some people physically active and others not? The Lancet. 2012;380(9838):258-271.
- 148. Pate RR, Mitchell JA, Byun W, Dowda M. Sedentary behaviour in youth. British Journal of Sports Medicine. 2011;45(11):906-913.
- 149. Van der Horst K, Paw M, Twisk JW, Van Mechelen W. A brief review on correlates of physical activity and sedentariness in youth. Medicine and Science in Sports and Exercise. 2007;39(8):1241-1250.
- 150. Juan FR, Bengoechea EG, Montes MEG, Bush PL. Role of individual and school factors in physical activity patterns of secondary-level Spanish students. Journal of School Health. 2010;80(2):88-95.
- 151. De Cocker K, Artero EG, De Henauw S, Dietrich S, Gottrand F, Béghin L, Hagströmer M, Sjöström M, Plada M, Manios Y, Mauro B, Molnár D, Moreno LA, Ottevaere C, Valtueña J, Maes L, De Bourdeaudhuij I. Can differences in physical activity by socio-economic status in European adolescents be explained by differences in psychosocial correlates? A mediation analysis within the HELENA (Healthy Lifestyle in Europe by Nutrition in Adolescence) Study. Public Health Nutrition. 2012;15(11):2100-2109.
- 152. Ding D, Sallis JF, Kerr J, Lee S, Rosenberg DE. Neighborhood environment and physical activity among youth: A review. American Journal of Preventive Medicine. 2011;41(4):442-455.
- 153. Timperio A, Salmon J, Ball K, Te Velde SJ, Brug J, Crawford D. Neighborhood characteristics and TV viewing in youth: Nothing to do but watch TV? Journal of Science and Medicine in Sport. 2012;15(2):122-128.
- 154. Salmon J, Tremblay MS, Marshall SJ, Hume C. Health risks, correlates, and interventions to reduce sedentary behavior in young people. American Journal of Preventive Medicine. 2011;41(2):197-206.
- 155. Davison KK, Lawson CT. Do attributes in the physical environment influence children's physical activity? A review of the literature. International Journal of Behavioral Nutrition and Physical Activity. 2006;3(1):19.
- 156. Casper M. A definition of "social environment". American Journal of Public Health. 2001;91(3):465.
- 157. Loprinzi P, Cardinal B, Loprinzi K, Lee H. Benefits and environmental determinants of physical activity in children and adolescents. Obesity Facts. 2012;5(4):597-610.

- 158. McMillan TE. The relative influence of urban form on a child's travel mode to school. Transportation Research Part A. 2007;41(1):69-79.
- 159. Li K, Liu D, Haynie D, Gee B, Chaurasia A, Seo D-C, Iannotti RJ, Simons-Morton BG. Individual, social, and environmental influences on the transitions in physical activity among emerging adults. BMC Public Health. 2016;16(1):682.
- 160. van Jaarsveld CH, Fidler JA, Simon AE, Wardle J. Persistent impact of pubertal timing on trends in smoking, food choice, activity, and stress in adolescence. Psychosomatic Medicine. 2007;69(8):798-806.
- 161. Luszczynska A, Cao DS, Mallach N, Pietron K, Mazurkiewicz M, Schwarzer R. Intentions, planning, and self-efficacy predict physical activity in Chinese and Polish adolescents: Two moderated mediation analyses. International Journal of Clinical and Health Psychology. 2010;10(2):265-278.
- 162. Madsen KA, McCulloch CE, Crawford PB. Parent modeling: Perceptions of parents' physical activity predict girls' activity throughout adolescence. Journal of Pediatrics. 2009;154(2):278-283.
- 163. Bauer KW, Nelson MC, Boutelle KN, Neumark-Sztainer D. Parental influences on adolescents' physical activity and sedentary behavior: Longitudinal findings from Project EAT-II. International Journal of Behavioral Nutrition and Physical Activity. 2008;5:12.
- 164. Boone-Heinonen J, Guilkey DK, Evenson KR, Gordon-Larsen P. Residential self-selection bias in the estimation of built environment effects on physical activity between adolescence and young adulthood. Internation Journal of Behavioral Nutrition and Physical Activity. 2010;7:70.
- 165. Velde SJ, Horst K, Oenema A, Timperio A, Crawford D, Brug J. Parental and home influences on adolescents' TV viewing: A mediation analysis. International Journal of Pediatric Obesity. 2011;6:e364-e372.
- 166. Owen N, Salmon J, Koohsari MJ, Turrell G, Giles-Corti B. Sedentary behaviour and health: Mapping environmental and social contexts to underpin chronic disease prevention. British Journal of Sports Medicine. 2014;48(3):174-177.
- 167. Dietrich J, Parker P, Salmela-Aro K. Phase-adequate engagement at the postschool transition. Developmental Psychology. 2012;48(6):1575-1593.
- 168. George LK. Sociological perspectives on life transitions. Annual Review of Sociology. 1993;19(1):353-373.
- 169. Nelson MC, Story M, Larson NI, Neumark-Sztainer D, Lytle LA. Emerging adulthood and college-aged youth: An overlooked age for weight-related behavior change. Obesity. 2008;16(10):2205-2211.
- 170. Edmonds MJ, Ferreira KJ, Nikiforuk EA, Finnie AK, Leavey SH, Duncan AM, Randall Simpson JA. Body weight and percent body fat increase during the transition from high school to university in females. Journal of the American Dietetic Association. 2008;108(6):1033-1037.
- 171. Pullman AW, Masters RC, Zalot LC, Carde LE, Saraiva MM, Dam YY, Randall Simpson JA, Duncan AM. Effect of the transition from high school to university on anthropometric and lifestyle variables in males. Applied Physiology, Nutrition and Metabolism. 2009;34(2):162-171.
- 172. Racette SB, Deusinger SS, Strube MJ, Highstein GR, Deusinger RH. Weight changes, exercise, and dietary patterns during freshman and sophomore years of college. Journal of American College Health. 2005;53(6):245-251.
- 173. Butler SM, Black DR, Blue CL, Gretebeck RJ. Change in diet, physical activity, and body weight in female college freshman. American Journal of Health Behavior. 2004;28(1):24-32.

- 174. Ball K, Crawford D. Socio-economic factors in obesity: A case of slim chance in a fat world? Asia Pacific Journal of Clinical Nutrition. 2006;15(SUPPL. 1):15-20.
- 175. Luthar SS, Cicchetti D, Becker B. The construct of resilience: A critical evaluation and guidelines for future work. Child Development. 2000;71(3):543-562.
- 176. Michie S, Ashford S, Sniehotta FF, Dombrowski SU, Bishop A, French DP. A refined taxonomy of behaviour change techniques to help people change their physical activity and healthy eating behaviours: The CALO-RE taxonomy. Psychology and Health. 2011;26(11):1479-1498.
- 177. Harrington NG, Noar SM. Reporting standards for studies of tailored interventions. Health Education Research. 2011;27(2):331-342.
- 178. Noar SM, Grant Harrington N, Van Stee SK, Shemanski Aldrich R. Tailored health communication to change lifestyle behaviors. American Journal of Lifestyle Medicine. 2011;5(2):112-122.
- 179. Keller C, Fleury J, Castro FG, Ainsworth B, Perez A. Moderators of physical activity in Hispanic women. Hispanic Health Care International. 2009;7(2):60-71.
- 180. van Bree RJH, van Stralen MM, Bolman C, Mudde AN, de Vries H, Lechner L. Habit as moderator of the intention-physical activity relationship in older adults: A longitudinal study. Psychology and Health. 2013;28(5):514-532.
- 181. St. George SM, Wilson DK, Lawman HG, Van Horn ML. Weight status as a moderator of the relationship between motivation, emotional social support, and physical activity in underserved adolescents. Journal of Pediatric Psychology. 2013;38(4):387-397.
- 182. Van Cauwenberg J, Cerin E, Timperio A, Salmon J, Deforche B, Veitch J. Park proximity, quality and recreational physical activity among mid-older aged adults: Moderating effects of individual factors and area of residence. International Journal of Behavioral Nutrition and Physical Activity. 2015;12(1):46.
- 183. St. George SM, Wilson DK, Schneider EM, Alia KA. Project SHINE: Effects of parent-adolescent communication on sedentary behavior in African American adolescents. Journal of Pediatric Psychology. 2013;38(9):997-1009.
- 184. Salmon J, Veitch J, Abbott G, ChinApaw M, Brug JJ, teVelde SJ, Cleland V, Hume C, Crawford D, Ball K. Are associations between the perceived home and neighbourhood environment and children's physical activity and sedentary behaviour moderated by urban/rural location? Health and Place. 2013;24:44-53.
- 185. Tram JM, Cole DA. Self-perceived competence and the relation between life events and depressive symptoms in adolescence: Mediator or moderator? Journal of Abnormal Psychology. 2000;109(4):753.
- 186. Campbell PT, Katzmarzyk PT, Malina RM, Rao DC, Pérusse L, Bouchard C. Prediction of physical activity and physical work capacity (PWC150) in young adulthood from childhood and adolescence with consideration of parental measures. American Journal of Human Biology. 2001;13(2):190-196.
- 187. Van Dyck D, De Bourdeaudhuij I, Deliens T, Deforche B. Can changes in psychosocial factors and residency explain the decrease in physical activity during the transition from high school to college or university? International Journal of Behavioral Medicine. 2015;22(2):178-186.
- 188. Popham F, Mitchell R. Leisure time exercise and personal circumstances in the working age population: Longitudinal analysis of the British household panel survey. Journal of Epidemiology and Community Health. 2006;60(3):270-274.

- 189. Deforche B, Van Dyck D, Deliens T, De Bourdeaudhuij I. Changes in weight, physical activity, sedentary behaviour and dietary intake during the transition to higher education: A prospective study. International Journal of Behavioral Nutrition and Physical Activity. 2015;12(1):16.
- 190. Nelson MC, Kocos R, Lytle LA, Perry CL. Understanding the perceived determinants of weight-related behaviors in late adolescence: A qualitative analysis among college youth. Journal of Nutrition Education and Behavior. 2009;41(4):287-292.
- 191. Deliens T, Deforche B, De Bourdeaudhuij I, Clarys P. Determinants of physical activity and sedentary behaviour in university students: A qualitative study using focus group discussions. BMC Public Health. 2015;15(1):201.
- 192. Koehn S, Gillison F, Standage M, Bailey J. Life transitions and relevance of healthy living in late adolescence. Journal of Health Psychology. 2016;21(6):1085-1095.
- 193. Simons D, Clarys P, De Bourdeaudhuij I, De Geus B, Vandelanotte C, Deforche B. Factors influencing mode of transport in older adolescents: A qualitative study. BMC Public Health. 2013;13(1):323.
- 194. Azar D, Ball K, Salmon J, Cleland VJ. Physical activity correlates in young women with depressive symptoms: A qualitative study. International Journal of Behavioral Nutrition and Physical Activity. 2010;7:3.
- 195. Tong A, Sainsbury P, Craig J. Consolidated criteria for reporting qualitative research (COREQ): A 32-item checklist for interviews and focus groups. International Journal for Quality in Health Care. 2007;19(6):349-357.
- 196. Morgan DL. Focus groups. Annual Review of Sociology. 1996;22(1):129-152.
- 197. Turner DW. Qualitative interview design: A practical guide for novice investigators. The Qualitative Report. 2010;15(3):754-760.
- 198. Strauss A, Corbin JM. Basics of qualitative research: Grounded theory procedures and techniques. Thousand Oaks C, USA, editor: Sage Publications, Inc; 1990.
- 199. Fenner Y, Garland SM, Moore EE, Jayasinghe Y, Fletcher A, Tabrizi SN, Gunasekaran B, Wark JD. Web-based recruiting for health research using a social networking site: An exploratory study. Journal of Medical Internet Research. 2012;14(1):e20.
- 200. Hansen EC. Successful qualitative health research: A practical introduction. Crows Nest, NSW: Allen & Unwin; 2006.
- 201. Punch KF. Introduction to social research: Quantitative and qualitative approaches. 3rd ed. London: Sage publications; 2014.
- 202. Handbook of research methods for nursing and health science. Frenchs Forest, NSW: Pearson Education Australia; 2003.
- 203. Huberman MA, Miles MB. The qualitative researcher's companion. Thousand Oaks, CA, USA: Sage publications Inc; 2002. 410 p.
- 204. Kisely S, Kendall E. Critically appraising qualitative research: A guide for clinicians more familiar with quantitative techniques. Australasian Psychiatry. 2011;19(4):364-367.
- 205. McKercher C, Schmidt M, Sanderson K, Patton G, Dwyer T, Venn A. Physical activity and depression in young adults. American Journal of Preventive Medicine. 2009;36(2):161-164.
- 206. Grieve FG. A conceptual model of factors contributing to the development of muscle dysmorphia. Eating Disorders. 2007;15(1):63-80.
- 207. Rowland TW. The biological basis of physical activity. Medicine and Science in Sports and Exercise. 1998;30(3):392-399.

- 208. Ridgers ND, Barnett LM, Lubans DR, Timperio A, Cerin E, Salmon J. Potential moderators of day-to-day variability in children's physical activity patterns. Journal of Sports Sciences. 2017;36(6):637-644.
- 209. Commonwealth of Australia. Australia's Physical Activity and Sedentary Behaviour Guidelines: 13-17 years. [Online]. Canberra: Department of Health; 2014. Available from: http://www.health.gov.au/internet/main/publishing.nsf/content/health-publithstrateg-phys-act-guidelines [Accessed Feb 13 2014].
- 210. Polit DF, Beck CT. Generalization in quantitative and qualitative research: Myths and strategies. International Journal of Nursing Studies. 2010;47(11):1451-1458.
- 211. Australian Bureau of Statistics. Education and Work, Australia, May 2016. Catalogue no. 6227.0. Canberra; 2016.
- 212. Steckler A, McLeroy KR, Goodman RM, Bird ST, McCormick L. Toward integrating qualitative and quantitative methods: An introduction. Health Education Quarterly. 1992;19(1):1-8.
- 213. Jose KA, Blizzard L, Dwyer T, McKercher C, Venn AJ. Childhood and adolescent predictors of leisure time physical activity during the transition from adolescence to adulthood: A population based cohort study. International Journal of Behavioral Nutrition and Physical Activity. 2011;8:54.
- 214. Trudeau F, Laurencelle L, Shephard RJ. Tracking of physical activity from childhood to adulthood. Medicine and Science in Sports and Exercise. 2004;36(11):1937-1943.
- 215. Australian Bureau of Statistics. Australian Statistical Geography Standard (ASGS): Volume 5 - Remoteness Structure, July 2011. Catalogue no. 1270.0.55.005. Canberra 2011.
- 216. Australian Bureau of Statistics. Census of Population and Housing: Socio-Economic Indexes for Areas (SEIFA), Australia, 2011. Catalogue no. 2033.0.55.001. Canberra; 2011.
- 217. Ottevaere C, Huybrechts I, De Bourdeaudhuij I, Sjöström M, Ruiz JR, Ortega FB, Hagströmer M, Widhalm K, Molnár D, Moreno LA. Comparison of the IPAQ-A and actigraph in relation to VO<sub>2</sub>max among European adolescents: The HELENA study. Journal of Science and Medicine in Sport. 2011;14(4):317-324.
- 218. Yore MM, Ham SA, Ainsworth BE, Macera CA, Jones DA, Kohl HW. Occupational physical activity: Reliability and comparison of activity levels. Journal of Physical Activity and Health. 2005;2(3):358-365.
- 219. Hallal PC, Gómez LF, Parra DC, Lobelo F, Mosquera J, Florindo A, Reis R, Pratt M, Sarmiento O. Lessons learned after 10 Years of IPAQ use in Brazil and Colombia. Journal of Physsical Activity and Health. 2010;7(Suppl 2):259-264.
- 220. Rosenberg DE, Bull FC, Marshall AL, Sallis JF, Bauman AE. Assessment of sedentary behavior with the International Physical Activity Questionnaire. Journal of Physical Activity and Health. 2008;5(SUPPL. 1):S30-S44.
- 221. Salmon J, Owen N, Crawford D, Bauman A, Sallis JF. Physical activity and sedentary behavior: A population-based study of barriers, enjoyment, and preference. Health Psychology. 2003;22(2):178-188.
- 222. Melkevik O, Torsheim T, Iannotti RJ, Wold B. Is spending time in screen-based sedentary behaviors associated with less physical activity: A cross national investigation. International Journal of Behavioral Nutrition and Physical Activity. 2010;7(1):46.
- 223. Healy GN, Wijndaele K, Dunstan DW, Shaw JE, Salmon J, Zimmet PZ, Owen N. Objectively measured sedentary time, physical activity, and metabolic risk:
The Australian Diabetes, Obesity and Lifestyle Study (AusDiab). Diabetes Care. 2008;31(2):369-371.

- 224. Tanaka C, Tanaka M, Okuda M, Inoue S, Aoyama T, Tanaka S. Association between objectively evaluated physical activity and sedentary behavior and screen time in primary school children. BMC Research Notes. 2017;10:175.
- 225. Norman GJ, Sallis JF, Gaskins R. Comparability and reliability of paper-and computer-based measures of psychosocial constructs for adolescent physical activity and sedentary behaviors. Research Quarterly for Exercise and Sport. 2005;76(3):315-323.
- 226. Giles-Corti B, Donovan RJ. The relative influence of individual, social and physical environment determinants of physical activity. Social Science and Medicine. 2002;54(12):1793-1812.
- 227. Marcus BH, Selby VC, Niaura RS, Rossi JS. Self-efficacy and the stages of exercise behavior change. Research Quarterly for Exercise and Sport. 1992;63(1):60-66.
- 228. Taylor WC, Sallis JF, Dowda M, Freedson PS, Eason K, Pate RR. Activity patterns and correlates among youth: Differences by weight status. Journal of Pediatric Exercise Science. 2002;14(4):418-431.
- 229. Ball K, Jeffery RW, Abbott G, McNaughton SA, Crawford D. Is healthy behavior contagious: Associations of social norms with physical activity and healthy eating. International Journal of Behavioral Nutrition and Physical Activity. 2010;7(1):86.
- 230. Sallis JF, Grossman RM, Pinski RB, Patterson TL, Nader PR. The development of scales to measure social support for diet and exercise behaviors. Preventive Medicine. 1987;16(6):825-836.
- 231. Rosenberg DE, Sallis JF, Kerr J, Maher J, Norman GJ, Durant N, Harris SK, Saelens BE. Brief scales to assess physical activity and sedentary equipment in the home. International Journal of Behavioral Nutrition and Physical Activity. 2010;7(1):10.
- 232. Telford A, Salmon J, Jolley D, Crawford D. Reliability and validity of physical activity questionnaires for children: The Children's Leisure Activities Study Survey (CLASS). Pediatric Exercise Science. 2004;16(1):64-78.
- 233. Mujahid MS, Roux AVD, Morenoff JD, Raghunathan T. Assessing the measurement properties of neighborhood scales: From psychometrics to ecometrics. American Journal of Epidemiology. 2007;165(8):858-867.
- 234. Rosenberg DE, Ding D, Sallis JF, Kerr J, Norman GJ, Durant N, Harris SK, Saelens BE. Neighborhood Environment Walkability Scale for Youth (NEWS-Y): Reliability and relationship with physical activity. Preventive Medicine. 2009;49:213–218.
- 235. Landis JR, Koch GG. The measurement of observer agreement for categorical data. Biometrics. 1977;33(1):159-174.
- 236. Cronbach LJ. Coefficient alpha and the internal structure of tests. Psychometrika. 1951;16(3):297-334.
- 237. Tavakol M, Dennick R. Making sense of Cronbach's alpha. International Journal of Medical Education. 2011;2:53-55.
- 238. Zach H. Psychometrics properties of the questionnaire surveys: Validity, reliability and feasibility. Sociologia. 2005;37(3):275-291.
- 239. Blischke WR, Murthy DP. Reliability: Modeling, prediction, and optimization. New York: John Wiley & Sons; 2011.
- 240. Shrout PE, Fleiss JL. Intraclass correlations: Uses in assessing rater reliability. Psychological Bulletin. 1979;86(2):420-428.

- 241. Kendzierski D, DeCarlo KJ. Physical activity enjoyment scale: Two validation studies. Journal of Sport and Exercise Psychology. 1991;13(1):50-64.
- 242. Carson V, Hunter S, Kuzik N, Gray CE, Poitras VJ, Chaput J-P, Saunders TJ, Katzmarzyk PT, Okely AD, Connor Gorber S. Systematic review of sedentary behaviour and health indicators in school-aged children and youth: An update. Applied Physiology, Nutrition, and Metabolism. 2016;41(6):S240-S265.
- 243. Hidding LM, Altenburg TM, Mokkink LB, Terwee CB, Chinapaw MJ. Systematic review of childhood sedentary behavior questionnaires: What do we know and what is next? Sports Medicine. 2017;47(4):677-699.
- 244. Walter S, Eliasziw M, Donner A. Sample size and optimal designs for reliability studies. Statistics in Medicine. 1998;17(1):101-110.
- 245. McHugh ML. Interrater reliability: The kappa statistic. Biochemia Medica. 2012;22(3):276-282.
- 246. Fleiss JL, Cohen J. The equivalence of weighted kappa and the intraclass correlation coefficient as measures of reliability. Educational and Psychological Measurement. 1973;33(3):613-619.
- 247. Streiner DL, Norman GR. Health measurement scales: A practical guide to their development and use. 4th ed. New York: Oxford University Press; 2008.
- 248. Atkinson G, Nevill AM. Statistical methods for assessing measurement error (reliability) in variables relevant to sports medicine. Sports Medicine. 1998;26(4):217-238.
- 249. Goldscheider FK, Thornton A, Yang LS. Helping out the kids: Expectations about parental support in young adulthood. Journal of Marriage and Family. 2001;63(3):727-740.
- 250. Johnson MK. Parental financial assistance and young adults' relationships with parents and well-being. Journal of Marriage and Family. 2013;75(3):713-733.
- 251. Ryan T, Xenos S. Who uses Facebook? An investigation into the relationship between the Big Five, shyness, narcissism, loneliness, and Facebook usage. Computers in Human Behavior. 2011;27(5):1658-1664.
- 252. Mellon J, Prosser C. Twitter and Facebook are not representative of the general population: Political attitudes and demographics of British social media users. Research and Politics. 2017;4(3):1-9.
- 253. Fleming J, Kamal A, Harrison E, Hamborg T, Stewart-Brown S, Thorogood M, Griffiths F, Robertson W. Evaluation of recruitment methods for a trial targeting childhood obesity: Families for Health randomised controlled trial. Trials. 2015;16(1):535.
- 254. Reaves AC, Bianchi DW. The role of social networking sites in medical genetics research. American Journal of Medical Genetics, Part A. 2013;161(5):951-957.
- 255. Da Silva TLN, Klein CH, De Moura Souza A, Barufaldi LA, De Azevedo Abreu G, Kuschnir MCC, de Vasconcellos MTL, Bloch KV. Response rate in the study of cardiovascular risks in adolescents ERICA. Revista de Saude Publica. 2016;50:1s-13s.
- 256. Hunt E, McKay EA, Dahly DL, Fitzgerald AP, Perry IJ. A person-centred analysis of the time-use, daily activities and health-related quality of life of Irish school-going late adolescents. Quality of Life Research. 2015;24(6):1303-1315.
- 257. Sharma B, Nam EW, Kim D, Yoon YM, Kim Y, Kim HY. Role of gender, family, lifestyle and psychological factors in self-rated health among urban adolescents in Peru: A school-based cross-sectional survey. BMJ Open. 2016;6(2).
- 258. Kelishadi R, Ardalan G, Qorbani M, Ataie-Jafari A, Bahreynian M, Taslimi M, Motlagh ME, Heshmat R. Methodology and early findings of the fourth survey of childhood and adolescence surveillance and prevention of adult non-

communicable disease in Iran: The CASPIAN-IV study. International Journal of Preventive Medicine. 2013;4(12):1451-1460.

- 259. Jekauc D, Reimers AK, Wagner MO, Woll A. Physical activity in sports clubs of children and adolescents in Germany: Results from a nationwide representative survey. Journal of Public Health (Germany). 2013;21(6):505-513.
- 260. Peterson MS, Lawman HG, Wilson DK, Fairchild A, Van Horn ML. The association of self-efficacy and parent social support on physical activity in male and female adolescents. Health Psychology. 2013;32(6):666-674.
- 261. Raustorp A, Lindwall M. Physical self-esteem-a ten-year follow-up study from early adolescence to early adulthood. International Journal of Adolescent Medicine and Health. 2015;27(1):31-39.
- 262. Sallis JF, Saelens BE. Assessment of physical activity by self-report: Status, limitations, and future directions. Research Quarterly for Exercise and Sport. 2000;71(Sup2):1-14.
- 263. Australian Bureau of Statistics. Australian Demographic Statistics, Mar 2019. Catalogue no. 3101.0. Canberra; 2019.
- 264. Australian Bureau of Statistics. Australian Historical Population Statistics, 2016. Catalogue no. 3105.0.65.001. Canberra; 2016.
- 265. Ryan T, Chester A, Reece J, Xenos S. The uses and abuses of facebook: A review of facebook addiction. Journal of Behavioral Addictions. 2014;3(3):133-148.
- 266. Mehdizadeh S. Self-presentation 2.0: Narcissism and self-esteem on facebook. Cyberpsychology, Behavior, and Social Networking. 2010;13(4):357-364.
- 267. Koc M, Gulyagci S. Facebook addiction among Turkish college students: The role of psychological health, demographic, and usage characteristics. Cyberpsychology, Behavior, and Social Networking. 2013;16(4):279-284.
- 268. Zhang X, Kuchinke L, Woud ML, Velten J, Margraf J. Survey method matters: Online/offline questionnaires and face-to-face or telephone interviews differ. Computers in Human Behavior. 2017;71:172-180.
- 269. Choi CS, Kim S. When smartphone responses are not unreliable. Communication Research Reports. 2017:307-315.
- 270. Martínez-Gómez D, Eisenmann JC, Gómez-Martínez S, Veses A, Marcos A, Veiga OL. Sedentary behavior, adiposity, and cardiovascular risk factors in adolescents. The AFINOS Study. Revista Española de Cardiología (English Edition). 2010;63(3):277-285.
- 271. Marshall SJ, Biddle SJ, Gorely T, Cameron N, Murdey I. Relationships between media use, body fatness and physical activity in children and youth: A metaanalysis. International Journal of Obesity. 2004;28(10):1238-1246.
- 272. Biddle SJH, Asare M. Physical activity and mental health in children and adolescents: A review of reviews. British Journal of Sports Medicine. 2011;45(11):886-895.
- 273. Rivière F, Aubert S, Omorou AY, Ainsworth BE, Vuillemin A. Taxonomybased content analysis of sedentary behavior questionnaires: A systematic review. PLOS ONE. 2018;13(3):e0193812.
- 274. Bellettiere J, Winkler EAH, Chastin SFM, Kerr J, Owen N, Dunstan DW, Healy GN. Associations of sitting accumulation patterns with cardio-metabolic risk biomarkers in Australian adults. PLOS ONE. 2017;12(6):e0180119.
- 275. Shephard RJ. Limits to the measurement of habitual physical activity by questionnaires. British Journal of Sports Medicine. 2003;37(3):197-206.
- 276. Busschaert C, De Bourdeaudhuij I, Van Holle V, Chastin SFM, Cardon G, De Cocker K. Reliability and validity of three questionnaires measuring contextspecific sedentary behaviour and associated correlates in adolescents, adults and

older adults. International Journal of Behavioral Nutrition and Physical Activity. 2015;12(1):117.

- 277. Amagasa S, Machida M, Fukushima N, Kikuchi H, Takamiya T, Odagiri Y, Inoue S. Is objectively measured light-intensity physical activity associated with health outcomes after adjustment for moderate-to-vigorous physical activity in adults? A systematic review. International Journal of Behavioral Nutrition and Physical Activity. 2018;15(1):65.
- 278. Füzéki E, Engeroff T, Banzer W. Health benefits of light-intensity physical activity: A Systematic review of accelerometer data of the National Health and Nutrition Examination Survey (NHANES). Sports Medicine. 2017;47(9):1769-1793.
- 279. Arundell L, Salmon J, Koorts H, Contardo Ayala AM, Timperio A. Exploring when and how adolescents sit: Cross-sectional analysis of activPAL-measured patterns of daily sitting time, bouts and breaks. BMC Public Health. 2019;19(1):653.
- 280. Rhodes RE, Mark RS, Temmel CP. Adult sedentary behavior: A systematic review. American Journal of Preventive Medicine. 2012;42(3):e3-e28.
- 281. Leppink J, Pérez-Fuster P. We need more replication research A case for testretest reliability. Perspectives on Medical Education. 2017;6(3):158-164.
- 282. Ball K, Jeffery, R.W., Crawford, D.A., Roberts, R.J., Salmon, J., Timperio, A.F. Mismatch between perceived and objective measures of physical activity environments. Preventive Medicine. 2008;47(3):294-298.
- 283. Timperio A, Jeffery RW, Crawford D, Roberts R, Giles-Corti B, Ball K. Neighbourhood physical activity environments and adiposity in children and mothers: A three-year longitudinal study. International Journal of Behavioral Nutrition and Physical Activity. 2010;7(1):18.
- 284. Timperio A, Ball K, Salmon J, Roberts R, Crawford D. Is availability of public open space equitable across areas? Health and Place. 2007;13(2):335-340.
- 285. Trizano-Hermosilla I, Alvarado JM. Best alternatives to Cronbach's Alpha reliability in realistic conditions: Congeneric and asymmetrical measurements. Frontiers in Psychology. 2016;7:769.
- 286. Okely AD, Salmon J, Vella S, Cliff D, Timperio A, Tremblay M, Trost S, Shilton T, Hinkley T, Ridgers N. A systematic review to update the Australian physical activity guidelines for children and young people. Canberra, Australia: Commonwealth of Australia; 2012.
- 287. Tabachnick BG, Fidell LS. Using multivariate statistics. 6th ed. Boston: Pearson; 2013.
- 288. UCLA: Statistical Consulting Group. Simple and Multiple Regression: 1.5 Transforming Variables. 2016. In: Regression with Stata [Internet]. Los Angeles: Institute for digital research and education, University of California. Available from: http://www.ats.ucla.edu/stat/stata/webbooks/reg/chapter1/statareg1.htm [cited
- January 24, 2015]. 289. Myers RH. Classical and modern regression with applications. 2nd ed. Boston, MA: Duxbury; 1990.
- 290. Rao P. Applied survival analysis: Regression modeling of time to event data. Journal of the American Statistical Association. 2000;95(450):681-681.
- 291. Greenland S, Pearce N. Statistical foundations for model-based adjustments. Annual Review of Public Health. 2015;36:89-108.
- 292. Montgomery DC, Peck EA, Vining GG. Introduction to linear regression analysis. Hoboken, New Jersey, United States of America: John Wiley & Sons; 2012.

- 293. Ridgers ND, Timperio A, Crawford D, Salmon J. What factors are associated with adolescents' school break time physical activity and sedentary time? PLOS ONE. 2013;8(2):e56838.
- 294. da Silva J, Andrade A, Capistrano R, Lisboa T, Andrade RD, Felden ÉPG, Beltrame TS. Insufficient levels of physical activity of adolescents associated with sociodemographic, environmental and school factors. Ciencia e Saude Coletiva. 2018;23(12):4277-4288.
- 295. Moral García JE, Agraso López AD, Pérez Soto JJ, Rosa Guillamón A, Tárraga Marcos ML, García Cantó E, Tárraga López PJ. Physical activity practice according to adherence to the Mediterranean diet, alcohol consumption and motivation in adolescents. Nutricion Hospitalaria. 2019;36(2):420-427.
- 296. Frömel K, Groffik D, Chmelík F, Cocca A, Skalik K. Physical activity of 15–17 years old adolescents in different educational settings: A Polish-Czech study. Central European Journal of Public Health. 2018;26(2):137-143.
- 297. Junger J, Salonna F, Bergier J, Junger A, Frömel K, Ács P, Bergier B. Physical activity and body-mass-index relation in secondary-school students of the Visegrad region. Journal of Physical Education and Sport. 2019;19:235-241.
- 298. Mitáš J, Sas-Nowosielski K, Groffik D, Frömel K. The safety of the neighborhood environment and physical activity in Czech and Polish adolescents. International Journal of Environmental Research and Public Health. 2018;15(1):126.
- 299. Morales Quispe J, Añez Ramos RJ, Suarez Oré CA. Level of physical activity among adolescents in a district of the Callao region. Revista Peruana de Medicina Experimental y Salud Publica. 2016;33(3):471-477.
- 300. Australian Bureau of Statistics. Australian Health Survey: Users' Guide, 2011-13. Catalogue no. 4363.0.55.001. Canberra; 2013.
- 301. Ridgers ND, Timperio A, Crawford D, Salmon J. Validity of a brief self-report instrument for assessing compliance with physical activity guidelines amongst adolescents. Journal of Science and Medicine in Sport. 2012;15(2):136-141.
- 302. Troiano RP, Berrigan D, Dodd KW, Masse LC, Tilert T, McDowell M. Physical activity in the United States measured by accelerometer. Medicine Science in Sports and Exercise. 2008;40(1):181-188.
- 303. Hallal PC, Andersen LB, Bull FC, Guthold R, Haskell W, Ekelund U, Group LPASW. Global physical activity levels: Surveillance progress, pitfalls, and prospects. The Lancet. 2012;380(9838):247-257.
- 304. Monge-Rojas R, Fuster-Baraona T, Garita-Arce C, Sánchez-López M, Colon-Ramos U, Smith-Castro V. How self-objectification impacts physical activity among adolescent girls in Costa Rica. Journal of Physical Activity and Health. 2017;14(2):123-129.
- 305. Spencer RA, Rehman L, Kirk SF. Understanding gender norms, nutrition, and physical activity in adolescent girls: A scoping review. International Journal of Behavioral Nutrition and Physical Activity. 2015;12(1):6.
- 306. Barnett LM, van Beurden E, Morgan PJ, Brooks LO, Beard JR. Gender differences in motor skill proficiency from childhood to adolescence: A longitudinal study. Research Quarterly for Exercise and Sport. 2010;81(2):162-170.
- 307. Eather N, Bull A, Young MD, Barnes AT, Pollock ER, Morgan PJ. Fundamental movement skills: Where do girls fall short? A novel investigation of objectcontrol skill execution in primary-school aged girls. Preventive Medicine Reports. 2018;11:191-195.

- 308. Deaner RO, Balish SM, Lombardo MP. Sex differences in sports interest and motivation: An evolutionary perspective. Evolutionary Behavioral Sciences. 2016;10(2):73-97.
- 309. Ferrar KE, Olds TS, Walters JL. All the stereotypes confirmed: Differences in how australian boys and girls use their time. Health Education and Behavior. 2012;39(5):589-595.
- 310. Sallis JF, Zakarian JM, Hovell MF, Hofstetter CR. Ethnic, socioeconomic, and sex differences in physical activity among adolescents. Journal of Clinical Epidemiology. 1996;49(2):125-134.
- 311. Martins J, Marques A, Peralta M, Palmeira A, Da Costa FC. Correlates of physical activity in young people. Challenges-New Trends in Physical Education Sport and Recreation. 2017(31):292-299.
- 312. Lisinskiene A, Juskeliene V. Links between adolescents' engagement in physical activity and their attachment to mothers, fathers, and peers. International Journal of Environmental Research and Public Health. 2019;16:866.
- 313. Parker KE, Salmon J, Brown HL, Villanueva K, Timperio A. Typologies of adolescent activity related health behaviours. Journal of Science and Medicine in Sport. 2019;22(3):319-323.
- 314. Duckworth A, Gross JJ. Self-control and grit: Related but separable determinants of success. Current Directions in Psychological Science. 2014;23(5):319-325.
- 315. Biddle SJ, Whitehead SH, O Donovan TM, Nevill ME. Correlates of participation in physical activity for adolescent girls: A systematic review of recent literature. Journal of Physical Activity and Health. 2005;2(4):423.
- 316. Rhodes R, Smith N. Personality correlates of physical activity: A review and meta-analysis. British Journal of Sports Medicine. 2006;40(12):958-965.
- 317. Allen MS, Magee CA, Vella SA, Laborde S. Bidirectional associations between personality and physical activity in adulthood. Health Psychology. 2017;36(4):332-336.
- 318. Giles-Corti B, Timperio A, Bull F, Pikora T. Understanding physical activity environmental correlates: Increased specificity for ecological models. Exercise and Sport Sciences Reviews. 2005;33(4):175-181.
- 319. Nelson NM, Woods CB. Neighborhood perceptions and active commuting to school among adolescent boys and girls. Journal of Physical Activity and Health. 2010;7(2):257-266.
- 320. Verhoeven H, Simons D, Van Dyck D, Van Cauwenberg J, Clarys P, De Bourdeaudhuij I, De Geus B, Vandelanotte C, Deforche B. Psychosocial and environmental correlates of walking, cycling, public transport and passive transport to various destinations in Flemish older adolescents. PLOS ONE. 2016;11(1):e0147128.
- 321. Van Dyck D, De Bourdeaudhuij I, Cardon G, Deforche B. Criterion distances and correlates of active transportation to school in Belgian older adolescents. International Journal of Behavioral Nutrition and Physical Activity. 2010;7(1):87.
- 322. Deforche B, Van Dyck D, Verloigne M, De Bourdeaudhuij I. Perceived social and physical environmental correlates of physical activity in older adolescents and the moderating effect of self-efficacy. Preventive Medicine. 2010;50:S24-S29.
- 323. Owen N, Humpel N, Leslie E, Bauman A, Sallis JF. Understanding environmental influences on walking; Review and research agenda. American Journal of Preventive Medicine. 2004;27(1):67-76.

- 324. Barnett DW, Barnett A, Nathan A, Van Cauwenberg J, Cerin E. Built environmental correlates of older adults' total physical activity and walking: A systematic review and meta-analysis. International Journal of Behavioral Nutrition and Physical Activity. 2017;14(1):103.
- 325. Cerin E, Nathan A, van Cauwenberg J, Barnett DW, Barnett A. The neighbourhood physical environment and active travel in older adults: A systematic review and meta-analysis. International Journal of Behavioral Nutrition and Physical Activity. 2017;14(1):15.
- 326. Royston P, Altman DG, Sauerbrei W. Dichotomizing continuous predictors in multiple regression: A bad idea. Statistics in Medicine. 2006;25(1):127-141.
- 327. Lister-Landman KM, Domoff SE, Dubow EF. The role of compulsive texting in adolescents' academic functioning. Psychology of Popular Media Culture. 2015;6(4):311.
- 328. Rehbein F, Psych G, Kleimann M, Mediasci G, Mößle T. Prevalence and risk factors of video game dependency in adolescence: Results of a German nationwide survey. Cyberpsychology, Behavior, and Social Networking. 2010;13(3):269-277.
- 329. Männikkö N, Billieux J, Kääriäinen M. Problematic digital gaming behavior and its relation to the psychological, social and physical health of Finnish adolescents and young adults. Journal of Behavioral Addictions. 2015;4(4):281-288.
- 330. Lemmens JS, Valkenburg PM, Peter J. Psychosocial causes and consequences of pathological gaming. Computers in Human Behavior. 2011;27(1):144-152.
- 331. Billieux J. Problematic use of the mobile phone: A literature review and a pathways model. Current Psychiatry Reviews. 2012;8(4):299-307.
- 332. Gentile D. Pathological video-game use among youth ages 8 to 18: A national study: Research article. Psychological Science. 2009;20(5):594-602.
- 333. Biddle SJ, Marshall SJ, Gorely T, Cameron N. Temporal and environmental patterns of sedentary and active behaviors during adolescents' leisure time. International Journal of Behavioral Medicine. 2009;16(3):278-286.
- 334. Onetti-Onetti W, Chinchilla-Minguet JL, Martins FML, Castillo-Rodriguez A. Self-concept and physical activity: Differences between high school and university students in Spain and Portugal. Frontiers in Psychology. 2019;10:1333.
- 335. Lopes VP, Gabbard C, Rodrigues LP. Physical activity in adolescents: Examining influence of the best friend dyad. Journal of Adolescent Health. 2013;52(6):752-756.
- 336. Omorou AY, Langlois J, Lecomte E, Briançon S, Vuillemin A. Cumulative and bidirectional association of physical activity and sedentary behaviour with health-related quality of life in adolescents. Quality of Life Research. 2016;25(5):1169-1178.
- 337. Houghton S, Hunter SC, Rosenberg M, Wood L, Zadow C, Martin K, Shilton T. Virtually impossible: Limiting Australian children and adolescents daily screen based media use. BMC Public Health. 2015;15(1):5.
- 338. Paudel S, Subedi N, Bhandari R, Bastola R, Niroula R, Poudyal AK. Estimation of leisure time physical activity and sedentary behaviour among school adolescents in Nepal. BMC Public Health. 2014;14(1):637.
- 339. Dollman J, Maher C, Olds TS, Ridley K. Physical activity and screen time behaviour in metropolitan, regional and rural adolescents: A cross-sectional study of Australians aged 9-16 years. Journal of Science and Medicine in Sport. 2012;15(1):32-37.

- 340. Olds T, Wake M, Patton G, Ridley K, Waters E, Williams J, Hesketh K. How do school-day activity patterns differ with age and gender across adolescence? Journal of Adolescent Health. 2009;44(1):64-72.
- 341. Bucksch J, Sigmundova D, Hamrik Z, Troped PJ, Melkevik O, Ahluwalia N, Borraccino A, Tynjälä J, Kalman M, Inchley J. International trends in adolescent screen-time behaviors from 2002 to 2010. Journal of Adolescent Health. 2016;58(4):417-425.
- 342. Rey-López JP, Vicente-Rodríguez G, Répásy J, Mesana MI, Ruiz JR, Ortega FB, Kafatos A, Huybrechts I, Cuenca-García M, León JF, González-Gross M, Sjöström M, De Bourdeaudhuij I, Moreno LA. Food and drink intake during television viewing in adolescents: The Healthy Lifestyle in Europe by Nutrition in Adolescence (HELENA) study. Public Health Nutrition. 2011;14(9):1563-1569.
- 343. Buck C, Loyen A, Foraita R, Van Cauwenberg J, De Craemer M, Donncha CM, Oppert JM, Brug J, Lien N, Cardon G, Pigeot I, Chastin S. Factors influencing sedentary behaviour: A system based analysis using Bayesian networks within DEDIPAC. PLOS ONE. 2019;14(1):e0211546.
- 344. O'Donoghue G, Perchoux C, Mensah K, Lakerveld J, Van Der Ploeg H, Bernaards C, Chastin SFM, Simon C, O'Gorman D, Nazare JA. A systematic review of correlates of sedentary behaviour in adults aged 18-65 years: A socioecological approach. BMC Public Health. 2016;16(1):163.
- 345. Shilts MK, Horowitz M, Townsend MS. Goal setting as a strategy for dietary and physical activity behavior change: A review of the literature. American Journal of Health Promotion. 2004;19(2):81-93.
- 346. Humpel N, Owen N, Leslie E, Marshall AL, Bauman AE, Sallis JF. Associations of location and perceived environmental attributes with walking in neighborhoods. American Journal of Health Promotion. 2004;18(3):239-242.
- 347. Rossow I, Rise J. Living arrangements and health behaviors in adolescence and young adulthood. Health Education Research. 1993;8(4):495-503.
- 348. Australian Bureau of Statistics. Employee Earnings and Hours, Australia, May 2016. Catalogue no. 6306.0. Canberra; 2017.
- 349. Akaike H. Information theory as an extension of the maximum likelihood principle. In: Petrov B, Csaki F, editors. Second International Symposium on Information Theory. Akademiai Kiado, Budapest 1973. p. 276-281.
- 350. Burnham KP, Anderson DR. Model selection and multimodel inference: A practical information-theoretic approach. 2nd ed. New York: Springer; 2002.
- 351. Giles-Corti B, Foster S, Shilton T, Falconer R. The co-benefits for health of investing in active transportation. New South Wales Public Health Bulletin. 2010;21(5-6):122-127.
- 352. Boyes E, Stanisstreet M, Skamp K, Rodriguez M, Malandrakis G, Fortner RW, Kilinc A, Taylor N, Chhokar K, Dua S, Ambusaidi A, Cheong IP-A, Kim M, Yoon H-G. An international study of the propensity of students to limit their use of private transport in light of their understanding of the causes of global warming. International Research in Geographical and Environmental Education. 2014;23(2):142-165.
- 353. Australian Bureau of Statistics. Education and Work, Australia, May 2018. Catalogue no. 6227.0. Canberra; 2018.
- 354. Bell S, Lee C. Emerging adulthood and patterns of physical activity among young Australian women. International Journal of Behavioral Medicine. 2005;12(4):227-235.

- 355. Bell S, Lee C. Does timing and sequencing of transitions to adulthood make a difference? Stress, smoking, and physical activity among young Australian women. International Journal of Behavioral Medicine. 2006;13(3):265-274.
- 356. Burton NW, Haynes M, Van Uffelen JGZ, Brown WJ, Turrell G. Mid-aged adults' sitting time in three contexts. American Journal of Preventive Medicine. 2012;42(4):363-373.
- 357. Gardiner JC, Luo Z, Roman LA. Fixed effects, random effects and GEE: What are the differences? Statistics in Medicine. 2009;28(2):221-239.
- 358. Hubbard A, Ahern J, Fleischer N, Van der Laan M, Lippman S, Jewell N, Bruckner T, Satariano W. To GEE or not to GEE: Comparing population average and mixed models for estimating the associations between neighborhood risk factors and health. Epidemiology (Cambridge, Mass). 2010;21(4):467.
- 359. Steinberg L, Silverberg SB. The vicissitudes of autonomy in early adolescence. Child Development. 1986;57(4):841-851.
- 360. Steinberg L, Monahan KC. Age differences in resistance to peer influence. Developmental Psychology. 2007;43(6):1531-1543.
- 361. Baumrind D. Current patterns of parental authority. Developmental Psychology. 1971;4(1):1-103.
- 362. Hughes SO, Papaioannou MA. Maternal predictors of child dietary behaviors and weight status. Current Nutrition Reports. 2018;7(4):268-273.
- 363. Suldo SM, Shaunessy-Dedrick E, Ferron J, Dedrick RF. Predictors of success among high school students in advanced placement and International Baccalaureate programs. Gifted Child Quarterly. 2018;62(4):350-373.
- 364. Guerrero LR, Dudovitz R, Chung PJ, Dosanjh KK, Wong MD. Grit: A potential protective factor against substance use and other risk behaviors among Latino adolescents. Academic Pediatrics. 2016;16(3):275-281.
- 365. Zahra J, Ford T, Jodrell D. Cross-sectional survey of daily junk food consumption, irregular eating, mental and physical health and parenting style of British secondary school children. Child: Care, Health and Development. 2014;40(4):481-491.
- 366. Hinckson E, Cerin E, Mavoa S, Smith M, Badland H, Stewart T, Duncan S, Schofield G. Associations of the perceived and objective neighborhood environment with physical activity and sedentary time in New Zealand adolescents. International Journal of Behavioral Nutrition and Physical Activity. 2017;14(1):145.
- 367. Deforche B, Van Dyck D, Verloigne M, De Bourdeaudhuij I. Perceived social and physical environmental correlates of physical activity in older adolescents and the moderating effect of self-efficacy. Preventive Medicine. 2010;50(SUPPL.):S24-S29.
- 368. Little R. Survey nonresponse adjustments for estimates of means. International Statistical Review. 1986;54:139-157.
- 369. Stierlin AS, De Lepeleere S, Cardon G, Dargent-Molina P, Hoffmann B, Murphy MH, Kennedy A, O'donoghue G, Chastin SF, De Craemer M. A systematic review of determinants of sedentary behaviour in youth: A DEDIPAC-study. International Journal of Behavioral Nutrition and Physical Activity. 2015;12(1):133.
- 370. Althubaiti A. Information bias in health research: Definition, pitfalls, and adjustment methods. Journal of Multidisciplinary Healthcare. 2016;9:211-217.
- 371. Terwee CB, Bot SD, de Boer MR, van der Windt DA, Knol DL, Dekker J, Bouter LM, de Vet HC. Quality criteria were proposed for measurement properties of health status questionnaires. Journal of Clinical Epidemiology. 2007;60(1):34-42.

- 372. Hearst MO, Patnode CD, Sirard JR, Farbakhsh K, Lytle LA. Multilevel predictors of adolescent physical activity: A longitudinal analysis. International Journal of Behavioral Nutrition and Physical Activity. 2012;9(1):8.
- 373. Matias TS, Silva KS, Silva JAD, Mello GTD, Salmon J. Clustering of diet, physical activity and sedentary behavior among Brazilian adolescents in the national school-based health survey (PeNSE 2015). BMC Public Health. 2018;18(1):1283.
- 374. Leech RM, McNaughton SA, Timperio A. The clustering of diet, physical activity and sedentary behavior in children and adolescents: A review. International Journal of Behavioral Nutrition and Physical Activity. 2014;11(1):4.
- 375. Murphy JJ, MacDonncha C, Murphy MH, Murphy N, Timperio A, Leech RM, Woods CB. Identification of health-related behavioural clusters and their association with demographic characteristics in Irish university students. BMC Public Health. 2019;19(1):121.
- 376. Murphy JJ, MacDonncha C, Murphy MH, Murphy N, Nevill AM, Woods CB. What psychosocial factors determine the physical activity patterns of university students? Journal of Physical Activity and Health. 2019;16(5):325-332.
- 377. Brownson RC, Hoehner CM, Day K, Forsyth A, Sallis JF. Measuring the built environment for physical activity: State of the science. American Journal of Preventive Medicine. 2009;36(4 (Suppl)):S99-S123. e112.
- 378. Badland HM, Opit S, Witten K, Kearns RA, Mavoa S. Can virtual streetscape audits reliably replace physical streetscape audits? Journal of Urban Health. 2010;87(6):1007-1016.
- 379. Sallis JF. Measuring physical activity environments: A brief history. American Journal of Preventive Medicine. 2009;36(4 (Suppl)):S86-S92.
- 380. Kwon S, Janz KF, Letuchy EM, Burns TL, Levy SM. Developmental trajectories of physical activity, sports, and television viewing during childhood to young adulthood: Iowa Bone Development Study. JAMA pediatrics. 2015;169(7):666-672.
- 381. Australian Bureau of Statistics. Children's Participation in Cultural and Leisure Activities, Australia, Apr 2012. Catalogue no. 4901.0. Canberra; 2012.
- 382. Haapala H, Hirvensalo M, Laine K, Laakso L, Hakonen H, Lintunen T, Tammelin T. Adolescents' physical activity at recess and actions to promote a physically active school day in four Finnish schools. Health Education Research. 2014;29(5):840-852.
- 383. Sutherland RL, Campbell EM, Lubans DR, Morgan PJ, Nathan NK, Wolfenden L, Okely AD, Gillham KE, Hollis JL, Oldmeadow CJ, Williams AJ, Davies LJ, Wiese JS, Bisquera A, Wiggers JH. The Physical Activity 4 Everyone cluster randomized trial: 2-year outcomes of a school physical activity intervention among adolescents. American Journal of Preventive Medicine. 2016;51(2):195-205.
- 384. Christiansen LB, Toftager M, Pawlowski CS, Andersen HB, Ersbøll AK, Troelsen J. Schoolyard upgrade in a randomized controlled study design-how are school interventions associated with adolescents' perception of opportunities and recess physical activity. Health Education Research. 2017;32(1):58-68.
- 385. Wheatley CM, Davies EL, Dawes H. Unspoken playground rules discourage adolescent physical activity in school: A focus group study of constructs in the prototype willingness model. Qualitative Health Research. 2018;28(4):624-632.
- 386. Hardy LL, Mihrshahi S, Gale J, Nguyen B, Baur LA, O'Hara BJ. Translational research: Are community-based child obesity treatment programs scalable? BMC Public Health. 2015;15(1):652.

- 387. Nelson NM, Foley E, O'Gorman DJ, Moyna NM, Woods CB. Active commuting to school: How far is too far? International Journal of Behavioral Nutrition and Physical Activity. 2008;5.
- 388. Ayala AMC, Sudholz B, Salmon J, Dunstan DW, Ridgers ND, Arundell L, Timperio A. The impact of height-adjustable desks and prompts to break-up classroom sitting on adolescents' energy expenditure, adiposity markers and perceived musculoskeletal discomfort. PLOS ONE. 2018;13(9):e0203938.
- 389. Hinckson E, Salmon J, Benden M, Clemes SA, Sudholz B, Barber SE, Aminian S, Ridgers ND. Standing Classrooms: Research and lessons learned from around the world. Sports Medicine. 2016;46(7):977-987.
- 390. Kariippanon KE, Cliff DP, Okely AD, Parrish A-M. Flexible learning spaces reduce sedentary time in adolescents. Journal of Science and Medicine in Sport. 2019;22(8):918-923.
- 391. Mnich C, Bachert P, Kunkel J, Wäsche H, Neumann R, Nigg CR. Stand up, students! Decisional cues reduce sedentary behavior in university students. Frontiers in Public Health. 2019;7:230.
- 392. Jerome M, Janz KF, Baquero B, Carr LJ. Introducing sit-stand desks increases classroom standing time among university students. Preventive Medicine Reports. 2017;8:232-237.
- 393. Plotnikoff RC, Costigan SA, Williams RL, Hutchesson MJ, Kennedy SG, Robards SL, Allen J, Collins CE, Callister R, Germov J. Effectiveness of interventions targeting physical activity, nutrition and healthy weight for university and college students: A systematic review and meta-analysis. International Journal of Behavioral Nutrition and Physical Activity. 2015;12(1):45.
- 394. Maselli M, Ward PB, Gobbi E, Carraro A. Promoting physical activity among university students: A systematic review of controlled trials. American Journal of Health Promotion. 2018;32(7):1602-1612.
- 395. Mackey E, Schweitzer A, Hurtado ME, Hathway J, Dipietro L, Lei KY, Klein CJ. The feasibility of an e-mail-delivered intervention to improve nutrition and physical activity behaviors in African American college students. Journal of American College Health. 2015;63(2):109-117.
- 396. Grim M, Hortz B, Petosa R. Impact evaluation of a pilot web-based intervention to increase physical activity. American Journal of Health Promotion. 2011;25(4):227-230.
- 397. Montagni I, Cariou T, Feuillet T, Langlois E, Tzourio C. Exploring digital health use and opinions of university students: Field survey study. JMIR mHealth and uHealth. 2018;6(3):e65.
- 398. Wadsworth DD, Hallam JS. Effect of a web site intervention on physical activity of college females. American Journal of Health Behavior. 2010;34(2):60-69.
- 399. Okanagan Charter: An International Charter for Health Promoting Universities and Colleges. VII International Congress on Health Promoting Universities and Colleges; 2015. Kelowna, British Columbia, Canada: The University of British Columbia; 2015.
- 400. Taylor P, Saheb R, Howse E. Creating healthier graduates, campuses and communities: Why Australia needs to invest in health promoting universities. Health Promotion Journal of Australia. 2018:1-5.
- 401. Strand BN, Egeberg J, Mozumdar A. The prevalence and characteristics of wellness programs and centers at two-year and four-year colleges and universities. Recreational Sports Journal. 2010;34(1):45-57.
- 402. Bopp M, Sims D, Matthews SA, Rovniak LS, Poole E, Colgan J. Development, implementation, and evaluation of active lions: A campaign to promote active

travel to a university campus. American Journal of Health Promotion. 2018;32(3):536-545.

- 403. Peachey AA, Baller SL. Perceived built environment characteristics of oncampus and off-campus neighborhoods associated with physical activity of college students. Journal of American College Health. 2015;63(5):337-342.
- 404. Brown DMY, Bray SR, Beatty KR, Kwan MYW. Healthy active living: A residence community-based intervention to increase physical activity and healthy eating during the transition to first-year university. Journal of American College Health. 2014;62(4):234-242.
- 405. Yu C-Y, Woo A. From perception to action: The mediating role of parental safety concerns on adolescents' physical activity. Journal of Physical Activity Health and Place. 2017;14(8):652-659.
- 406. Bounova A, Michalopoulou M, Agelousis N, Kourtessis T, Gourgoulis V. The parental role in adolescent screen related sedentary behavior. International Journal of Adolescent Medicine and Health. 2018;30(2):0031.
- 407. Bergh IH, Van Stralen MM, Bjelland M, Grydeland M, Lien N, Klepp KI, Anderssen SA, Ommundsen Y. Post-intervention effects on screen behaviours and mediating effect of parental regulation: The HEalth in Adolescents study - A multi-component school-based randomized controlled trial. BMC Public Health. 2014;14(1):200.
- 408. Gersh E, Richardson LP, Katzman K, Spielvogle H, Arghira AC, Zhou C, McCarty CA. Adolescent health risk behaviors: Parental concern and concordance between parent and adolescent reports. Academic Pediatrics. 2018;18(1):66-72.
- 409. Pearson N, Salmon J, Crawford D, Campbell K, Timperio A. Are parental concerns for child TV viewing associated with child TV viewing and the home sedentary environment? International Journal of Behavioral Nutrition and Physical Activity. 2011;8:102.
- 410. Jackson M, Crawford D, Campbell K, Salmon J. Are parental concerns about children's inactivity warranted, and are they associated with a supportive home environment? Research Quarterly for Exercise Sports Medicine. 2008;79(3):274-282.
- 411. Sigfusdottir ID, Kristjansson AL, Gudmundsdottir ML, Allegrante JP. Substance use prevention through school and community-based health promotion: A transdisciplinary approach from Iceland. Global Health Promotion. 2011;18(3):23-26.
- 412. Barnekow-Bergkvist M, Hedberg G, Janlert U, Jansson E. Physical activity pattern in men and women at the ages of 16 and 34 and development of physical activity from adolescence to adulthood. Scandinavian Journal of Medicine and Science in Sports. 1996;6(6):359-370.
- 413. Riddoch C, Savage JM, Murphy N, Cran GW, Boreham C. Long term health implications of fitness and physical activity patterns. Archives of Disease in Childhood. 1991;66(12):1426-1433.
- 414. Pereira M, FitzerGerald S, Gregg E. A collection of physical activity questionnaires for health-related research. Kriska and Caspersen, Eds. Centers for Disease Control and Prevention. Medicine and Science in Sports and Exercise. 1997;29(6 Suppl):S1-205.
- 415. Glenmark B, Hedberg G, Jansson E. Prediction of physical activity level in adulthood by physical characteristics, physical performance and physical activity in adolescence: An 11-year follow-up study. European Journal of Applied Physiology and Occupational Physiology. 1994;69(6):530-538.

416. Telama R, Yang X, Laakso L, Viikari J. Physical activity in childhood and adolescence as predictor of physical activity in young adulthood. American Journal of Preventive Medicine. 1997;13(4):317-323.

Appendices

Appendix 2.1 Summary of prospective longitudinal studies of tracking of physical activity during the transition from adolescence to adulthood

Study (if			Age at	Follow-up			
applicable) and			baseline	duration			
authors	Country	п	(years)	(years)	Measures	Tracking	Mean change
Norwegian Longitudinal Health Behavior Study. Anderssen, Wold and Torsheim, 2005 <sup>132</sup>	Norway	557	13	8 (6 follow-ups at intervals of 1-2y)	Self-reported frequency and duration of PA.	Weak tracking of PA frequency (females: r=0.18, $p<0.01$ ; males: r=0.22, $p<0.001$ ) and duration (females: r=0.25, $p<0.001$ ; males: $r=0.27$ ,	Proportion active ≥2 times/wk were 72% at 13y, 56%, at 16y, 46% at 19y, and 50% at 21y (females) vs 85%, 66%, 52%, and 53% (males).
Barnekow- Bergkvist et al., 1996 <sup>412</sup>	Sweden	373	15-18	18	Self-reported survey on number of leisure-time sports activities and memberships of sports clubs, and attitude to sports activities, sports performances and PE lessons.	p < 0.001). No tracking in females (r=0.0) or males (r=0.1).	Proportion of LTPA participation ↑51-63% (females) and ↓69-65% (males).
Young Hearts Project. Boreham et al., 2004 <sup>106</sup>	Northern Ireland	476	15	7	Self-reported PA (transport, during school breaks, after school sport).	Poor tracking of PA in females (k=0.021) and low (k=0.202) in males.	Incomparable PA scores were calculated at baseline <sup>413</sup> and follow- up <sup>414</sup> .

#### Appendix 2.1 Continued

Study (if applicable)			Age at baseline	Follow-up duration			
and authors	Country	n	(years)	(years)	Measures	Tracking	Mean change
Quebec Family study. Campbell et al., 2001 <sup>186</sup>	Canada	153	8-18	12	Self-reported 3-day recall survey.	No tracking of MVPA in females (r=0.22; NS) or males (r=0.14; NS).	Not reported.
Childhood Determinants of Adult Health study. Cleland, Dwyer and Venn, 2011 <sup>136</sup>	Australia	2 201	9-15	20	Self-reported questionnaire and pedometer.	Weak tracking of PA (r=-0.08-0.14).	Total PA 13-15y ↑333 mins/wk (females) and 241 mins/wk (males). LTPA ↓102 mins/wk (females) and 164 mins/wk (males).
Glenmark et al., 1994 <sup>415</sup>	Sweden	105	16	11	Self-reported LTPA.	Baseline duration of PA, frequency of PA, frequency of competitive PA, and PA index (range: 5.5- 18) were associated with PA index at follow-up.	Sport club membership ↓52-45% (females) and 79- 52% (males).
Childhood Determinants of Adult Health study. Jose et al., 2011 <sup>213</sup>	Australia	2 048	7-15	19-21	Self-reported LTPA, PA attitudes, and sport competency.	Not reported.	30.3% remained inactive, 49.8% remained variably active and 19.9% remained active.

Appendix 2.1 Continued

				Follow-			
Study (if applicable) and authors	Country	п	Age at baseline (years)	up duration (years)	Measures	Tracking	Mean change
Norwegian Longitudinal Health Behaviour study. Kjønniksen, Torsheim and Wold, 2008 <sup>133</sup>	Norway	630	13	10	Self-reported LTPA and recreational PA (sport).	Not reported.	LTPA↓ in females (B=-0.1; SE=0.01) and males (B=-0.17; SE=0.01). Recreational PA↓ in females (B=-0.09; SE=0.01) and males (B=-0.12; SE=0.01). Those active $\geq$ 4 days/wk at baseline, reduced this level to 2-2.5 days/wk at follow-up. Types of PA↓ from 5.7±3.49 to 3.5±2.33 (females) and 7.5±5.22 to 4.0±2.83 (males).
Canada's National Population Health Survey. Kwan et al., 2012 <sup>135</sup>	Canada	640	12-15	14 (6 follow- ups at 2- year intervals)	Self-reported types of LTPA.	Tracking coefficient = - 0.105, SE=0.03, p < 0.01)	Duration of LTPA↓ by 17% (females) and 30% (males).
Northern Finland birth cohort. Tammelin et al., 2003 <sup>134</sup>	Finland	7 794	14	17	Self-reported participation in leisure- time sport, club membership and PE grade. At follow-up the survey asked about frequency, intensity and duration or LTPA.	Low PE grade associated with adult low PA in males (OR= 1.29; 95%CI=1.07– 1.56; adj. for adolescent PA).	LTPA participation ≥4 times/wk↓ 27.5- 11.5% (females) and 47.9-13.4% (males).

Appendix 2.1	Continued
--------------	-----------

Study (if applicable) and authors	Country	п	Age at baseline (years)	Follow-up duration (years)	Measures	Tracking	Mean change
Cardiovascular Risk in Young Finns study. Telama et al., 1997 <sup>416</sup>	Finland	2 309	9-18	9 and 12	Self-reported school sport participation, school PE, and LTPA type, duration and intensity.	Significant but low tracking. Correlation coefficients after 9y ranged from 0.18- 0.47, and 0.00-0.27 after 12y. Strongest associations with later PA were with baseline competitive sport participation and school PE grade.	Not reported.
Cardiovascular Risk in Young Finns study. Telama et al., 2005 <sup>131</sup>	Finland	1 563	3-18	18 (5 follow-ups at 3-year intervals)	Self-reported active commute, school PE, and LTPA type, duration and intensity.	High PA predicted high adulthood PA. Females less likely than males to track PA into adulthood. Tracking coefficient from 9-24y was 0.21 for females and 0.31 for males.	Not reported.
Cardiovascular Risk in Young Finns study. Telama et al., 2006 <sup>137</sup>	Finland	1 606	9-18	21	Self-reported validated survey about organised sport participation and its training frequency. Follow-up survey contained frequency, duration, and intensity of PA.	Frequent organised sport participation at baseline increased the probability of being highly active at follow-up: OR=6, 95%CI=2.38-15.14 in females; and OR=5.11, CI=2.88-9.08 in males.	Not reported.

Appendix 2.1 Continued

				Follow-			
Study (if			Age at	up			
applicable)			baseline	duration			
and authors	Country	п	(years)	(years)	Measures	Tracking	Mean change
Trois-	Canada	166	10-12	23-25	Self-reported weekly	Significant but weak tracking	Not reported.
Rivières					duration of total,	between childhood and	
Growth and					vigorous-intensity,	adulthood total PA (r=0.2),	
Development					light-intensity, and	vigorous PA (r=0.18),	
study.					organised PA.	organised PA (r=0.12).	
Trudeau et					-	_	
al., 2004 <sup>214</sup>							
Amsterdam	Netherlands	181	13	14	Interviews. Self-	Daily PA tracked low-	Mean daily PA change (1 000
Growth and					reported frequency,	moderately and significantly:	METs/wk) decreased 3.81±1.7-
Health study.					intensity and duration	stability coefficient was 0.34	$3.18\pm1.9$ (females) and
Twisk et al.,					of total PA of the last	(95%CI=0.19-0.49); OR=3.6	4.99±2.0-2.91±2.3 (males).
$2000^{85}$					three months.	(CI=2.4-5.4).	

±: standard deviation; CI: confidence interval; LTPA: leisure-time physical activity; METs: metabolic equivalent tasks; MVPA: moderate- to vigorous-intensity physical activity; NS: not significant; OR: odds ratio; PA: physical activity; PE: physical education; SE: standard error; y: years.

Appendix 3.1 Consolidated criteria for reporting qualitative research (COREQ) guidelines<sup>195</sup>

No	Item	Guide questions/description

### Domain 1: Research team and reflexivity

Personal Characteristics

1. 2.	Interviewer/facilitator Credentials	Jennifer Hatt. BEx&SS(Hons).
3.	Occupation	PhD candidate.
4. 5.	Gender Experience and training	Sex: female. Gender: I identify as a woman. Research Assistant experience with Quantitative studies. Limited Qualitative training.
Rela	tionship with participants	
6.	Relationship established	One participant was a first cousin once removed and another participant was a cousin-in-law. Otherwise, no.
7.	Participant knowledge of the interviewer	Occupation.
8.	Interviewer characteristics	Gender and undergraduate degree qualification.

#### Domain 2: study design

Theo	oretical framework	
9.	Methodological orientation and Theory	The ecological model <sup>20, 21</sup> and the life transition model <sup>18</sup> .
Parti	cipant selection	
10.	Sampling	Convenience and snowball.
11.	Method of approach	Convenience sampling, flyers, snowball sampling
		and social media.
12.	Sample size	29
13.	Non-participation	None.

#### Setting

Sell	ing	
14.	Setting of data	Interviewer at place of study; participant at home
	collection	( <i>n</i> =24), travelling ( <i>n</i> =1) or face-to-face at place of study ( <i>n</i> =4).
15.	Presence of non-	Yes, for four interviews: One Mum, two Dads,
	participants	and members of the public on a train.
16.	Description of sample	See 3.4.1. Gender equal. All lived in an urban
		area.
		Most were 19-years-old, finished secondary
		school >1-year ago, living in a least-
		disadvantaged neighbourhood (the highest tertile
		of socio-economic area), and were concurrently
		working and studying.

### Appendix 3.1 Continued

Data	collection	
17.	Interview guide	Appendix 3.2. Questions, prompts, guides were pilot-tested and used.
18.	Repeat interviews	No.
19.	Audio/visual	Audio recording.
	recording	
20.	Field notes	Few made during and/or after the interviews.
21.	Duration	Range: 11-30 mins. Mean 20.9±6.1 mins (SD)
22.	Data saturation	Yes. Rolling recruitment ceased at <i>n</i> =29.
23.	Transcripts returned	No.

## Domain 3: analysis and findings

Data analysis

24.	Number of data coders	Two for 10% of sample. Dr. Felicity Pendergast and Jennifer Hatt. Otherwise, one.
25.	Description of the coding tree	Appendix 3.3. 75 codes sectioned into six categories.
26.	Derivation of themes	Derived from data.
27.	Software	QSR NVivo 12
28.	Participant checking	No.
Rep	orting	Vac and quotes identified via participant
29.	Quotations presented	demographic characteristics.
30.	Data and findings consistent	Yes.
31.	Clarity of major themes	Major themes clearly presented in the findings.
32.	Clarity of minor	Diverse cases described and discussion.
	themes	

#### **Appendix 3.2** Qualitative study interview topic guide

#### [Preamble]

Hi.

This is Jenny Hatt from the LEAP study at Deakin University. How are you? I am just calling to conduct the LEAP interview that you volunteered for. This interview is expected to take between 20-40 minutes; is that okay? Okay, great. [*If no, arrange alternative time to call*]

Thank you for agreeing to take part in this study.

This interview is about your experiences and viewpoints about physical activity and sedentary behaviour over your transition out of high school. Very few studies have researched experiences during the transition out of high school.

I will record the interview, but your personal details will be kept confidential. There are no right or wrong answers. You do not need to answer questions that you would rather not. You can end the interview at any time for any reason. Is all of that okay? Do you have any questions before we start?

Okay great. [*Start recording*]

You indicated that you left school in 20xx. Can you tell me in what ways your life changed after leaving school?

[*Clarification, examples, prompts if needed:*]

- Did you go straight to University or get a job?
- *Travel, take time off?*
- *Did you move out of home?*
- Did expectations of you at home change [if applicable]? Do you have more responsibilities now, do more chores or pay board?
- Did your friendship groups and social life change?
- Did you get your license and a car?

I am now going to ask you about physical activity which means 'any bodily movement produced by skeletal muscles that requires energy expenditure'. Examples are walking, jogging, cycling, playing sport, swimming, non-organised things such as playing with a basketball or Frisbee, or other tasks like shopping and chores. Can you please describe your physical activity when you were still at school? /How active were you when you were went to school?

[Clarification, examples, prompts if needed:]

- Did you walk or cycle for transport - to/from school, to shops, friends' houses?
- Physical activity at school: Physical education classes, active during recess and lunch?
- Were you on any sports teams or have a gymnasium membership?
- Did you do any non-organised physical activity so going for walks or bike rides, shooting hoops?
- Did you have a job while you were at school and did you mainly sit, stand or move?
- *Physical activity at home; e.g. chores, gardening?*

Can you please describe your physical activity now? Have there been any changes? More/less/same? / How active are you now?

[Clarification, examples, prompts if needed:]

- Do you walk or cycle for transport to/from university, work, shops, friends' houses? More, less or the same as when at school?
- Are you on any sports teams or have a gymnasium membership? More, less or the same as when at school?
- Do you do any non-organised physical activity in your leisure-time so walking, cycling, shooting hoops with family or friends? More, less or the same as when at school?
- Do you have a job now? Do you mainly sit, stand or move at your work? More, less or the same as when at school?
- *Physical activity at home; e.g. chores? Mostly sit, stand, or move? More, less or the same as when at school?*
- *Physical activity at university: mostly sit, stand, or move? More, less or the same as when at school?*

Can you please summarise how your physical activity has changed since you left high school?

It is interesting that you mentioned that you did different physical activity while you were at school, compared to now. I'd like to pick up on that and hear a bit more about it. Can you list some reasons for that change?

Can you think of any other reasons?

Was there one particular influence that was more important than the others? [If few self-identified reasons provided, prompt for the situational transitions after

school (from the start of the interview) if not provided as reasons.]

- *How does your university influence how much physical activity you do?*
- How does your work influence how much physical activity you do?
- How does your moved to a new house/change in home responsibilities influence how much physical activity you do?
- How do your friends influence how much physical activity you do?
- How does your family influence how much physical activity you do?
- *How does having your license now/driving influence how much physical activity you do? More of a commute now?*
- *How does your gender, remoteness, socio-economic area/income influence how much physical activity you do?*

Can you think of anything that your school could have been done to help make you do more physical activity while you were there?

/after you left school?

Can you think of anything that could be done at university/ work/ home/ friends/ family can do now that would make you do more physical activity?

I'm now going to switch from physical activity to sedentary behaviour. Sedentary behaviour means any activity that you do while you are awake and sitting or reclining that requires low energy expenditure. Examples include time spent sitting at a desk, sitting to read, study, watch television, use a computer/tablet/mobile, play music or electronic games consoles, and sitting while travelling, such as when driving or on public transport.

Can you please describe your sedentary behaviour when you were still at school? / What kinds of sedentary behaviour did you do? / How much time did you spend sitting when you went to school?

[Clarification, examples, prompts if needed:] In different settings:

- Leisure-time
- At school
- At home
- At work?
- Travelling driving? Commute? Train?
- How much do you watch television, read, watch movies, play video games, use a portable music player/ mobile/ tablet?
- When you sat, was it for a prolonged period (more than 30 minutes)?

Can you please describe your sedentary behaviour now? Have there been any changes? More/less/same? What kinds of sedentary behaviour do you do? How much time do you spend sitting now?

[Clarification, examples, prompts if needed:] In different settings:

- Leisure-time
- At university
- At home
- At work
- Travelling driving? Commute? Train?
- How much do you watch television, read, watch movies, play video games, use a portable music player/ mobile/ tablet?
- and is that more, less or the same as when you were in high school?
- When you sit, is it for a prolonged period (more than 30 minutes)?
- If you were to break up prolonged sitting time, would it only happen if you consciously decided to or do you naturally spontaneously break up your sitting time?

Can you please summarise how your sedentary behaviour changed after you left school? It is interesting that you mentioned that you did different sedentary behaviour while you were at school, compared to now. I'd like to pick up on that and hear a bit more about it.

What were the reasons for that change?

Can you think of any other reasons?

Was there one particular influence that was more important than the others? [If few self-identified reasons provided, prompt for the situational transitions after school (from the start of the interview) if not provided as reasons.]

- How does your university influence how much you sit?
- *How does your work influence how much you sit?*
- *How does your moved to a new house/change in home responsibilities influence how much you sit?*
- How do your friends influence how much you sit?
- How does your family influence how much you sit?
- *How does having your license now/driving influence how much you sit? More of a commute now?*
- *How does your gender, remoteness, socio-economic area /income influence how much you sit?*

Can you think of anything that your school could have been done to help make you sit less while you were there?

/after you left school? Can you think of anything that could be done at university/ work/ home/ friends/ family that might help you sit less?

[Closing]

That is all of the questions that I have for you. Is there anything extra that you can think of that you feel we should have talked about, but we have missed? I am after a clear picture of how your physical activity and sedentary behaviour has changed since you left school, and why it has changed. Is there anything extra that you can add?

Do you have any questions for me or about the study?

In closing, has there been anyone else present while you have being doing this interview?

[*If they ask why*] I ask because I need to report that, as having other people around might bias some people's answers.

Could you please describe your setting right now? For example, are you at home?

[*If they ask why*] I ask because I need report that, as the setting you are in can bias some people's answers.

I would like to now offer you a \$20 Coles Group and Myer voucher as compensation for participating in this study? Would you like one of those?

I will post it to you and you should get it soon.

If you want to get in touch again later, just email the LEAP email address. Thank you so much for your time and for contributing to this study. Have a good day.

Code	Description of what the categories and codes referred to
Category: Gender 'GEN'	The young adult's self-identified gender.
Woman 'GEN WOM'	The young adult self-identified as a woman.
Man 'GEN MAN'	The young adult self-identified as a man.
Category: Changes 'CHA'	The young adults' experiences and viewpoints about what has changed since they have left secondary school that are not directly identified as being associated with PA or SB, including moving to a new house, changes in friendship groups and commencing employment.
Category: Levels 'LEV'	PA and SB they engage in during secondary school and now.
Leisure-time physical activity 'LEV LEI PA'	Anything related to PA during leisure-time (after school and on weekends) including going to public gymnasiums, dog walking, non-organised PA such as walking, cycling, shooting hoops, as well as activities at home including chores, gardening and home gymnasiums.
Leisure-time sedentary behaviour 'LEV LEI SB'	Anything related to SB during leisure-time (after school and on weekends) including playing video games, reading, studying at home, doing homework, watching television, and watching movies at home or at a cinema.
Occupation physical activity 'LEV OCC PA'	Anything related to standing, moving and physical activities at work.
Occupation sedentary behaviour 'LEV OCC SB'	Anything related to sitting at work.
Prolonged sitting 'LEV PRO'	Anything related to unbroken prolonged sitting >30 mins and not being able to break up prolonged sitting without reminders or prompts.
Prolonged sitting breaks 'LEV PRO BR'	Anything related to prolonged sitting that is broken up into smaller bouts naturally and spontaneously without the need for reminders or prompts.
School physical activity 'LEV SCH PA'	Anything related to PA levels at school, including active breaks in the classroom, PE lessons, school sport, school sport carnivals and being active during school breaks (recess and lunchtime).
School sedentary behaviour 'LEV SCH SB'	Anything related to SB levels at school, including sitting during class and breaks (recess and lunchtime), and studying at school.
Sport 'LEV SPO'	Anything related to sport, including being a member of a sport team at school and/or in the community.
Technology 'LEV TEC'	Anything related to technology and electronic devices, including using social media, internet, mobile phones, computers, laptops, tablets, portable music players and video games, and whether they are used at home, school, place of tertiary study or work.

# Appendix 3.3 Qualitative study coding tree

Diago of toutions	DA levels at the place of tertiamy study (university and
Place of tertiary	PA levels at the place of tertiary study (university and
study physical	vocational training), including standing or moving during
activity	classes (lectures, tutorials and practicals) and moving
'LEV TER PA'	during breaks between classes.
Place of tertiary	SB levels at the place of tertiary study (university and
study sedentary	vocational training), including sitting during classes
behaviour	(lectures, tutorials and practicals), sitting to study on
'LEV TER SB'	campus, and sitting during breaks between classes.
Transport physical	Anything related to active transport, including walking or
activity	cycling for transport, and standing on public transport.
'LEV TRA PA'	
Transport sedentary	Anything related to SB and transport, including sitting on
behaviour	public transport, driving, car ownership and having a
'LEV TRA SB'	driver's license.
Category: Reasons	The young adults' experiences and viewpoints about what are
'REA'	the reasons and influences that they have attributed to their
ILL/I	changes in PA and SB over their transition out of secondary
	school
Age	Anything related to age as a reason or influence on PA.
'REA AGE PA'	
AGE	Anything related to age as a reason or influence on SB.
'REA AGE SB'	
Availability	Anything related to availability as a reason or influence on
ΈΕΔ ΔΥΔ ΡΔ'	PA including the availability of and access to technology
	resources and facilities
Avoilability	A nuthing related to availability as a reason or influence on
Availability	SD including the queilability of and access to technology
KEA AVA SB	SB, including the availability of and access to technology,
	resources and facilities.
	Anything related to food, drink or energy intake as a reason
•REA FOO PA'	or influence on PA.
Food	Anything related to food, drink or energy intake as a reason
'REA FOO SB'	or influence on SB.
Gender	Anything related to gender as a reason or influence on PA.
'REA GEN PA'	
Gender	Anything related to gender as a reason or influence on SB.
'REA GEN SB'	
Habits	Anything related to habits, routines, structure and
'REA HAB PA'	regimented days as a reason or influence on PA.
Habits	Anything related to habits, routines, structure and
'REA HAB SB'	regimented days as a reason or influence on SB.
Home	Anything related to home, including moving out of the
'REA HOM PA'	family home/changing residences, having more
	responsibilities now, having to do more chores and having
	to pay board as a reason or influence on PA.
Home	Anything related to home, including moving out of the
'REA HOM SR'	family home/changing residences having more
	responsibilities now having to do more chores and having
	to pay board as a reason or influence on SR
Income	Anything related to income and money as a reason or
DEA INC DAY	influence on DA
	A mathing related to income or dimensional managements
income	Anything related to income and money as a reason or
TKEA INC SB	influence on SB.

Independence	Anything related to independence levels and freedom as a
Independence	A systhing related to independence levels and freedom as a
'REA IND SB'	reason or influence on SB.
Internal 'REA INT PA'	Related to anything internal as a reason or influence on PA, including maturity levels, being positively or negatively motivated internally (not externally by social networks, for example), feelings of shame or embarrassment, mental health and depression, personal preference (likes, dislikes, being bored, attention span, interest and enjoyment levels) and attitude.
Internal	Related to anything internal as a reason or influence on SB,
'REA INT SB'	including maturity levels, being positively or negatively motivated internally (not externally by social networks, for example), feelings of shame or embarrassment, mental health and depression, personal preference (likes, dislikes, being bored, attention span, interest and enjoyment levels) and attitude.
Main reason	The reason that is the most important and influential for
'REA MAI PA'	their PA.
Main reason 'REA MAI SB'	The reason that is the most important and influential for their SB.
Occupation	Anything related to the occupation, including the choice of
'REA OCC PA'	occupation and changes in occupation (commencing full-
	time work, part-time work and casual work) as a reason or
	influence on DA
	influence on PA.
Occupation 'REA OCC SB'	Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full- time work, part-time work and casual work) as a reason or influence on SB.
Occupation 'REA OCC SB' Pet ownership 'REA PET PA'	Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full- time work, part-time work and casual work) as a reason or influence on SB. Anything related to having a pet, especially walking a dog as a reason or influence on PA.
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB'	Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full- time work, part-time work and casual work) as a reason or influence on SB. Anything related to having a pet, especially walking a dog as a reason or influence on PA. Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB.
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Iniury	Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full- time work, part-time work and casual work) as a reason or influence on SB. Anything related to having a pet, especially walking a dog as a reason or influence on PA. Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB. Anything related to an injury, a physical impairment or
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA'	Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full- time work, part-time work and casual work) as a reason or influence on SB. Anything related to having a pet, especially walking a dog as a reason or influence on PA. Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB. Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma allergies and a lack of sleep/being tired as a reason
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA'	<ul> <li>Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full-time work, part-time work and casual work) as a reason or influence on SB.</li> <li>Anything related to having a pet, especially walking a dog as a reason or influence on PA.</li> <li>Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on PA.</li> </ul>
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA'	Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full- time work, part-time work and casual work) as a reason or influence on SB. Anything related to having a pet, especially walking a dog as a reason or influence on PA. Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB. Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on PA. Anything related to an injury a physical impairment or
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA' Injury 'REA PHY SB'	<ul> <li>Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full-time work, part-time work and casual work) as a reason or influence on SB.</li> <li>Anything related to having a pet, especially walking a dog as a reason or influence on PA.</li> <li>Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on PA.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on PA.</li> </ul>
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA' Injury 'REA PHY SB'	<ul> <li>Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full-time work, part-time work and casual work) as a reason or influence on SB.</li> <li>Anything related to having a pet, especially walking a dog as a reason or influence on PA.</li> <li>Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as a sthma, allergies, and a lack of sleep/being tired as a reason or influence on PA.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as a sthma, allergies, and a lack of sleep/being tired as a reason or influence on PA.</li> </ul>
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA' Injury 'REA PHY SB' Priorities	<ul> <li>Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full-time work, part-time work and casual work) as a reason or influence on SB.</li> <li>Anything related to having a pet, especially walking a dog as a reason or influence on PA.</li> <li>Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on PA.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on SB.</li> <li>Anything related to planning, time management, priorities</li> </ul>
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA' Injury 'REA PHY SB' Priorities 'REA PRI PA'	<ul> <li>Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full-time work, part-time work and casual work) as a reason or influence on SB.</li> <li>Anything related to having a pet, especially walking a dog as a reason or influence on PA.</li> <li>Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on PA.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on SB.</li> <li>Anything related to planning, time management, priorities or goal setting as a reason or influence on PA.</li> </ul>
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA' Injury 'REA PHY SB' Priorities 'REA PRI PA' Priorities	<ul> <li>Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full-time work, part-time work and casual work) as a reason or influence on SB.</li> <li>Anything related to having a pet, especially walking a dog as a reason or influence on PA.</li> <li>Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as a sthma, allergies, and a lack of sleep/being tired as a reason or influence on PA.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as a sthma, allergies, and a lack of sleep/being tired as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as a sthma, allergies, and a lack of sleep/being tired as a reason or influence on SB.</li> <li>Anything related to planning, time management, priorities or goal setting as a reason or influence on PA.</li> <li>Anything related to planning, time management, priorities</li> </ul>
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA' Priorities 'REA PHY SB' Priorities 'REA PRI PA'	<ul> <li>Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full-time work, part-time work and casual work) as a reason or influence on SB.</li> <li>Anything related to having a pet, especially walking a dog as a reason or influence on PA.</li> <li>Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on PA.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on SB.</li> <li>Anything related to planning, time management, priorities or goal setting as a reason or influence on SB.</li> <li>Anything related to planning, time management, priorities or goal setting as a reason or influence on SB.</li> </ul>
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA' Priorities 'REA PHY SB' Priorities 'REA PRI PA' Priorities 'REA PRI SB' Remoteness	<ul> <li>Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full-time work, part-time work and casual work) as a reason or influence on SB.</li> <li>Anything related to having a pet, especially walking a dog as a reason or influence on PA.</li> <li>Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on PA.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on SB.</li> <li>Anything related to planning, time management, priorities or goal setting as a reason or influence on SB.</li> <li>Anything related to planning, time management, priorities or goal setting as a reason or influence on SB.</li> <li>Anything related to planning, time management, priorities or goal setting as a reason or influence on SB.</li> </ul>
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA' Priorities 'REA PHY SB' Priorities 'REA PRI PA' Priorities 'REA PRI SB' Remoteness 'REA REM PA'	<ul> <li>Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full-time work, part-time work and casual work) as a reason or influence on SB.</li> <li>Anything related to having a pet, especially walking a dog as a reason or influence on PA.</li> <li>Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as a sthma, allergies, and a lack of sleep/being tired as a reason or influence on PA.</li> <li>Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as a sthma, allergies, and a lack of sleep/being tired as a reason or influence on SB.</li> <li>Anything related to planning, time management, priorities or goal setting as a reason or influence on SB.</li> <li>Anything related to planning, time management, priorities or goal setting as a reason or influence on SB.</li> <li>Anything related to planning, time management, priorities or goal setting as a reason or influence on SB.</li> </ul>
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA' Priorities 'REA PHY SB' Priorities 'REA PRI PA' Priorities 'REA PRI SB' Remoteness 'REA REM PA'	Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full- time work, part-time work and casual work) as a reason or influence on SB. Anything related to having a pet, especially walking a dog as a reason or influence on PA. Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB. Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on PA. Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on PA. Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on SB. Anything related to planning, time management, priorities or goal setting as a reason or influence on PA. Anything related to planning, time management, priorities or goal setting as a reason or influence on SB. Anything related to remoteness as a reason or influence on PA, including changes from living in a rural area to an urban area.
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA' Priorities 'REA PHY SB' Priorities 'REA PRI PA' Priorities 'REA PRI SB' Remoteness 'REA REM PA'	Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full- time work, part-time work and casual work) as a reason or influence on SB. Anything related to having a pet, especially walking a dog as a reason or influence on PA. Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB. Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as a sthma, allergies, and a lack of sleep/being tired as a reason or influence on PA. Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as a sthma, allergies, and a lack of sleep/being tired as a reason or influence on PA. Anything related to planning, time management, priorities or goal setting as a reason or influence on SB. Anything related to planning, time management, priorities or goal setting as a reason or influence on SB. Anything related to planning, time management, priorities or goal setting as a reason or influence on SB. Anything related to planning, time management, priorities or goal setting as a reason or influence on SB.
Occupation 'REA OCC SB' Pet ownership 'REA PET PA' Pet ownership 'REA PET SB' Injury 'REA PHY PA' Priorities 'REA PHY SB' Priorities 'REA PRI PA' Priorities 'REA PRI SB' Remoteness 'REA REM PA'	Anything related to the occupation, including the choice of occupation and changes in occupation (commencing full- time work, part-time work and casual work) as a reason or influence on SB. Anything related to having a pet, especially walking a dog as a reason or influence on PA. Anything related to having a pet, including owning a cat, rather than a dog as a reason or influence on SB. Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on PA. Anything related to an injury, a physical impairment or physical health, including conditions and illnesses such as asthma, allergies, and a lack of sleep/being tired as a reason or influence on PA. Anything related to planning, time management, priorities or goal setting as a reason or influence on PA. Anything related to planning, time management, priorities or goal setting as a reason or influence on SB. Anything related to planning, time management, priorities or goal setting as a reason or influence on SB. Anything related to remoteness as a reason or influence on PA, including changes from living in a rural area to an urban area. Anything related to remoteness as a reason or influence on SB, including changes from living in a rural area to an

School 'REA SCH PA'	Anything related to school as a reason or influence on PA, including size and quality of grounds, play equipment, oval, sport teams and facilities, or studying, assignments, exams and homework
<u>C</u> -1 1	A mething malete d to ache al esta margin flagment an CD
'REA SCH SB'	Anything related to school as a reason or influence on SB, including size and quality of grounds, play equipment,
	exams and homework.
Social support 'REA SOC PA'	Anything related to the young adult's social network, including their friends, family, housemates, peers and
	health practitioners as a reason or influence on PA
Social support	Anything related to the young adult's social network
'DEA SOC SD'	including their friends, family, housemates, nears and
KEA SUC SD	health prestitioners as a reason or influence on SP
	nearin practitioners as a reason or influence on SB.
Technology	Anything related to technology-use as a reason or influence
'REA TEC SB'	on SB, including video games, social media, computer,
	mobile phone, portable music players and televisions.
Study	Anything related to tertiary study as a reason or influence
'REA TER PA'	on PA, including time spent studying, time on campus
	(contact hours), the choice of course, being educated and
	the grounds/facilities.
Study	Anything related to tertiary study as a reason or influence
'REA TER SB'	on SB including time spent studying time on campus
ILLIT ILLIT OD	(contact hours) the choice of course being educated and
	the grounds/facilities
Time	A nuthing related to time as a reason or influence on DA
	Anything feated to time as a feason of influence of FA,
KEA I IIVI PA	including free time, spare time, routines, regimented days
<b>—</b>	and nondays.
I ime	Anything related to time as a reason or influence on SB,
KEA HM SB	including free time, spare time, routines, regimented days
	and holidays.
Transport	Anything related to transport as a reason or influence on
'REA TRA PA'	PA, including walking or cycling for transport, public
	transport, driving, car ownership, driver's license and
	commute duration/distance.
Transport	Anything related to transport as a reason or influence on
'REA TRA SB'	SB, including walking or cycling for transport, public
	transport, driving, car ownership, driver's license and
	commute duration/distance.
Weather	Anything related to weather as a reason or influence on
'REA WEA PA'	PA including hot and cold temperature rain and sunshine
Woother	Anything related to weather as a reason or influence on SP
	Anything related to weather as a reason of influence on SD,
	A worthing malated to main ht are a manual sufficiency on DA
weight	Anything related to weight as a reason or influence on PA,
KEA WEI PA	including body image, weight levels, weight loss and weight gain
Catagoriu I. C	The young edulte? experiences and viewer sints about what
'INF'	influences their PA and SB in a positive or negative way.
Negative influences	All types of influences that resulted in PA decreasing or SR
'INF NEG'	increasing.
	<b>0</b> .

No influence 'INF NON'	Anything that was deemed not an influence.
Positive influences 'INF POS'	All types of influences that resulted in PA increasing or SB decreasing.
Category: Action 'ACT'	Any suggested actions and strategies that the young adults suggested could be changed at school, university, home, work or within their social network that could result in more PA and less SB.
Home 'ACT HOM PA'	Any strategies to increase PA that could be implemented at home.
Home 'ACT HOM SB'	Any strategies to decrease SB that could be implemented at home.
Occupation 'ACT OCC PA'	Any strategies to increase PA that could be implemented at their place of occupation.
Occupation 'ACT OCC SB'	Any strategies to decrease SB that could be implemented at their place of occupation.
Other 'ACT OTH PA'	Any strategies to increase PA where there has not been a setting or context specified, or are general, overall, other, random or uncommon.
Other 'ACT OTH SB'	Any strategies to decrease SB where there has not been a setting or context specified, or are general, overall, other, random or uncommon.
School 'ACT SCH PA'	Any strategies to increase PA that could be implemented at their secondary school.
School 'ACT SCH SB'	Any strategies to decrease SB that could be implemented at their secondary school.
Social network 'ACT SOC PA'	Any strategies to increase PA that could be implemented by their social network, including their friends, family, housemates and peers.
Social network 'ACT SOC SB'	Any strategies to decrease SB that could be implemented by their social network, including their friends, family, housemates and peers.
Study 'ACT TER PA'	Any strategies to increase PA that could be implemented at their place of tertiary study, including university or vocational training.
Study 'ACT TER SB'	Any strategies to decrease SB that could be implemented at their place of tertiary study, including university or vocational training.

PA: physical activity; SB: sedentary behaviour.

#### Appendix 4.1 ProjectADAPT two-year follow-up telephone survey

Participant ID: \_\_\_\_\_ (4 DIGITS ONLY)

Interviewer ID: \_\_\_\_\_ (2 DIGITS)

**SECTION 1: Eating Behaviour** 

The first section of the interview is about things that you eat.

1 How many serves of VEGETABLES do you USUALLY eat each day? This can include fresh, dried, frozen and tinned vegetables, but not potatoes, hot chips or fried potato (a serve is equal to <sup>1</sup>/<sub>2</sub> cup of cooked vegetables or 1 cup of salad vegetables).

[DO NOT READ OPTIONS] (SELECT ONE ONLY)

- None/don't eat vegetables
  - Less than one serve a day
- o 1 serve
- o 2 serves
- 3 serves
- 4 serves
- 5 serves
- 6 serves or more
- 2 How many serves of FRUIT do you USUALLY eat each day? This can include all fresh, dried, frozen, and tinned fruit, <u>but not fruit juice</u> (a serve is equal to 1 medium piece or 2 small pieces of fruit, or 1 cup of diced pieces of fruit).

[DO NOT READ OPTIONS] (SELECT ONE ONLY)

- None/don't eat fruit
- Less than one serve a day
- o 1 serve
- o 2 serves
- o 3 serves
- 4 serves
- o 5 serves
- $\circ$  6 serves or more

#### **3** What type of BREAD do you USUALLY eat?

(If more than one type, ask which one they eat most often). [DO NOT READ OPTIONS] (SELECT ONE ONLY)

None/Don't eat bread
High fibre white bread
Regular white bread
Wholemeal (brown) bread
Rye bread
Multigrain bread
Other bread

# 4 What type of MILK do you USUALLY drink? (If more than one type, ask which type they drink most often).

[DO NOT READ OPTIONS] (SELECT ONE ONLY)		••	• •		• ,
	DO NOT	READ OPTIC	ONS] (SE	LECT	ONE ONLY)

0	None/don't drink milk
0	Whole milk (full-cream)
0	Trim, low or reduced-fat milk (eg. Rev)
0	Skim (no fat) milk
0	Flavoured milk (eg. chocolate)
0	Other type of milk (eg. soy, rice, goat)
0	Evaporated or sweetened condensed milk

5 How much WATER do you USUALLY drink each day? This can be plain tap water, mineral water or bottled water (*Clarify per WEEK or per DAY*; 1 cup = 250ml, 1 average bottle = 2 cups, 1 litre bottle = 4 cups)..

#### [DO NOT READ OPTIONS] (SELECT ONE ONLY)

- None
- Less than 1 cup/glass a day (<250ml)
- ~1 cup/glass a <u>day</u> (<250ml)
- $\circ$  ~2 cups/glasses a <u>day</u> (~500ml)
- $\circ$  ~3 cups/glasses a <u>day</u> (<750ml)
- ~4 cups/glasses a <u>day</u> (~1litre)
- $\circ$  5 or more cups/glasses a <u>day</u> (>=1.25litre)

#### 6 How much FRUIT JUICE do you USUALLY drink in a day or week?

(*Clarify per WEEK or per DAY;* 1 cup = 250ml, a Prima / Popper / Tetra pack = 1 cup).

- None/don't drink fruit juice
  - Less than 1 cup/glass a week
  - ~1-3 cups/glasses a <u>week</u>
  - ~4-6 cups/glasses a <u>week</u>
  - $\circ$  ~1-2 cups/glasses a <u>day</u>
  - $\circ$  ~3-4 cups/glasses a **<u>day</u>**
  - $\circ$  5 or more cups a **<u>day</u>**

#### 7 How much <u>DIET</u> SOFT DRINKS, DIET CORDIAL or DIET SPORTS DRINKS do you USUALLY drink in a week or day? (*Clarify per WEEK or per*

DAY; 1 cup = 250ml, one can of diet soft drink =  $1 \frac{1}{2}$  cups).

[DO NOT READ OPTIONS] (SELECT ONE ONLY)

0	None
0	Less than 1 cup a <u>week</u>
0	~1-3 cups a <u>week</u>
0	~4-6 cups a <u>week</u>
0	~1-2 cups a <u>day</u>
0	~3-4 cups a <u>day</u>
0	5 or more cups a <b>day</b>

#### 8 How much <u>REGULAR</u> SOFT DRINK, CORDIAL, ENERGY DRINK or SPORTS DRINK do you USUALLY drink in a week or day? (*Clarify per WEEK or per DAY;* 1 cup = 250ml, one can of soft drink = 1 ½ cups) Do not include diet soft drinks.

[DO NOT READ OPTIONS] (SELECT ONE ONLY)

- o None
- Less than one cup a <u>week</u>
- ~1-3 cups a <u>week</u>
- ~4-6 cups a <u>week</u>
- ~1-2 cups a <u>day</u>
- ~3-4 cups a <u>day</u>
- 5 or more cups a <u>day</u>

I'm no How o	ow going to ften do yo	ask you a u:	bout how	often you	ı eat a range	of foods	
[D	O NOT RE	EAD RESP	ONSE OP	TIONS] (.	SELECT ON	E OPTION	I ONLY
FC	OR EACH	ITEM)					
a) Ea	t RED ME	EAT such	as beef or	lamb? Th	is includes	all steaks,	chops,
roa	asts, mince	, stir fries	and casser	oles, but n	ot pork or ch	icken?	
	<1	1-2	3-4	5-6	Once/dav/	2 times/	<u>≥</u> 3
Never	time/	times/	times/	times/	evervdav	dav	times/
	week	week	week	week			day
b) Ea	t MEAT F	PRODUC	<b>FS</b> such as	sausages,	hot dogs, ha	am, devon,	sausage
ro	lls, salami,	meat pies	or bacon?				
	<1	.1-2	3-4	5-6	Once/dav/	2 times/	.≥3
Never	time/	times/	times/	times/	evervdav	dav	times/
	week	week	week	week			day
c) Ea	tt FISH, ind	cluding fre	sh, canned	or frozen	, but not batt	ered or cru	imbed?
[1]	NCLUDES	TUNA					
	<1	.1-2	3-4	5-6	Once/dav/	2 times/	.≥3
Never	time/	times/	times/	times/	evervdav	dav	times/
	week	week	week	week		<i>a.a.j</i>	day
<i>d</i> ) Ha	ave ICE C	REAM, IC	CY POLES	S or ICE 1	BLOCKS?		
	<1	1-2	3-4	5-6	Once/dav/	2 times/	≥3
Never	time/	times/	times/	times/	evervdav	dav	times/
	week	week	week	week			day
<i>e)</i> Ea	t HOT CH	IIPS, FRE	NCH FRI	ES, WED	GES or FRI	ED POTA	TOES?
	<1	1-2	3-4	5-6	Once/dav/	2 times/	≥3
Never	time/	times/	times/	times/	evervdav	dav	times/
	week	week	week	week			day
f) Ea	t POTAT	O CRISPS	S/CHIPS of	or OTHE	R SALTY S	NACKS (s	such as
Tv	visties, cor	n chips)?					
	<1	1-2	3-4	5-6	Once/dav/	2 times/	$\geq 3$
Never	time/	times/	times/	times/	evervdav	dav	times/
	week	week	week	week	0.019.0003	<i>a.a.j</i>	day
g) Ha	ave meals of	or snacks s	uch as BU	RGERS,	PIZZA, CH	ICKEN or	CHIPS
fro	om places l	like McDo	nalds, Hur	ngry Jacks	Burger King	g, Pizza Hı	it, KFC,
Re	ed Rooster	or local tal	ke away fo	od places	?		
	<1	.1-2	3-4	5-6	Once/dav/	2 times/	.≥3
Never	time/	times/	times/	times/	evervdav	dav	times/
- · -	week	week	week	week			day
<i>h)</i> Eat CONFECTIONERY, such as Iollies and chocolates?							
	<1	1-2	3-4	5-6	Once/dav/	2 times/	<u>≥</u> 3
Never	time/	times/	times/	times/	evervdav	dav	times/
	week	week	week	week	- · · · · · · · · · · · · · · · · · · ·	J	day
<i>i)</i> Eat SWEET FOODS, such as sweet biscuits, cakes or muffins?							
	<1	1-2	3-4	5-6	Once/dav/	2 times/	<u>≥</u> 3
Never	time/	times/	times/	times/	evervdav	dav	times/
	week	week	week	week	c ; ci ; uu j	any	day

**10** How often do you add salt to your food either during cooking or after it's cooked? Would you say: [READ OPTIONS] (SELECT ONE ONLY)

	[ <u>KEAD</u> OPTIONS] (SELECT ONE ONL.
0	Never or rarely
0	Sometimes
0	Usually OR
0	Always

11 How many MEALS (including breakfast, lunch and dinner) do you usually eat in a typical day, including evenings? [DO NOT READ] (SELECT ONE ONLY)

	[DU NUT KEAD] (SELECT
0	None
0	1
0	2
0	3
0	4 or more

#### 12 How many times did you snack or eat between meals <u>vesterday</u>?

\_\_\_\_\_ times

**13** On how many days per week do you usually.....

	0	1	2	3	4	5	6	7
a. Have something to eat for breakfast?	0	0	0	0	0	0	0	0
b. Eat lunch?	0	0	0	0	0	0	0	0
c. Eat dinner?	0	0	0	0	0	0	0	0
d. Eat meals in front of the TV?	0	Ο	0	0	0	0	0	0
e. Eat snacks in front of the TV?	0	0	0	0	0	0	0	0
f. Drink any soft drink in front of the TV?	0	0	0	0	0	0	0	0

	Never	Someti mes	Usually OR	Always
a. Fruit	0	0	0	0
b. WHAT ABOUT Vegetables	0	0	0	0
c. Potato chips or other salty snack foods	0	0	0	0
d. Chocolate or other lollies	0	0	0	0
e. Sweet biscuits, cakes or muffins	0	0	0	0
f. Ice cream, icy poles or ice blocks	0	0	0	0
g. Reduced or no-fat milk	0	0	0	0
h. DIET Soft drink, cordial or sports drinks	0	0	0	0
i. REGULAR soft drink, cordial, sports drinks or energy drinks	0	0	0	0

#### **14** About how often is FRUIT available in your HOME? Would you say..... [READ OPTIONS] [CHOOSE ONE OPTION]

#### **15** How confident are you that you could eat healthy foods when you are..... [READ OPTIONS] (SELECT ONE FOR EACH)

	Not at all	Slightl y	Moder ately	Very OR	Extreme ly confiden t
a. At the shops? Would you say you are:	0	0	0	0	0
Using the same scale, how conformed to the same scale, how conformed to the same scale, how conformed to the same scale, how configuration of the same scale, h	onfident a	re you tha	at you coul	d eat hea	lthy
University/TAFE or work?	0	0	0	0	0
c. With your friends?	0	0	0	0	0
d. Feeling down, bored or stressed?	0	0	0	0	0
e. Eating out?	0	0	0	0	0
f. Alone?	0	0	0	0	0

#### 16a About how confident are you that you ....

<b>[DO NOT READ OPTIONS</b>	UNLESS NECESSARY]	(SELECT ONE FOR EACH)
	••••=•=•=•=•	(======================================

	Not at all	Slightly	Moderately	Very	Extremely confident
a. Could prepare and cook vegetables at home? ( <i>includes vegies cooked in</i> <i>microwave</i> )	0	0	0	0	0
b. Could cook a range of dishes?	0	Ο	Ο	0	Ο
c. Could read and understand food labels?	0	0	Ο	0	0
<b>16</b> ł	Do you feel confide	nt cooking:			
-------------	-----------------------	-------------	----		
		Yes	No		
a.	Red meat	0	0		
b.	WHAT ABOUT Chicken	0	0		
c.	Fish	0	0		
d.	Beans and lentils	0	Ο		
e.	Pasta	0	0		

f. Rice

g. Potatoes

# **17 How many times in the past month did you....** [DO NOT READ UNLESS NECESSARY] (CHOOSE ONE OPTION)

0

0

	Never/ rarely	<once <br="">wk</once>	Once/wk	About 2-3 times/ wk	About 4-6 times/ wk	Every- day
a. Shop for food, or help shop for food, for your household?	Ο	Ο	0	Ο	Ο	0
b. Plan ahead what you would eat at meals at home?	0	0	0	0	0	0
c. Make a grocery shopping list for your household?	Ο	Ο	0	0	0	Ο
d. Prepare a meal for your household on your own?	0	Ο	0	Ο	0	0
e. Help to prepare a meal for your household?	0	0	0	0	0	0

0

0

#### How often over the past 6 months have you been hungry because you or 18 your family couldn't afford more food? Would you say ..... (CHOOSE ONE ONLY)

Almost every month	Some months, but not every month	One or two months, OR	You have not been hungry for this reason	I don't want to answer
0	0	0	0	0

#### 19a Are you currently a vegetarian?

0	Yes
0	No- Skip to SECTION 2

19b	As a vegetarian, do you eat any of the following?
(SELECT	AS MANY AS APPLY)

1	/
0	Eggs
0	Dairy food (such as milk or cheese)
0	Chicken
0	Fish
	None of the above

**SECTION 2: Physical activity** 

The next section is about the physical activity you do in your everyday life. We are interested in two types of physical activity – vigorous and moderate. VIGOROUS PHYSICAL ACTIVITY takes hard physical effort and makes you breathe much harder than normal.

MODERATE PHYSICAL ACTIVITY makes you breathe somewhat harder than normal.

### TRANSPORTATION PHYSICAL ACTIVITY

Now, think about how you travel from place to place, including travel to work and to other places like shops and the movies etc.

**20a On how many days in a usual week do you travel in a motor vehicle like a train, bus, car or tram?** [DO NOT READ] (SELECT ONE ONLY)

None	1 day	2 days	3 days	4 days	5 days	6 days	7 days
0	0	0	0	0	0	0	0
SKIP to Q20c							

20b On <u>one of those days</u>, how much time in total do you usually spend *traveling in a car, bus, train or other kind of motor vehicle?* 

\_\_\_\_\_hours per day

and/or \_\_\_\_\_minutes per day

For the next few questions we are only interested in physical activity you do for at least 10 *uninterrupted* minutes.

**20c On how many days in a usual week do you cycle** *to go from place to place* **for at least 10 minutes at a time?** [DO NOT READ] (SELECT ONE ONLY)

None 1 day 2 days 3 days 4 days 5 days 6 days 7 days 0 0 0 0 0 0 0 0 SKIP to *Q20e* 

20d On <u>one of those days</u>, how much time in total do you usually *cycle from place to place*?

\_\_\_\_\_hours per day

and/or \_\_\_\_\_minutes per day

**20e On how many days in a usual week do you walk** *to go from place to place* **for at least 10 minutes at a time?** [DO NOT READ] (SELECT ONE ONLY)

None 1 day 2 days 3 days 4 days 5 days 6 days 7 days 0 0 0 0 Ο 0 0 0 SKIP to Q21a

20f On <u>one of those days</u>, how much time in total did you usually spend walking *from place to place*?

\_\_\_\_\_hours per day

and/or \_\_\_\_\_minutes per day

**RECREATION, SPORT, AND LEISURE-TIME PHYSICAL ACTIVITY** Now, think about all the physical activities that you do in a usual week solely <u>for</u> <u>recreation, sport, exercise or leisure</u>.

21a On how many days in a usual week, do you walk for at least 10 minutes at a time *in your leisure time*, <u>not counting any walking you have already</u> <u>mentioned</u>?

[DO NOT READ] (SELECT ONE ONLY)

None 5 days 1 day 2 days 3 days 4 days 6 days 7 days 0 0 0 0 0 0 0 0 SKIP to *Q21c* 

21b On <u>one of those days</u>, how much time in total do you usually spend *walking in your leisure time?* 

\_\_\_\_\_hours per day

and/or \_\_\_\_\_minutes per day

**21c On how many days in a usual week do you do vigorous physical activities** *in your leisure time* for at least 10 minutes at a time? [DO NOT READ] (SELECT ONE ONLY)

None	1 day	2 days	3 days	4 days	5 days	6 days	7 days
0	0	0	0	0	0	0	0
SKIP to							
Q21e							

## 21d On <u>one of those days</u>, how much time in total do you usually spend doing vigorous physical activities *in your leisure time*?

\_\_\_\_\_hours per day

and/or \_\_\_\_\_minutes per day

**21e On how many days in a usual week do you do <u>moderate</u> physical activities** *in your leisure time* **for at least 10 minutes at a time? [DO NOT READ] (SELECT ONE ONLY)** 

None 1 dav 2 days 3 days 4 days 5 days 6 days 7 days 0 0 0 0 0 0 0 0 SKIP to *Q22a* 

21f On <u>one of those days</u>, how much time in total do you usually spend doing moderate physical activities *in your leisure time*?

\_\_\_\_hours per week

and/or \_\_\_\_\_minutes per day

*HOUSEWORK, HOUSE MAINTENANCE AND CARING FOR FAMILY* Now think about physical activity that you do in a usual week in and around the house

22a On how many days in a usual week do you do physical activities that take moderate or vigorous effort in the garden or in your home for at least 10 minutes at a time. Examples include carrying heavy loads, scrubbing floors, sweeping

[DO NOT READ] (SELECT ONE ONLY)

None	1 day	2 days	3 days	4 days	5 days	6 days	7 days
0	0	0	0	0	0	0	0
SKIP to							
Q23a							

## 22b. On one of those days, how much time in total did you usually spend doing moderate or vigorous activities in the garden or in your home?

\_\_\_\_ hours \_\_\_\_ minutes

23a	Over the past 6 months, have you been	[DO NOT R	EAD OPTIONS]
		Yes	No
a.	A member of a gym or fitness club?	Ο	Ο
b.	A participant in a sporting club or squad?	0	Ο
c.	A participant in any other organised physical activity?	0	0

**23b** Over the past 6 months, how many sports teams have you been part of? (If you play for more than 1 team of the same sport or play across 2 seasons (e.g., two football teams), count this as 2.)

\_\_\_\_\_ teams

•••••	[REAL	) OPTI	<u>'ONS] (CH</u>	OOSE ONE F	OR EA	CH ITEM)
		Not at all	Slightly	Moderately	Very	Extremely confident
a.	You are tired? Would you say	0	0	Ο	0	
On	the same scale, how confid	lent ar	e you that	you could do	physic	al
acti	vity even when you					
b.	You feel sad, stressed or in a bad mood?	0	Ο	Ο	0	Ο
с.	You are on holiday?	0	0	0	0	Ο
d.	You feel you don't have time?	0	0	Ο	0	Ο
e.	It's raining?	0	0	0	0	0

### **24 How confident are you that you could do physical activity even when** ..... [*READ OPTIONS*] (CHOOSE ONE FOR EACH ITEM)

## 25 How many times in the past month did you...

[DO NOT READ OPTIONS UNLESS NECESSARY] (CHOOSE ONE OPTION)

	Never/ rarely	<once <br="">week</once>	Once/ week	About 2-3 times/ week	About 4-6 times/ week	Every- day
a. Set a goal for how much physical activity you would like to do?	Ο	Ο	0	Ο	Ο	Ο
<ul> <li>b. Plan particular days on which you would do physical activity?</li> </ul>	0	0	0	Ο	Ο	0
c. Meet someone to do physical activity with?	0	0	0	0	0	0

**SECTION 3: Sedentary behaviour** 

The next questions are about the time that you spend sitting while you are at work, University, TAFE, at home, while studying and during your free time. This may include time spent sitting at a desk, visiting friends, reading or sitting or lying down to watch TV. Do not include any time spent sitting in a motor vehicle that you have already told me about.

26a In a usual week, how much time in total do you spend *sitting* on a <u>week</u> <u>day</u>? This includes the day and evening. Include sitting at all locations and time spent lying down while you are awake as well as sitting.

\_\_\_\_hours (Mon-Fri)

and/or \_\_\_\_\_minutes (Mon-Fri)

**26b** In a usual week, how much time in total do you spend sitting on a weekend day?

\_\_\_\_\_hours (Sat and Sun)

and/or \_\_\_\_\_minutes (Sat and Sun)

The next questions are about your LEISURE TIME – the time when you are not at work, University/TAFE, travelling or sleeping. These questions ask you to add up time from Monday to Friday and across weekends.

27a What is the <u>total time</u> from Monday to Friday that you usually spend SITTING watching TV, DVDs or videos?

\_\_\_\_hours (Mon-Fri)

and/or \_\_\_\_\_minutes (Mon-Fri)

27b What about on weekends, so Saturday and Sunday?

\_\_\_\_\_hours (Sat and Sun)

and/or \_\_\_\_\_minutes (Sat and Sun)

28a What is the total time from Monday to Friday that you usually spend SITTING using a computer, laptop or tablet such as an IPAD for entertainment, so not for work or study (e.g. streaming)?

\_\_\_\_hours (Mon-Fri)

and/or \_\_\_\_\_minutes (Mon-Fri)

28b What about on weekends, so Saturday and Sunday?

\_\_\_\_hours (Sat and Sun)

and/or \_\_\_\_\_minutes (Sat and Sun)

sequence

#### 29a Do you usually play electronic games such as Nintendo, X-Box or Playstation while SITTING? ALERT: Skip

[DO NOT READ]

Yes 0 No SKIP TO Q30a 0

29b What is the total time from Monday to Friday that you usually spend playing electronic games while SITTING?

\_hours (Mon-Fri)

and/or \_\_\_\_\_minutes (Mon-Fri)

#### **29c** What about on weekends, so Saturday and Sunday?

\_hours (Sat and Sun)

and/or \_\_\_\_minutes (Sat and Sun)

How many TVs do you have in your house? **30a** 

> TV's IF ZERO (0), SKIP TO Q32

ALERT: Skip sequence

**30b** Do you have a TV in your bedroom?

0	Yes	
0	No	

#### 31 How confident are you that you could... [READ OPTIONS]

	Not at all	Slightly	Moderately	Very OR	Extremely confident			
a. Watch less TV even when you are bored?	Ο	0	0	0	0			
Using the same scale, how confident are you that you could								
b. Turn off the TV even when there is a program you enjoy?	Ο	0	0	0	0			
c. Watch less TV even when it is raining?	Ο	0	Ο	0	О			
d. Watch less TV even if others want to watch it?	Ο	0	Ο	0	Ο			
e. Turn off the TV when you are doing something else?	Ο	0	Ο	0	0			
f. Not eat meals while watching TV?	0	0	0	0	Ο			
g. Not eat snacks while watching TV?	Ο	0	Ο	0	Ο			

	Yes	No
a. Mobile phone	0	0
b. Laptop	0	0
c. Tablet/IPAD	0	0
d. Desktop computer [eg. PC or Mac]	0	0
e. Active games console, like a Wii or Kinect	0	0
f. Other games console	0	0
g. Treadmill or other gym equipment	0	0
h. Basketball/netball ring	0	0
i. Bike	0	0
j. Tennis court	0	0
k. Swimming pool	0	0

## **32 Do you have access to the following AT HOME...**

## **33 Do you do any of the following things in your bedroom?** [DO NOT READ OPTIONS]

	[D0]	) NO1
	Yes	No
a. Watch TV shows or movies	0	0
b. Play electronic games	0	0
c. Use the internet	0	0

# **34** Do you agree, strongly agree, disagree, strongly disagree or neither agree nor disagree with the following statements?

[DO NOT READ OPTIONS] (CHOOSE ONE FOR EACH ITEM)

		Strongly agree	Agree	Neither agree nor disagree	Disagree	Strongly disagree	N/A
a.	I enjoy cooking and preparing meals	0	0	0	0	0	0
b.	I enjoy trying new foods	0	0	0	0	0	0
c.	I enjoy the taste of many vegetables	0	0	0	0	0	0
d.	I regularly go without food or other necessities because I cannot afford them	Ο	0	Ο	Ο	Ο	0
e.	I enjoy doing physical activity	0	0	0	0	0	0
f.	I enjoy playing sport	0	0	0	0	0	0
g.	I enjoy going for walks or walking places	0	0	0	0	0	0
h.	I am good at physical activity	0	0	0	0	0	0

i. At home, there are enough supplies and pieces of sports equipment (like       O <th>_</th> <th></th> <th></th> <th></th> <th></th> <th></th> <th>Арр</th> <th>enuices</th>	_						Арр	enuices
j. My parents always pick me up and drop me off when I go places k. I am too busy to eat healthy foods 0 0 0 0 0 0 0 1. I am too rushed in the morning to eat a healthy breakfast m. Eating healthy meals just takes too 0 0 0 0 0 0 0 0 much time 1. I don't have time to think about healthy 0 0 0 0 0 0 0 eating 0. I don't have time to to think about healthy 0 0 0 0 0 0 0 p. Doing physical activity or exercise 0 0 0 0 0 0 0 0 0 r. I don't have time to to rother physical sport sees are too 0 0 0 0 0 0 0 0 0 me r. I don't have enough time to do more physical sport sees are too 0 0 0 0 0 0 0 0 0 we mensely to buy 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		i. At home, there are enough supplies and pieces of sports equipment (like balls, bikes and racquets) to use for physical activity	0	0	0	Ο	0	0
k. I am too busy to eat healthy foods       0       0       0       0       0       0         1. I am too rushed in the morning to eat o a healthy breakfast       0       0       0       0       0       0         m. Eating healthy meals just takes too       0       0       0       0       0       0       0         n. I don't have time to think about healthy       0       0       0       0       0       0       0         o. I don't have time to cook meals       0       0       0       0       0       0       0         p. Doing physical activity or exercise       0       0       0       0       0       0       0         q. Eating a healthy diet is important to me       0       0       0       0       0       0       0         q. Eating a healthy diet is important to me       0       0       0       0       0       0       0         s. Memberships and sports fees are too       0       0       0       0       0       0       0         s. Memberships and sports fees are too       0       0       0       0       0       0       0         with y parents give me money to buy       0       0       0		j. My parents always pick me up and drop me off when I go places	0	0	0	Ο	0	0
1. I am too rushed in the morning to eat a healthy breakfast       0       0       0       0       0       0       0         m. Eating healthy meals just takes too       0       0       0       0       0       0       0         n. I don't have time to think about healthy       0       0       0       0       0       0       0         o. I don't have time to cook meals       0       0       0       0       0       0       0         p. Doing physical activity or exercise       0       0       0       0       0       0       0         q. Eating a healthy diet is important to me       0       0       0       0       0       0       0         q. Eating a healthy diet is important to do more physical activity, exercise or sport       0       0       0       0       0       0         s. Memberships and sports fees are too       0       0       0       0       0       0         s. Memberships and sports fees are too       0       0       0       0       0       0         u. My parents give me money to buy       0       0       0       0       0       0       0         u. My parents help me to pay for sport, gym memberships       0 <td></td> <td>k. I am too busy to eat healthy foods</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td>		k. I am too busy to eat healthy foods	0	0	0	0	0	0
m. Eating healthy meals just takes too O O O O O O O O much time n. I don't have time to think about healthy O O O O O O O O eating o. I don't have time to cook meals O O O O O O O O p. Doing physical activity or exercise O O O O O O O O is important to me q. Eating a healthy diet is important to O O O O O O O O me r. I don't have enough time to do more physical O O O O O O O O sport s. Memberships and sports fees are too O O O O O O O t. My parents give me money to buy O O O O O O O O u. My parents help me to pay for sport, gym memberships O O O O O O O O or other physical activity.		<ol> <li>I am too rushed in the morning to eat a healthy breakfast</li> </ol>	0	Ο	0	0	0	0
<ul> <li>n. I don't have time to think about healthy</li> <li>o. I don't have time to cook meals</li> <li>o. O</li> <li>o. I don't have time to cook meals</li> <li>o. O</li> <li>o. O</li> <li>o. O</li> <li>o. I don't have time to cook meals</li> <li>o. O</li> <li>o. O</li></ul>		m. Eating healthy meals just takes too much time	0	0	0	0	0	0
o. I don't have time to cook mealsOOOOOOp. Doing physical activity or exerciseOOOOOOis important to meOOOOOOq. Eating a healthy diet is important to meOOOOOOr. I don't have enough time to do more physical activity, exercise or sportOOOOOOs. Memberships and sports fees are too me money to buy foodOOOOOOOu. My parents give me to pay for sport, gym membershipsOOOOOOOu. My parents help me to pay for sport, activityOOOOOOO		n. I don't have time to think about healthy eating	0	0	0	0	0	0
<ul> <li>p. Doing physical activity or exercise O O O O O O O O</li> <li>q. Eating a healthy diet is important to O O O O O O O O me</li> <li>r. I don't have enough time to do more physical O O O O O O O O O activity, exercise or sport</li> <li>s. Memberships and sports fees are too O O O O O O O O expensive</li> <li>t. My parents give me money to buy O O O O O O O O food</li> <li>u. My parents help me to pay for sport, gym memberships O O O O O O O O or other physical activity</li> </ul>		o. I don't have time to cook meals	0	0	0	0	0	0
q. Eating a healthy diet is important to meOOOOOr. I don't have enough time to do more physical sportOOOOOOactivity, exercise or sportS. Memberships and sports fees are too expensiveOOOOOOt. My parents give me money to buy foodOOOOOOOOu. My parents help me to pay for sport, gym membershipsOOOOOOOu. My parents help me to pay for sport, gym membershipsOOOOOOOu. My parents help me to pay for sport, gym membershipsOOOOOO		<ul> <li>p. Doing physical activity or exercise is important to me</li> </ul>	Ο	Ο	0	0	0	0
<ul> <li>r. I don't have enough time to do more physical O O O O O O O O activity, exercise or sport</li> <li>s. Memberships and sports fees are too O O O O O O O O expensive</li> <li>t. My parents give me money to buy O O O O O O O O food</li> <li>u. My parents help me to pay for sport, gym memberships O O O O O O O O or other physical activity</li> </ul>		q. Eating a healthy diet is important to me	0	0	0	0	0	0
<ul> <li>s. Memberships and sports fees are too O O O O O O O O expensive</li> <li>t. My parents give me money to buy O O O O O O O O food</li> <li>u. My parents help me to pay for sport, gym memberships O O O O O O O O or other physical activity</li> </ul>		r. I don't have enough time to do more physical activity, exercise or sport	0	0	0	0	0	0
<ul> <li>t. My parents give me money to buy O O O O O O O food</li> <li>u. My parents help me to pay for sport, gym memberships O O O O O O O or other physical activity</li> </ul>		s. Memberships and sports fees are too expensive	Ο	Ο	0	0	0	0
u. My parents help me to pay for sport, gym memberships O O O O O O or other physical activity		t. My parents give me money to buy food	Ο	Ο	0	0	0	0
		u. My parents help me to pay for sport, gym memberships or other physical activity	0	0	0	0	0	0

### **SECTION 4: Social environment**

The next section of the survey is about your family, friends, colleagues and people you live with. First I will ask you about your family. Family members include: parents/step-parents, siblings, cousins, aunts, uncles and grandparents.

[REAL	[READ OPTIONS] (SELECT ONE ONLY)					
	Never	Rarely	A few times	Often OR	Very often	
a. Do physical activity with you? Would you say	0	0	0	0	0	
During the past 6 months, how often did a family member						
b. Encourage you to be physically active?	0	0	0	0	0	
c. Take you to places where you can do physical activity or sport?	0	0	Ο	Ο	0	
d. Discourage you from sitting too much (e.g. watching too much TV)?	0	0	Ο	0	0	
e. Watch TV or DVDs with you?	0	0	0	0	0	
f. Play electronic games with you?	0	0	0	0	0	
g. Encourage you to eat healthy, low-fat foods?	0	0	0	0	0	
h. Discourage you from eating unhealthy foods?	0	0	0	0	0	
i. Encourage you to cook or prepare a meal?	0	0	0	0	0	

#### 35 Over the past 6 months, how often did a family member:

#### 36 How many times per week do you usually eat dinner together with your family?

[DO NOT READ OPTIONS] (SELECT ONE ONLY)

0	Rarely/Never
0	Less than once a week
0	~1-3 times a <u>week</u>
0	~4-6 times a <u>week</u>
0	Everyday
0	N/A

#### 37a Do you live together with family members?

IDO NOT READ	) (SELECT ONE ONLY)	)
DO NOI KLAD	(	/

[DO NOT ]	<u>READJ (SELECT ONE O</u>	
0	Yes <u>SKIP TO</u>	
	<u>Q39a</u>	sequence
0	No	

The next questions are about the people that you live with.

**37b** How often do you usually eat dinner together with all, or most, of the people living in your household? Would you say...

0	I live alone SKIP TO Q39a	
0	Rarely/Never	
0	Less than once a <u>week</u>	
0	~1-3 times a <u>week</u>	ALERT: Skin
0	~4-6 times a <u>week</u>	Sequence
0	Everyday	sequence

### **38 During the past 6 months, how often did <u>people that you live with</u>: [READ OPTIONS] (SELECT ONE ONLY)**

	Neve r	Rare ly	A few time s	Ofte n OR	Very often
a. Do physical activity with you?	0	0	0	0	0
During the past 6 months, how often d	id people	e that yo	ou live v	with	
b. Encourage you to be physically active?	0	0	0	0	0
c. Take you to places where you can be active?	Ο	0	0	0	0
d. Discourage you from sitting too much?	Ο	0	0	0	0
e. Watch TV or DVDs with you?	0	0	0	Ο	0
f. Play electronic games with you?	0	0	0	Ο	0
g. Encourage you to eat healthy, low-fat foods?	Ο	0	0	0	0
h. Discourage you from eating unhealthy foods?	Ο	0	0	0	0
i. Encourage you to cook or prepare a meal?	0	0	0	0	0

### **39a** Do you currently have a boyfriend or girlfriend?

[DO NOT READ] (SELECT ONE ONLY)

0	Yes
0	No <u>SKIP TO Q40</u>

\_\_\_\_

ALERT: Skip
sequence

## **39b** How long have you been in this relationship? \_\_\_\_\_years AND/OR \_\_\_\_\_ months

### **During the past 6 months, how often did your <b>boyfriend or girlfriend:** [READ OPTIONS] (SELECT ONE ONLY) **39c**

[READ OF HONS] (SELEC	I ONE ON	LI)			
	Never	Rarel y	A few tim es	Ofte n OR	Very often
a. Do physical activity with you? Would you say	0	0	0	0	0
During the past 6 months, how often die	d your boy	friend or	r girlfr	iend	•
b. Encourage you to be physically active?	0	0	0	0	0
c. Take you to places where you can be active?	0	0	0	0	0
d. Discourage you from sitting too much?	0	0	0	0	0
e. Watch TV or DVDs with you?	0	0	0	0	0
f. Play electronic games with you?	0	0	0	0	0
g. Encourage you to eat healthy, low-fat foods?	0	0	0	0	0
h. Discourage you from eating unhealthy foods?	0	0	0	0	0
i. Encourage you to cook or prepare a meal?	0	0	0	0	0

## The next questions are about your friends or work colleagues.40 During the past 6 months, how often did friends or work

#### л .

40	During the past 6 months, now often did <u>irlends or work colleagues</u> :	
	[READ OPTIONS] (SELECT ONE ONLY)	
	A Ofte Vor	

		r	ly	time s	n OR	often
a.	Do physical activity with you? <b>Would</b> you say	0	0	0	0	0
D	uring the past 6 months, how often did f	riends o	or work	colleag	ues	
b.	Encourage you to be physically active?	Ο	Ο	Ο	0	Ο
c.	Take you to places where you can be active?	0	0	0	0	0
d.	Discourage you from sitting too much?	Ο	Ο	Ο	0	Ο
e.	Watch TV or DVDs with you?	0	0	0	0	Ο
f.	Play electronic games with you?	Ο	0	0	0	Ο
g.	Encourage you to eat healthy, low-fat foods?	0	0	0	0	0
h.	Discourage you from eating unhealthy foods?	0	0	0	0	0
i.	Encourage you to cook or prepare a meal?	0	0	0	0	0

## **SECTION 5: Neighbourhood context**

The following questions are about your 'neighbourhood' or the area in which you live. Some people may live in two households or residences.

<b>41</b>	<b>Do you h</b> a	ave more	e <b>than one residence?</b>	ALERT: Skip
[DO No	OT READ	] (SELEC	CT ONE ONLY)	sequence
	0	Yes	NOTE THIS ON THE INTERVIEW	W FORM

0	res	NOIE INIS ON THE INTERVIEW FORM
0	No	SKIP TO Q44

## 42 What are the postcodes of the places in which you live?

- a. Postcode 1: \_\_\_\_\_
- b. Postcode 2: \_\_\_\_\_
- c. Postcode 3: \_\_\_\_\_

## 43 In which of these postcodes do you live most of the time?

If this is equal, answer for the household you are living in today.

\_\_\_\_\_

Please answer the next questions for the place you live in most of the time. *SKIP TO Q45* 

## 44 What is your postcode?

\_ \_\_\_ \_\_\_ \_\_\_

45 About how long have you lived at this address?

\_\_\_\_\_ months and/or \_\_\_\_\_ years

46 **Do you currently live in...** [*READ OPTIONS*]

		Small town	Rural property,
Major city	Large town	OR	such as a farm?
0	Ο	0	0

# 47 The next question is about how long it would take to get from your home to the nearest business or facility if you <u>WALKED</u> to them.

		1-5 mins	6-10 mins	11- 20 mins	21- 30 mins	>30 mins	N/A	Don't know
a. N	Milkbar, convenience or small grocery store	0	0	0	0	0	0	0
b. S	Supermarket	0	0	0	0	0	0	0
c. I	Fruit & vegetable shop or narket	0	0	0	0	0	0	0
d. I	Fast food outlet	0	0	0	0	0	Ο	Ο
e. (	Café or restaurant	0	0	0	0	0	0	0
f. I t	Public transport stop (bus, ram, train)	0	0	0	0	0	0	0
g. I	Park	0	0	0	0	0	0	0
h. I	Bike or walking track	0	0	0	0	0	0	0
i. I e g	Indoor recreation or exercise facility, such as a gym	0	0	0	0	0	0	0
j. S	Swimming pool	0	Ο	Ο	0	0	0	0
k. H f	Playing fields, such as Football ovals or tennis courts	0	0	0	0	0	0	0
1. Y	Your University, TAFE or college	0	0	0	0	0	Ο	0
m. \	Your work if you have one	0	0	0	0	0	0	0

About how long would it take to get from your home to the nearest.... [DO NOT READ OPTIONS UNLESS NECESSARY] (SELECT ONE ONLY)

48	Do you agree, strongly agree, disagree, strongly disagree or neither agree
nor	disagree with the following statements about the area in which you live?
	[DO NOT READ OPTIONS] (CHOOSE ONE FOR EACH ITEM)

		Strongly agree	Agree	Neither agree nor disagree	Disagree	Strongly disagree
a.	A large selection of fresh fruits and vegetables is available in my neighbourhood	0	0	0	0	0
b.	The fresh fruits and vegetables in my neighbourhood are of high quality	0	0	0	Ο	Ο
c.	There are many opportunities to purchase fast food in my neighbourhood	Ο	0	0	Ο	Ο
d.	I do not buy many fruits because they cost too much	0	0	0	0	0
e.	I do not buy many vegetables because they cost too much	0	0	0	0	0
f.	Local sports clubs and other facilities in my neighbourhood offer many opportunities to get exercise	0	0	0	0	0
g.	I often see other people walking in my neighbourhood	0	0	0	0	0
h.	I often see other people exercising (for example, jogging, bicycling, playing sports) in my neighbourhood	Ο	0	О	Ο	0
i.	I feel safe walking in my neighbourhood, day or night	0	0	0	0	0
j.	My neighbourhood is safe from crime	0	0	0	0	Ο
k.	People around here can be trusted	0	0	0	0	0
1.	People around here are willing to help their neighbours	0	Ο	0	0	Ο
m	There is a lot of noise in my neighbourhood	0	0	0	0	0
n.	My neighbourhood is attractive	0	0	0	0	0

o. In my neighbourhood it is easy to walk places	0	Ο	0	0	0
p. My neighbourhood has heavy traffic	0	0	0	0	Ο
q. This is a close-knit neighbourhood	0	0	Ο	0	0
r. People in this neighbourhood generally don't get along with each other	0	0	0	Ο	Ο
s. people in this neighbourhood do not share the same values	Ο	Ο	0	0	0
t. It is cheaper for me to buy unhealthy foods than healthy foods	Ο	0	0	0	0
u. Soft drink is good value for money	0	0	0	0	0
v. Fast food is good value for money	Ο	0	0	0	0

### **SECTION 6: Changes in your life**

49	On a scale of 1 to 10, where one means you have "little or no stress" and
10 m	eans you have "a great deal of stress," how would you rate your average
level	of stress during the <u>past month</u> ?
(SEL	ECT ONE ONLY)

		ONLI	)						
1	2	3	4	5	6	7	8	9	10
0	0	Ο	0	Ο	0	0	0	Ο	0

50 Compared to the same time last year, would you say that your average level of stress is currently more, less or about the same?

(SELECT	ONE ONLI)	
	Μ	

More	Less	About the same
0	0	0

**51 How much do you agree or disagree with the following statements?** (CHOOSE ONE FOR EACH ITEM)

		Strongly agree	Agree	Neither agree nor disagree	Disagree	Strongly disagree
a.	I have lots of spare time	0	0	0	0	0
b.	I feel like I am always on the go	0	0	0	0	0
c.	I am dependent on my parents for money	0	0	0	0	0
d.	I have too many competing demands on my time	0	Ο	0	0	0
e.	I am good at managing my time	0	0	0	0	0

52a How much sleep do you usually get at night on weekdays?

\_\_\_\_\_ hours & \_\_\_\_\_ minutes

52b How much sleep do you usually get at night on weekends?

\_\_\_\_\_ hours & \_\_\_\_\_ minutes

**52c** Compared to the same time last year, on average, would you say that you currently get more, less or about the same amount of sleep? (*SELECT ONE ONLY*)

Less	About the same
0	0
	Less O

53 In general, how well do you feel you have adjusted to changes in your life over the last 12 months? [READ OPTIONS] (SELECT ONE ONLY)

Very well	Well	Not well	Not very well at all
0	0	0	0

54a How many adults aged 18 years or over usually live in your household? If you live in more than one residence, answer for the household you spend most of your time in.

If this is equal, answer for the household you are living in today. \_\_\_\_\_ adults

54b How many children or young people aged less than 18 years? (Include yourself if you are under 18 years)

children

#### 54c How many times have you moved house in the last 12 months? times

54d	Can you tell	me who is	presently	living	with y	<u>you</u> ?
(SEI	ΕСТ ΔΥ ΜΔΝΥ	AS APPIV	7)			

(SEL	LEC	T AS MANY A	SAPPLY	<sup>7</sup> )			ALERT: Sk	kin
	0	Live alone	SKIP TO	<u>) Q54</u>	f		seauence	<b>r</b>
	0	Mother						
	0	Father						
	0	Stepmother						
	0	Stepfather						
	0	House maste	er/mistres	S				
	0	Partner/spou	ise					
	0	Boyfriend/g	irlfriend					
	-	Sister(s)		$\rightarrow$	How	,		
	0	many?						
	0	Brother(s)	$\rightarrow$		How many?			
	0	Cousin(s)	$\rightarrow$		How many?			
	0	Friend	$\rightarrow$		How many?			
	0	Own child	$\rightarrow$		How many?			
	0	Other	relative		(Please	specify)		

#### 54e Do you have your own bedroom for yourself?

[DO NOT READ] (SELECT ONE ONLY)

0	Yes	
0	No	

#### 54f Where do you live for the majority of the time?

- 0 Parents home (rent free) Parents home (pay board) 0
- Rent or share rent 0
  - Residence hall 0
- Other relative's home 0
- Other (please 0
  - specify)

\_

#### 55a Are you currently attending secondary school?

0	Yes- SKIP TO Q55c	
0	No	

ALERT: Skip sequence

55h	When did you leave secondary	Confirm coding decision with
JJD 	when the you leave secondary	participant if necessary. Eg. If
school?		participant responds with a tim

scho	ol?		participant responds with a timeframe such as two months ago, ask 'So
	0	Last year This year during Term 1	<i>during Term X?' OR 'Was that Term 3 or Term 4?)</i>
	0	This year, during Term 2	TERM GUIDE: Term 1: Feb-April + April holidays
	0	This year, during Term 3 This year, during Term 4	Term 2: April-June + June/July holidays
	0	This year, after Term 4 exams Refused	Term 3: July-September + Sept/Oct holidays
55c part	-time?	Are you currently studying full or	Term 4: Oct-Nov + study break + exam period Code school holidays as the preceding Term. Eg. July holidays = Term 2).

Full time	Part time	Deferred	No study
0	0	0	0

SKIP to Q56a

SKIP to Q56a

55d How many hours a week do you spend at University or TAFE (e.g. in class and studying on campus)?

- 0 hours/ Week (Off-campus/ online) SKIP to 55e
- 1-5 Hours/ Week SKIP to 55f
- $\circ$  6-10 Hours/ Week-SKIP to 55f
- 11-15 Hours/ Week- SKIP to 55f
- 16-20 Hours/ Week- SKIP to 55f
- 21-25 Hours/ Week- SKIP to 55f
- 26-30 Hours/ Week- SKIP to 55f
- $\circ$  36+ Hours/ Week- SKIP to 55f

55e How many hours, on average, do you usually spend studying (e.g. completing modules/ coursework, ect)?

\_\_\_\_\_ Hours/ week SKIP to Q56a

55f How many hours, on average, do you usually spend studying while not on campus?

\_\_\_\_\_ Hours/ week

## 56a Do you currently have a....

(CAN SELECT BOTH PAID & VOLUNTEER)

- o .....a paid job OR
  - ....a volunteer job? IF <u>ONLY</u> 'VOLUNTEER' SELECTED, <u>SKIP TO Q56c</u>
- I don't work <u>SKIP TO Q57a</u>

56b Do you have an apprenticeship? • Yes • No ALERT: Skip sequence

**56c** Are you currently working on a full-time, part-time or a casual basis? (SELECT AS MANY AS NEEDED)

Full-time	Part-time	OR casual basis	
0	0	0	

### 56d What is your MAIN job outside of school?

## 56e How many hours per week do you spend in this job? \_\_\_\_\_\_ hours per week

 56f
 Do you currently have more than one job?

 o
 Yes

 o
 No

 SKIP TO 056i

ALERT: Skip sequence

- 56g How many jobs do you have? \_\_\_\_\_ jobs
- 56h How many hours in total do you usually spend in these jobs per week?
- 56i When you are at work, which of the following best describes what you do? Would you say....

If you have more than one job, answer for your main job. *[READ OPTIONS] (SELECT ONE ONLY)* 

Mostly standing	Mostly sitting	Mostly walking OR	Mostly heavy labour or physically demanding
			work
0	0	0	0

### 57a Are you currently looking for a... [SELECT AS MANY AS NEEDED]

Casual	job Pa	rt-time job	full-time job	Not looking for a
				job
0		0	Ο	O SKIP to Q57c
				ALERT: Skip
				sequence
57b I	How long have	you been looking	for a job?	_
	-		-	

\_\_\_\_\_ months and/or \_\_\_\_\_\_weeks

## **57c** What are you <u>MAINLY</u> doing this year? (by mainly we mean things that occupy most of your time.

(SELECT AS MANY AS APPLY)

Studying at TAFE
Studying at University
Have a part-time job
Have a full-time job
Looking for a job
Doing an apprenticeship
Volunteering
Doing a traineeship
Defence Forces: Army, navy, air force
Other (please specify)

58 Please indicate in which bracket your current weekly income falls before tax. Include all sources of income, such as work income and government benefits. Is it:

	[READ OPTIONS] (SELECT ONE ONLY)
0	Less than \$50 per week?
0	\$51- \$150 per week?
0	\$151-\$250 per week?
0	\$251-\$350 per week?
0	\$351-\$450 per week?
0	\$451 -\$550 per week? OR
0	More than \$550 per week?
0	Don't know
0	Refused

## 59a How many close friends would you say you have? \_\_\_\_\_\_ friends

## 59b Compared to the same time last year, would you say your network or close friends; increased, decreased or stayed about the same?

Increased	Decreased	Stayed about the same
0	0	Ο

## **59c On average, how much time do you usually spend during a typical week socializing with friends?** [DO NOT READ OPTIONS] (SELECT ONE ONLY)

0	None
0	Less than 1 hour
0	1-2 hours
0	3-5 hours
0	6-10 hours
0	11-15 hours
0	16-20 hours
0	Over 20 hours

# 59d Compared to the same time last year, would you say the time you usually spend socializing with friends increased, decreased or stayed about the same?

Increased	Decreased	Stayed about the same
0	0	0

#### 60 What is your date of birth? 19 Dav Month Year **4** April 5 May 1 January 2 February 3 March 6 June 12 December 7 July 8 August **9** September 10 October 11 November **61**a Do you currently have a..... [READ OPTIONS] (SELECT AS MANY AS APPLY) Probationary or full Driver's licence? 0

- Probationary or full Motorbike licence?Car for transport?
- Motorbike for transport?

61b	What is your main form of transport?	[READ OPTIONS] (SELECT
ONE	ONLY)	

Public transport	Car	Walking/cycling	Refused
0	0	0	0

61c Compared to last year, do you spend more, less or about the same time travelling by the following modes? [READ OPTIONS] (SELECT ONE PER LINE)

	More	Less	About the same
Public transport	0	0	0
Car (as driver)	0	0	0
Car (as passenger)	0	Ο	0
Walk	0	0	0
Bike	0	Ο	0

The last few questions are about your general health.

### 62a What is your current height, without shoes?

- \_\_\_\_\_ metres & \_\_\_\_\_ cm
- 62b What is your current weight, without shoes?

\_\_\_\_\_ kg

### 63a Are you currently trying to:

[READ OPTIONS] (SELECT ONE ONLY)

С	)	Lose weight	
С	)	Gain weight	
С	)	Maintain your current weight	
С	)	Not doing anything for your	
		weight	

## 64a Do you have a serious illness, long-term injury or disability that prevents you from being physically active?

0	Yes	
0	No	SKIP TO Q65

ALERT: Skip	
sequence	

### 64b What is the illness, long-term injury or disability?

(ALLOW FOR 70 CHARACTERS)

### 65 Would you say your health is: [READ OPTIONS] (SELECT ONE ONLY)

0	Excellent
0	Very good
0	Good
0	Fair OR
0	Poor

## 66 Are you willing for us to contact you about further research in the future?

0	Yes	
0	No	

### **SIGN-OFF**

That is the end of the survey. Thank you very much for your participation and for all the information you have provided. We really appreciate it. In a couple of weeks, we will send you a \$25 gift voucher as a token of our appreciation.

This was the final survey as part of ProjectADAPT. We cannot thank you enough for your involvement in this study as without you it's not possible. Have a great day/night.

# **Appendix 4.2**Sample profile (n=1 022) according to remoteness andarea-level SEP

	Remoteness		Area-level SEPs			
	Urban	Rural	Lowest	Mid	Highest	
	%	%	%	%	%	
(n)	(721)	(301)	(243)	(307)	(471)	
Recruitment method,						
schools, $n=1$ 022	31.6***	51.2	48.6***	41.0	29.3	
Survey delivery mode,						
online, n=1 022	95.7***	85.0	88.1**	92.5	94.9	
Age (mean years±SD,						
n=1 008)	16.9±0.4	$16.8 \pm 0.4$	16.9±0.4	$16.8 \pm 0.5$	16.9±0.4	
Gender, girl, n=1 021	75.4	70.1	70.4	73.5	75.8	
Remoteness, urban,						
<u>n=1 022</u>	-	-	39.5***	65.8	89.6	
Tertiles of area-level SEP, n	=1 020					
Lowest	13.3***	48.8	-	-	-	
Mid	28.1***	34.9	-	-	-	
Highest	58.6***	16.3	-	-	-	
Parent tertiary educated						
Mother, <i>n</i> =1 018	56.9***	48.5	44.0***	47.2	64.5	
Father, <i>n</i> =1 017	49.4***	30.6	26.7***	35.5	58.0	
Employment						
Not employed, $n=1$ 016	46.2***	30.6	35.8	42.3	43.9	
Casual, <i>n</i> =1 014	38.7	51.2	46.5	42.7	40.1	
Part-time, <i>n</i> =1 014	19.8	23.3	23.0	17.6	21.9	
Income <\$150/wk,						
n=805	67.3	71.1	69.5***	63.5	70.9	
Birth country (Australia)	1		1			
Participant, $n=1$ 015	80.9***	95.0	86.8	84.2	84.6	
Mother, $n=1 011$	52.0***	84.7	66.3***	63.5	58.0	
Father, $n=1\ 004$	50.9***	85.7	66.7	59.9	59.0	
English as primary						
language spoken at home,		–				
n=1 018	80.4***	96.7	82.7	86.3	85.8	

\*\*p<0.01, \*\*\*p<0.001: Continuous variables by binary subgroups used independent t-test or one-way ANOVA for the non-binary subgroup, and categorical variables used Pearson's  $\chi^2$  test of significance.

SD: standard deviation.

	Survey delivery mode		<b>Recruitment method</b>	
	Telephone	Online	Schools	Social media
n	76	947	383	640
Recruitment method,				
schools, $n=1$ 022	93.4***	32.9	-	-
Survey delivery mode,				
online, n=1 022	-	-	81.4***	99.2
Age (mean years±SD,				
n=1 008)	16.5±0.4***	16.9±0.4	16.8±0.4***	$16.9 \pm 0.4$
Gender (%), girls, n=1 022	73.7	73.9	69.9*	76.2
Remoteness (%), urban,				
<i>n</i> =1 022	40.8***	72.9	<b>59.7</b> ***	77.0
<i>Tertiles of area</i> $SEP(\%)$ , $n=1$	020			
Lowest	38.2**	22.6	30.9***	19.6
Mid	30.3**	30.1	33.0***	28.3
Highest	31.6**	47.3	36.1***	52.1
Parent tertiary educated (%)				
Mother $n-1.018$	46 1	55 1	53.4	55.0
Father $n=1.017$	39.5	44 2	39.8	463
1 unior, <i>n</i> =1 017	57.5	11.2	57.0	10.5
<i>Employment (%), n=1 016</i>			1	
Not employed	34.2	42.2	46.3*	38.8
Casual, $n=1 014$	44.7	42.2	39.5	44.1
Part-time, <i>n</i> =1 014	18.4	21.0	17.8	22.7
Income <\$150/wk, n=805	84.2	67.1	68.1	68.6
Birth country (Australia) (%)				
Participant, <i>n</i> =1 015	84.2	85.1	83.9	85.7
Mother, $n=1 011$	75.0*	60.6	64.7	59.8
Father, $n=1\ 004$	73.7*	60.1	63.6	59.7
English as primary				
language spoken at home				
(%), <i>n</i> =1 018	88.2	85.0	84.6	85.6

**Appendix 4.3** Sample profile (*n*=1 022) according to survey delivery mode and recruitment method

p<0.05, p<0.01, p<0.001: Independent t-test for continuous variables between binary subgroups or Pearson's chi-squared test of significance.

SD: standard deviation.

Survey items		Карра	Weighted kappa	ICC	Percent agreement
		±SE	±SE	(95%CI)	(%)
PA variables					
LTPA:					
MPA	(day/wk)	-	$0.46 \pm 0.07$	0.78 (0.65, 0.86)	85.7
	(mins/day)	-	-	0.57 (0.33, 0.72)	-
VPA	(day/wk)	-	$0.70\pm0.07$	0.88 (0.81, 0.92)	90.2
	(mins/day)	-	-	0.79 (0.67, 0.86)	-
Walking	(day/wk)	-	$0.46 \pm 0.07$	0.69 (0.52, 0.80)	85.5
	(mins/day)	-	-	0.59 (0.36, 0.73)	-
AT:					
Cycling	(day/wk)	-	$0.68 \pm 0.08$	0.85 (0.77, 0.90)	96.6
	(mins/day)	-	-	0.39 (0.05, 0.60)	-
Walking	(day/wk)	-	0.59±0.08	0.84 (0.76, 0.90)	83.6
	(mins/day)	-	-	0.66 (0.47, 0.78)	-
School PA:					
Main activity level of	luring school breaks (%)	-	0.65±0.08	0.84 (0.76, 0.90)	90.0
PE classes/wk (%)		-	$0.82 \pm 0.08$	0.94 (0.91, 0.96)	95.4
Participate in school sport (%)		0.76±0.11	_	-	88.0
School sport teams count (%)		0.62±0.11	-	-	75.0
Occupational PA:					
Main activity level a	tt work (%)	$0.54\pm0.09$	-	-	74.1

## Appendix 4.4 Individual survey items' test-retest reliability scores

Survey items		Kappa ±SE	Weighted kappa ±SE	ICC (95%CI)	Percent agreement (%)
SB variables				· · · · ·	
SB (excluding transport)	(h/wkday)	-	-	0.60 (0.37, 0.74)	-
	(h/wkend day)	-	-	0.74 (0.59, 0.83)	-
Recreational screen-time:					
Computer, laptop or tablet	(h/wkday)	-	-	0.63 (0.43, 0.76)	-
	(h/wkend day)	-	-	0.78 (0.65, 0.86)	-
E-games	(h/wkday)	-	-	0.47 (0.18, 0.66)	-
	(h/wkend day)	-	-	0.04 (-0.5, 0.38) <sup>NS</sup>	-
TV, DVD, videos	(h/wkday)	-	-	0.65 (0.46, 0.78)	-
	(h/wkend day)	-	-	0.67 (0.48, 0.79)	-
Sedentary transport	(day/wk)	-	0.55±.07	0.83 (0.73, 0.89)	88.6
	(h/day)	-	-	0.34 (-0.02, 0.58)	-
School SB:			· ·		·
Main activity level during school breaks (%)		-	0.65±0.08	0.84 (0.76, 0.90)	90.0
Occupational SB:				,	
Main activity level at work	(%)	0.54±0.09	-	_	74.1

				Percent
Survey items	Kappa	Weighted kappa	ICC	agreement
	±SE	±SE	(95%CI)	(%)
Individual-level variables (scores)				
PA enjoyment	-	$0.68 \pm 0.08$	0.77 (0.65, 0.85)	90.1
	-	0.73±0.08	0.76 (0.63, 0.84)	91.7
	-	0.51±0.07	0.57 (0.34, 0.72)	88.9
PA goal setting	-	0.54±0.07	0.86 (0.78, 0.91)	84.1
	-	0.57±0.07	0.80 (0.70, 0.87)	84.6
	-	0.59±0.08	0.70 (0.54, 0.81)	88.8
PA competence	-	0.72±0.08	0.75 (0.62, 0.84)	90.1
PA self-efficacy	-	0.50±0.07	0.81 (0.71, 0.88)	82.5
	-	0.59±0.07	0.82 (0.73, 0.89)	84.5
	-	0.47±0.07	0.75 (0.61, 0.84)	82.6
	-	0.41±0.07	0.71 (0.55, 0.81)	81.4
	-	0.59±0.07	0.84 (0.75, 0.89)	84.8
TV avoidance self-efficacy	-	0.52±0.07	0.68 (0.51, 0.80)	82.4
	-	0.49±0.07	0.70 (0.54, 0.81)	83.1
	-	0.56±0.07	0.74 (0.60, 0.83)	83.3
	-	0.44±0.08	0.63 (0.43, 0.76)	79.1
	-	0.50±0.08	0.71 (0.55, 0.81)	82.1

Survey items	Kappa	Weighted kappa	ICC	Percent agreement
	±SE	±SE	(95%Cl)	(%)
Social-level variables (scores)				
Family:				
E-games co-participation	-	0.37±0.08	0.58 (0.36, 0.73)	86.0
PA co-participation	_	0.61±0.07	0.71 (0.55, 0.81)	85.9
TV/DVDs co-participation	_	$0.48 \pm 0.07$	0.72 (0.57, 0.82)	82.2
PA social support	-	0.61±0.07	0.86 (0.78, 0.91)	85.9
		$0.47 \pm 0.07$	0.75 (0.61, 0.84)	80.3
	-	$0.60{\pm}0.08$	0.89 (0.83, 0.93)	86.3
SB discouragement	-	$0.47 \pm 0.07$	0.71 (0.55, 0.81)	85.5
Friends/colleagues:				
E-games co-participation	-	$0.50{\pm}0.08$	0.39 (0.06, 0.61)	84.4
PA co-participation	-	$0.43 \pm 0.08$	0.79 (0.67, 0.86)	79.7
TV/DVDs co-participation	-	0.35±0.08	0.13 (-0.35, 0.43)	79.4
PA social support	-	$0.43 \pm 0.08$	0.76 (0.62, 0.84)	79.7
		0.34±0.07	0.67 (0.49, 0.79)	78.4
	-	$0.37{\pm}0.08$	0.64 (0.44, 0.77)	84.1
SB discouragement	-	0.33±0.08	0.15 (-0.32, 0.45)	86.6
Social network count	-	-	0.96 (0.93, 0.97)	-
	-	-	0.93 (0.89, 0.96)	-
	-	-	$0.00 (-0.56, 0.36)^{NS}$	-
Gym membership	0.76±0.11	-	-	91.5

				Percent
Survey items	Карра	Weighted kappa	ICC	agreement
	±SE	±SE	(95%CI)	(%)
Environment-level variables (scores)				
Home environment:				
E-devices, no. of	$0.00 \pm 0.00$	-	-	98.8
	0.55±0.11	-	-	96.3
	0.88±0.11	-	-	95.1
	0.57±0.11	-	-	82.7
	0.81±0.11	-	-	92.6
	0.70±0.11	-	-	85.2
PA equipment, no. of	0.78±0.11	-	-	88.9
	0.78±0.11	-	-	88.9
	0.70±0.11	-	_	88.9
	0.36±0.11	-	-	92.6
	0.79±0.11	-	-	91.4
TVs, no. of	-	-	0.96 (0.94, 0.98)	-
Neighbourhood environment:				
Noise	-	0.30±0.08	0.44 (0.14, 0.64)	80.0
Walking environment	-	0.19±0.08	0.48 (0.19, 0.66)	75.9
	-	0.46±0.08	0.69 (0.53, 0.80)	81.6
	_	0.29±0.08	0.69 (0.52, 0.80)	78.8
	-	0.36±0.08	0.46 (0.16, 0.65)	80.6

Survey items	Kappa +SE	Weighted kappa	<b>ICC</b>	Percent agreement
Safety		0.50+0.08		80.9
Salety		0.55+0.08	0.60 (0.39, 0.74)	81.9
Social cohesion	-	0.45±0.09	0.71 (0.55, 0.81)	77.8
	-	0.46±0.09	0.65 (0.46, 0.77)	79.1
Land use mix diversity	-	0.76±0.08	0.94 (0.91, 0.96)	93.5
	-	0.71±0.07	0.92 (0.88, 0.95)	89.4
	-	0.54±0.07	0.81 (0.70, 0.87)	87.9
	-	0.59±0.07	0.80 (0.70, 0.87)	87.3
	-	0.65±0.07	0.85 (0.76, 0.90)	87.2
	-	0.63±0.08	0.85 (0.77, 0.90)	93.1
	-	0.68±0.07	0.89 (0.83, 0.93)	92.9
	-	0.80±0.07	0.94 (0.91, 0.96)	93.8
Recreation facilities	-	0.62±0.08	0.86 (0.78, 0.91)	90.9
	-	$0.67 \pm 0.08$	0.88 (0.82, 0.93)	89.1
	-	0.68±0.08	0.89 (0.83, 0.93)	91.3
	-	0.58±0.07	0.85 (0.76, 0.90)	87.1
	-	0.65±0.08	0.83 (0.74, 0.89)	90.4

CI: confidence interval; AT: active transport; DVDs: digital versatile discs; e-devices: electronic devices; e-games: electronic games; h: hours; ICC: internal correlation coefficient; LTPA: leisure-time physical activity; min: minutes; MPA: moderate-intensity physical activity; no.: number; NS: not significant; PA: physical activity; PE: physical education; SB: sedentary behaviour; SE: standard error; TV: television; VPA: vigorous-intensity physical activity; wk: week.

	Remot	eness	1	Area-level SEP		
	Urban	Rural	Lowest	Mid	Highest	
Individual-level variables (scores)						
PA enjoyment	11.7±2.6	$11.8 \pm 2.6$	11.6±2.7	$11.8 \pm 2.6$	$11.8 \pm 2.6$	
PA goal setting	8.6±4.0	8.7±3.6	8.1±3.6**	$9.0{\pm}3.8$	$8.7 \pm 4.0$	
PA competence	3.5±1.2*	3.5±1.3	3.4±1.3	3.5±1.2	3.6±1.1	
PA self-efficacy	14.2±5.0	$14.7 \pm 5.0$	13.7±4.9	$14.5 \pm 5.0$	$14.6 \pm 5.0$	
TV avoidance self-efficacy	16.1±4.7	$15.9 \pm 4.7$	15.2±4.9	15.6±4.6	$16.8 \pm 4.6$	
Social-level variables Family (scores):	1	1 < 1 0		10.10	15.00	
E-games co-participation	1./±1.1*	$1.6 \pm 1.0$	1.8±1.0**	1.8±1.2	1.5±0.9	
PA co-participation	2.6±1.3	$2.7 \pm 1.3$	2.5±1.2*	$2.7 \pm 1.3$	$2.7 \pm 1.3$	
TV/DVDs co-participation	3.4±1.2**	3.4±1.1	3.4±1.2	$3.5 \pm 1.2$	$3.3 \pm 1.2$	
PA social support	9.4±3.3	9.8±3.2	9.0±3.3	$9.7 \pm 3.2$	$9.7 \pm 3.2$	
SB discouragement	3.1±1.3	$3.0 \pm 1.3$	3.0±1.3	$3.1 \pm 1.3$	$3.0 \pm 1.3$	
Friends/colleagues (scores):						
E-games co-participation	2.1±1.3	2.1±1.3	2.0±1.2	$2.2 \pm 1.3$	$2.0{\pm}1.2$	
PA co-participation	$2.8 \pm 1.4$	$3.0{\pm}1.4$	2.9±1.4	$2.9 \pm 1.4$	2.9±1.3	
TV/DVDs co-participation	2.8±1.2	$2.8 \pm 1.2$	2.8±1.2	$2.8{\pm}1.2$	$2.8 \pm 1.2$	
PA social support	7.4±3.2	$7.8 \pm 3.2$	$7.5 \pm 3.2$	$7.5 \pm 3.4$	$7.6 \pm 3.1$	
SB discouragement	$1.7{\pm}1.0$	$1.7{\pm}1.0$	$1.8{\pm}1.1$	$1.8 \pm 1.1$	$1.7\pm0.9$	
Social network count	11.6±9.8	12.5±10.6	12.7±12.9	$10.9 \pm 7.3$	12.0±9.9	
Gym membership	27.4%	30.9%	28.4%	26.1%	29.9%	

## Appendix 4.5 Baseline distribution (mean±SD) for independent variables according to remoteness and area-level SEP

	Remote	eness	Area-level SEP			
	Urban	Rural	Lowest	Mid	Highest	
Environmental-level variables (scores) Home environment:						
E-devices. no. of	4.6±1.2	$4.4{\pm}1.1$	4.5±1.2	$4.6 \pm 1.2$	$4.6 \pm 1.1$	
PA equipment, no. of	1.9±1.2***	$2.4{\pm}1.2$	$2.2 \pm 1.2$	2.1±1.2	$2.0{\pm}1.2$	
TV, no. of	2.5±1.3**	$3.0 \pm 3.6$	2.9±4.0*	2.8±1.5	2.4±1.2	
Neighbourhood environment:	'					
Noise	2.6±1.0***	2.3±1.0	2.6±1.0	2.6±1.0	$2.5 \pm 1.0$	
Walking environment	16.0±2.8***	15.1±3.4	15.4±2.9**	$15.3 \pm 3.2$	$16.3 \pm 2.8$	
Safety	$7.2{\pm}1.9$	7.5±1.9	6.9±2.0***	7.1±2.0	7.7±1.7	
Social cohesion	7.2±1.6**	7.7±1.7	7.0±1.7***	7.2±1.8	7.7±1.5	
Land use mix diversity	3.4±2.1***	$2.0\pm2.3$	2.6±2.3***	2.7±2.3	$3.4 \pm 2.2$	
Recreation facilities	2.6±1.4***	2.0±1.6	2.3±1.6***	2.1±1.5	$2.6 \pm 1.4$	

\*p<0.05, \*\*p<0.01, \*\*\*p<0.001: Pearson's  $\chi^2$  test of significance by remoteness or area-level SEP for categorical variables; independent t-test for continuous variables by remoteness; or one-way ANOVA for continuous variables by area-level SEP.

DVD: digital versatile disc; e-devices: electronic devices; e-games: electronic games; PA: physical activity; SB: sedentary behaviour; SD: standard deviation; SEP: socio-economic position; TV: television.

	Gender		<b>Remoteness</b> (Rural <i>vs</i> urban)		Area-level SEP		
	(Women vs n	nen)			(Mid vs lowest, highest v	s lowest)	
	B (95%CI)	р	B (95%CI)	р	B (95%CI)	р	
Discretionary PA (mins/day)	-2.9 (-14.8, 9.1)	0.638	-1.4 (-12.6, 9.7)	0.803	-3.1 (-17.3, 11.1)	0.666	
					9.4 (-3.5, 22.4)	0.153	
LTPA (mins/day)	1.8 (-7.2, 10.8)	0.697	2.3 (-6.0, 10.7)	0.585	-0.9 (-11.6, 9.7)	0.862	
					4.8 (-4.9, 14.5)	0.330	
Walking	2.0 (-2.0, 6.0)	0.318	2.1 (-1.6, 5.8)	0.269	1.0 (-3.7, 5.8)	0.674	
					1.6 (-2.7, 5.9)	0.469	
Other MPA	-1.1 (-5.2, 3.0)	0.594	1.1 (-2.7, 4.9)	0.575	-1.8 (-6.6, 3.1)	0.470	
					0.1 (-4.3, 4.5)	0.951	
VPA	0.9 (-3.8, 5.7)	0.699	-0.5 (-4.9, 3.9)	0.823	0.3 (-5.3, 5.9)	0.922	
					3.2 (-2.0, 8.3)	0.228	
Active transport (mins/day)	-4.1 (-10.2, 2.0)	0.187	-3.0 (-8.7, 2.7)	0.298	-1.9 (-9.2, 5.3)	0.599	
					5.0 (-1.6, 11.6)	0.140	
Cycling for transport	0.9 (-3.1, 1.2)	0.382	-1.0 (-3.0, 0.9)	0.304	-1.7 (-4.2, 0.8)	0.182	
					1.6 (-0.7, 3.9)	0.173	
Walking for transport	-3.2 (-8.7, 2.4)	0.266	-2.0 (-7.2, 3.2)	0.454	-0.2 (-6.9, 6.4)	0.947	
					3.4 (-2.7, 9.5)	0.271	
Total SB (h/day)	-0.3 (-0.9, 0.3)	0.305	-0.9 (-1.5, -0.4)	0.001	0.1 (-0.6, 0.9)	0.689	
					0.5 (-0.1, 1.2)	0.103	
Sitting (excluding transport)	-0.2 (-0.8, 0.3)	0.383	-0.4 (-0.9, 0.1)	0.123	0.0 (-0.6, 0.7)	0.976	
					0.4 (-0.2, 1.0)	0.202	
Sedentary transport	-0.1 (-0.3, 0.1)	0.360	-0.5 (-0.7, -0.3)	0.000	0.1 (-0.1, 0.3)	0.356	
					0.1 (-0.1, 0.3)	0.189	

**Appendix 7.1** Time by gender, remoteness or area-level SEP interactions for changes in physical activity and sedentary behaviour between baseline and the two-year follow-up<sup>1</sup>

	Gender		Remoteness		Area-level SEP		
	(Women vs n	(Women vs men)		(Rural vs urban)		(Mid vs lowest, highest vs lowest)	
	B (95%CI)	р	B (95%CI)	р	B (95%CI)	р	
Total recreational screen time (h/day)	-0.1 (-0.7, 0.4)	0.575	-0.5 (-1.0, -0.1)	0.027	0.2 (-0.4, 0.8)	0.471	
					0.6 (0.1, 1.2)	0.020	
Watching TV, DVDs or videos	-0.1 (-0.4, 0.2)	0.418	-0.2 (-0.4, 0.1)	0.202	0.3 (0.0, 0.7)	0.045	
-					0.5 (0.2, 0.8)	0.001	
Computer, laptop or tablet	0.0 (-0.3, 0.3)	0.841	-0.2 (-0.5, 0.1)	0.178	-0.1 (-0.4, 0.3)	0.733	
					0.2 (-0.1, 0.5)	0.206	
E-games	0.0 (-0.1, 0.1)	0.901	-0.1 (-0.2, -0.0)	0.030	-0.0 (-0.1, 0.1)	0.909	
-					0.1 (-0.1, 0.2)	0.350	

<sup>1</sup> Baseline to 1-year follow-up data not shown and there were two p-values <0.05 for time by area-level SEP highest vs lowest tertiles for discretionary PA and total recreational screen time.

*B:* correlation coefficient; CI: confidence interval; DVD: digital versatile disc; e-games: electronic games; h: hours; LTPA: leisure-time physical activity; mins: minutes; MPA: moderate-intensity physical activity; PA: physical activity; SB: sedentary behaviour; TV: television; VPA: vigorous-intensity physical activity.
Situational transition at the 2-year follow-up		Changes in discretionary PA			Changes in total SB				
n (%)		T2-1 ( <i>p<sup>a</sup></i> )	<b>T3-1</b> ( <i>p<sup>a</sup></i> )	AIC <sup>b</sup>	AIC <sup>a</sup>	T2-1 ( <i>p<sup>a</sup></i> )	T3-1 ( <i>p<sup>a</sup></i> )	AIC <sup>b</sup>	AIC <sup>a</sup>
Tertiary student status				24312.47	24313.12			10420.75	10395.48
Yes	631 (79.3)	0.153	0.087			0.405	0.000		
No (ref.)	165 (20.7)								
Work hour				24479.00	24479.98			10511.66	10465.45
≥20 h/wk	219 (27.2)	0.425	0.584			0.866	0.000		
<20 h/wk	401 (49.9)	0.552	0.789			0.813	0.602		
Not working (ref.)	184 (22.9)								
Lived with parents status				24160.34	24162.50			10411.27	10407.56
Yes	638 (80.9)	0.183	0.662			0.659	0.009		
No (ref.)	167 (19.1)								
Study and work combinations				24283.21	24289.84			10390.20	10367.09
Worked exclusively	145 (18.2)	0.775	0.900			0.624	0.276		
Studied exclusively	163 (20.5)	0.729	0.612			0.728	0.349		
Concurrently studied and worked	468 (58.9)	0.848	0.446			0.896	0.718		
Neither studied nor worked (ref.)	19 (2.4)								
Studied and lived with parents status				24002.37	24005.29			10729.23	10701.49
Studying while living with parents	493 (62.8)	0.082	0.071			0.561	0.000		
Studying while living independently	127 (16.2)	0.779	0.357			0.495	0.177		
Not studying (ref.)	165 (21.0)								
Worked and lived with parents status				24244.32	24247.99			10840.00	10834.58
Working while living with parents	515 (64.8)	0.727	0.498			0.818	0.024		
Working while living independently	96 (12.1)	0.085	0.659			0.640	0.001		
Not working (ref.)	184 (23.1)								

Appendix 7.2 Interactions between situational transitions and changes in discretionary physical activity and total sedentary behaviour

<sup>*a*</sup> Situational transition was an interaction term in models; <sup>*b*</sup> Situational transition was an independent variable in models. Mixed-effects multi-level linear regression models adjusted for English as the primary language at home, age, gender and survey mode, and accounted for school clusters.

**Appendix 8.1** Interaction effects<sup>1</sup> between baseline levels of the independent variables and situational transitions on discretionary physical activity (mins/day) after leaving secondary school

	Tertiary student	status	Working≥20 h/wk		Living with parents status		
	(Ref: no)		(Ref: working <2	(Ref: working <20 h/wk)		(Ref: no)	
	B (95%CI) <sup>1</sup>	р	B (95%CI) <sup>1</sup>	р	B (95%CI) <sup>1</sup>	р	
Individual-level variables (scores)							
PA enjoyment	-12.5 (-38.3, 13.4)	0.345	24.8 (-0.3, 49.9)	0.053	8.4 (-18.9, 35.8)	0.546	
PA goal setting	-19.2 (41.1, 2.8)	0.087	18.7 (-0.9, 38.2)	0.061	5.0 (-17.1, 27.2)	0.655	
PA competence	-10.5 (-32.3, 11.3)	0.346	14.7 (-5.0, 34.3)	0.144	6.3 (-15.6, 28.2)	0.571	
PA self-efficacy	-14.6 (-36.2, 7.1)	0.187	27.4 (8.1, 46.6)	0.005	18.9 (-2.9, 40.7)	0.089	
TV avoidance self-efficacy	12.9 (-8.9, 34.6)	0.247	6.2 (-13.3, 25.8)	0.532	-0.0 (-22.6, 22.5)	0.999	
Social-level variables							
Family (scores):							
E-games co-participation	-10.0 (-52.3, 32.3)	0.642	20.0 (-17.9, 57.9)	0.302	25.4 (-18.2, 69.0)	0.254	
PA co-participation	-9.7 (-34.7, 15.3)	0.446	31.3 (9.3, 53.4)	0.005	7.2 (-16.6, 31.0)	0.554	
TV/DVDs co-participation	-20.3 (-42.1, 1.5)	0.068	3.3 (-16.3, 23.0)	0.738	2.1 (-19.8, 24.0)	0.850	
PA social support	1.3 (-20.2, 22.8)	0.906	8.4 (-10.9, 27.7)	0.392	14.8 (-7.4, 36.9)	0.192	
SB discouragement	-27.0 (-49.6, -4.5)	0.019	-5.5 (-25.7, 14.8)	0.598	-22.7 (-45.2, -0.3)	0.047	
Friends/colleagues (scores):							
E-games co-participation	-11.8 (-39.5, 15.8)	0.402	21.0 (-6.0, 48.1)	0.127	40.3 (5.1, 75.4)	0.025	
PA co-participation	-0.3 (-22.3, 21.6)	0.976	3.6 (-16.5, 23.6)	0.728	11.0 (-11.4, 33.5)	0.335	
TV/DVDs co-participation	-3.7 (-26.4, 19.0)	0.750	10.6 (-9.8, 31.1)	0.308	24.8 (0.9, 48.7)	0.042	
PA social support	-3.3 (-27.0, 20.4)	0.786	4.3 (-17.6, 26.2)	0.700	21.4 (-3.5, 46.4)	0.092	
SB discouragement	-33.9 (-79.7, 11.8)	0.146	62.6 (21.8, 103.4)	0.003	9.4 (-33.6, 52.5)	0.667	
Social network count	3.5 (-18.1, 25.1)	0.750	-4.0 (-23.3, 15.4)	0.689	-1.5 (-23.4, 20.4)	0.891	
Gym membership	8.1 (-15.4, 31.7)	0.497	13.7 (-6.9, 34.3)	0.193	17.6 (-5.7, 41.0)	0.139	

## Appendix 8.1 Continued

	Tertiary study	y status	Working ≥20 h/wk		Living with parents status	
	(Ref: no)		(Ref: working <20 h/wk)		(Ref: no)	
	B (95%CI) <sup>1</sup>	р	B (95%CI) <sup>1</sup>	р	B (95%CI) <sup>1</sup>	р
Environmental-level variables						
Home environment:						
E-devices, no. of	-7.3 (-35.0, 20.4)	0.606	26.2 (1.7, 50.8)	0.036	11.1 (-16.1, 38.4)	0.423
PA equipment, no. of	15.8 (-6.0, 37.7)	0.156	1.9 (-17.9, 21.7)	0.850	22.4 (0.2, 44.6)	0.048
TVs, no. of	-23.5 (-45.2, -1.8)	0.034	9.1 (-10.6, 28.7)	0.365	-0.3 (-22.7, 22.1)	0.979
Neighbourhood environment (scores):						
Noise	-5.9 (-32.5, 20.7)	0.663	4.7 (-19.8, 29.2)	0.707	1.8 (-24.9, 28.6)	0.894
Walking environment	-1.3 (-32.7, 30.1)	0.935	15.2 (-12.5, 42.9)	0.282	-0.3 (-29.9, 29.3)	0.985
Safety	2.0 (-20.5, 24.5)	0.862	9.2 (-11.3, 29.7)	0.380	0.4 (-22.4, 23.2)	0.972
Social cohesion	11.0 (-12.6, 34.5)	0.361	11.6 (-9.1, 32.4)	0.271	6.0 (-17.4, 29.5)	0.614
Land use mix diversity	15.0 (-6.4, 36.4)	0.169	-20.1 (-39.4, -0.8)	0.041	-2.0 (-24.2, 20.2)	0.859
Recreation facilities	-6.7 (-28.5, 15.1)	0.546	12.5 (-7.0, 31.9)	0.209	1.8 (-20.3, 24.0)	0.872

*B:* correlation coefficient; CI: confidence interval; DVD: digital versatile disc; e-devices: electronic devices; e-games: electronic games; mins: minutes; no.: number; PA: physical activity; ref: reference group; SB: sedentary behaviour; TV: television.

Generalised estimation equations: PA models adjusted for baseline PA, maternal education, English as the primary language spoken at home, arealevel SEP, age, gender, survey mode and school clusters; SB models adjusted for baseline SB, paternal education, English as the primary language spoken at home, participant born in Australia, remoteness, age, gender, survey mode and school clusters.

<sup>1</sup> Coefficient of the interaction term (independent variable\*situational transition).

Appendix 8.2	Interaction effects <sup>1</sup> between baseline levels of the independent variables and situational transitions on total sedentary
behaviour (h/day) af	iter leaving secondary school

	<b>Tertiary study status</b> (Ref: no)		Working ≥20 h/wk (Ref: working <20 h/wk)		Living with parents status (Ref: no)	
	B (95%CI) <sup>1</sup>	р	B (95%CI) <sup>1</sup>	р	B (95%CI) <sup>1</sup>	р
Individual-level variables (scores)						
PA enjoyment	0.5 (-0.9, 1.9)	0.510	-0.7 (-2.1, 0.7)	0.311	0.4 (-1.2, 1.9)	0.639
PA goal setting	0.1 (-1.1, 1.3)	0.877	0.3 (-0.8, 1.3)	0.621	-0.8 (-2.0, 0.5)	0.220
PA competence	-0.5 (-1.7, 0.7)	0.430	-0.1 (-1.2, 0.9)	0.809	0.3 (-0.9, 1.5)	0.645
PA self-efficacy	-0.1 (-1.3, 1.1)	0.849	0.8 (-0.3, 1.8)	0.158	-0.1 (-1.3, 1.2)	0.920
Social-level variables						
Family (scores):						
E-games co-participation	-0.9 (-3.3, 1.4)	0.431	1.3 (-0.7, 3.4)	0.198	0.8 (-1.6, 3.2)	0.521
PA co-participation	0.8 (-0.6, 2.1)	0.274	-0.1 (-1.3, 1.1)	0.822	1.4 (0.1, 2.8)	0.037
TV/DVDs co-participation	0.6 (-0.6, 1.8)	0.358	0.3 (-0.8, 1.3)	0.614	-0.1 (-1.3, 1.1)	0.880
PA social support	-0.1 (-1.2, 1.1)	0.931	0.4 (-0.7, 0.4)	0.471	0.2 (-1.0, 1.5)	0.721
SB discouragement	1.1 (-0.2, 2.3)	0.090	-0.7 (-1.8, 0.4)	0.193	-0.2 (-1.5, 1.0)	0.726
Friends/colleagues (scores):						
E-games co-participation	0.3 (-1.2, 1.8)	0.715	0.4 (-1.1, 1.9)	0.583	-0.1 (-2.0, 1.8)	0.927
PA co-participation	0.0 (-1.2, 1.2)	0.985	0.8 (-0.3, 1.8)	0.173	0.6 (-0.6, 1.9)	0.331
TV/DVDs co-participation	1.0 (-0.3, 2.2)	0.128	-0.1 (-1.2, 1.0)	0.838	-0.4 (-1.8, 0.9)	0.531
PA social support	-0.4 (-1.7, 0.9)	0.514	0.8 (-0.3, 2.0)	0.165	0.9 (-0.4, 2.3)	0.180
SB discouragement	0.9 (-1.7, 3.4)	0.508	-3.5 (-5.6, -1.3)	0.002	1.8 (-0.6, 4.2)	0.146
Social network count	0.6 (-0.6, 1.8)	0.293	0.4 (-0.7, 1.5)	0.459	0.0 (-1.2, 1.3)	0.941
Gym membership	0.0 (-1.3, 1.3)	0.964	0.1 (-1.0, 1.3)	0.816	0.3 (-1.0, 1.6)	0.630

## Appendix 8.2 Continued

	Tertiary study status (Ref: no)B (95%CI)1p		<b>Working ≥20</b> (Ref: working <2	<b>h/wk</b> 20 h/wk)	<b>Living with parents status</b> (Ref: no)	
			B $(95\% CI)^1$ p		B (95%CI) <sup>1</sup>	р
Environmental-level variables						
Home environment:						
E-devices, no. of	-1.2 (-2.8, 0.3)	0.126	0.0 (-1.3, 1.4)	0.964	1.1 (-0.5, 2.6)	0.173
PA equipment, no. of	0.6 (-0.6, 1.8)	0.337	-0.7 (-1.8, 0.3)	0.182	-0.2 (-1.4, 1.1)	0.793
TVs, no. of	-0.3 (-1.5, 0.9)	0.640	0.1 (-1.0, 1.2)	0.870	0.0 (-1.2, 1.3)	0.991
Neighbourhood environment (scores):						
Noise	-0.5 (-2.0, 1.0)	0.504	-0.3 (-1.7, 1.1)	0.670	-1.1 (-5.6, 0.4)	0.156
Walking environment	0.3 (-1.5, 2.1)	0.754	0.5 (-1.1, 2.0)	0.561	1.1 (-0.6, 2.8)	0.215
Safety	-0.3 (-1.6, 1.0)	0.647	0.1 (-1.0, 1.3)	0.837	-0.4 (-1.7, 0.9)	0.515
Social cohesion	-1.0 (-2.3, 0.4)	0.151	0.6 (-0.5, 1.8)	0.287	-0.4 (-1.7, 1.0)	0.596
Land use mix diversity	0.2 (-1.0, 1.4)	0.777	-0.7 (-1.7, 0.4)	0.221	0.3 (-0.9, 1.5)	0.630
Recreation facilities	0.2 (-1.0, 1.4) 0.777		-0.2 (-1.3, 0.8)	0.657	-0.2 (-1.5, 1.0)	0.724

*B:* correlation coefficient; CI: confidence interval; DVD: digital versatile disc; e-devices: electronic devices; e-games: electronic games; no.: number; PA: physical activity; ref: reference group; SB: sedentary behaviour; TV: television.

Generalised estimation equations: PA models adjusted for baseline PA, maternal education, English as the primary language spoken at home, arealevel SEP, age, gender, survey mode and school clusters; SB models adjusted for baseline SB, paternal education, English as the primary language spoken at home, participant born in Australia, remoteness, age, gender, survey mode and school clusters.

<sup>1</sup> Coefficient of the interaction term (independent variable\*situational transition).

**Appendix 8.3** Associations between situational transitions and difference in adjusted marginal means of discretionary physical activity and total sedentary behaviour at the two-year follow-up, stratified by significant baseline moderators

		Discretionary PA	Total SB
		(mins/day)	(h/day)
		B (95%CI)	B (95%CI)
Studied (Ref: No), n=796			
SB discouragement from	Low (≤3)	9.6 (-3.6, 22.8)	-
family score	High (>3)	-15.1 (-34.7, 4.5)	-
TVs no of	Low (≤2)	15.2 (0.3, 30.1)*	-
1 v 3, 10. 01	High (>2)	-12.6 (-29.6, 4.3)	-
Worked >20 h/wk (Ref: Wo	rked <20 h/wi	k). n=804	
	Low (<15)	6.3 (-4.6, 17.1)	_
PA self-efficacy score	High (>15)	30.2 (12.0, 48.4)**	-
PA co-participation with	Low (≤3)	13.2 (2.2, 24.3)*	-
family score	High (>3)	36.1 (14.7, 57.6)**	-
¥	L	12 5 (2 4 22 () **	-1.8 (-2.4, -
SB discouragement from	Low (≤3)	13.5 (3.4, 23.6)**	1.3)***
friends/colleagues score	High (>2)	113.9 (65.2,	-5.8 (-7.9, -
	Tingii (>3)	162.6)***	3.7)***
E devices no of	Low (≤3)	-2.4 (-21.6, 16.8)	-
E-devices, no. or	High (>3)	24.2 (13.1, 35.3)***	-
I and use mix diversity	Low (≤2)	24.9 (10.7, 39.2)**	-
Land use mix diversity	High (>2)	10.2 (-3.7, 24.1)	-
Lived with parents (Ref: No	p), $n=805$		
PA co-participation with	Low (≤3)	-	-0.2 (-0.9, 0.6)
family score	High (>3)	-	1.5 (0.4, 2.7)**
SB discouragement from	Low (≤3)	12.7 (-1.1, 26.4)	_
family score	High (>3)	-6.5 (-24.3, 11.3)	-
E-games co-participation	Low (≤3)	-4.1 (-15.0, 6.8)	-
with friends/colleagues	High (>3)	41.6 (-2.6, 85.9)	-
TV/DVD co-participation	Low (≤3)	-5.5 (-18.6, 7.5)	-
with friends/colleagues score	High (>3)	21.3 (2.3, 40.2)*	-
DA aquinmant ra of	Low (≤2)	-7.0 (-20.3, 6.3)	_
rA equipment, no. of	High (>2)	11.6 (-7.2, 30.3)	-

\**p*<0.05, \*\**p*<0.01, \*\*\**p*<0.001: generalised estimating equation.

*B:* correlation coefficient; CI: confidence interval; DVD: digital versatile disc; *e*-devices: electronic devices; *e*-games: electronic games; *h*: hours; mins: minutes; no.: number; ref: reference group; PA: physical activity; SB: sedentary behaviour; TV: television; wk: week.